BUSINESS AFFAIRS AND HUMAN RESOURCES NOVEMBER 29 – DECEMBER 1, 2006

TAB	DESCRIPTION	ACTION
1	BOISE STATE UNIVERSITY Facilities Financing Plan	Information Item
2	BOISE STATE UNIVERSITY Student Health, Wellness, Counseling, Nursing Building	Motion to approve
3	BOISE STATE UNIVERSITY Student Union Building Expansion	Motion to approve
4	IDAHO STATE UNIVERSITY CAES Facility/Idaho Falls	Motion to approve
5	IDAHO STATE UNIVERSITY 2006 General Revenue Bonds	Motion to approve
6	UNIVERSITY of IDAHO Amendments of Faculty-Staff Handbook	Motion to approve
7	UNIVERSITY of IDAHO Parking Lot Site Lease	Motion to approve
8	UNIVERSITY of IDAHO Contract for Vending Rights	Motion to approve
9	UNIVERSITY of IDAHO Pierce Property Re-conveyance	Motion to approve
10	LEWIS-CLARK STATE COLLEGE Management Agreement for Residence Hall	Motion to approve
11	AMENDMENT OF BOARD POLICY Policy Section III.T.4 - 1 st Reading - Intercollegiate Athletics	Motion to approve

BUSINESS AFFAIRS AND HUMAN RESOURCES NOVEMBER 29 – DECEMBER 1, 2006

TAB	DESCRIPTION	ACTION
12	IDAHO PROMISE SCHOLARSHIP Increase Category B Award	Motion to approve
13	COLLEGE & UNIVERSITY AUDIT PRESENTATION Moss Adams, LLP	Motion to approve

INSTITUTION / AGENCY AGENDA BOISE STATE UNIVERSITY

SUBJECT

Information item by Boise State University (BSU) regarding future financing plans, as they relate to debt capacity and debt management.

REFERENCE

December 2003	Single Bond System and Financing Plan for New Projects
February 2004	BSU Bonding Discussion, Series 2004A
December 2004	Long-Term Facilities Funding Plan
April 2005	Refinancing and Revenue Bonds, Series 2005A
October 2005	Board approved 2005 Campus Master Plan Update
June 2006	Financing Update

APPLICABLE STATUTE, RULE OR POLICY

Idaho State Board of Education Governing Policies & Procedures, Sections V.F. and V.K.3.

BACKGROUND

As of April 30, 2006, Boise State University has approximately \$145,000,000 in outstanding debt, including leases, notes, and bank loans. \$127,600,000 (92%) of current debt is in the form of fixed rate bonds. The current weighted average life of the bonded debt of the University is relatively short at 11.2 years.

The University anticipates issuing a significant amount of debt during the next two years. Variables impacting the amount include the planning timeline of the various projects, as some may be ready to begin construction sooner than others. In addition, the amount of donations, grants, and/or state funds secured could impact the amount of the borrowing. An increase in these funds could increase scope, thereby increasing the debt, or substitute for financing and decrease debt.

DISCUSSION

This information item is being presented to the Board to disclose the amount of debt Boise State will incur pending Board approval of proposed new buildings (see Attachment 1). The attachment provides the following:

- cost of pending projects,
- estimate of the debt financed portion of each project
- breakdown of the annual debt service for each project
- estimate of debt service as a percentage of operating revenue

INSTITUTION / AGENCY AGENDA BOISE STATE UNIVERSITY – continued

Pending Board approval of the projects, the University would issue debt for the Parking Deck (already approved by the Board), Student Union Building (SUB) expansion, and the Student Health, Wellness, Counseling Center/Nursing Building in January of 2007. In February 2007, the University would issue debt for the Stadium Suites project. The University would issue debt for the CESED/Research Facility, College of Business and Economics Building, and Student Housing twelve to eighteen months hereafter.

IMPACT

This amount of new debt will likely reduce the University's credit rating to A2/A. While aggressive, Boise State has the inherent financial capacity to incur this debt without negatively impacting access to future capital, and believes this level of borrowing is necessary to complete pending new buildings consistent with the University's strategic plan.

ATTACHMENTS

Attachment 1 - Ten Year Debt Projection Page 4

STAFF COMMENTS AND RECOMMENDATIONS

As noted above, BSU is beginning a substantial new building construction phase, which will require the issuance of bonds as the major, or only, financing mechanism for the new buildings. Staff requested that BSU officials make a more detailed presentation to the Board in November 2006 as a follow-up to the June 2006 presentation on the same subject.

The Attachment, which will be discussed by BSU officials at the meeting, presents the following information:

Lines 1 – 12	Existing and future building plans
Line 13	Total amount of new debt financing, and by fiscal year
Lines 14 – 17	Type of new building (coded by color, Lines $1 - 12$)
Lines 18-23	Beginning and ending facilities fee reserve through FY 2015
Line 24	Amount of projected annual debt service
Line 25	BSU operating budget, all funds (excluding direct student loans)
Line 26	Debt service as a percentage of operating budget
Lines 30 – 35	Economic and other assumptions used in analysis

Line 23, the Ending Facilities Fee Reserve, shows a steady increase through the life of the spreadsheet. This reserve is important for several reasons, most importantly to help with the University's position with the bond rating agencies.

INSTITUTION / AGENCY AGENDA BOISE STATE UNIVERSITY – continued

Line 26, the percentage of debt service compared to the overall university operating budget, is important to review. As BSU begins to issue additional debt in FY 2008, that percentage increases, and continues to increase through FY 2010. That percentage begins to decrease in FY 2011 as the overall university operating budget increases (using the assumptions noted at the bottom of the Attachment, including an increased facilities fee being in place in FY 2009).

Most importantly, the attached schedule allows the Board to become aware of the overall new construction position of the university, and allow for a very open discussion of future building needs, revenue and debt to be incurred. If any of the economic conditions change (state appropriations, enrollment/fee revenue, programmatic needs, etc.), the analysis can be revised as necessary.

The attached schedule shows that BSU will request increases in their strategic facilities fee by \$65 in FY 2008 and \$95 in FY 2009. Staff points out that as BSU increasingly dedicates its student-paid fees to bonds, options to the institution with respect to funding are reduced. The Board and State may consider developing initiatives that might, for example, dedicate a portion of student fees to other areas besides paying off bonds. For example, the institution, Board or State may want to have the institution set aside a portion of fees for financial aid, as have many institutions in the country have done. Basically, funding any new initiatives are hindered when a large percentage of an institution's fees are already reserved for future bond payments.

Later in this agenda are two facilities which will be partially or wholly financed with bonds: the Student Health, Wellness, Counseling and Nursing building; and the Student Union Building expansion. In future meetings, the Board will be requested to authorize additional buildings, as noted in the attachment.

Staff commends BSU officials, who will be present to offer additional comments on their financing and construction plans, for offering to undertake such a comprehensive planning process regarding debt management.

BOARD ACTION

This item is for informational purposes only.

THIS PAGE INTENTIONALLY LEFT BLANK

Boise State University Ten Year Debt Projection November 13, 2006

		_			FY 2006	FY 2007	FY 2008	FY 2009	FY 2010	FY 2011	FY 2012	FY 2013	FY 2014	FY 2015	Total
			Est. Debt												
1 N	lew Buildings	Cost	Financed	terms				Ann	ual Debt Servic	9					
2	Parking Deck	\$12,000,000	\$2,400,000	30, 5%	1st payment Oct	2008	\$156,000	\$156,000	\$156,000	\$156,000	\$156,000	\$156,000	\$156,000	\$156,000	\$1,248,000
3	SUB Expansion	\$30,650,000	\$30,000,000	30, 5%	1st payment Oct	2008	\$1,952,000	\$1,952,000	\$1,952,000	\$1,952,000	\$1,952,000	\$1,952,000	\$1,952,000	\$1,952,000	\$15,616,000
4	Student HWC/Nursing	\$25,000,000	\$23,400,000	30, 5%	1st payment Oct	2008	\$1,522,000	\$1,522,000	\$1,522,000	\$1,522,000	\$1,522,000	\$1,522,000	\$1,522,000	\$1,522,000	\$12,176,000
5	Stadium Suites	\$35,000,000	\$30,000,000	30, 6% (tax)	1st payment Oct	2008	\$2,179,467	\$2,179,467	\$2,179,467	\$2,179,467	\$2,179,467	\$2,179,467	\$2,179,467	\$2,179,467	\$17,435,736
6	CESED/Research Facility	\$35,000,000	\$31,000,000		1st payment Oct			\$2,250,000	\$2,250,000	\$2,250,000	\$2,250,000	\$2,250,000	\$2,250,000	\$2,250,000	\$15,750,000
7	College of Business & Economics	\$30,000,000	\$15,000,000		1st payment Oct			\$1,090,000	\$1,090,000	\$1,090,000	\$1,090,000	\$1,090,000	\$1,090,000	\$1,090,000	\$7,630,000
8	Student Housing	\$30,000,000	\$30,000,000	30, 6%	1st payment Oct	2009		\$2,180,000	\$2,180,000	\$2,180,000	\$2,180,000	\$2,180,000	\$2,180,000	\$2,180,000	\$15,260,000
9	Next Science Building	\$35,000,000	undetermined												\$0
10	Student Services Center	\$16,000,000	undetermined												\$0
11	Parking Deck	\$13,000,000	undetermined												\$0
12	College of Health Science Building	\$30,000,000	undetermined												\$0
13	New Debt Financing		\$161,800,000		\$0	\$0	\$5,809,467	\$11,329,467	\$11,329,467	\$11,329,467	\$11,329,467	\$11,329,467	\$11,329,467	\$11,329,467	\$85,115,736
14	Infrastructure Facility	6%	\$15,400,000												
15	Student Facility	35%	\$89,900,000												
16	Academic Facility	47%	\$120,500,000												
17	Athletics Facility	12%	\$30,000,000												
18 B	eginning Facilities Fee Reserve				\$0	\$1,950,000	\$6,192,000	\$9,335,615	\$11,266,324	\$13,289,760	\$15,406,851	\$17,618,532	\$19,925,750	\$22,329,460	\$0
19	Debt Service				\$0	\$0	-\$5,809,467	-\$11,329,467	-\$11,329,467	-\$11,329,467	-\$11,329,467	-\$11,329,467	-\$11,329,467	-\$11,329,467	-\$85.115.736
20	SFF Fee revenue				\$1,950,000	\$4,242,000	\$6,273,615	\$9,272,709	\$9,365,436	\$9,459,090	\$9,553,681	\$9,649,218	\$9,745,710	\$9,843,167	\$79,354,628
21	Pledges for debt service				\$ 1,000,000	¢.,2.2,000	\$500,000	\$500,000	\$500,000	\$500,000	\$500,000	\$500,000	\$500,000	\$500,000	\$4,000,000
22	Project Revenues for debt service						\$2,179,467	\$3,487,467	\$3,487,467	\$3,487,467	\$3,487,467	\$3,487,467	\$3,487,467	\$3,487,467	\$26,591,736
	nding Facilities Fee Reserve				\$1,950,000	\$6,192,000	\$9,335,615	\$11,266,324	\$13,289,760	\$15,406,851	\$17,618,532	\$19,925,750	\$22,329,460	\$24,830,628	\$24,830,628
	5				, ,,	+-, - ,	+ - , ,	+)) -	+ -,,	¥ - , ,	+))	+ - , ,	* ,,	*))	+ ,,
24 T	otal Projected Debt Service				\$0	\$12,454,126	\$18,298,526	\$23,852,182	\$23,882,673	\$22,880,208	\$22,206,767	\$22,251,571	\$22,294,103	\$22,342,388	
-															
25 Operating Budget (less direct loans) \$219,000			\$219,000,000	\$227,520,000	\$238,529,100	\$254,452,295	\$274,891,687	\$295,639,183	\$316,702,171	\$338,088,243	\$359,805,200	\$381,861,057			
00 P															
26 🛛	ebt Service as a % of Operating Budg	et (note 1)			5.71%	5.5%	7.7%	9.4%	8.7%	7.7%	7.0%	6.6%	6.2%	5.9%	

27 Note 1:

28 As presented in June, 2006, a ratio of 6% likely drops the University's bond rating to A/A2 and a ratio of 10% would significantly hinder access to capital in the bond markets.

29 Assumptions:

- 30 1. 1% growth in enrollment
- 31 2. 3% growth in state support
- 32 3. \$1 SFF equals \$30,000 in annual funding, \$65 new fee in FY 2008 and \$95 new fee in FY 2009

33 4. Athletics revenues will equal debt service

5. Housing will pay 60% of debt service from operations

35 6. No new source of state funding for academic buildings

Attachment 1

THIS PAGE INTENTIONALLY LEFT BLANK

REFERENCE – APPLICABLE STATUTE, RULE OR POLICY

Idaho State Board of Education GOVERNING POLICIES AND PROCEDURES SECTION: V. FINANCIAL AFFAIRS Subsection: F. Bonds and Other Indebtedness

April 2002

F. Bonds and Other Indebtedness

1. General Powers

The University of Idaho, Idaho State University, Lewis-Clark State College, and Boise State University may, by a majority vote of all the members of the Board, borrow money with or without the issuance of bonds pursuant to Chapter 38, Title 33, Idaho Code. The Board must act by formal resolution. Such indebtedness is not an obligation of the state of Idaho but is an obligation solely of the respective institutions and the respective board of trustees. Any indebtedness is to be used to acquire a project, facility, or other asset that may be required by or be convenient for the purposes of the institution. Student fees, rentals, charges for the use of the projected facility, or other revenue may be pledged or otherwise encumbered to pay the indebtedness. Refunding bonds also may be issued.

Eastern Idaho Technical College is not authorized to borrow money under Chapter 38, Title 33, Idaho Code.

2. Attorney General's Opinion

The Board or the institution may request the Attorney General of Idaho to review and pass upon the validity of a proposed bond issue. If found valid, the bond is an incontestable, binding obligation on the institution.

3. Private Sale

A private sale of bonds is permitted only with the prior approval of the Board as the governing body of the institution. The chief executive officer of the institution must justify why a public sale is not desirable and explain the benefits of a private sale of bonds.

4. Responsibility of the Chief Executive Officer

The chief executive officer of the institution is responsible for compliance with state law and these provisions when any indebtedness is incurred.

5. Expenditure of Excess Revenue

Expenditure of project revenues over and above that pledged or otherwise encumbered to meet the indebtedness is limited to expenditures for projects identified in the bond's Official Statement. Expenditure of excess revenue for other projects requires prior Board approval. Expenditures between two hundred fifty thousand dollars (\$250,000) and five hundred thousand dollars (\$500,000) require prior approval from the executive director and expenditures greater than five hundred thousand dollars (\$500,000) require prior Board approval.

Idaho State Board of Education GOVERNING POLICIES AND PROCEDURES SECTION: V. FINANCIAL AFFAIRS Subsection: K. Construction Projects

April 2002

K. Construction Projects

3. Fiscal Revisions to Previously Approved Projects

Project revisions that substantially alter the use of the project causing changes in project costs between two hundred fifty thousand dollars (\$250,000) and five hundred thousand dollars (\$500,000) must first be submitted to the executive director for review and approval. Changes in project costs of more than five hundred thousand dollars (\$500,000) must first be submitted to the Board for its review and approval. Requests must be supported by a revised detailed project budget and justification for changes.

INSTITUTION/AGENCY AGENDA BOISE STATE UNIVERSITY

SUBJECT

Boise State University requests approval to complete design development and proceed with construction of the Student Health, Wellness and Counseling Center (SHWCC) with the inclusion of additional program space for the Department of Nursing for a total project cost not to exceed \$25,000,000.

REFERENCE

January 2004	Board approved 2004-2005 student fees, including a \$40/semester facility fee
April 2004	Board approved request to proceed with the RFP process planning and design of the Health, Wellness and Counseling Services Center project, at a cost not to exceed \$932,025.
June 2004	Board approved programming and design for an estimated project budget of \$13,600,000 and authorized design work through design development.
November 2005	Interim Executive Director of the Board approved a \$495,000 increase to the project budget to add design and materials to meet higher energy efficiency standards such as LEED.
June 2006	Capital projects update included discussion that the SHWCC project was delayed to add space for nursing education to be funded from private gifts and other sources.

APPLICABLE STATUTE, RULE, OR POLICY

Idaho State Board of Education Governing Policies & Procedures, Sections B.8-9 and V.K.

Sections 33-3804, 67-5710, 67-5710A, 67-5711-C, Idaho Code

BACKGROUND

As noted in the reference section above, the Board has approved the planning and design for a new Student Health, Wellness, and Counseling Center (SHWCC) to be funded from a student facility fee. The approved project budget for this facility is currently \$14,095,000, which includes \$495,000 to pursue additional energy efficiency measures.

It is well known that additional academic space for nursing education is needed. University staff has determined the most efficient way to meet the most urgent space needs is to build additional space in the SHWCC building, which is already in the planning and design process, and to modify and increase the capacity of the SHWCC to incorporate the nursing academic space utilizing private gifts, state and university funds.

INSTITUTION/AGENCY AGENDA BOISE STATE UNIVERSITY - continued

DISCUSSION

The decision to expand this project reflects the critical shortage of nurses in Idaho and across the nation. The national nursing shortage is expected to exceed 800,000 by 2020. Idaho is currently ranked 49th in the nation for registered nurses per capita. The health care community is looking to Boise State University to increase enrollments in its nursing program to meet looming workforce shortages in the nursing profession. Preliminary programming studies indicate the new facility will allow for the graduation of 30 to 50 additional nursing students per year.

A flexible, integrated facility that expands the previously approved Student Health, Wellness and Counseling Center to incorporate space needed to support the nursing program is proposed to meet this demand. The nursing component of the new facility will encompass approximately 20,000 square feet of net assignable area. The added program space will support the full range of instructional and support space required by the Department including classrooms, seminar rooms, practice labs, and offices. A simulation lab, which will allow the nursing faculty to simulate an acute care clinical experience for students, is also planned.

The original program for the Student Health, Wellness and Counseling facility included approximately 35,000 square feet of net assignable area. Ongoing refinement of program needs has led to the reduction or elimination of some program elements, resulting in a complete current program space list that consists of approximately 21,000 net assignable square feet. The new integrated facility will be approximately 70,000 total gross square feet. Attachment 3 provides a detailed program listing.

Expanding the scope of the SHWCC will enhance the nursing program at Boise State in three ways. First, it will increase the nursing program's capacity to allow for a 30% increase in the number of graduates. Second, the addition of the simulation lab will provide an alternative to practice in clinical space for acute care that is very difficult to find as schools are increasing capacity. Finally, this addition will allow the Nursing Program to better collaborate with the Student Health, Wellness, and Counseling Center to foster improved health on our campus.

INSTITUTION/AGENCY AGENDA BOISE STATE UNIVERSITY - continued

IMPACT

A detailed cost estimate based on the comprehensive programming estimates the construction contract cost at \$20,500,000. Contingencies (\$1,939,249), design fees (\$2,255,000) and other administrative and soft costs (\$305,751) bring the total project cost to \$25,000,000 (See Attachment 1). This project will be brought back to the Board for financing approval prior to contract award.

Current funding sources for the project are as follows:

Student Facilities Fee Bonds	\$13,900,000
Facilities Fee Reserve	1,000,000
Private gifts	5,100,000
Private, state or facility fee bonds	5,000,000*

Total \$25,000,000

*The University is pursuing both private gifts and state funds for this amount. However, there is sufficient debt capacity in the current strategic facilities fee to bond this additional amount should a portion or all of these funds not materialize. The University anticipates requesting Board approval to issue bonds to finance the construction of this facility in fall 2007.

ATTACHMENTS

Attachment 1 – Capital Facilities Approval Processing	Page 5
Attachment 2 – Project Budget Worksheet	Page 6
Attachment 3 – Capital Project Tracking Sheet	Page 7
Attachment 4 – Programming Information	Page 8

STAFF COMMENTS AND RECOMMENDATIONS

Staff has included a "Capital Facilities Approval Processing" chart on page 5, which shows the process used by the State Board of Education and the Division of Public Works for new facility planning and approval. BSU's request is outlined as Step 3 as the current Board request to approve the budget and preliminary design.

BSU's intention to increase the scope of this project beyond the original plans is in response to the need for additional nursing graduates in Idaho.

INSTITUTION/AGENCY AGENDA BOISE STATE UNIVERSITY - continued

BSU will request permission to issue bonds if some or all of the \$5.0 million noted above does not materialize. As indicated in the following paragraph, Board members need to be adequately briefed regarding likely sources of financing, including the University's ability to service potential debt, before granting authority to proceed with design development, financing and construction.

Staff has requested from BSU a comprehensive report outlining its medium-term (10+ years) capital facilities situation: needs, funding, bond payments and reserves, etc. This report was presented in the previous agenda item. The report is designed to provide the Finance Committee, and all Board members, with a foundation regarding BSU's overall capital facilities situation, and will be used as a template for the other institutions, with modifications as required.

Staff notes BSU's intention to pursue private and/or state funding for the expanded portion (nursing education) part of this building project. Governor Risch has suggested that additional nursing buildings be constructed at Lewis-Clark State College and the College of Southern Idaho. This recommendation, and that of incoming Governor Otter, will be acted upon by the legislature in the upcoming session. Final outcome for these two projects, and the potential for BSU receiving additional state funding for this building, is not known.

Staff recommends Board approval for BSU to complete design development and proceed with construction, contingent upon Board approval to issue bonds to finance this project.

BOARD MOTION

A motion to approve the request by Boise State University to proceed with construction of the Student Health, Wellness and Counseling Center with the inclusion of additional program space for the Department of Nursing and to authorize the Division of Public Works to award a design-build contract to Leatham, Krohn, Van Ocker Architects, the selected design-build team. The motion includes funding that comes from bonds, facility fee reserves, and private gifts, and the entire cost of the project shall not exceed \$25,000,000. The University may not proceed with construction of this project prior to Board approval to issue debt.

Moved by Seconded by Carried Yes No

Idaho State Board of Education (SBOE) and Permanent Building Fund Advisory Council (PBFAC)

Capital Facilities Approval Process (excluding UI)

Step	SBOE Actions	PBFAC/DPW Actions
1.	Approves project concept submitted by Institution	
2.	Staff sends Board approval letter to DPW	DPW sets up project, selects Architecture/ Engineering firm (& other consultants) PBFAC approves selection & proceeding with preliminary design.
3.	Approves project budget & preliminary design (through design development) and project budget *	
4.	Staff sends Board approval letter to DPW	Approves preliminary design (through design development, authorizes proceeding with construction documents
5.	Approves final budget financing and final design **	
6.	Staff sends Board approval letter to DPW	Approves final design and authorizes bidding and award of contract for construction
7.		DPW reviews bids, determines responsive contractor
8.		DPW awards construction contract, if within budget, as requested by institution
9.	If bids exceed budget review, approve request for increase as presented by institution	
10.	Staff sends Board approval letter to DPW if necessary	DPW awards construction contract after budget increase is authorized by SBOE
11.		DPW confirms all contract documents are correct, then issues Notice to Proceed
12.	Review and approve increase in budget during course of construction, if needed	
13.	Staff sends Board approval letter to DPW if necessary	DPW establishes Substantial Completion
14.		Review final accounting of project expenditures, compare to approved budget
15.		Submit report to SBOE

* If Project Delivery Method has been determined, report method to the Board.

** Project Delivery Method will be known and reported to the Board.

Note: PBFAC and/or SBOE staff provide project updates as determined to be needed, or at request by SBOE members. University of Idaho follows these guidelines for projects using State General Funds or Bonds.

PROJECT BUDGET WORKSHEET

PROJECT NUMBER:

PROJECT TITLE:

DPW04-214

STUDENT HEALTH, WELLNESS AND COUNSELING DEPARTMENT OF NURSING BUILDING

TOTAL BUDGET: \$25,000,000

CATEGORY		Estimated Budget
A/E Fees		\$2,050,000
A/E Reimbursable		\$205,000
Construction Manager		\$0
CM Reimbursable		\$0
Consultant - Estimator		\$10,000
Consultant Reimbursable		\$0
Testing during construction		\$70,000
Construction Contract	\$20,500,000	
Construction Contingency	\$1,025,000	

Subtotal Construction Contract & Contin	\$21,525,000	\$21,525,000
Commissioning		\$70,000
Builder's Risk		\$82,000
Utility Hook-up Fees		\$30,000
Project Contingency		\$914,249
Plan Check Fee		\$27,651
Advertising		\$100

TOTAL	#05 000 000
Soil Investigation	\$8,000
Site Survey	\$8,000
5	

TOTAL

\$25,000,000

Office of the Idaho State Board of Education Capital Project Tracking Sheet Nov-06

History Narrative

	Institution/Agency: Project Description:	Nev incl Nur	lude sp rsing f	ding pace facul	to ho es to Ity of	ouse Stu accomo	date nd in	t He con	nprehensive	Nui s an heal	Ident Health, We rsing Building Id Counseling S Ith services inclu spaces will be	ervi udin	ices and Dep g general m	partment of Nur edicine, counse	rsing. The integeling and wellne	grated	d facility will omponents.
3	Project Use:	Ne	w inteç	grate	ed fac	ilities fo	r Stu	iden	t Health, Wel	Ines	s and Counselir	ng S	Services and	the Department	t of Nursing		
4	Project Size:	70,	000 g.	s.f. (of Ne	w Space	Э										
5									<u>, - , </u>			1					
6 7						S	ourc	es o	f Funds		Total			Use of Funds	f Funds		Total
8			PBF			ISBA			Other *		Sources		Planning	Const	Other		Uses
	Initial Cost of Project	\$		-	\$			\$	932,025	\$	932,025	\$	932,025			\$	932,025
12	History of Revisions																
13	Revision #1 (November 2005)	\$		-	\$		-	\$	495,000	\$	495,000	\$	50,000	\$ 445,000		\$	495,000
15		rials	to inci	reas	se ene	ergy effi	cieno	су									
	Proposed Revision #2 (November 2006) Incorporate Planning for Departn	\$ nent	of Nur	- rsina	\$ 1 and	Constru	- Ictior	\$ nofi	23,572,975 ntegrated fac		23,572,975	\$	1,272,975	\$ 21,080,000	\$ 1,220,000	\$ 2	23,572,975
18 19 20 21		ioni		Unig		Constre	otion			,y.							
	Total Project Costs	\$		-	\$		-	\$	25,000,000	\$	25,000,000	\$	2,255,000	\$ 21,525,000	\$ 1,220,000	\$ 2	25,000,000
23	-				-						· · ·						
24 25								h	l nstitutional		* Other Source Student	es o	f Funds	 Total	Total		
	History of Funding:		PBF			ISBA			Funds		Revenue		Other	Other	Funding		
26 27 28	11/17/05 - Increase Approved 10/31/06 - Requested Increase	\$ \$ \$		-	\$ \$ \$		l	\$ \$ \$	-	\$ \$ \$	932,025 495,000 13,472,975	\$	10,100,000	\$ 932,025 \$ 495,000 \$ 23,572,975	\$ 495,000)	
29 30		\$		-	\$		-	\$	-	\$	14.900.000	\$	- 10.100.000		\$ 25.000.000	5	

PROGRAM SUMMARY BY GROUP	Proposed Program
Department/Use	NASF
STUDENT HEALTH, WELLNESS AND COUNSELING	
Group A	
Administration	
Medical Records	
Central Services	
Insurance	
Computer Services	
Group B	
Health Promotion	
Group C	
General Medicine Clinics	
Specialty Clinics	
Urgent Care Clinic (Included with General Medicine Clinics)	
Group D	
Laboratory	
Phlebotomy	
Pharmacy	
Group E- Optional Ancillary Services	
Dental Clinic	
Group F	
Counseling	
Group G	
Common Areas	
Subtotal Student Health and Wellness	2
DEPARTMENT OF NURSING	
Group H	
Administration and Support	
Faculty Offices	
Group I	1
Instructional Learning Areas	1
Group J	
Nursing Workforce Center	
Group K	
Common Areas	
Subtotal Department of Nursing	1
COMBINED HWC AND NURSING TOTAL AREA	4
Unassignable Spaces	2
+	
ESTIMATED BUILDING GROSS SQUARE FEET	7

REFERENCE: APPLICABLE STATUTE, RULE, OR POLICY

Idaho State Board of Education GOVERNING POLICIES AND PROCEDURES SECTION: V. FINANCIAL AFFAIRS B. Budget Policies

April 2002

8. Major Capital Improvement Project -- Budget Requests

For purposes of Item 8., the community colleges (NIC and CSI), the State Historical Society, and the State Library are included, except as noted in V.B.8.b. (2).

a. Definition

A major capital improvement is defined as the acquisition of an existing building, construction of a new building or an addition to an existing building, or a major renovation of an existing building. A major renovation provides for a substantial change to a building. The change may include a remodeled wing or floor of a building, or the remodeling of the majority of the building's net assignable square feet. An extensive upgrade of one (1) or more of the major building systems is generally considered to be a major renovation.

- b. Preparation and Submission of Major Capital Improvement Requests
 - (1) Permanent Building Fund Requests

Requests for approval of major capital improvement projects to be funded from the Permanent Building Fund are to be submitted to the Office of the State Board of Education on a date and in a format established by the executive director. Only technical revisions may be made to the request for a given fiscal year after the Board has made its recommendation for that fiscal year. Technical revisions must be made prior to November 1.

(2) Other Requests

Requests for approval of major capital improvement projects from other fund sources are to be submitted in a format established by the executive director. Substantive and fiscal revisions to a requested project are resubmitted to the Board for approval. This subsection shall not apply to the community colleges.

c. Submission of Approved Major Capital Budget Requests

The Board is responsible for the submission of major capital budget requests for the institutions, school and agencies under this subsection to the Division of Public Works. Only those budget requests which have been formally approved by the Board will be submitted by the office to the executive and legislative branches.

9. Approval by the Board

Requests for approval of major capital improvement projects must be submitted for Board action. Major capital improvement projects, which are approved by the Board and for which funds from the Permanent Building Fund are requested, are placed in priority order prior to the submission of major capital budget requests to the Division of Public Works.

REFERENCE: APPLICABLE STATUTE, RULE, OR POLICY

Idaho State Board of Education GOVERNING POLICIES AND PROCEDURES SECTION: V. FINANCIAL AFFAIRS K. Construction Projects

April 2002

K. Construction Projects

1. Major Project Approvals - Proposed Plans

Without regard to the source of funding, before any institution, school or agency under the governance of the Board begin formal planning to make capital improvements, either in the form of renovation or addition to or demolition of existing facilities, when the cost of the project is estimated to exceed five hundred thousand dollars (\$500,000), must first be submitted to the Board for its review and approval. All projects identified on the institutions', school's or agencies' six-year capital plan must receive Board approval.

2. Project Approvals

Without regard to the source of funding, proposals by any institution, school or agency under the governance of the Board to make capital improvements, either in the form of renovation or addition to or demolition of existing facilities, when the cost of the project is estimated to be between two hundred fifty thousand dollars (\$250,000) and five hundred thousand dollars (\$500,000), must first be submitted to the executive director for review and approval. Without regard to the source of funding, proposals by any institution, school or agency under the governance of the Board to make capital improvements, either in the form of renovation or addition to or demolition of existing facilities or construction of new facilities, when the cost of the project is estimated to exceed five hundred thousand dollars (\$500,000), must first be submitted to the Board for its review and approval. Project cost must be detailed by major category (construction cost, architecture fees, contingency funds, and other). When a project is under the primary supervision of the Board of Regents or the Board and its institutions, school or agencies, a separate budget line for architects, engineers, or construction managers and engineering services must be identified for the project cost. Budgets for maintenance, repair, and upkeep of existing facilities must be submitted for Board review and approval as a part of the annual operating budget of the institution, school or agency.

3. Fiscal Revisions to Previously Approved Projects

Project revisions that substantially alter the use of the project causing changes in project costs between two hundred fifty thousand dollars (\$250,000) and five hundred thousand dollars (\$500,000) must first be submitted to the executive director for review and approval. Changes in project costs of more than five hundred thousand dollars (\$500,000) must first be submitted to the Board for its review and approval. Requests must be supported by a revised detailed project budget and justification for changes.

4. Project Acceptance

Projects under the supervision of the Department of Administration are accepted by the Department on behalf of the Board and the state of Idaho. Projects under the supervision of an institution, school or agency are accepted by the institution, school or agency and the project architect. Projects under the supervision of the University of Idaho are accepted by the University on behalf of the Board of Regents.

- 5. Statute and Code Compliance
 - a. All projects must be in compliance with Section 504 of the Rehabilitation Act of 1973 and must provide access to all persons. All projects must be in compliance with applicable state and local building and life-safety codes and applicable local land-use regulations as provided in Chapter 41, Title 39, and Section 67-6528, Idaho Code.
 - b. In designing and implementing construction projects, due consideration must be given to energy conservation and long-term maintenance and operation savings versus short-term capital costs.

REFERENCE – APPLICABLE STATUTE, RULE OR POLICY – continued

Idaho Statutes

TITLE 33 EDUCATION CHAPTER 38 STATE INSTITUTIONS OF HIGHER EDUCATION BOND ACT

33-3804. POWERS AND DUTIES OF STATE INSTITUTIONS.

Every institution shall have power in its proper name as aforesaid:

- (a) To have a corporate seal and alter the same at pleasure;
- (b) To sue and be sued;
- (c) To acquire by purchase, gift or the exercise of the right of eminent domain and hold and dispose of real or personal property or rights or interests therein and water rights;
- (d) To make contracts and to execute all instruments necessary or convenient;
- (e) To acquire any project or projects, and to own, operate, and maintain such project;
- (f) To accept grants of money or materials or property of any kind from a federal agency, upon such terms and conditions as such federal agency may impose;
- (g) To borrow money, with or without the issuance of bonds and to provide for the payment of the same and for the rights of the holders of such bonds and/or of any other instrument of such indebtedness, including the power to fix the maximum rate of interest to be paid thereon and to warrant and indemnify the validity and tax exempt character;
- (h) To perform all acts and do all things necessary or convenient to carry out the powers herein granted, to obtain loans or grants or both from any federal agency, and to accomplish the purposes of sections 33-3801--33-3813, Idaho Code, and secure the benefits of the Recovery Act;
- To issue refunding bonds, for the purpose of paying, redeeming, or refunding any (i) outstanding bonds theretofore issued under authority of this chapter. Refunding bonds so issued shall have such details, shall bear such rate or rates of interest and shall be otherwise issued and secured as provided by the board authorizing the issuance of such bonds and as otherwise provided in this chapter, provided, however, that such changes in the security and revenues pledged to the payment thereof may be made by such board as may be provided by it in the proceedings authorizing such bonds, but in no event shall such refunding bonds ever be secured by revenues not authorized by this chapter to be pledged to the payment of bonds issued for other than refunding purposes. Refunding bonds issued hereunder may be exchanged for a like principal amount of the bonds to be refunded, may be sold in the manner provided in this chapter for the sale of other bonds, or may be exchanged in part and sold in part. If sold, the proceeds of such bonds may be deposited in escrow for the payment of the bonds to be refunded, provided such bonds mature or are callable for redemption under their terms within six (6) months from the date of the delivery of the refunding bonds.

No refunding bonds may be issued hereunder in a principal amount in excess of the principal amount of the bonds to be refunded nor may any bonds not maturing or callable for redemption under their terms as above provided be refunded hereunder without the consent of the holders thereof. Refunding bonds so authorized and issued may in the discretion of the board be combined with other bonds to be authorized and issued under this chapter, and a single issue of bonds may be so authorized in part for improvement and in part for refunding purposes.

(j) In connection with borrowing without the issuance of bonds, to fix fees, rents or other charges for utilization of any facility or project being financed by said borrowing and to pledge the same, together with any other revenue from such project or facility, as collateral for repayment of principal and interest in the same manner and to the same extent as provided in this chapter for securing the payment of bonds issued pursuant to this chapter.

REFERENCE – APPLICABLE STATUTE, RULE OR POLICY – continued

Idaho Statutes TITLE 67 STATE GOVERNMENT AND STATE AFFAIRS CHAPTER 57 DEPARTMENT OF ADMINISTRATION

67-5710. PERMANENT BUILDING FUND ADVISORY COUNCIL -- APPROVAL OF USE OF FUND -- DUTIES OF ADMINISTRATOR OF PUBLIC WORKS.

There is hereby created in he division of public works a permanent building fund advisory council which shall be appointed by the governor. This council shall be composed of one (1) member of the senate, one (1) member of the house of representatives, a citizen engaged in the contracting business, a citizen engaged in the banking business, and a citizen who is a member of the business community not engaged in contracting or banking. The senate member and house of representative member shall be appointed for a fixed term of two (2) years. All other council members shall be appointed for a fixed term of three (3) years. The terms of office of members of the council holding office prior to July 1, 1996, shall expire on the following dates: contracting business member on July 1, 1996; senate member and house of representative member on December 1, 1996; business community member on July 1, 1997; and banking member on July 1, 1998. On or after July 1, 1996, the governor shall appoint members of the council as terms of existing members expire. All members of the council shall serve at the pleasure of the governor. The administrator of public works and the responsible heads of the agencies for which appropriations for construction, renovations, remodelings or repairs are made pursuant to chapter 11, title 57, Idaho Code, shall consult, confer and advise with the permanent building fund advisory council in connection with all decisions concerning the administration of these appropriations and the planning and construction or execution of work or works pursuant thereto. The approval of the permanent building fund advisory council shall be a condition precedent to the undertaking of planning or construction.

The administrator of public works is hereby directed to work in close cooperation with the responsible heads of institutions and agencies for which appropriations are made herein and no building proposals shall be approved by the administrator of public works nor any planning or work undertaken by that officer pursuant to these appropriations without the prior approval of the responsible chief officer of the institutions and agencies for whom appropriations are made herein. **67-5710A**. REQUIREMENT OF PLANS AND SPECIFICATION APPROVAL BY PERMANENT

BUILDING FUND ADVISORY COUNCIL AND DELEGATION OF PROJECT OVERSIGHT BY THE

ADMINISTRATOR FOR THE DIVISION OF PUBLIC WORKS.

(1) (a) Unless an emergency exists as defined in section 67-5711B, Idaho Code, an existing public works may not be altered, repaired, constructed or improved on property owned or occupied by any state institution, department, commission, board or agency, if the estimated cost of work exceeds the limit established in section 67-5711, Idaho Code, and except for those institutions and agency exemptions listed in section 67-5711, Idaho Code, without regard to source of funding, until the location, design, plans and specifications are approved by the permanent building fund advisory council and the project supervised by the division of public works or its designee.

(b) Facilities to be built with funds under the control of a nonstate entity, and owned or occupied by state entities, must have plans and specifications prepared, and all plans and specifications must be reviewed and approved by the permanent building fund advisory council prior to the advertising, bidding, construction and/or negotiation for construction of the facilities.

(2) (a) The administrator for the division of public works may delegate control over design, construction and all other aspects of a public works or maintenance project which costs less than one hundred fifty thousand dollars (\$150,000), to agencies of state government on a project-by-project basis, if a responsible party of the state agency requests that delegation in writing and the permanent building fund advisory council approves the delegation.

(i) The state agency to whom control is delegated shall assume all responsibility for project budgets and shall receive funds appropriated for the project upon application and approval by the permanent building fund advisory council.

(ii) Delegation of project control does not exempt the state agency from complying with public works statutes, life safety and building codes or other applicable codes and regulations. The state agency also must comply with any guidelines or procedures for design and construction adopted by the division of public works and the permanent building fund advisory council.

(iii) State agencies that receive delegated projects may not have access to permanent building fund advisory council contingency funds unless approved by the permanent building fund advisory council or authorized by appropriation.

(iv) Prior, written approval from the administrator must be granted for any public works utilizing sole source or limited competition. No agency will be delegated the ability to declare an emergency as defined in section 67-5711B, Idaho Code. (v) The permanent building fund advisory council may elect to audit any project for compliance with applicable codes and policies.

(vi) The delegated state agency will use standard documents for professional services contracts and for construction contracts as adopted by the division of public works.

(vii) Delegation is subject to cancellation by the administrator for the division of public works with the concurrence of the permanent building fund advisory council.

67-5711C. CONSTRUCTION OF PUBLIC PROJECTS -- COMPETITIVE SEALED BIDDING.

(1) All construction contracts for public works shall be awarded to the lowest responsible and responsive bidder after receipt of competitive sealed bidding except as otherwise provided in sections 67-5711B, 67-5711D and 67-5713, Idaho Code.

(2) An invitation for bids shall be issued and shall include a project description and all contractual terms and conditions applicable to the public works.

(3) Adequate public notice of the invitation for bids shall be given at least fourteen (14) days prior to the date set forth therein for the opening of bids. Such notice shall include publication at least fourteen (14) days prior to bid opening in a newspaper of general circulation in the area where the work is located.

(4) When pregualification is deemed in the best interest of the state, competitive bidding procedures shall be open only to licensed public works contractors that meet preliminary supplemental gualifications. The solicitation for bids in a pregualified bidder public works project shall consist of two (2) stages, an initial stage for identifying pregualified contractors, either prime or specialty contractors, followed by a stage during which bid prices will be accepted only from pregualified contractors. Notice of the pregualification stage shall be given in the same manner that notice of open competitive bidding is provided. Prequalification standards must be premised upon demonstrated technical competence, experience constructing similar facilities, prior experience with the state, available nonfinancial resources, equipment and personnel as they relate to the subject project, and overall performance history based upon a contractor's entire body of work. Any request for qualifications must include the standards for evaluating the qualifications of prospective bidders. Licensed contractors desiring to be prequalified to bid on a project must submit a written response to a request for qualifications. After a review of qualification submittals, licensed contractors that meet the pregualification standards shall be notified. Thereafter, bids may be solicited from contractors that meet the pregualification standards. The department may promulgate rules or develop procedures to implement the pregualification process.

(5) Bids shall be opened publicly at the time and place designated in the invitation for bids. The amount of each bid and such other relevant

information as may be specified by rules, together with the name of each bidder, shall be entered on a record and the record shall be open to public inspection. After the time of the award all bids and bid documents shall be open to public inspection in accordance with the provisions of sections 9-337 through 9-347 and 67-5725, Idaho Code.

(6) With respect to a project having a written cost estimate of greater than twenty-five thousand dollars (\$25,000) but less than the public works limit established in section 67-5711, Idaho Code, the agency, if it does not perform the work with existing physical plant staff, must award a written contract to the lowest responsible and responsive bidder after soliciting at least three (3) documented informal bids from contractors licensed in Idaho to perform public works contracts, if reasonably available. Adequate public notice of the invitation for informal bids shall be given at least seven (7) days prior to the date set forth therein for the receipt of the informal bids. Such notice may include publication at least seven (7) days prior to bid opening in a newspaper of general circulation in the area where the work is located; or the agency may advertise the invitation for bids in appropriate trade journals, and otherwise notify persons believed to be interested in the award of a contract. Informal bids must be submitted by the contractor in writing in response to a prepared written document describing the project's scope of work in sufficient detail so as to enable a contractor familiar with such work to prepare a responsible bid. Nothing herein exempts an agency from the responsibility of utilizing formal plans and specifications if the work involves the public health or safety as described in chapters 3 and 12, title 54, Idaho Code. The agency must document receipt of the informal bids in the project file.

(7) Any personal property including goods, parts, supplies and equipment which is to be supplied or provided by a state agency for use in any public work, project, or preventive maintenance programs, whether the public work, project, or preventive maintenance program is constructed, undertaken or performed by agency in-house personnel, or by delegation pursuant to section 67-5710A, Idaho Code, or otherwise provided or supplied by the agency to a contractor, the personal property, goods, parts, supplies or equipment supplied or provided by the agency must be purchased or procured by the agency through the division of purchasing in accordance with the Idaho Code.

INSTITUTION / AGENCY AGENDA BOISE STATE UNIVERSITY

SUBJECT

Boise State University requests approval to proceed with the construction of the Student Union Building Expansion for a project budget not to exceed \$30,650,000.

REFERENCE

March 2001	Boise State University submitted an information item regarding planning for an expansion of the existing Student Union Building.
April 2001	Boise State University requested approval to proceed with the expansion of its Student Union Building for a project budget of \$23 million.
April 2001	Student fee increase request of \$25 for Student Union and Housing Operations fee to fund the expansion.
April 2002	Request to reallocate the \$25 student union and housing operations fee to matriculation to minimize the impact of the appropriated budget holdbacks.
October 2005	Request to procure architectural services not to exceed \$650,000 to complete project plans through design development for the Student Union Building.

APPLICABLE STATUTE, RULE OR POLICY

Idaho State Board of Education Governing Policies & Procedures, Sections B.8-9 and V.K.

Sections 33-3804, 67-5710, 67-5710A, 67-5711-C, Idaho Code

BACKGROUND

This project encompasses an addition to and remodel of the Student Union Building. In 2001 the State Board of Education approved the project to expand the facility. In October of 2005 the State Board approved a request to complete design development. In January 2006 Lombard Conrad Architects of Boise, Idaho and MHTN Architects of Salt Lake City, Utah were selected to design the addition and remodel. In March 2006 Kreizenbeck Construction of Boise, Idaho, was chosen as the construction manager. In October 2006 design development was completed.

INSTITUTION / AGENCY AGENDA BOISE STATE UNIVERSITY - continued

DISCUSSION

The SUB expansion project will provide for expanded food service, bookstore, meeting rooms and other student activity spaces consistent with the Board's project approval in April 2001 and October 2005. The project includes the addition of 66,000SF of new space and also includes renovation of approximately 27,000SF of the current Student Union Building.

Consistent with the October 2005 Board approval, the design work through design development phase has been completed, including detailed programming, complete design development documents and two reconciled cost estimates which support the total project cost of \$30,650,000.

Due to increases in the cost of construction since the project was first conceptualized in 2001, significant design effort was expended to address the most pressing needs outlined in the earlier project approval within the funds available from financing supported by the approved strategic facilities fee. While the earlier assessment of the needs projected a 99,000SF addition, the current developed plan encompasses 66,000SF of new space and the renovation of 27,000SF to meet a majority of the needs originally outlined. The final design includes the following elements:

- Expansion of Table Rock Café (residence hall dining) adding approximately 150 seats
- Expansion of the Bookstore adding approximately 9,440 nasf
- New ballroom adding approximately 10,540 nasf
- Expansion of student services organization space approximately 4,800 nasf
- Study lounge spaces adding approximately 8,000 nasf
- Expansion of programming/common space associated with first floor dining space adding approximately 4,200 sf
- New patio performance and seating area with adjacent green space
- Entrances which are oriented toward the new parking structure and the Student Recreation Center.

Forecasts of continuing volatility in the construction market present significant risk to the project budget. Accordingly, a range of bid alternates have been identified to allow the project to bid successfully within the authorized budget amount.

NSTITUTION / AGENCY AGENDA BOISE STATE UNIVERSITY - continued

IMPACT

The project budget for the Student Union expansion totals \$30,650,000. The source of funding for this expansion project is the strategic facilities fee that will be used to service revenue bonds for this and other university capital projects The University plans to request Board approval to issue debt to finance the SUB Expansion at the next available Board meeting.

ATTACHMENTS

Attachment 1 – Project Budget	Page 5
Attachment 2 – Capital Project Tracking Sheet	Page 7
Attachment 3 – Program Summary	Page 9

STAFF COMMENTS AND RECOMMENDATIONS

This project will be funded by revenue bonds that are anticipated to be issued in early 2007, following Board approval. The bonds will be serviced with proceeds from the strategic facilities fee, which will also be used to finance other capital projects at BSU. The existing Student Union Building is operating at maximum capacity with respect to student and community groups; in fact, non-student groups which have historically been allowed to use the facility (for a fee), have been displaced because of the increasing need of students and student-related groups.

BSU staff will be present at the meeting to describe this project in greater detail.

Staff recommends approval.

BOARD ACTION

A motion to approve the request by Boise State University to proceed with construction of the Student Union Building expansion project, in conjunction with the state Division of Public Works. The motion includes funding that comes from the strategic facilities fee, and the entire cost of the project shall not exceed \$30,650,000. The University may not proceed with construction of this project prior to Board approval to issue debt.

Moved by	Seconded by	Carried Yes	No
----------	-------------	-------------	----

THIS PAGE INTENTIONALLY LEFT BLANK

Architectural & Engineering Services PROJECT BUDGET

DPW06-210									
Student Union Expansion	on								
	Dept. ID No.	886L103706							
Douglas Suddreth									
6/22/2006 rev. 9/22/06									
ory	Budget	Revised							
		\$1,929,800							
		\$40,000							
		\$1,226,300							
rsables		\$1,248,078							
		\$57,000							
		\$-							
		\$100,000							
		\$23,600,000							
		\$1,652,000							
		\$120,000							
		\$120,000							
		\$75,000							
		\$429,141							
		\$31,681							
		\$200							
		\$14,800							
		\$6,000							
Subtotal	\$0.00	\$30,650,000							
Sustou	+ 0								
	I I								
	Student Union Expansion	Student Union Expansion Douglas Suddreth 6/22/2006 rev. 9/22/06 ory Budget ory Budget student Union Expansion Student Union Expansion ory Budget Student Expansion rsables Image: Student Expansion Image: Student Expansion rsables Image: Student Expansion Image: Student Expansion Image: Student Expansion Image: Student Expansion Image: Student Expansion Image: Student Expansion Image: Student Expansion Image: Student Expansion Image: Student Expansion Image: Student Expansion Image: Student Expansion Image: Student Expansion Image: Student Expansion Image: Student Expansion Image: Student Expansion Image: Student Expansion Image: Student Expansion Image: Student Expansion Imag							

 Total
 \$0.00
 \$
 30,650,000

THIS PAGE INTENTIONALLY LEFT BLANK

Office of the Idaho State Board of Education Capital Project Tracking Sheet As of November 2006

History Narrative

2 3	Institution/Agency: Project Description: Project Use: Project Size:	Thi hal bal Ade	I dining Iroom, p ditional	roc roc batic spa	ncom om, t o peri ce fo	passes bookstore formance or studen	e, st e and ts, fa	expa ude d se acult	nt organizaito ating area, ar y, staff and co	mc on nd om	tudent Union E odel of the Stu space, study the relocation/ munity organi et to the existin	ide loi /exj zat	nt Union Buil unges and fi pansion of th ions and eve	lding irst e W ents	g. It includes floor retail c /omen's Cen	the linir	ng space, as		
6						Sou	rces	of	Funds						Use of	Fu	nds		
7						000		011	undo		Total			Us	se of Funds	14			Total
8			PBF			ISBA			Other *		Sources		Planning		Const		Other		Uses
9	October 2005 - Planning	\$		-	\$		-	\$	650,000	\$	650,000	\$	650,000					\$	650,000
10	November 2006 - Construction							\$	30,000,000	\$	30,000,000	\$	1,319,800	\$	27,883,378	79	6822	\$ 3	0,000,000
11																			
12																			
13																			
14																			
14																			
15																			
16																			
17																			
17																			
18																			
19																			
20																			
21																			
	Total Project Costs	\$		-	\$			\$	30,650,000	¢	30,650,000	¢	1,969,800	¢	27,883,378	¢	796,822	\$ 2	0,650,000
		φ		-	φ		-	ψ	30,030,000	φ	5 50,050,000	φ	1,909,000	φ	21,000,010	φ	190,022	φΟ	0,000,000
23											* 04h a # 0 4		of Frinds						
24 25								- I-			* Other Source	ces	s of Funas		•		Total		
									nstitutional		Student		-		Total		Total		
	History of Funding:		PBF			ISBA			Funds		Revenue		Other		Other		Funding		
	October 2005 - Planning	\$		-	\$		-	\$	650,000					\$	650,000		650,000		
	November 2006 - Construction										30,000,000			\$	30,000,000	\$	30,000,000		
28																			
29 30	Total	\$		-	\$		-	\$	650,000	\$	30,000,000	\$	-	\$	- 30,650,000	\$	- 30,650,000		
30	iulai	ψ		-	ψ		-	ψ	000,000	φ	50,000,000	ψ	-	ψ	30,030,000	ψ	30,030,000		

THIS PAGE INTENTIONALLY LEFT BLANK
Attachment 3

Boise State University Student Union Expansion Program Summary 10-31-2006

Space Descript	ion	Sq. Ft.					
First Floor Bookstore							
	Includes Retail Offices & Support	9,440					
Food Service	Includes Kitchen, Dining & Support	17,330					
Study / Lounge	Areas	4,400					
	<u>es / Grossing Factor / Circ.</u> cludes Restrooms, Stairs, M.E. Loading Dock	9,580 40,750 1ST FLOOR TOTAL					
Second Floor							
<u>Ballroom</u>	Ballroom Proper and Support	10,540					
Student Service	<u>26</u>	4,800					
Study Lounges	/ Ballroom Prefunction	3,780					
Building Serive	s / Grossing Factor / Circ.	6,005 25,125 2ND FLOOR TOTAL					
	[65,875 OVERALL BUILDING TOTAL					

REFERENCE: APPLICABLE STATUTE, RULE, OR POLICY

Idaho State Board of Education GOVERNING POLICIES AND PROCEDURES SECTION: V. FINANCIAL AFFAIRS B. Budget Policies

April 2002

8. Major Capital Improvement Project -- Budget Requests

For purposes of Item 8., the community colleges (NIC and CSI), the State Historical Society, and the State Library are included, except as noted in V.B.8.b. (2).

a. Definition

A major capital improvement is defined as the acquisition of an existing building, construction of a new building or an addition to an existing building, or a major renovation of an existing building. A major renovation provides for a substantial change to a building. The change may include a remodeled wing or floor of a building, or the remodeling of the majority of the building's net assignable square feet. An extensive upgrade of one (1) or more of the major building systems is generally considered to be a major renovation.

b. Preparation and Submission of Major Capital Improvement Requests

(1) Permanent Building Fund Requests

Requests for approval of major capital improvement projects to be funded from the Permanent Building Fund are to be submitted to the Office of the State Board of Education on a date and in a format established by the executive director. Only technical revisions may be made to the request for a given fiscal year after the Board has made its recommendation for that fiscal year. Technical revisions must be made prior to November 1.

(2) Other Requests

Requests for approval of major capital improvement projects from other fund sources are to be submitted in a format established by the executive director. Substantive and fiscal revisions to a requested project are resubmitted to the Board for approval. This subsection shall not apply to the community colleges.

c. Submission of Approved Major Capital Budget Requests

The Board is responsible for the submission of major capital budget requests for the institutions, school and agencies under this subsection to the Division of Public Works. Only those budget requests which have been formally approved by the Board will be submitted by the office to the executive and legislative branches.

9. Approval by the Board

Requests for approval of major capital improvement projects must be submitted for Board action. Major capital improvement projects, which are approved by the Board and for which funds from the Permanent Building Fund are requested, are placed in priority order prior to the submission of major capital budget requests to the Division of Public Works.

REFERENCE: APPLICABLE STATUTE, RULE, OR POLICY

Idaho State Board of Education GOVERNING POLICIES AND PROCEDURES SECTION: V. FINANCIAL AFFAIRS K. Construction Projects

April 2002

K. Construction Projects

1. Major Project Approvals - Proposed Plans

Without regard to the source of funding, before any institution, school or agency under the governance of the Board begin formal planning to make capital improvements, either in the form of renovation or addition to or demolition of existing facilities, when the cost of the project is estimated to exceed five hundred thousand dollars (\$500,000), must first be submitted to the Board for its review and approval. All projects identified on the institutions', school's or agencies' six-year capital plan must receive Board approval.

2. Project Approvals

Without regard to the source of funding, proposals by any institution, school or agency under the governance of the Board to make capital improvements, either in the form of renovation or addition to or demolition of existing facilities, when the cost of the project is estimated to be between two hundred fifty thousand dollars (\$250,000) and five hundred thousand dollars (\$500,000), must first be submitted to the executive director for review and approval. Without regard to the source of funding, proposals by any institution, school or agency under the governance of the Board to make capital improvements, either in the form of renovation or addition to or demolition of existing facilities or construction of new facilities, when the cost of the project is estimated to exceed five hundred thousand dollars (\$500,000), must first be submitted to the Board for its review and approval. Project cost must be detailed by major category (construction cost, architecture fees, contingency funds, and other). When a project is under the primary supervision of the Board of Regents or the Board and its institutions, school or agencies, a separate budget line for architects, engineers, or construction managers and engineering services must be identified for the project cost. Budgets for maintenance, repair, and upkeep of existing facilities must be submitted for Board review and approval as a part of the annual operating budget of the institution, school or agency.

3. Fiscal Revisions to Previously Approved Projects

Project revisions that substantially alter the use of the project causing changes in project costs between two hundred fifty thousand dollars (\$250,000) and five hundred thousand dollars (\$500,000) must first be submitted to the executive director for review and approval. Changes in project costs of more than five hundred thousand dollars (\$500,000) must first be submitted to the Board for its review and approval. Requests must be supported by a revised detailed project budget and justification for changes.

4. Project Acceptance

Projects under the supervision of the Department of Administration are accepted by the Department on behalf of the Board and the state of Idaho. Projects under the supervision of an institution, school or agency are accepted by the institution, school or agency and the project architect. Projects under the supervision of the University of Idaho are accepted by the University on behalf of the Board of Regents.

- 5. Statute and Code Compliance
 - a. All projects must be in compliance with Section 504 of the Rehabilitation Act of 1973 and must provide access to all persons. All projects must be in compliance with applicable state and local building and life-safety codes and applicable local land-use regulations as provided in Chapter 41, Title 39, and Section 67-6528, Idaho Code.
 - b. In designing and implementing construction projects, due consideration must be given to energy conservation and long-term maintenance and operation savings versus short-term capital costs.

REFERENCE – APPLICABLE STATUTE, RULE OR POLICY – continued

Idaho Statutes

TITLE 33 EDUCATION CHAPTER 38 STATE INSTITUTIONS OF HIGHER EDUCATION BOND ACT

33-3804. POWERS AND DUTIES OF STATE INSTITUTIONS.

Every institution shall have power in its proper name as aforesaid:

- (a) To have a corporate seal and alter the same at pleasure;
- (b) To sue and be sued;
- To acquire by purchase, gift or the exercise of the right of eminent domain and hold and dispose of real or personal property or rights or interests therein and water rights;
- (d) To make contracts and to execute all instruments necessary or convenient;
- (e) To acquire any project or projects, and to own, operate, and maintain such project;
- (f) To accept grants of money or materials or property of any kind from a federal agency, upon such terms and conditions as such federal agency may impose;
- (g) To borrow money, with or without the issuance of bonds and to provide for the payment of the same and for the rights of the holders of such bonds and/or of any other instrument of such indebtedness, including the power to fix the maximum rate of interest to be paid thereon and to warrant and indemnify the validity and tax exempt character;
- (h) To perform all acts and do all things necessary or convenient to carry out the powers herein granted, to obtain loans or grants or both from any federal agency, and to accomplish the purposes of sections 33-3801--33-3813, Idaho Code, and secure the benefits of the Recovery Act;
- To issue refunding bonds, for the purpose of paying, redeeming, or refunding any (i) outstanding bonds theretofore issued under authority of this chapter. Refunding bonds so issued shall have such details, shall bear such rate or rates of interest and shall be otherwise issued and secured as provided by the board authorizing the issuance of such bonds and as otherwise provided in this chapter, provided, however, that such changes in the security and revenues pledged to the payment thereof may be made by such board as may be provided by it in the proceedings authorizing such bonds, but in no event shall such refunding bonds ever be secured by revenues not authorized by this chapter to be pledged to the payment of bonds issued for other than refunding purposes. Refunding bonds issued hereunder may be exchanged for a like principal amount of the bonds to be refunded, may be sold in the manner provided in this chapter for the sale of other bonds, or may be exchanged in part and sold in part. If sold, the proceeds of such bonds may be deposited in escrow for the payment of the bonds to be refunded, provided such bonds mature or are callable for redemption under their terms within six (6) months from the date of the delivery of the refunding bonds.

No refunding bonds may be issued hereunder in a principal amount in excess of the principal amount of the bonds to be refunded nor may any bonds not maturing or callable for redemption under their terms as above provided be refunded hereunder without the consent of the holders thereof. Refunding bonds so authorized and issued may in the discretion of the board be combined with other bonds to be authorized and issued under this chapter, and a single issue of bonds may be so authorized in part for improvement and in part for refunding purposes.

(j) In connection with borrowing without the issuance of bonds, to fix fees, rents or other charges for utilization of any facility or project being financed by said borrowing and to pledge the same, together with any other revenue from such project or facility, as collateral for repayment of principal and interest in the same manner and to the same extent as provided in this chapter for securing the payment of bonds issued pursuant to this chapter.

REFERENCE – APPLICABLE STATUTE, RULE OR POLICY – continued

Idaho Statutes TITLE 67 STATE GOVERNMENT AND STATE AFFAIRS CHAPTER 57 DEPARTMENT OF ADMINISTRATION

67-5710. PERMANENT BUILDING FUND ADVISORY COUNCIL -- APPROVAL OF USE OF FUND -- DUTIES OF ADMINISTRATOR OF PUBLIC WORKS.

There is hereby created in he division of public works a permanent building fund advisory council which shall be appointed by the governor. This council shall be composed of one (1) member of the senate, one (1) member of the house of representatives, a citizen engaged in the contracting business, a citizen engaged in the banking business, and a citizen who is a member of the business community not engaged in contracting or banking. The senate member and house of representative member shall be appointed for a fixed term of two (2) years. All other council members shall be appointed for a fixed term of three (3) years. The terms of office of members of the council holding office prior to July 1, 1996, shall expire on the following dates: contracting business member on July 1, 1996; senate member and house of representative member on December 1, 1996; business community member on July 1, 1997; and banking member on July 1, 1998. On or after July 1, 1996, the governor shall appoint members of the council as terms of existing members expire. All members of the council shall serve at the pleasure of the governor. The administrator of public works and the responsible heads of the agencies for which appropriations for construction, renovations, remodelings or repairs are made pursuant to chapter 11, title 57, Idaho Code, shall consult, confer and advise with the permanent building fund advisory council in connection with all decisions concerning the administration of these appropriations and the planning and construction or execution of work or works pursuant thereto. The approval of the permanent building fund advisory council shall be a condition precedent to the undertaking of planning or construction.

The administrator of public works is hereby directed to work in close cooperation with the responsible heads of institutions and agencies for which appropriations are made herein and no building proposals shall be approved by the administrator of public works nor any planning or work undertaken by that officer pursuant to these appropriations without the prior approval of the responsible chief officer of the institutions and agencies for whom appropriations are made herein. **67-5710A**. REQUIREMENT OF PLANS AND SPECIFICATION APPROVAL BY PERMANENT

BUILDING FUND ADVISORY COUNCIL AND DELEGATION OF PROJECT OVERSIGHT BY THE

ADMINISTRATOR FOR THE DIVISION OF PUBLIC WORKS.

(1) (a) Unless an emergency exists as defined in section 67-5711B, Idaho Code, an existing public works may not be altered, repaired, constructed or improved on property owned or occupied by any state institution, department, commission, board or agency, if the estimated cost of work exceeds the limit established in section 67-5711, Idaho Code, and except for those institutions and agency exemptions listed in section 67-5711, Idaho Code, without regard to source of funding, until the location, design, plans and specifications are approved by the permanent building fund advisory council and the project supervised by the division of public works or its designee.

(b) Facilities to be built with funds under the control of a nonstate entity, and owned or occupied by state entities, must have plans and specifications prepared, and all plans and specifications must be reviewed and approved by the permanent building fund advisory council prior to the advertising, bidding, construction and/or negotiation for construction of the facilities.

(2) (a) The administrator for the division of public works may delegate control over design, construction and all other aspects of a public works or maintenance project which costs less than one hundred fifty thousand dollars (\$150,000), to agencies of state government on a project-by-project basis, if a responsible party of the state agency requests that delegation in writing and the permanent building fund advisory council approves the delegation.

(i) The state agency to whom control is delegated shall assume all responsibility for project budgets and shall receive funds appropriated for the project upon application and approval by the permanent building fund advisory council.

(ii) Delegation of project control does not exempt the state agency from complying with public works statutes, life safety and building codes or other applicable codes and regulations. The state agency also must comply with any guidelines or procedures for design and construction adopted by the division of public works and the permanent building fund advisory council.

(iii) State agencies that receive delegated projects may not have access to permanent building fund advisory council contingency funds unless approved by the permanent building fund advisory council or authorized by appropriation.

(iv) Prior, written approval from the administrator must be granted for any public works utilizing sole source or limited competition. No agency will be delegated the ability to declare an emergency as defined in section 67-5711B, Idaho Code. (v) The permanent building fund advisory council may elect to audit any project for compliance with applicable codes and policies.

(vi) The delegated state agency will use standard documents for professional services contracts and for construction contracts as adopted by the division of public works.

(vii) Delegation is subject to cancellation by the administrator for the division of public works with the concurrence of the permanent building fund advisory council.

67-5711C. CONSTRUCTION OF PUBLIC PROJECTS -- COMPETITIVE SEALED BIDDING.

(1) All construction contracts for public works shall be awarded to the lowest responsible and responsive bidder after receipt of competitive sealed bidding except as otherwise provided in sections 67-5711B, 67-5711D and 67-5713, Idaho Code.

(2) An invitation for bids shall be issued and shall include a project description and all contractual terms and conditions applicable to the public works.

(3) Adequate public notice of the invitation for bids shall be given at least fourteen (14) days prior to the date set forth therein for the opening of bids. Such notice shall include publication at least fourteen (14) days prior to bid opening in a newspaper of general circulation in the area where the work is located.

(4) When pregualification is deemed in the best interest of the state, competitive bidding procedures shall be open only to licensed public works contractors that meet preliminary supplemental gualifications. The solicitation for bids in a pregualified bidder public works project shall consist of two (2) stages, an initial stage for identifying pregualified contractors, either prime or specialty contractors, followed by a stage during which bid prices will be accepted only from pregualified contractors. Notice of the pregualification stage shall be given in the same manner that notice of open competitive bidding is provided. Prequalification standards must be premised upon demonstrated technical competence, experience constructing similar facilities, prior experience with the state, available nonfinancial resources, equipment and personnel as they relate to the subject project, and overall performance history based upon a contractor's entire body of work. Any request for qualifications must include the standards for evaluating the qualifications of prospective bidders. Licensed contractors desiring to be prequalified to bid on a project must submit a written response to a request for qualifications. After a review of qualification submittals, licensed contractors that meet the pregualification standards shall be notified. Thereafter, bids may be solicited from contractors that meet the pregualification standards. The department may promulgate rules or develop procedures to implement the pregualification process.

(5) Bids shall be opened publicly at the time and place designated in the invitation for bids. The amount of each bid and such other relevant

information as may be specified by rules, together with the name of each bidder, shall be entered on a record and the record shall be open to public inspection. After the time of the award all bids and bid documents shall be open to public inspection in accordance with the provisions of sections 9-337 through 9-347 and 67-5725, Idaho Code.

(6) With respect to a project having a written cost estimate of greater than twenty-five thousand dollars (\$25,000) but less than the public works limit established in section 67-5711, Idaho Code, the agency, if it does not perform the work with existing physical plant staff, must award a written contract to the lowest responsible and responsive bidder after soliciting at least three (3) documented informal bids from contractors licensed in Idaho to perform public works contracts, if reasonably available. Adequate public notice of the invitation for informal bids shall be given at least seven (7) days prior to the date set forth therein for the receipt of the informal bids. Such notice may include publication at least seven (7) days prior to bid opening in a newspaper of general circulation in the area where the work is located; or the agency may advertise the invitation for bids in appropriate trade journals, and otherwise notify persons believed to be interested in the award of a contract. Informal bids must be submitted by the contractor in writing in response to a prepared written document describing the project's scope of work in sufficient detail so as to enable a contractor familiar with such work to prepare a responsible bid. Nothing herein exempts an agency from the responsibility of utilizing formal plans and specifications if the work involves the public health or safety as described in chapters 3 and 12, title 54, Idaho Code. The agency must document receipt of the informal bids in the project file.

(7) Any personal property including goods, parts, supplies and equipment which is to be supplied or provided by a state agency for use in any public work, project, or preventive maintenance programs, whether the public work, project, or preventive maintenance program is constructed, undertaken or performed by agency in-house personnel, or by delegation pursuant to section 67-5710A, Idaho Code, or otherwise provided or supplied by the agency to a contractor, the personal property, goods, parts, supplies or equipment supplied or provided by the agency must be purchased or procured by the agency through the division of purchasing in accordance with the Idaho Code.

INSTITUTION / AGENCY AGENDA IDAHO STATE UNIVERSITY

SUBJECT

Boise State University, Idaho State University, and University of Idaho request approval to proceed with the lease agreement and construction of the Center for Advanced Energy Studies (CAES) building in cooperation with the Idaho National Laboratory.

REFERENCES

April 2005	Overview of CAES at SBOE regular meeting
August 2005	Update information on CAES at SBOE regular meeting
December 2005	Approved motion to issue a Request for Proposal (RFP) through the Division of Public Works (DPW) for selection of an architect and proceed with the pre-design/programming for the CAES Facility.

APPLICABLE STATUTE, RULE, OR POLICY

Idaho State Board of Education Governing Polices & Procedures, Sections V.B.8 and V.K.

BACKGROUND

The Battelle Energy Alliance (BEA) was awarded the contract from the Department of Energy to manage the new Idaho National Laboratory (INL) for nuclear energy research. As part of the contract, Battelle committed to establish CAES which is designed to become a nationally and internationally recognized focal point for the advancement of education in energy science and technology. A Memorandum of Agreement was signed by the Governor, the University Presidents, and Battelle Memorial Institute to work towards the establishment of a joint laboratory/university center which would serve as a research center for the INL with the construction of a State-owned building to house the CAES.

At the December 1, 2005 SBOE meeting, a motion was approved to allow ISU to issue a Request for Proposal (RFP) through the Division of Public Works for selection of an architect to design the Center for Advanced Energy Studies facility, and proceed with the pre-design/programming. Through the bid process, Big-D Construction was selected as the design/build contractor.

DISCUSSION

The CAES facility is planned to open in June 2008 and will have a minimum of 55,000 square feet. To satisfy the BEA requirement for an operating lease as opposed to a capital lease, the lease agreement must show that BEA has use of

INSTITUTION / AGENCY AGENDA IDAHO STATE UNIVERSITY - continued

70% of the building space. This 70% includes common space (hallways, restrooms, conference rooms, maintenance, foyers, shared laboratories, and office space for staff joint appointments). BEA has agreed to cover the occupancy costs for 70% of the building. The three universities will share the remaining space on a per use basis. The facility is envisioned to be a two-story, structural steel building with a brick façade and will be located on state property north of the ISU/UI Center for Higher Education. When fully occupied, the CAES facility will house a total of up to 175 people, including approximately 100 faculty, researchers, and staff; 50 graduate students; and 25 undergraduate students. The interior design will be cooperatively planned among all the users with the architect.

IMPACT

The design and construction of the facility is estimated to cost approximately \$17 million. Funding for the facility is based on: (1) \$4.9 million (\$100,000 previously expended) from the INEEL Settlement Fund, as defined in the Idaho Code 67-806A, for use according to the terms of the agreement for the construction of the Center for Science and Technology in Idaho Falls (dated June 29, 2001), between the Office of the Governor of the State of Idaho and the Regents of the University of Idaho and the Trustees of Idaho State University; (2) \$1,942,756 from two grants from the U.S. Department of Housing and Urban Development (HUD) to the University of Idaho: HUD Grant B-00-SP-ID-0116 in the amount of \$925,000 and HUD Grant B-01-SP-ID-0172 in the amount of \$1,017,756 for use according to the terms of the grant (approximately \$300,000 has been expended from one of the grants for preliminary designs for the previous CST facility); and (3) \$10 million through the issuance of bonds, using ISU's bond capacity, to be retired over 20 years supported by rent paid by BEA for occupancy of approximately 70% of the CAES facility.

The BEA lease payments will be equivalent to the amount required to amortize the bond. Repayment is being guaranteed by Battelle Memorial Institute (\$8.8 million) and Washington Group International (\$1.2 million). It is anticipated that the universities will make a subsequent request for additional funds from the INEEL Settlement Fund to further support this project. The amount is yet to be determined.

Cost of infrastructure (roadway, utilities, and parking) is estimated to be approximately \$2.5 million, and will be funded by a federal appropriation to install a utility corridor from Fremont Avenue to the new building location north of the railroad tracks. This funding was in the House Energy and Water Bill as a plusup. It is envisioned that the utility corridor would have all the necessary utilities, communications and a roadway.

INSTITUTION / AGENCY AGENDA IDAHO STATE UNIVERSITY - continued

Maintenance and occupancy costs will be covered according to the assignment of space with 70% of the total M&O being paid as service rent by BEA as part of the rental agreement and the three universities paying their respective shares. The specific amounts for the three universities will be worked out later when actual shared spaces are defined.

Each organization occupying the building will provide furniture and laboratory equipment for the spaces they occupy. The universities will cooperatively determine their equipment needs based on their specific programs and opportunities for sharing equipment and space. Grants and donations will be used to cover the costs of the equipment and furnishings.

ATTACHMENTS

Attachment 1:	Capital Project Tracking Sheet
Attachment 2:	Business Plan
Attachment 3:	CAES Lease Agreement
Attachment 4:	DOE Lease Letter
Attachment 5:	First Amendment

STAFF COMMENTS AND RECOMMENDATIONS

This project is complex in terms of concept, scope and financing because it necessarily involves the collaboration of various parties. As noted under "Background", CAES is a component of BEA's commitment as prime contractor to the U.S. Department of Energy at the Idaho National Laboratory. Idaho State University is the lead educational institution for purposes of financing the facility and gathering the other two universities in Idaho to participate in the programmatic activities of CAES.

The Board has had several previous presentations on this project. The first two were information-only; on December 1, 2005, the Board provided approval for ISU to select an architect and proceed with pre-design/design development, with a construction estimate, in conjunction with the state Division of Public Works. The allowable expenditure for that component was approximately \$1.7 million. ISU has expended approximately \$625,000 to date for design-related costs, from one of the two HUD grants that were originally obtained by the University of Idaho and transferred to ISU. The second HUD grant must be used for construction only.

ISU has submitted a draft Program Planning Guide (PPG) to the Board office for review. Following approval of a final PPG document, the Office of the Governor will be requested to release \$4.9 million from the INEEL Settlement fund, which ISU can immediately use to continue design and begin construction.

INSTITUTION / AGENCY AGENDA IDAHO STATE UNIVERSITY - continued

A Business Plan has been provided as part of this agenda, and does not require Board approval. Staff has reviewed the Business Plan, and has discussed what appear to be shortcomings in the plan. ISU officials suggest the plan is a 'living document', and will continue to be revised as time goes on. However, because this document outlines the purpose of, need for, and organization of the project, the document should be more focused upon the programmatic aspects of the project and less on the construction/facility details.

Following Board approval, ISU will issue bonds in the amount of \$10 million, the payment for which will become the responsibility of BEA under the lease agreement. Repayment is guaranteed by the Battelle Memorial Institute and Washington Group International.

University officials will be available at the meeting to discuss additional details of this project with the Board.

Staff recommends approval, with the comments noted.

BOARD ACTION

A motion to approve the request by Idaho State University to proceed with financing and construction of the Center for Advanced Energy Studies in Idaho Falls and to authorize the Division of Public Works to award a design-build contract to Big-D Construction, the selected design-build team. The motion includes funding that comes from a combination of INEEL Settlement funds, U.S. Department of Housing and Urban Development grants, and bond proceeds, and the entire cost of the project shall not exceed \$17.0 million.

In addition, the motion provides approval for the lease agreement of the CAES building in cooperation with the Idaho National Laboratory.

Moved by _____ Seconded by _____ Carried Yes ____ No ____

Office of the Idaho State Board of Education Capital Project Tracking Sheet

As of: 26-Oct-06

History Narrative

		facility Center researc	cility is will pro for Hig chers, s	for a omote gher l staff,	a cooper e, perfor Educatio	ative c m, anc n at Ui duate a	d rev nive	er with Battel vitalize resea rsity Place ir	le Energy A Irch disciplir Idaho Falls	Allian nes. s. It	ce It wil	and the thr will be loca I ultimately	ated on state house a total	ersit own of 1	ies, BSU, ISI ed land north 75 people, ir	U, and UI. The h of the ISU/UI ncluding faculty, ct will be under
3	-															other national nd other events.
4	-		•	•			-	FY2008 and fifices and la					re feet with Ba	ittelle	ə having 709	% of the space.
6					Sou	rces o	f Ei	Inde					Use d	of Er	unde	
7					300			111113	Total				Use of Fund		1105	Total
8		PE	BF		ISBA			Other *	Sources	5	Р	Planning	Const	3	Other*	Uses
9	Initial Cost of Project	\$	-	\$		- \$		16,562,756				1,680,000	\$ 13,362,756	5 \$		\$ 16,562,756
10								-,,	+ - , , -			,,			,,	+ -,,
11 12	History of Revisions:															
	Proposed Revision	\$		\$												\$-
		φ	-	φ		-										φ -
14																
15																
16	*Estimated Bond issuance cost.															
17																
18																
19																
20																
21																
22	Total Project Costs	\$	-	\$		- \$;	16,562,756	\$16,562,75	56	\$	1,680,000	\$ 13,362,756	5\$	1,520,000	\$ 16,562,756
23																•
24 25									* Other So Student		s o	of Funds			Total	
	Liston, of Funding,	PE	- -					stitutional		-		Other	Total Other		Total	
	History of Funding:	<u>•</u>	ЪГ	\$	ISBA			Funds	Revenue		\$	Other 16,562,756		6 \$	Funding 16,562,756	
26 27		Φ	-	φ							φ	10,002,700	\$ 16,562,75 \$ -	ь \$	- 10,002,750	
28													Ŧ	Ψ		
29												-	-		-	
30	Total	\$	-	\$		- \$		-	\$-	-	\$	16,562,756	\$ 16,562,75	6\$	16,562,756	I

THIS PAGE INTENTIONALLY LEFT BLANK

Center for Advanced Energy Studies (CAES) Preliminary Business Plan

October 2006



Center for Advanced Energy Studies





INL/EXT-06-11508, Rev. 1

Center for Advanced Energy Studies (CAES) Preliminary Business Plan

Boise State University Idaho State University University of Idaho Idaho National Laboratory

October 2006

Center for Advanced Energy Studies Idaho Falls, Idaho 83415

Prepared for the U.S. Department of Energy Under DOE Idaho Operations Office Contract DE-AC07-05ID14517





EXECUTIVE SUMMARY

Energy challenges of the 21st century—energy demand growth with prosperity, energy security, and climate protection—require every available energy option as well as the consideration of new options. The Center for Advanced Energy Studies (CAES) is a collaborative initiative by the State of Idaho and its academic institutions and the federal government through the U.S. Department of Energy (DOE) and the Idaho National Laboratory (INL) that will serve as a nexus to advance energy security for our nation. CAES will expand the research portfolios of the Idaho universities in these key areas, creating new capabilities (nuclear engineering, chemical sciences, material sciences, system modeling, energy policy, and environmental sciences), and working with national laboratories, other universities and research centers, and industry for technology and workforce development.

To achieve the desired outcomes—knowledge transfer leading to a new trained energy workforce, enhanced institutional capabilities leading to scientific advancement, and end-use application resulting in economic expansion—CAES will create a unique and collaborative environment for our energy future.

CAES will provide benefits to DOE and other federal agencies, university, and industrial partners by facilitating technical collaborations. Operating as a jointly managed research center, comprised initially of four charter member institutions—Boise State University (BSU), Idaho State University (ISU), University of Idaho (UI), and INL—CAES will maximize the utilization of the energy-related capabilities of its member institutions and sponsors. Cross-organizational, peer-to-peer technical collaboration in areas of nuclear, renewable, fossil and alternative energy will be encouraged and other institutions will be invited to partner and participate as appropriate.

The CAES member institutions have established CAES under the aegis of INL in order to build on a firm foundation. The financial, legal, business and organizational mechanisms of each institution will continue to be used in support of CAES activities.

To facilitate technical activities, a new high performance building will be constructed. This facility will provide the necessary laboratories, offices, and collaborative space for the CAES research teams to accomplish their research and educational mission. The CAES member institutions and associated sponsors have secured the necessary funding to finance this building, including approximately \$7M in State of Idaho grants and \$10M via ISU bonds guaranteed by the Battelle Memorial Institute and the Washington Group International.

Finances are readily available to cover the total annual CAES building lease, maintenance and operating costs given the present INL and university operations. These costs are estimated to total \$1130K, which includes the 20-year lease payments by INL and the estimated maintenance and operating costs.

The financial viability of the CAES program which will include joint research, joint appointments, business development efforts, seminars and workshops is also projected to be quite favorable given the anticipated institutional support. Financial support from DOE, INL, and the State of Idaho will lay the foundation for future success. The ultimate success of CAES is dependent on researchers of the member institutions successfully competing for research funding in the public and private R&D arena.



This page intentionally left blank.



CONTENTS

EXEC	CUTIVE	E SUMMARYiii
1.	INTRC	DUCTION1
2.	BACK	GROUND2
3.	PROBI	LEM STATEMENT
4.	SOLU	ΓΙΟΝ DESCRIPTION
	4.1	Desired Outcomes
	4.2	Products and Services
5.	VALU	E PROPOSITION
	5.1	Public Benefit
	5.2	University Benefits
	5.3	Industry Benefits
	5.4	INL Benefits
	5.5	DOE Benefits
6.	BUSIN	IESS DESCRIPTION
	6.1	Joint Institute
	6.2	Transformation Schedule
	6.3	Responsibilities of the CAES member institutions
7.	ORGA	NIZATIONAL MODEL
	7.1	Organizational Structure
	7.2	Infrastructure and Equipment Plan
8.	MARK	ET ANALYSIS
	8.1	Energy Market Description
	8.2	Energy Investments



9.	MARK	ET PLAN
	9.1	Initial Research Themes
	9.2	Capture Strategies
10.	FINAN	CIAL PROJECTIONS
	10.1	Current Operating Scenario
	10.2	Incremental Investment Scenario – "Achieving the Vision" 19
	10.3	Projected Expenses
	10.4	Financial Analysis
APPE	ENDIX A	A – APPROVED MEMORANDUM OF AGREEMENT
APPE	ENDIX I	3 – ENERGY MARKET ANALYSIS
	B.1 En	ergy Policy Act 2005
	B.2 FY	-2007 Proposed DOE Budget
	B.3 En	vironmental Protection Agency
	B.4 Na	tional Science Foundation
	B.5 Pri	vate Foundations
	B.6 Co	mpetitors



CENTER FOR ADVANCED ENERGY STUDIES (CAES) PRELIMINARY BUSINESS PLAN

1. INTRODUCTION

This report presents the preliminary business plan for the Center for Advanced Energy Studies (CAES). Included in this plan is an introduction to the CAES program, a description of the CAES business model, an overview of its organizational structure, an analysis of the relative markets, and a summary of the initial CAES financial model, including the anticipated costs and revenue streams associated with its program and support infrastructure.

CAES will bring together Idaho academic institutions, Idaho National Laboratory (INL), and other partners to facilitate regional, national, and international research and education in energy sciences, with a focus on nuclear energy, and public policy covering all forms of energy. Operating as a jointly managed and financed center, CAES will leverage the expertise of its member institutions and will expand their capabilities by focusing on the energy challenges of the 21st Century.

CAES will maximize the utilization of its member institutions energy-related capabilities to provide technology and policy solutions and help develop the next generation workforce for the national and global energy enterprise. The founding CAES member institutions include Boise State University (BSU), Idaho State University (ISU), University of Idaho (UI), and Idaho National Laboratory (INL)^a (see Figure 1).



Figure 1. CAES will integrate the capabilities of its member institutions to expand all research and development opportunities for the mutual benefit of the institutions.

^a Battelle Energy Alliance, LLC, manages and operates the INL under Contract # DE-AC07-05ID14517 to the U.S. Department of Energy.



2. BACKGROUND

National and global energy challenges of the 21st century—energy demand growth with prosperity, energy security, and climate protection—require development of policies and technologies to provide the required breadth of energy options. The Center for Advanced Energy Studies (CAES) is a critical resource in the national effort to respond to these challenges.

Meeting today's energy demands will require developing every available energy option and creation of new options. Acceptable energy options for the 21st century must be economical, abundant, secure, and environmentally friendly. Each option presents challenges. Coal, while abundant and cheap in the U.S., requires the ability to capture and sequester greenhouse gasses (GHG) and appropriately eliminate or capture and acceptably dispose of pollutants. Oil and natural gas are in limited supply globally; the growing demand is straining already volatile supply lines, and these sources have climate implications-transportation is the single fastest growing source of CO₂ emissions globally. Renewable energy systems including biomass have attractive security and environmental attributes but have intermittency problems, and have not been deployed at strategic levels at acceptable costs. Nuclear energy has the potential to provide a virtually limitless supply of secure energy without carbon emissions, but new generations of nuclear power must meet cost, proliferation, security, environmental, and safety requirements in a socially, politically, and technically sustainable manner, or they may not be supportable.

In 2002, the U.S. Department of Energy (DOE) consolidated its capabilities and created Idaho National Laboratory (INL). This new

laboratory has been charged with the missions of leading the national and global agenda for nuclear energy development, providing a broad set of energy security options, and applying resources to national and homeland security needs. The INL energy mission covers all sources of energy and features a leadership role in nuclear energy. INL is charged with putting in place the programs, people, capabilities, partnerships and depth, and recognition in signature technologies for energy and other applications. This mission requires joint planning with a national and international constituency of national laboratory, academic, and industrial partners. It also requires communicating energy issues and the role of solutions such as nuclear in the context of national and global challenges to the public, to provide a foundation for the decisions and investments that must be made.

CAES is a public/private partnership by the State of Idaho and its academic institutions and the federal government through INL and DOE. With the 21st century energy challenges and opportunities, Idaho has the prospect to serve as a nexus to advance energy security for our nation. This leadership role provides a technical core for the State for decades to come. CAES will provide research, policy analysis, a venue for the energy dialogue, and education and training for the future energy workforce focused intensely on energy issues and priorities. INL and the Idaho academic community will take leadership roles in energy science and technology critical for the future of the nation, expanding the research portfolios of the universities in these key areas, creating new capabilities, and working with national laboratories, industry, and other partners in technology and workforce development.



3. PROBLEM STATEMENT

Critical energy challenges of growth, security, and climate protection create a need for action. World energy demands are at an all time high and are expected to triple by 2050. Never in human history has energy become so important to the welfare of humanity. National security is linked to energy through the dependence of this country and many others on imported oil—much of it located in politically troubled parts of the globe. Because of this dependence, large-scale failures in the global production and distribution system present a real domestic energy threat. Achieving a secure, affordable, and sustainable energy supply mix with minimal environmental impact is essential to meeting U.S. energy security and economic stability needs. Since the major growth in energy demand in this century will occur within developing nations, technology and policy solutions must support a burgeoning global energy demand in ways that support economic prosperity, stability, and stewardship of the planet. Energy independence for the USA can only occur by working from these foundations and in the global context provide technologies that support sustainable development.

Solutions to today's critical energy issues require substantial research and development (R&D), technological advancements, and renewed educational emphasis in science and engineering. The global energy expansion of the 21st century must be built on a new generation of energy concepts and technologies and a greatly expanded and more technically sophisticated workforce. CAES will serve as a resource to assist in determining a national energy research agenda that will provide for technology and policy advancement and workforce development. At the same time, CAES will provide a venue for its member institutions to jointly provide a new generation of capabilities to respond to these challenges.

BSU, ISU, UI, and INL with their talented professionals and researchers, have achieved numerous institutional successes. These institutions, however, are each faced with the challenge of limited resources in terms of technical staff, equipment, and funding. The competition for limited R&D funds, talented students, and trained workforce is strong, with federal laboratories, national universities, private industry, and non-profit institutions all aggressively competing. As a result of these limitations and challenges, these institutions are often unable to adequately compete for larger multi-disciplinary R&D programs given only their individual capabilities. New and innovative interdisciplinary teaming approaches are required to broaden the available capabilities so as to more successfully compete on a national and international level. The expanded partnership provided by the formation of CAES will allow each of these institutions to expand the core programs within their own institutions to create the skills, capabilities, and results needed for success.



4. SOLUTION DESCRIPTION

CAES is a public/private partnership for achieving results that member institutions may take decades to achieve individually.

Functionally, CAES consists of a series of technical capabilities focused in four energy research areas as illustrated in Figure 2. By focusing the collective efforts and investments of the four institutions on the energy-related challenges of the intermountain region, the nation, and the world, CAES member institutions will create a unique and collaborative environment for the energy future.

- At the most fundamental level, CAES provides an intellectual venue intensely focused on regional, national and global energy issues and challenges. This focus provides a common forum to support the energy dialogue, engage the world's leaders in energy policy and technology, and jointly carry out the planning that will help set directions and priorities for the global energy agenda.
- b) The CAES facility will allow colocation of researchers, educators, visitors, and other collaborators. The facility will provide state-of-the-art laboratories with instrumentation to provide new capabilities to CAES researchers and its affiliate partners.
- c) CAES will serve as a conduit to the capabilities of INL (i.e., Advanced Test

Reactor, Actinide Chemistry Laboratory, etc.) and the other member institutions, their partners, and to the broader capabilities infrastructure of the DOE national enterprise. These capabilities will include national technical user facilities, computation and information facilities nationally and globally, and engineering testbeds for emerging energy technologies in nuclear and other energy arenas.

- d) CAES will provide an integrated education and training environment whereby the academic member institutions and others can engage the emerging workforce in the issues and concepts of the energy industry.
- e) CAES will provide the administrative framework for joint appointments to improve the ease of movement of people and skill sets within the member institutions, and provide attractive new positions for the acquisition of new skills and talent.
- f) CAES will enable the development of joint proposals, hosted by appropriate member institutions, to allow a transparent integration of staff skills and institutional capabilities to allow the member institutions to grow into new technical and scientific areas.



Figure 2. CAES consists of a series of functional capabilities focused in four energy research areas.



- g) Through the co-location of several INL centers-initially the Center for Advanced Modeling and Simulation (CAMS) and the Center for Space Nuclear Research (CSNR)—CAES will provide a valuable synergism for the development and application of new capabilities. The unique capabilities of the INL Distinctive Scientific Signatures (Material & Nuclear Fuel Sciences; Instrumentation, Controls, & Intelligent Systems; Theory, Modeling & Simulation: Separations & Actinide Sciences, and Microbiological & Geological Sciences), which are the focal point for INL investment, will have critical links within CAES in terms of capabilities, investments, and program development.
- h) CAES will provide an academic linkage to important DOE programs, such as the

Climate Change Technology Program (CCTP), the Global Nuclear Energy Partnership (GNEP), the Advanced Fuel Cycle Initiative (AFCI), and other national and international organizations (International Atomic Energy Agency, World Nuclear University, Dalton Nuclear Institute, etc.).

4.1 Desired Outcomes

As a jointly managed center, CAES will provide a number of organizational, technical, infrastructure, and programmatic resources and activities to facilitate the building of new programs and providing credible solutions to the energy customer set. Achievements will be measured in terms of Knowledge Transfer, Enhanced Capabilities and Programs, and regional Economic Expansion (see Figure 3).







4.1.1 Knowledge Transfer

Knowledge transfer involves the successful conveyance of ideas, experience, research results, and professional skills to others for the purposes of individual, organizational, or societal benefit. Knowledge can be tangible and intangible. This transfer can occur within organizations (faculty to students), across organizational lines (laboratory to university), as a business transaction (company to company), or to and from the wider community (public awareness).

CAES will support its member institutions with regard to educational development, trained professionals, multidisciplinary research, innovative products and services, and development of publications with high impact. The intellectual environment and focused energy agenda of CAES will help to further attract talented faculty and staff allowing its member institutions to attract the best students thus educating a new generation of scientists and engineers.

4.1.2 Enhanced Capabilities and Programs

Capacity building involves the development, enhancement, and promotion of sustainable skills, human resources, institutional capabilities, organizational structures, and physical infrastructure to support identified programs and services.

CAES will support its affiliated researchers by facilitating access to the collective energy-related capabilities of its member institutions. Additionally, CAES will support its member institutions by identifying needed capabilities and will work with these institutions to obtain new and expanded institutional capabilities. Through the co-location of talent and joint R&D efforts, CAES member institutions will gradually attain a higher level of functionality and sophistication in energyrelated research (i.e., larger, multi-disciplinary research and development programs) and produce high impact publications.

4.1.3 Economic Expansion

Economic expansion involves a robust and diverse economy, new employment opportunities, a growth of commerce and industry, and an increase in the standard of living. Research and development and educational efforts resulting from this CAES joint institute will enhance economic development through the commercialization and transfer of advanced technology.

The development, protection, and commercialization of intellectual property will support regional economic expansion. Cooperative research and development agreements (CRADAs), licenses, technical assistance efforts, and start-up companies are mechanisms that will be employed to facilitate economic expansion.

4.2 Products and Services

CAES, via the staff of its member institutions, will provide services and products described below that support the CAES affiliated researchers and the member institutions in achieving their desired outcomes. These products and services will serve as enhancements to existing institutional processes as CAES works with the member institutions to better align their individual talents for mutual benefit.

4.2.1 Joint Research

CAES member institutions will conduct joint research via their mutually enhanced capabilities to provide solutions to their energy customers. This research will maximize the utilization of the member institutions' capabilities by eliminating barriers that currently prevent or hinder collaborations. Joint research within CAES will:

- involve multiple member institutions
- include a multi-discipline approach
- utilize and create unique capabilities
- be substantial in size and duration
- include a strong educational component.



4.2.2 Facilities/Equipment Access

CAES will facilitate researcher access to unique facilities and equipment that would otherwise be inaccessible or overly problematic. This service will include facilitating researcher, faculty, and student access to existing facilities, encourage joint activities involving available member institution's research facilities, supporting the development of new facilities, and acquiring the unique and necessary equipment for the CAES member institutions to accomplish their CAES research objectives.

The CAES member institutions are working together with the State of Idaho and DOE to complete construction of the CAES facility at University Place in Idaho Falls by June 2008. This new CAES facility will further enable the collaboration of CAES affiliated researchers. The CAES facility will be a premier user facility that promotes energy research, education, and training (see Section 7.2.1). The facility is expected to encompass 55,000 ft² with approximately half of the facility being dedicated to laboratory space. The proposed preliminary equipment is listed in Section 7.2.2.

4.2.3 Joint Appointments

CAES will help to establish joint appointment positions between its member institutions and affiliated institutions. The intent of these joint positions is to mutually benefit and strengthen the member institutions' ability to retain and attract outstanding scientists and engineers.

4.2.4 Business Development

CAES member institutions have agreed that a collaborative approach will expand available research opportunities that may otherwise be unavailable to a single institution. CAES will work with the member institutions to identify these opportunities in the area of advanced energy studies. Through joint business development efforts, CAES will support the member institutions in capturing opportunities that align with the participating institutions' capabilities and objectives.

4.2.5 Relationship Building

The CAES member institutions recognize that other institutions and organizations possess considerable talent and capabilities. These organizations include the member institutions of the National University Consortium, private industry and other national and international institutions. CAES member institutions will build appropriate working relationships with these institutions to achieve the CAES mission.

4.2.6 Workshops

CAES will sponsor and organize key technical workshops focused on various energy systems. Representatives of the member institutions and other organizations will attend these workshops to develop plans, specifications, or other documentation that will guide the CAES program. These workshops will be designed to facilitate technical and intellectual interaction and the exchange of information between individuals and groups.

4.2.7 Seminars

CAES will establish and host a seminar series that focuses on advanced energy studies and topics of specific interest to the member institution's representatives. The purpose of these seminars is to enrich the skills and knowledge of the participants. Identified experts and specialists will be invited to present their ideas and share their expertise for the betterment of students, faculty, researchers and the public.

4.2.8 Conferences

CAES affiliated researchers will actively participate in, and, at times, organize professional conferences to further disseminate their research results. These professional conferences will serve as a means of obtaining peer-review feedback of ongoing research as well as the facilitation of information exchange.



5. VALUE PROPOSITION

The need for CAES, in collaboration with the activities of its member institutions, to address key energy challenge issues is evident given the current U.S. and global energy situation, U.S. science and engineering educational challenges, and the combination of current energy industry workforce demographics and future workforce projections.

CAES will provide benefits to the public, its universities and industrial partners, INL, DOE, and other federal agencies by facilitating technical collaborations. By promoting multidisciplinary approaches, CAES will help satisfy customer needs in the areas of advanced energyrelated research, education and training, and energy policy.

5.1 Public Benefit

CAES will foster public dialogue, communications, and innovation which is needed to form the energy options foundation associated with the difficult energy decisions of the next decade.

- a.) Public dialogue of energy related issues will be promoted, leading to more sustainable energy solutions and national energy security.
- b.) Resources will be cost effectively managed and coordinated across government, academia, and industry.
- c.) Educational opportunities will be expanded for current and future generations of students and INL staff.

5.2 University Benefits

CAES will provide an organizational and infrastructure conduit to INL, and unique capabilities for a research, education, and training partnership among the Idaho universities and INL, intensely focused on the energy agenda of the 21st century. CAES will allow optimal development and application of member institution capabilities and skills.

- a.) Idaho's academic expertise and capabilities in energy science, technology, and policy will be strengthened through the involvement of INL staff, which will provide the opportunity for the development of new academic programs, growth in the faculty, and a significant increase in research activity.
- b.) Modern research laboratories at the Idaho Falls University Place campus, and other associated locations, will become available via CAES.
- c.) A network of Idaho, national, and international affiliated universities (initially involving Boise State University, Idaho State University, University of Idaho, Massachusetts Institute of Technology, North Carolina State University, the Ohio State University, Oregon State University, and University of New Mexico) will serve as an integrating function to advance U.S. nuclear and other energyrelated education.
- d.) The unique research and development capabilities and facilities of INL will be accessible to the CAES network of universities.

5.3 Industry Benefits

CAES will provide a means for industry to interact with a unique blend of national laboratory/university researchers focused on the energy agenda, and a resource for development of the workforce for the future.

- a.) The next generation of energy-related scientists, engineers, and technicians will be educated to support industrial needs.
- b.) A unique education and research facility will facilitate the co-location and collaboration of government-universityindustry energy-related interests, which



will expedite the commercialization of advanced energy-related technology.

5.4 INL Benefits

CAES will provide key capabilities to serve as an intellectual and organizational tie between INL and Idaho and other national and international partners to assist in establishing the national energy agenda and providing solutions.

- a.) The integration of the research and educational capabilities of participating institutions will strengthen existing programs by facilitating the infusion of additional talent.
- b.) Access and utilization of advanced research equipment will be facilitated to the benefit of member institutions in responding to energy program needs.
- c.) Visiting researchers will be located in CAES to create a positive research atmosphere while satisfying security requirements.
- d.) Broader research funding opportunities will be available to INL technical staff.

5.5 DOE Benefits

CAES will provide a unique environment for bringing together and developing national

laboratory, university, and industrial skills, capabilities, and programs to foster a new generation of concepts and technologies for national energy challenges.

- a.) U.S. Department of Energy (DOE) initiatives, such as the Generation IV International Forum and the Global Nuclear Energy Partnership (GNEP), will be supported through a network of affiliated institutions (e.g. DOE National Laboratories, International Atomic Energy Agency, World Nuclear University, and Dalton Nuclear Institute, etc.).
- b.) Cross-organizational research that is critical to resolving the technical challenges of advanced energy sources will be facilitated.
- c.) Sound energy policy leading the U.S. towards improved energy security will be advanced through coordinated and multi-disciplinary efforts.
- d.) Through the co-location and collaboration of government-universityindustry energy-related interests, CAES will help expedite the commercialization of advanced energy-related technology.



6. **BUSINESS DESCRIPTION**

The operations of CAES will be transparent to its member institutions and participating processes, faculty, and staff, and are being designed to simplify technical cooperation and foster a positive researcher-friendly culture. CAES products and services will directly support the member institutions by fostering intellectual and economic synergy between the member institutions in the form of multiorganizational joint research projects, unique educational assignments, student/ faculty exchanges, sabbaticals, and joint employment in energy research and policy.

6.1 Joint Institute

CAES will operate as a jointly funded and managed research institute, comprised initially of four charter member institutions:

- Boise State University (BSU)
- Idaho State University (ISU)
- University of Idaho (UI)
- Idaho National Laboratory (INL).

These four member institutions have agreed to establish this joint advanced energy Center and aspire to be world-class by addressing, from a scientific perspective, problems that are critical to the nation. Appendix A contains the Memorandum of Agreement approved by the CAES member institutions.

6.2 Transformation Schedule

CAES will initially operate as a public/private partnership characterized as a relationship of mutual cooperation and responsibilities for the achievement of the specified goals.

CAES member institutions may consider moving from this strategic partnership stage to a formal joint venture. If desired, this joint venture will be characterized as a formal business relationship, a legal entity, in which the member institutions agree to contribute equity and share in the revenues, expenses, and control of the new enterprise. During the formulation of this joint venture entity, the member institutions will collectively determine if they should structure it to satisfy the requirements of section 501 (c)(3) of the Internal Revenue Code.

For the purposes of this Business Plan, this entire CAES relationship is referred to as the joint institute. An additional working assumption is that the transition from the strategic partnership to the joint venture may occur in 2008 (see Figure 4). It is anticipated that a more detailed MOA will be developed and will supersede the one signed in January 2006. The more detailed MOA is planned to be in place before the CAES facility is occupied in July 2008.



Figure 4. The CAES jointly managed institute may involve two distinct stages.



6.3 Responsibilities of the CAES member institutions

The responsibilities of the CAES member institutions include:

- 1.) Seeking opportunities for collaboration on research projects that require the use of CAES membership resources
- 2.) Participating in educational and research activities designed to accomplish the CAES mission

- 3.) Sharing research and education resources as appropriate
- 4.) Leading and participating in CAES efforts to secure funding in support of CAES activities
- 5.) Providing space, working with INL, as appropriate and to the extent available, for CAES staff, equipment, and visitors
- 6.) Exploring and examining possible future funding mechanisms.



7. ORGANIZATIONAL MODEL

7.1 Organizational Structure

CAES member institutions have initially established CAES under the aegis of INL in order for CAES to be built on a firm foundation. The final organizational structure is currently being developed.

CAES member institutions have initially established CAES under the aegis of INL in order for CAES to be built on a firm foundation. As such, CAES is managed by an INL Director, accountable to the CAES Steering Committee. The CAES Steering Committee consists of senior representatives from each of the four member institutions. The CAES Steering Committee provides oversight for the formulation phase of CAES including facility development and scientific direction.

Three Associate Directors, one from each of the Idaho Universities and an INL Deputy Director support the CAES Director in execution of the research, education, and policy activities. The three CAES Associate Directors will serve as the program leads for their respective Universities and will assist in the development and coordination of CAES sponsored activities. Additionally, *ad hoc* advisory committees will be formed as needed for scientific advice and assistance. Figure 4 illustrates the CAES organizational structure.



Figure 4. Organizational structure of the CAES joint institute as of October 2006.


If the out year decision is made to transition CAES to a formal joint venture entity, it is anticipated that a CAES Board of Governors would replace the CAES Steering Committee. At that time this Board would then be accountable for CAES operations and other fiduciary responsibilities.

7.1.1 Staffing Plans

The technical staff of the member institutions and other affiliated organizations will develop and implement the CAES research, education, and policy agenda. These technical research staff will remain employees of their respective organizations. Arrangements are being made to create appropriate mechanisms to facilitate interaction and collaboration within the CAES structure.

A Joint Appointment program has been established between BSU, ISU, UI, and INL to improve the ability of CAES member institutions to share technical staffing capabilities across organizational lines. This program also improves the member institutions' ability to attract and retain the outstanding talents (i.e., key strategic hires) needed to fulfill the mission of CAES.

The CAES member institutions anticipate defining and offering a number of research positions. Funds described in an FY06 DOE Field Work Proposal (FWP), totaling \$2,970,000, are being used by the university members to place a number of new hires.. Through this FWP, the university member institutions will recruit and hire at least six new research-active faculty in nuclear engineering and other energy-related disciplines, including energy policy, by September 30, 2007. Where possible, joint appointment positions with INL will be considered in support of INL/DOE programs. Also, during academic year 2006-2007, the university member institutions will provide support for a minimum of 15 Idaho university graduate students in nuclear engineering and other energy-related disciplines, including energy policy. Where possible, these new faculty and graduate students will provide direct support to the CAES research agenda and to INL programs and projects.

7.2 Infrastructure and Equipment Plan

7.2.1 CAES Facility Plans

The State of Idaho is providing approximately 7 acres of land on which the CAES facility can be constructed. The primary location for co-locating the CAES joint institute team is within the proposed CAES facility (see Figure 5). Additional aligned activities developed on the main university campuses will engage a wider grouping of faculty/technical staff and specific equipment.

This new CAES laboratory/office facility is currently in the design phase, will be located on the joint ISU/UI campus in Idaho Falls, and will be a key element within the developing INL Idaho Falls integrated National Laboratory/ University campus. Construction of this facility is scheduled for completion by July 2008. The expected general location is shown in Figure 6.



Figure 5. An artist's conception of the proposed CAES facility (northern exposure).





Figure 6. Proposed location of the CAES facility in Idaho Falls.

7.2.2 CAES Equipment Plans

Technical capabilities will be integrated for use by the CAES member institutions and affiliated members in fulfillment of its mission. Some existing capabilities will remain at their current location and will be accessed as necessary by participating researchers, while other existing equipment will be relocated into the new CAES facility. New state-of-the-art equipment will be obtained to supplement existing institutional capabilities. These new capabilities will be located within the new CAES facility. In addition, CAES will support increased access to existing and future facilities at INL and the universities.

A facility operations guide and an equipment management plan will be developed to further define equipment operations, maintenance and user costs. Maintenance and replacement of the equipment is expected to be the responsibility of the equipment owners in coordination with the CAES Director. Equipment use will be coordinated with the CAES Director. The equipment anticipated to be located within the new CAES building includes:

- High-Speed Computing Network Connectivity
 - o Lambda Rail Hub
- Distance Learning/Visualization Capabilities
 - o Research Collaboration Network
 - o Power Grid Wall
- Advanced Imaging Capabilities
 - Scanning Transmission Electron Microscope
 - Dual Beam Scanning Electron Microscope/Focused Ion Beam System
- Radiochemistry Laboratory
- Chemistry Laboratory
- Advanced Materials Laboratory
- Hydrogen Laboratory
- Analytical Laboratory
 - Basic Radiation Measurement Capabilities.



8. MARKET ANALYSIS

8.1 Energy Market Description

The need for energy will continue to rise as the population and economic factors within the nation and the world increase. In the U.S. and throughout the world, the energy market includes many different sectors. Most energy sectors show an increase in production over time. The energy sectors that have the highest projected growth in production from now until 2030 are coal, renewable energy, and nuclear power, whereas crude oil production shows a decline. These projected trends are the result of changing economic conditions; geopolitical considerations; governmental subsidies and policies; and environmental awareness.

Consumption in all energy sectors is increasing, projecting a constant increase in demand for energy. U.S. energy production does not currently meet the domestic demand. Importation of energy, therefore, will continue to be necessary to meet domestic demand, unless new technologies arise or greater investment into domestic production of energy occurs. Areas such as nuclear, fossil fuel, and renewable energy are areas that have the potential to boost the domestic production with technological innovation and increased investment.

8.2 Energy Investments

Federal, State, private industry, and private foundations provide substantial R&D

investments into the energy sector. These sources are opportunities for CAES to pursue in support of its research, training/education, and policy business objectives.

Federal programs, DOE in particular, provide CAES with the most direct and widest range of energy R&D options. The Energy Policy Act of 2005 defines a number of national initiatives that CAES can directly support. Additionally, the President's Fiscal Year 2007 Budget request details his administration's energy priorities which include basic energy science, high energy physics, nuclear energy, hydrogen, biomass/biofuels, solar, clean coal, and wind. The CAES member institutions will aggressively pursue specific research areas within these federal programs.

Many private foundations are focused on environmentally friendly energy solutions. Foundations are also actively involved in educational initiatives. CAES member institutions will continue to approach private foundations for scholarship support and educational activities.

Appendix B provides a more detailed analysis of specific and relevant opportunities. A preliminary overview of potential competitors is also included in this analysis.



9. MARKET PLAN

CAES will pursue research opportunities approved by its Steering Committee. These opportunities will be focused within specified research themes. These themes will be in areas where there is synergy between multiple member institutions so as to provide value to individual researchers (i.e., transparent collaboration, project management services, etc.) as well as the member institutions.

Further, these research themes are areas in which the CAES member institutions have an existing expertise that can be built upon as well as in areas in which member institutions desire to build new expertise for emerging markets.

The CAES research themes will nominally involve at least 6-8 senior scientific staff from multiple institutions (at least 3), will have bold objectives, and will span a multi-year time period (~5 years). Each of these themes will involve an educational component to actively engage graduate students. These research areas will also present an opportunity for active engagement of industrial partners (i.e., engineering centers, contribute resources, etc.), including technology transfer and commercialization.

9.1 Initial Research Themes

An initial set of workshops has been completed to initiate the development of the CAES research themes. These workshops have identified a number of research areas that could potentially serve as the initial foundation for CAES affiliated research. The identified areas and technical leaders are as follows:

- Carbon Management (Bob Smith, UI)
- Synthetic Liquid Fuels (Lyman Frost, INL)
- Energy-Water Nexus (Gary Johnson, UI)
- Hydrogen Economy/Production (Darryl Butt, BSU)

- Renewable Energy (Darryl Butt, BSU)
- Energy Storage (Bob Cherry, INL)
- Energy Policy (John Freemuth, BSU)
- Advanced Nuclear Fuel Cycle (ACE Committee)
- Post Irradiation Examinations (Rory Kennedy, INL)
- Material Science (Darryl Butt, BSU)
- System Modeling (Mary Lou Dunzik Gougar, ISU)
- Instrumentation and Controls (Leonard Bond, INL)
- Waste Management (Kevin Kostelnik, INL)
- Siting Integration (Jay Kunze, ISU)
- Thermal Hydraulics (John Crepeau, UI).

9.2 Capture Strategies

The technical leaders will investigate the identified themes. This analysis includes a high level survey across the member institutions of strengths and capabilities, a list of senior investigators currently working (or interested in working) in the area, and a summary of existing research efforts (e.g., 2 projects and 3 graduate students).

This analysis will also include consideration of the cross cutting issues, such as environmental consequences, policy implications, and educational/training needs and opportunities. This information will be used to:

a.) Conduct a SWOT (Strengths, Weaknesses, Opportunities, and Threats) analysis



- b.) Focus our assessment of federal and other sources of research funding opportunities
- c.) Design more inclusive topical workshops to fully develop the CAES research, policy, and education agenda.

This information will be integrated into the

CAES Strategic Plan, which will be completed by November 30, 2006. Within this Strategic Plan, the CAES member institutions will work to align their capabilities to best achieve the desired outcomes. Figure 7 illustrates a preliminary strategy map for the alignment of key capabilities.



Figure 7. CAES preliminary strategy map for aligning the applicable technical capabilities of its member institutions towards specific energy challenges and customer expectations.



10. FINANCIAL PROJECTIONS

10.1 Current Operating Scenario

The current CAES operating model is a public/private partnership in which the member institutions agree to cooperate in areas of mutual benefit. The financial, legal, business, and organizational mechanisms of each institution will continue to be used in support of CAES affiliated activities. Investments, in terms of financial resources, staff, and equipment will continue to be managed within the defined policies of the individual member institutions. The CAES Steering Committee will serve as the institutional representatives.

10.1.1 Key Assumptions

The following key assumptions established this CAES investment scenario:

- All CAES partners agree to the proposed business model and seek to develop the program needed to implement staffing/revenue and expense projections
- CAES expenses and revenue projects are equal
- INL support for CAES will increase between FY06 FY10
- Competitively awarded research grants generated by all CAES partners will increase substantially beginning in FY09
- State appropriations will be requested to begin in FY08 to support CAES-affiliated research faculty
- DOE support for the Idaho universities will continue in FY07 and will be requested for FY08
- Not all CAES-affiliated research faculty and activities will be housed in the CAES building

• A portion of university indirect charges on CAES related research revenue will be reinvested into their individual CAES business development efforts.

10.1.2 Projected Building Costs

An important financial area of mutual interest for the CAES member institutions involves the financing of the CAES building and subsequent occupation and operation of the joint institute. The financing of this building involves multiple fund sources. The following funding mechanisms have been secured for construction of the CAES facility:

- \$4,900,000 from the INEEL Settlement Fund, as defined in the Idaho Code 67-806A
- \$625,000 in Housing and Urban Development (HUD) Grants #B-00-SP-ID-0116
- \$1,017,756 in HUD Grants #B-01-SP-ID-0172
- \$10,000,000 via ISU bonds to be retired by an INL lease agreement, with Battelle Memorial Institute (BMI) and Washington Group International as the guarantor of the bonds.

Only one of these sources, the \$10M via ISU bonds, represents a long-term financial obligation. These obligations are projected to begin in the third quarter of FY08. The majority of CAES building expenses are attributed to INL, which will lease approximately 71% of the building for 20 years. These INL lease payments are estimated to total approximately \$890K per year. The final lease rate will be dependent on the acquired interest rate.

The only CAES building expense attributed to the university partners is a pro-rated portion of the annual maintenance and operating (M&O) cost which will be based on the square footage occupied by each institution. The total building M&O costs are projected to total \$240.4K/year.



For the purposes of this plan, the combined university share of annual CAES building M&O costs is estimated to be \$99.6K. The INL share of the annual CAES building M&O costs is estimated to be \$140.8K. These estimates are based on a total CAES building square footage of 55,000 square feet.

Finances to cover the total annual CAES building costs of approximately \$1130K appear to be readily available given the present INL and university operations.

10.2 Incremental Investment Scenario – "Achieving the Vision"

The vision for CAES is that it will develop into an enduring and sustaining entity—a joint institute collectively operating the building, with laboratory space and equipment shared between university and laboratory researchers—that will enhance capabilities and grow new academic and research activities. It is anticipated that by 2009, approximately 150 people will occupy the new building, and by 2015 a Phase II addition may be required to accommodate the projected growth.

10.2.1 Projected Revenue

CAES projected revenue by member institution for the period FY06 through FY15 is shown in Figure 8. These revenue projections are based on the best estimates of each member institution. During FY06 and FY07, a significant portion of CAES revenue is shown to be associated with the FY06 DOE university FWP and INL support. For the purposes of this analysis, it is assumed that additional DOE university support will be received in FY07 and requested for FY08.

In FY08, it is projected that CAES-affiliated faculty and staff from member institutions will begin acquiring research grants through competitive proposals. It is expected that each university CAES-affiliated faculty member (excluding Energy Policy Institute faculty) will generate \$250-\$300K annually in competitively awarded research projects by FY10. It is expected that the Energy Policy Institute faculty will generate approximately \$125K each in competitively awarded research projects by FY10. It is also expected that strategic hires will be made by the INL and that these INL CAESaffiliated senior researchers will generate approximately \$500K each in competitively awarded research projects.



Figure 8. Projected CAES revenue by member institution.



Beginning in FY08, INL and/or DOE support for CAES is projected to increase substantially to support occupancy of the CAES building and the hiring of INL CAES-affiliated senior research personnel. The development of CAES assumes that INL will provide indirect support as well as program development and Laboratory Directed Research and Development (LDRD) funding through FY13. These INL funds will help support CAES personnel and provide funding for CAES business development activities. INL support is expected to peak in FY09 then steadily decline as external research funding grows. Figure 9 shows the projected level of investment into CAES by member institutions for the period FY06-FY15. Additional State of Idaho appropriations will be requested to begin in FY08 to cover a portion of CAES-affiliated university faculty salaries. Also illustrated on this figure are the funding targets associated with specialty research equipment that will be located within the CAES facility.

The projected CAES external funding amounts are assumed to be new research funds awarded, not a commitment of existing research funding from INL, DOE, or affiliated universities to CAES.



Figure 9. Projected member institution investments in CAES.

10.3 Projected Expenses

The projected expenses of each CAES member institution for the period FY06 through FY15 are illustrated in Figure 10. These projections are based on the best estimates of the CAES members. The majority of the expenses anticipated are related to the labor cost of CAES-affiliated faculty, graduate students, INL research staff, and CAES management and staff. CAES labor costs escalate over the period as staffing levels increase to their peak in FY10. A substantial portion of INL expenses in FY09 and FY10 are anticipated to result from the purchase of specialized research equipment in those years. CAES building costs for this scenario are the same as discussed in Section 10.1.2. Business development expenses, research grants, and other non-labor charges account for the remainder of INL CAES related expenses. Travel, materials, and indirect costs make up the remainder of university CAES related expenses.





Figure 10. Projected expenses for CAES member institutions.

10.4 Financial Analysis

The financial viability of CAES is favorable as illustrated in Figure 11. These financial projections rely on CAES member institutions successfully competing for research funding in the highly competitive public and private R&D environment. Significant financial support from DOE and INL is crucial in the early years to lay the groundwork for future CAES success. This financial support will provide the seed money necessary to attract and hire key personnel including senior researchers and faculty. DOE and INL investments will continue to be necessary to fund CAES business development efforts and ensure that CAES gains exposure at key national and international professional conferences and symposia.

The early financial support will gradually build an integrated CAES portfolio of competitively awarded grants and contracts. Figure 11 depicts the estimated growth in CAES external funding and the corresponding decline in investments. Projected CAES external funding is the estimated amount of competitively awarded contracts and projects secured by CAES-affiliated faculty, fellows, and researchers, and State of Idaho appropriations for university faculty salaries. As Figure 11 shows, the annual amount of CAES external revenue and State of Idaho CAES-affiliated salary lines surpasses the annual amount of DOE and CAES member institution investments in FY09.





Figure 11. Projected total annual investment vs. CAES targeted external funding.



APPENDIX A – APPROVED MEMORANDUM OF AGREEMENT

MEMORANDUM OF AGREEMENT

AMONG BATTELLE ENERGY ALLIANCE, LLC, IDAHO STATE UNIVERSITY, BOISE STATE UNIVERSITY AND UNIVERSITY OF IDAHO

To Establish "The Center for Advanced Energy Studies"

This Agreement is entered into by and among Battelle Energy Alliance, LLC, Idaho State University, Boise State University and the University of Idaho.

RECITALS

WHEREAS, the parties wish to establish a world class advanced energy center to address from a scientific perspective problems that are critical to the Nation, which center is hereafter referred to as The Center for Advanced Energy Studies ("CAES" or "the Center"), with these charter members: Idaho State University, Boise State University, the University of Idaho and Battelle Energy Alliance, LLC, which operates the Idaho National Laboratory for the Department of Energy (hereafter referred to as "INL"), with all the heretofore named entities hereafter referred to as "The CAES Members;" and

WHEREAS, the mission of CAES (hereafter referred to as the "CAES Mission") is to become a world class, advanced energy organization with an emphasis on nuclear energy—recognized for contributions to advanced energy education, research, policy studies and training; and

WHEREAS, a secure and affordable energy supply is essential for achieving U.S. national security, in continuing U.S. prosperity, and to lay the foundations to enable future economic growth; and

WHEREAS, advanced nuclear energy provides the potential to meet a significant fraction of the growing U.S. electricity demand, give methods to economically produce hydrogen to be used to enrich heavy crude oils, and in the longer term, fuel the Hydrogen Economy; and

WHEREAS, nuclear power also offers technologies to address global energy needs to achieve sustainable development with increasing global population and to globally address critical environmental and energy-water needs; and

WHEREAS, U.S. participation and leadership in these science and technology areas is critical to ensure national security and economic health, and for growing U.S. industry through providing high-value, advanced nuclear technologies that are proliferation resistant; and

WHEREAS, the Center will address areas that are critical to U.S. energy security and leadership in the global energy arena;

NOW, THEREFORE, The CAES Members mutually desire, subject to final approval of their respective governing boards or entities, to establish CAES, initially

Page 1 of 5



under the aegis of Battelle Energy Alliance, LLC, as an appropriately constituted business relationship agreed to by The CAES Membership for the purpose of accomplishing the CAES Mission, and to establish the necessary administrative procedures, in accordance with the articles set forth below, to ensure the long-term success of CAES.

Article 1 Purpose

The goal of The CAES Membership is to develop the means to strengthen and support science-related research with The CAES Membership and to develop CAES as a national research and education center, consistent with the CAES Mission and with the missions of the member entities of The CAES Membership.

As part of the objective to foster a nationally respected reputation for CAES, The CAES Membership will ensure that other academic institutions are invited to participate as appropriate in CAES programs and initiatives.

In support of the CAES Mission and with CAES serving as collaborator and facilitator, The CAES Members will cooperate by sharing access, as appropriate, to their scientific facility resources. Thereby, through CAES, researchers within The CAES Membership and other institutions participating in CAES may access state-of-the-art facilities and intellectual community interactions that may otherwise be unavailable to them.

Article 2 Organization

It is the intent of The CAES Membership to initially establish CAES under the aegis of BEA in order for CAES to be built on a firm foundation, in a cost effective manner. It is the intent of The CAES Membership to examine the possibility of establishing CAES as a separate, independent entity, and The CAES Members will determine when it is appropriate to initiate the appropriate procedures to accomplish that goal.

CAES will be managed by a Director who will be hired by BEA and will be a BEA employee so long as CAES is operated under the aegis of BEA. The Director will report to, among others, the CAES Steering Committee (as defined in Article 3).

The Director and the CAES Steering Committee may jointly form *ad hoc* advisory committees as needed for scientific advice and assistance. The selection and duties of the CAES Steering Committee and the Director are defined in Articles 3 and 4, respectively.

Article 3 CAES Steering Committee

The CAES Steering Committee shall consist of one representative from each CAES Member. The Steering Committee will provide policy oversight and scientific direction for the operation of CAES. It will also provide ongoing oversight of the development of the initial CAES facility to be located on the University Place Campus. The CAES Steering Committee will be chaired by the INL Chief Research Officer, or his/her

Page 2 of 5

24



designee. The Director shall serve as an *ex officio* nonvoting member of the CAES Steering Committee.

The Steering Committee will establish or approve such policies, procedures and guidelines as are necessary for the conduct of CAES operations and Steering Committee activities, acknowledging that such policies, procedures and guidelines shall be subject to The CAES Members' contracts with the U. S. Department of Energy, other research sponsors and to any statutory or regulatory requirements.

Article 4 CAES Director

It is envisioned that CAES will be managed by a Director who will be hired by BEA and will be employed by BEA for so long as CAES is operated under the aegis of BEA. The Director will be hired in consultation with the other CAES members. The Director shall manage the operations and activities of CAES according to the policies and procedures established by the CAES Steering Committee, or recommended by the Director and approved by the CAES Steering Committee.

The Director is responsible for the activities, management and operation of all aspects of CAES. The Director shall present an annual budget and an annual report to the CAES Steering Committee.

Article 5 Responsibilities of the CAES Membership

The responsibility to work actively towards fulfilling the CAES Mission and to cooperate with or on behalf of CAES rests with The CAES Membership. Such responsibilities shall include, but shall not be limited to, the following activities:

- Seek opportunities for collaboration on research projects that require the use of CAES Membership resources;
- Participate in educational and research activities designed to accomplish the CAES Mission;
- Share research and education resources as appropriate;
- Lead and participate in CAES efforts to secure funding in support of CAES activities;
- Provide space, working with BEA/INL, as appropriate and to the extent available, for CAES staff, equipment and visitors;
- Explore and examine possible future funding mechanisms.

Page 3 of 5



Article 6 Infrastructure Support

It is intended that BEA will initially be the fiscal agent for CAES, receiving and distributing funds for the operation of CAES. It is intended that CAES will, at a later date, occupy office space in a facility built on land contributed by the State of Idaho, in a building constructed with State of Idaho funds and owned by Idaho State University, the specific form of ownership to be determined. It is expected that the State of Idaho or Idaho State University will act as the facilities agent for space made available in this facility, on such terms as will hereafter be established. No CAES activities will be undertaken in the facility that are inconsistent with the laws or missions applicable to DOE.

Article 7 Term and Approval

This Memorandum of Agreement does not create legal rights or obligations, impose legal obligations on any party hereto, or confer legal rights on any party hereto or any third party. To the extent any of the clauses and provisions of this MOA conflict with the terms of BEA's Contract No. DE-AC07-05-ID14517 with DOE the terms of such Contract will prevail.

The parties intend to use best efforts to cooperate and to support the activities of CAES and fulfill their responsibilities set forth in Article 5.

It is the intent of The CAES Membership that this be a continuing joint effort for the benefit of all parties. However, any party may withdraw from this Memorandum of Agreement by providing written notice to the CAES Director not later than March 1 of the year in which withdrawal is intended. The CAES Membership will explore a funding mechanism for CAES, such as appropriate Member payments, grants or other sources. Withdrawal shall terminate that Member's obligations under this Agreement as of August 31 of the year of withdrawal, but shall not be a basis for return of any portion of the Member's payments under any agreed upon funding provision.

The Memorandum of Agreement and any modifications thereto shall be subject to the approval of the duly authorized representatives of The CAES Membership. In consideration of the provisions and mutual promises herein contained, the parties have caused to be affixed hereto their duly authorized signatures as of the dates indicated. This Memorandum of Agreement may be executed in several counterparts, each of which shall be an original, and all of which together shall constitute one and the same instrument.

Page 4 of 5



IN WITNESS WHEREOF, the Parties hereto have signed this Agreement.

BATTELLE ENERGY ALLIANCE, LLO By: LANNAG 2006 25 Date: 1 BOISE STATE UNIVERSITY By: Date:

IDAHO STATE UNIVERSITY By: Date: 12-20-05

UNIVERSITY OF IDAMO unot By: 1.17.06 Date:

Page 5 of 5



APPENDIX B – ENERGY MARKET ANALYSIS

B.1 Energy Policy Act 2005

The Energy Policy Act of 2005 outlines a number of strategic initiatives that are aligned with key research and educational objectives of CAES.

Energy Efficiency

Provisions are included to expand the Energy Star program, a government/industry partnership for promoting energy efficient products. It establishes new energy efficiency standards for many new commercial and consumer products that use large amounts of energy – providing significant savings on monthly energy costs.

• Oil and Gas

The EPA Act authorizes expansion of the Strategic Petroleum Reserve's (SPR) capacity from 700 million to 1 billion barrels and filling the SPR to that capacity during periods of stability. The act clarifies the federal government's role in the siting of liquefied natural gas terminals. Several streamlining provisions have been included to encourage efficient review of applications at the federal level; also requires consultation with the state in the area of state and local safety.

The Act authorizes funding in support of a number of strategic areas including: (1) Seventy-five million dollars per year to support research involving ultra-deepwater programs; (2) Fifty-two million dollars per year for research of unconventional energy resources, including, but not limited to, gas shale, deep drilling, etc.; and (3) Fifteen million dollars per year in support of small producers of natural gas and related products. The Refinery Revitalization Act of 2005 designates certain areas as refinery revitalization zones based on an area's previous refining or manufacturing experience and current unemployment rate; the EPA act provides an accelerated review and approval process for new refinery facilities in a refinery revitalization zone.

• Clean Coal Technologies

The EPA authorizes the Secretary of Energy to provide up to \$ 1.8 billion in support of research and demonstration of advanced coal-based power generating technologies that achieve significant reductions in air emissions. It further requires that at least sixty percent of the \$1.8 billion will be utilized in coal-based gasification technologies projects that meet stringent environmental performance standards and vastly increased efficiency standards.

Another \$3 billion is authorized for the Secretary of Energy to carry out a Clean Air Coal Program to facilitate production and generation of coal-based power and accelerate the development and deployment of pollution control equipment to meet current and future obligations of coal-fired generation units regulated under the Clean Air Act. Loan guarantees for five petroleum coke gasification projects are authorized by the act.



Vehicles and Fuels

The new Act requires all "dual-fueled" vehicles acquired under the Energy Policy Act of 1992 programs to be operated on alternative fuels; also requires a study to review the vehicle purchase requirements under the Energy Policy Act to ensure the program works more effectively in the future. The Act further establishes a requirement to develop and implement a Hybrid Retrofit and Electric Conversion program to provide incentive to develop innovative technologies for use in retrofitting conventional internal combustion engines.

• Automobile Efficiency/CAFE

The Act increases Department of Transportation funding to six million dollars over five years (2006-2010) to continue research and development of technologies to improve the Corporate Average Fuel Economy (CAFE) standards. The National Highway Traffic Safety Administration (NHTSA) is required to evaluate the impact of vehicle safety and automobile industry employment when establishing new CAFE requirements. When setting CAFE standards, the National Highway Traffic Safety Administration (NHTSA) is required to consider the impact on vehicle safety and automobile industry jobs. Finally, the Act further requires the NHTSA to undertake a study to evaluate CAFE alternative.

Hydrogen

The Act requires the establishment of development and implementation program to encourage production and use of hydrogen-powered automobiles by 2020. The program also supports development and deployment of the necessary infrastructure required to safely fuel hydrogen vehicles. It also establishes an interagency task force on hydrogen as well as an outside advisory committee with a budget authorized at \$2.15 billion over five fiscal years.

The Department of Energy is further required to develop a road map plan that outlines: (1) Technical and non-technical hurdles for hydrogen fueled vehicles; (2) Infrastructure requirements; and (3) Technical milestones required to meet the 2020 time frame. The hydrogen research and development programs are to be carried out through public/private partnerships and will focus on: (1) Producing hydrogen from diverse sources, including fossil fuels, hydrogen-carrier fuels and renewable energy resources including biomass and nuclear energy; (2) Transportation and storage of hydrogen; (3) Advanced fueling systems; and (4) Development of codes and standards.

• Nuclear Energy

The EPA emphasizes the government support for a renewal of the Nuclear Energy industry through renewal of the Price-Anderson Nuclear Industries Indemnity Act for an additional twenty years. Furthermore, the Act includes provisions to encourage research and development of advanced modular reactors, and for nuclear contractor, subcontractor and supplier financial accountability. The Price-Anderson legislation is to indemnify the nuclear industry against liability claims arising from nuclear incidents while still ensuring compensation for the public. The Act further establishes a no fault insurance-type system wherein the first \$10 billion in claims would be industry-funded according to a formula and all claims in excess of this amount would be covered by the federal government.



The EPA Act provides requirements for increased security of nuclear facilities, including improved federal oversight of plant security and the expansion of federal statutes for sabotage of nuclear facilities. The Act further outlines three additional areas of emphasis: (1) establishment of a permanent nuclear waste disposal site; (2) establishment of nuclear fuel recycling to replace current fuel disposal; (3) creation of Advanced Reactor Hydrogen Cogeneration Project; and (4) establishment of a research program for an Advanced Reactor Hydrogen Cogeneration Project.

Studies and Program Support

The act provides extensive authorization for the Department of Energy to increase the efficiency of all energy intensive sectors, promotes diversity in energy supply, improves energy security and decreases the environmental impact of energy-related activities. Specific authorizations are provided for energy-efficiency efforts, a next-generation lighting initiative, national building performance initiative, programs regarding renewable energy, bioenergy and solar power, oil and gas programs, including ultra-deepwater and unconventional natural gas.

The plan authorizes nearly \$4 billion for energy and conservation activities; more than \$3 billion in funding over the next five years for renewable energy efforts; and more than \$2.5 billion for fossil fuel efforts. It also establishes a competitive grant program for 15 different geographic locations/programs based on the acquisition of alternatively fueled, fuel cell, hybrid, and/or ultra-low sulfur vehicles, including programs dedicated to installation of infrastructure necessary for the maintenance and support of such vehicles, amounting to no more than 20 million dollars per applicant.

The act establishes a training and fellowship program for nuclear sciences, health physics, and engineering fields, with a budget of one million dollars per FY until 2009. It also establishes a program of study for the secondary use of spent batteries and a program for study of the storage capacity of petroleum and natural gas.

Electrical Power Generation

The Act promotes investment to increase an upgrade critical electric transmission capacity and efficiency measures. The Federal Energy Regulatory Commission (FERC) is required to implement an incentive rate rule and to provide for participant funding. FERC is further required to implement policies for expedited siting processes on both federal and private lands; and provides for the development and deployment of advanced transmission technologies.

• Alternative Fuels

Under the provisions of the EPA, renewable fuels' standard is set at 5 billion gallons to be available in the marketplace by 2010. It further establishes operator-training programs to be created in areas where they do not already exist. In addition, it bans the use of MTBE in motor fuels after 2014.



B.2 FY-2007 Proposed DOE Budget

The Department of Energy Budget request for fiscal year 2007 emphasizes energy diversification with increased funding for:

- o Solar America—\$148 million
- o Biomass/Biofuels Initiative—\$150 million
- o Wind—\$44 million
- o Hydrogen Fuel Initiative—\$289 million
- Clean Coal—Future Gen—\$54 million
- o Nuclear Power 2010—\$54 million
- o Nuclear Energy, Science & Technology-\$347 million
- Fusion Energy Science—\$319 million
- High Energy Physics—\$776 million
- Nuclear Physics—\$454 million
- State Energy Grant Programs—\$49 million
- Basic Energy Science—\$1,421 million
- o Advanced Scientific Computing—\$318 million

The budgetary requests listed above are, in general, largely mortgaged funding for ongoing programs. However, there are areas where additional and/or new research is contemplated. Some of the areas of increased research emphasis include: (1) Biotechnology;

(2) Nanotechnology; (3) Materials Science; (4) High-Speed Computing; (5) High-Intensity Light sources; and (6) Fusion Energy.

B.3 Environmental Protection Agency

The Environmental Protection Agency (EPA) has a number of grant-related programs focused in the areas of energy efficiency and renewable energy including: (1) Pollution prevention; (2) Healthy Communities; and (3) Clean Air. State and local governments, businesses, public organizations (e.g. schools), nonprofit organizations, and consumers may be eligible to participate in the EPA grant's program. In general, projects selected for funding will be in support of: (1) Reduction of energy-related costs, (2) Improvement in air quality and public health; and (3) Enhanced economic development. Funding opportunities are published through the EPA's website. http://www.epa.gov/

B.4 National Science Foundation

The National Science Foundation (NSF) offers funding and grant-related programs in support of sciencerelated projects. NSF funds research and education in most fields of science and engineering. It does this through grants and cooperative agreements to more than 2,000 colleges, universities, K-12 school systems, businesses, informal science organizations and other research organizations throughout the United States. NSF also supports cooperative research between universities and industry, US participation in international scientific and engineering efforts, and educational activities at every academic level. Grants and awards provided through NSF are generally directed to basic science, computing and engineering research projects. Funding opportunities are published through NSF's website. http://www.nsf.gov/funding/



B.5 Private Foundations

• Energy Foundation

The energy foundation provides funding in support of energy efficiency and renewable energy use. Grants are allocated in six areas: 1) Power generation; 2) Transportation; 3) National Policy and Analysis; 4) Climate and are to be used in improving energy efficiency and renewable energy use. Funds may not be used for R&D projects, community energy projects, or demonstration projects. Eligibility is restricted to non-profit charitable organizations. http://www.energyfoundation.org/home.cfm

• Kirsch Foundation

The Foundation has a focus in reducing air pollution from mobile sources, supports the development and use of alternative, clean fuel technologies, elimination of industrial and consumer dependence on diesel fuel and/or eradicates global worming. Has an annual funding budget of ~\$1.5 million. http://www.kirschfoundation.org/how/environmental/environment.html

Strategic Environmental Project Pipeline Foundation

Under the StEPP Foundation, funding is provided to support organizations and/or projects in the U.S. that demonstrate positive benefits of renewable energy, energy efficiency, and pollution prevention projects. The Foundation is dedicated to helping organizations realize their vision of a clean and safe environment by nationally matching projects with fund sources. The StEPP Foundation provides project oversight to enhance the success of projects increasing the number of energy efficiency, clean energy and pollution prevention projects implemented at the local, state and national levels for the public. Government entities (local, state and national), non-profit organizations, academic institutions and other organizations are all encouraged to enter project ideas into the StEPP Project Pipeline for funding consideration. http://www.steppfoundation.org/

Bulitt Foundation

The Bulitt Foundation supports projects/initiatives that encourage ultra-efficient energy use, environmentally sustainable urban growth, and other environmental objectives. It also provides funding for training programs that support these areas. The Foundation focuses on non-profit organizations that serve the States of Washington, Alaska, Oregon, Idaho, Montana, and the Canadian Province of British Colombia. <u>http://www.bulitt.org/</u>

Bonneville Environmental Foundation

The Bonneville Environmental Foundation provides grants, loans, convertible loans, guarantees, and direct investment in renewable energy projects. Projects of interest include, but not limited to, solar, wind and geothermal projects. Projects eligibility is limited to individuals, organizations, local or tribal governments located in Idaho, Montana, Oregon or Washington. <u>http://ww.b-e-f.org/grants/</u>



B.6 Competitors

Following is a preliminary and partial list of potential competitors to CAES. Included within this analysis are other National Laboratories, Universities, Energy Centers, and Training Centers.

National Laboratories

- 1. Argonne National Laboratory <u>http://www.anl.gov/</u>
- 2. Sandia National Laboratory <u>http://www.sandia.gov/</u>
- 3. Los Alamos National Laboratory <u>http://www.lanl.gov/</u>
- 4. Oak Ridge National Laboratory <u>http://www.ornl.gov/</u>
- 5. Pacific Northwest National Laboratory http://www.pnnl.gov/
- 6. National Renewable Energy Laboratory <u>http://www.nrel.gov/</u>
- 7. National Energy Technology Laboratory <u>http://www.netl.gov/</u>
- 8. Lawrence Livermore National Laboratory <u>http://www.llnl.gov/</u>
- 9. Brookhaven National Laboratory <u>http://www.bnl.gov/</u>

University Centers

A number of research and educational centers and institutes have been established in support of alternative energy research. The majority of the more established centers have a fossil energy focus whereas some of the more recently established programs have taken on emphases in the renewable energy sector. Most of the programs have established funding sources and constituencies that provide sustainability to their programs.

1. *Energy & Environmental Research Center*—University of North Dakota. The Center has 280 scientists, engineers and instructors and has plans to expand to 500 by 2010. The Institute does research in advanced power and energy. Key areas include: (a) hydrogen production & distribution; (b) fuel cells; (c) fossil fuels; and (d) biomass. Centers of excellence include: (a) Center of biomass utilization; (b) Coal Ash Research Center; (c) Coal Utilization Center; (d) National Center for Hydrogen Technology; and (e) Wind Energy Resources Center. It also operates a national alternative fuels laboratory. http://www.undeerc.org/

2. *Center for Applied Energy Research*—University of Kentucky. The Center conducts research and development in: (a) carbonaceous materials; (b) clean coal technologies; and (c) clean fuels and chemicals. <u>http://www.car.uky.edu/</u>

3. *Appalachian Oil & Natural Gas Research Consortium*—West Virginia University. The Center's is one of the leading fossil energy research institutions with emphasis in coal. In addition to fossil energy research, the Center operates a Supercomputing Science Consortium and operates an EPSCoR (e.g. Experimental Program to Stimulate Competitive Research). <u>http://www.nrcce.wvu.edu</u>

4. *Energy, Environment & Resources Center*—University of Tennessee-Knoxville. The Center has a staff of 75 and is funded through grants, contributions, funding and appropriations from DOE, DOD, EPF, TVA, state and private sector. The Center's research is focused primarily in research and development and environmental issues related to carbonaceous fuels. <u>http://eerc.ra.utk.edu/</u>



5. *Future Energy Laboratory*—Imperial College, UK. The Institute focuses on major cross-cutting, interdisciplinary research programs in key scientific, technological or commercial interest related to energy research. <u>http://www3.imperial.ac.uk/energyfutureslab/</u>

6. *The Public Utility Research Center*—University of Florida. The Center is funded by energy & telecommunications companies and the public service commission. The Center conducts energy policy forums both domestic and international. <u>http://www.purc.ufl.edu/</u>

7. *Energy Policy Initiatives Center*—University of San Diego. The Center is a non-profit academic research center. The Center emphasizes energy policy as well as analysis of energy trends. <u>http://www.sandiego.edu/usdlaw/</u>

8. *Center for Energy & the Global Environment*—Virginia Tech University. The Center conducts research in energy policy. The Center conducts seminars, workshops, training programs and joint research activities in collaboration with scientists and researchers from the U.S. and international institutions. It is linked with industry groups and universities in more than 30 nations, making it one of the most internationally active research organizations of its kind. <u>http://www.ncr.vt.edu/research/</u>

9. *Joint Global Change Research Institute*—University of Maryland. The Institute is a partnership with the Pacific Northwest National Laboratory. While housed with the University, the core research staff primarily consists of PNNL employees. Climate change economics, technology and policy impacts on adaptation and vulnerability, and international, long-term energy systems planning are the three main areas of focus. The Institute builds on the University strengths in public policy and the social sciences areas, and researchers who have been studying climate change. <u>http://www.globalchange.umd.edu/</u>

10. *The Laboratory for Energy and the Environment*—Massachusetts Institute of Technology. This is a university-based affiliation of programs and projects sharing information about the impacts of energy sourcing and usage on the environment. The program strengths include studying issues on energy alternatives, energy infrastructure, transportation energy, economic and climate impacts. Research is internationally focused with an academic audience. The Laboratory supports 40 staff, mostly Ph.Ds in engineering fields. <u>http://www.lfee.mit.edu/</u>

Training Centers

Centers/Institutes involved in training vary considerably in their offerings and their area of expertise. Many of the established programs already provided accredited training to the oil & gas, utility, and nuclear power industries. Programs and training are offered as web-based training and/or traditional classroom training.

1. *United States Association for Energy Economics*. The Center conducts training for professional involved in the energy industry as well as studies issues related to energy economics. http://www.usaee.org/

2. *British Institute of Energy Economics*. The international Institute's primary charter is to conduct studies related to energy economics and national energy policies. <u>http://www.biee.org/</u>

3. *Canadian Energy Research Institute*. The Institute has three primary activities: a) to conduct energy economic analyses; b) to evaluate national energy polices; and c) to carryout energy market assessments/forecasts. Their activities are focused primarily in the oil, natural gas and electrical power generation sectors. <u>http://www.eri.ca/</u>



4. *National Alternative Fuel Training Consortium*—West Virginia University. The Consortium currently supports 25-National Training Centers in the U.S. The Consortium conducts training in the use of alternative fuels in transportation applications. <u>http://www.naftc.wvu.edu/</u>

5. *General Physics Corporation*—Privately owned training center that is focused primarily on training and certifying nuclear operators for the nuclear power industry. <u>http://gpworldwide.com/nuclear</u>

6. *Nuclear Engineering Teaching Laboratory*—University of Texas. The Laboratory educates nuclear science engineers. <u>http://www.me.utexas.edu/</u>

7. *Nuclear Training Center*—Owned and operated by Duke Power and Westinghouse Electric Company at the Catawba Nuclear Station. The Center trains operators and employees of nuclear power utilities. Training emphasis is focused primarily on instrumentation, controls, and maintenance.

8. *The Public Utility Research Center*—University of Florida. The Center is funded by energy & telecommunications companies and the public service commission. The Center provides training for operators and employees of energy & telecommunications companies. <u>http://www.purc.ufl.edu/</u>

9. *Bismarck State Energy Education*—Bismarck State University. The Center is partnered with EPRI and provides online education training for energy employees and operators of the utilities. <u>http://www.bismarkstate.edu/energy/</u>

BAHR – SECTION II

ATTACHMENT 3

LEASE AGREEMENT NO. 00049377

BETWEEN

IDAHO STATE UNIVERSITY

AND

BATTELLE ENERGY ALLIANCE, LLC

FOR

THE CENTER FOR ADVANCED ENERGY STUDIES

TAB 4 Page 51

TABLE OF CONTENTS

ARTICLE 1 – DESCRIPTION OF PREMISES	1
ARTICLE 2 – TERM	2
ARTICLE 3 – PAYMENT	2
ARTICLE 4 – TITLE	3
ARTICLE 5 - APPLICABLE LAWS, CODES AND ORDINANCES	3
ARTICLE 6 - WARRANTY AS TO MECHANICAL EQUIPMENT AND UTILITIES	3
ARTICLE 7 – USE OF PREMISES	3
ARTICLE 8 – TERMINATION	4
ARTICLE 9 - COMMON AREAS AND ALTERATIONS TO PREMISES BY LESSOR .	4
ARTICLE 10 – IMPROVEMENTS	5
ARTICLE 11 – JANITORIAL SERVICES, MAINTENANCE AND REPAIR	6
ARTICLE 12 – ASSUMPTION OF RISK	7
ARTICLE 13 – LIENS	7
ARTICLE 14 – LIABILITY INSURANCE	7
ARTICLE 15 – PROPERTY INSURANCE	8
ARTICLE 16 – CONDEMNATION	8
ARTICLE 17 – DAMAGE OR DESTRUCTION	8
ARTICLE 18 – CLOSURE AND SURRENDER OF PREMISES	9
ARTICLE 19 – ACCESS TO PREMISES.	10
ARTICLE 20 – HOLDOVER	10
ARTICLE 21 – DISPUTES AND GOVERNING LAW	10
ARTICLE 22 – DEFAULT	10
ARTICLE 23 - LIMITATION OF LIABILITIES	11
ARTICLE 24 – NOTICES	12
ARTICLE 25 – WAIVER OF RIGHTS	12
ARTICLE 26 – TRANSFER OF OBLIGATION	12
ARTICLE 27 - HAZARDOUS SUBSTANCES TRACKING AND COMMUNICATION	13
ARTICLE 28 – ORDER OF PRECEDENCE	13
ARTICLE 29 – SHARED USE OF SPACE AND EQUIPMENT	13
ARTICLE 30 - ARTICLES INCORPORATED BY REFERENCE	13
ARTICLE 31 – EXAMINATION OF RECORDS	14
ARTICLE 32 – ASSIGNMENT	14
ARTICLE 33 – STATUS OF BATTELLE MEMORIAL INSTITUTE AND	
WASHINGTON GROUP INTERNATIONAL	14
ARTICLE 34 – APPLICATION OF CONSTRUCTION CONTINGENCY FUNDS	17
ARTICLE 35 – ENTIRE AGREEMENT	17
ARTICLE 36 – BOARD AND CONTRACTING OFFICER APPROVAL	17
ARTICLE 37 – COUNTERPART SIGNATURES	18

LEASE AGREEMENT

THIS LEASE, made and entered into this <u>33.</u>day of October 2006, by and between the **IDAHO STATE UNIVERSITY** an institution of higher education and agency of the State of Idaho (Lessor), with the acknowledgment and consent of the University of Idaho and Boise State University, all three collectively referred to herein as the Idaho Universities Consortium, and **BATTELLE ENERGY ALLIANCE, LLC**, a limited liability company organized under the laws of the State of Delaware, (Lessee or BEA) collectively referred to as the **Parties**.

WHEREAS, BEA is the management and operating contractor for the Idaho National Laboratory (INL) under Contract with the U.S. Department of Energy (DOE); and

WHEREAS, the Idaho State Legislature, through appropriation of funds for the design and ultimate construction of a facility that will contain the Center for Advanced Energy Studies (CAES) at the Idaho Falls Center for Higher Education campus, has approved a facility to be jointly occupied by the Idaho Universities Consortium (Consortium) and BEA in a collaboration intended to foster the deployment of new technology beneficial to the residents of the state, to encourage further collaborative research between the Consortium and BEA, and to enhance educational offerings at the Idaho Falls Center for Higher Education campus;

WHEREAS, the Consortium contemplates the construction of a multipurpose building at the Idaho Falls Center for Higher Education campus in the City of Idaho Falls, Bonneville County, Idaho; and

WHEREAS, BEA, with the intent of collaborating with the Consortium, is desirous of renting general office, and dry and wet laboratory space in the facility, and BEA is willing to lease said space upon the terms and conditions and for the purposes hereinafter set forth;

NOW, THEREFORE, it is agreed:

ARTICLE 1 - DESCRIPTION OF PREMISES

release or disposal of any hazardous or regulated substance but excludes reasonable wear and tear), Lessor agrees to reimburse Lessee for such damage. All parking areas provided for Lessee's use are included in and subject to the terms and conditions of this Lease.

ARTICLE 2 - TERM

A. The term of this Lease shall be, subject to the provisions of Article 8 herein, for twenty (20) years beginning on the _____ day of _____, 200_ (Effective Date) and expiring on the _____ day of _____, 202_.

B. The Effective Date of this Lease shall be established by mutual agreement of the Parties signing a "Commencement Date Agreement" (see Exhibit B). The commencement date established in Exhibit B shall be when the requirements for beneficial occupancy have been met. The term "requirements for beneficial occupancy" means (1) the Lessor has met all requirements established in the Lease in order to place the Premises in a condition of occupancy and (2) a "Certificate of Occupancy" for the Premises has been issued by the City of Idaho Falls.

C. Rental payments, as described in Article 3, shall not accrue until the first day prescribed in the Commencement Date Agreement. However, if the Parties have not signed the Commencement Date Agreement by March 1, 2009 (assuming this Lease has not been terminated prior to March 1, 2009), Base Rent, established in section 3.A below, shall be payable monthly by Battelle Memorial Institute (**Battelle**); provided however, that Battelle shall have no obligation to pay Base Rent (or any other obligation) if the requirements for beneficial occupancy have not been met as a result of the actions or inactions of Lessor or its contractors responsible for the construction of the Premises.

ARTICLE 3 - PAYMENT

A. <u>Base Rent</u>. The Parties agree to amend this Lease by mutual agreement by signing Exhibit C of this Lease, which shall establish the base monthly rental (**Base Rent**). The Lessee covenants and agrees to pay Lessor as rental for the Premises the Base Rent to be established in Exhibit C. Base Rent shall be due on the fifth business day of each month, payable in arrears. Base Rent payments shall be made directly to the trustee identified in Article 24 (**Trustee**) for bonds to be issued by Lessor to finance the Premises (**Bonds**). Lessor agrees that Lessee's Base Rent payments to trustee are deemed to be payments to Lessor for purposes of satisfying Lessee's Base Rent obligations under this Lease.

B. <u>Service Rent</u>. In addition to the Base Rent, the Lessee shall pay a pro rata share (based on the relative percentage of the Parties' occupancy of the Premises as indicated in Article 1) of all actual maintenance and operations costs, utilities, parking (if charged in the future), and insurance (subject to change) on an annual basis. Such monthly payment shall be based initially on one twelfth (1/12) of the estimated annual cost, with payment due at the same time as the Base Rent, and the Service Rent subsequently being adjusted to actual annual cost within sixty (60) days after the end of each Lessor fiscal year. If the actual cost was less than the amount that the Lessee paid, such excess shall be allowed as a credit against the Service Rent charges next coming due or if at the end of the Lease such excess repaid to Lessee.

C. If Lessee fails to make payment to Trustee as required by Article 3.A, Lessor agrees to provide notice to Battelle and WG (as these terms are defined in Article 33.A) to the persons listed in Article 24 within ten (10) business days of such failure. Notwithstanding the foregoing requirement, a failure of the Lessor to give such notice shall not impact in any way (except as provided in the following proviso) the obligations of Battelle or WG hereunder; provided however, the obligations of Battelle or WG to make guaranty payments under this Lease shall not be triggered until Lessor or Trustee provide the Notice(s) required in Article 33.D.

ARTICLE 4 - TITLE

The Lessor warrants it has title to the Premises, or sufficient interest and rights in the Premises to guarantee the Lease with no interference to the Lessee's rights of possession under the Lease. Should the Lessee suffer any damages or expenses as the result of any defect in the Lessor's title or rights and interests in the Premises, the Lessor shall reimburse the Lessee for all such damages or expenses.

ARTICLE 5 - APPLICABLE LAWS, CODES AND ORDINANCES

The Lessor, as part of the Lease consideration, agrees to comply with all applicable laws, codes, ordinances, rules, regulations and requirements of all federal, state and municipal governments and their appropriate departments, commission, boards and officials applicable to the ownership and establishment of the Premises and at its own expense, to obtain all necessary permits and related items. The Lessee agrees to comply with all laws, codes, ordinances, rules, regulations and requirements of all federal, state and municipal governments and their appropriate departments, commissions, boards and officials applicable to its tenancy and use of the said Premises.

ARTICLE 6 - WARRANTY AS TO MECHANICAL EQUIPMENT AND UTILITIES

The Lessor warrants that the mechanical equipment and utilities will be in good serviceable and proper operating condition on the Effective Date, and agrees Lessor will maintain the equipment and utilities and perform regular preventative maintenance on all equipment and devices as recommended by the original equipment manufacturer during the Lease term, and any extension periods of this Lease. The equipment and utilities include all plumbing, heating, cooling systems, and electrical and mechanical devices and fixtures.

ARTICLE 7 - USE OF PREMISES

A. Lessee agrees that the Premises are to be used as laboratory facilities and associated offices. Lessee shall not allow use of the Premises in a manner which would increase insurance premiums (unless Lessee pays the increased premium through Service Rent) or for any illegal purpose.

B. Lessor shall provide Lessee with written notice of Lessor's requirements imposed upon the Premises by Lessor's environmental permits applicable to activities conducted on the Premises. Lessee shall ensure that all Lessee activities conducted on the Premises are in full compliance with Lessor's requirements contained in such written notification. In return, Lessor shall provide

adequate advance notice and opportunity to Lessee to comment upon any proposed change(s) to such environmental permits and shall cooperate fully with Lessee in making reasonable changes and accommodations with regard to such permits to facilitate Lessee's continued use of the Premises.

C. Lessee shall cooperate with the Lessor in the Lessor's development of general Premises policies and procedures (i.e., hazardous chemicals handling and disposal, reporting and tracking, fitness for duty, safety, etc.) for the management and operation of the Premises. Any subsequent modification or adjustment to such policies and procedures shall be in writing and shall be developed by Lessor in cooperation with Lessee. The Lessee, its agents, officers, employees, subcontractors, licensees, and invitees shall comply with such Premises policies and procedures for the management and operation of CAES and the Premises. Lessee's failure to comply with such policies and procedures shall be grounds for default as set forth in Article 22, "Default", including Lessee's continued non-compliance after Lessor's 60 days' notice.

D. Lessor and Lessee agree that each Party will allow the other to use designated areas of the Premises on a basis that does not interfere with planned activities of the other Party.

ARTICLE 8 - TERMINATION

A. <u>Termination by Lessee</u>. Lessee shall have the right upon 365 days' advance written notice to terminate this Lease for any reason.

B. <u>Battelle Memorial Institute's/Washington Group International's Obligations</u>. If Lessee (or its successors or assignees) terminates this lease under Article 8.A. of this Lease prior to payment in full of the principal and interest of the Bonds, the terms of Article 33 of this Lease shall apply effective the day after the last day of Lessee's occupation of the Premises.

C. <u>Termination by Lessor</u>. Lessor may terminate this Lease as allowed by Article 22 of this Lease.

ARTICLE 9 - COMMON AREAS AND ALTERATIONS TO PREMISES BY LESSOR

A. The Lessee shall have nonexclusive use of all areas of the Premises designated by the Lessor as common areas for the use generally of the tenants of the Premises. The Lessor shall maintain the common areas in good condition.

B. Prior to the commencement of the original leasehold term, in order to ready the Premises for Beneficial Occupancy, the Lessor shall, at its own expense, complete the work that is necessary to establish Beneficial Occupancy as generally described in Exhibit B.

C. With the approval of Lessor, the Lessee has the right during the Lease term or any option or extension to make alterations or to attach fixtures and erect signs in or upon the Premises; and Lessor's approval shall not be unreasonably withheld except as may be required by law. The fixtures, alterations and/or signs placed in, upon or attached to the Premises are and remain the

property of the Lessee or the United States Government (Government) and may be removed by the Lessee prior to or upon the expiration of this Lease. At the option of Lessee, and with the written consent of Lessor, Lessee's improvements may be left on the Premises upon termination or expiration of the term or any option of this Lease. If left after Lessee vacates the Premises, the improvements become the property of the Lessor. If the improvements are removed by the Lessee, the Lessee agrees to restore the Premises to their condition prior to installation of Lessee's property, reasonable wear and tear excepted. Plans for structural change will be submitted to the Lessor for approval and approval will not be unreasonably withheld.

D. Title to property of the Government will not be affected by the incorporation of the property or its attachment to any property not owned by the Government. Government property will not become a fixture or lose its identity as personalty because of affixation to realty. Any damage to the Premises caused by the removal of Government property will be repaired and restored by Lessee to its condition prior to attachment of government property, less normal wear and tear.

E. If major capital alterations or modifications are desired and mutually agreed upon the Lessor will provide them and a new rental rate will be negotiated to cover the cost of installation. Any real estate taxes or assessments, if any, resulting from capital improvement of the Premises by Lessor are the responsibility of Lessor.

ARTICLE 10 - IMPROVEMENTS

A. After the commencement of any leasehold term, and at Lessee's own expense, the Lessee may make additions or improvements to the Premises after having obtained Lessor's prior written approval to do so. Lessor's concurrence of any additions or improvements shall eliminate any right to request restoration or removal upon termination of this Lease.

B. Lessee may, during any leasehold term, install in the Premises such furnishings, machinery, equipment and fixtures as Lessee deems necessary for its use of the Premises, provided such furnishings, machinery, equipment, or fixtures do not materially damage the Premises, are not hazardous to other tenants or users of the property upon which the Premises are situated, are not in conflict with Lessor's permits or any other applicable regulatory requirements, and do not unduly interfere with any other tenant's use and enjoyment of their use of the Premises. Should Lessee's installation of any such furnishings, machinery, equipment and/or fixtures require service upgrades to the Premises to support the same, Lessee shall be responsible for all costs related to such service upgrades. During the term of the Lessee. Upon early termination or expiration of the respective leasehold term, if there is no default by the Lessee in the Lease, the Lessee shall have the right to remove all such furnishings, machinery, equipment and fixtures from the Premises regardless of whether this personal property is attached to the Premises by piping, wiring, bolts, or otherwise. If so removed, Lessee shall repair any damage to Premises caused by the removal.

ARTICLE 11 - JANITORIAL SERVICES, MAINTENANCE AND REPAIR

A. All matters regarding maintenance and repair of the Premises (and Common areas if applicable) shall be referred to:

Name:	Greg Horton, Manager, Facilities Operations	
Office Address:	ISU/UI Center for Higher Education, Room #100 P.O. Box 8150 Idaho Falls, ID 83402	
Work Phone Number	: (208) 282-7823	
E-mail Address:	hortgreg@isu.edu	
Cell Number:	(208) 221-4061	
Home Address:	1850 Carmel Drive Idaho Falls, ID 83402	
Emergency Phone Number: (208) 282-2515		

Said individual or his/her designee shall be available at all times to receive such contacts.

B. The Lessor shall provide and pay for janitorial services, maintenance, repair and replacement of the Premises, including the building interior, exterior, grounds, and all equipment, fixtures, and appurtenances furnished by the Lessor under this Lease in order to keep the same in good repair and habitable condition, except for damage resulting from willful abuse or negligence of the Lessee. The appropriate share of these expenses (as provided in Article 3.B) will be billed to Lessee in the Service Rent. The Lessor shall have the right to enter upon the Premises at reasonable times in order to inspect the same and to perform such maintenance and repair, as well as replacement, but this right shall be exercised in a manner that does not unreasonably interfere with Lessee's use of the Premises.

C. Lessee shall operate and maintain the information technology (IT) infrastructure of the Premises. The cost for providing this service shall be allocated between the Parties pursuant to the terms of Article 3.B above and shall be accounted for in calculating Lessee's Service Rent. Lessor agrees that Lessee shall have 7 x 24 access to the entire Premises in order to operate and maintain the IT infrastructure.

D. Janitorial services, maintenance, repair, and replacement services by the Lessor, in accordance with Articles 11.B and 11.C, shall be agreed to in writing by the Parties. Such written agreement, which shall become Exhibit D of this Lease, shall specify the (i) scope and level of maintenance services to be performed and (ii) the Parties' pro-rata share of the costs of such services.

ARTICLE 12 - ASSUMPTION OF RISK

A. Lessor shall not be responsible for any injuries or damages incurred by Lessee, its agents, officers, employees, invitees or licensees arising from acts or omissions of any sublessee of the Lessee or from any cause other than the negligence or willful misconduct of Lessor or its employees.

B. Lessor shall take reasonable measures to maintain the Premises in safe and habitable condition and shall be responsible for injuries or damages incurred by Lessee, its agents, officers, employees, invitees or licensees arising out of or resulting from, and to the extent of, the negligence or willful misconduct of Lessor or its employees. Lessee shall be responsible for occupying and utilizing the space leased hereunder in a safe manner and shall be responsible for injuries or damages incurred by Lessor, its agents, officers, employees, invitees or licensees arising out of or resulting from, and to the extent of, the negligence or willful misconduct of Lessor or its employees, invitees or licensees arising out of or resulting from, and to the extent of, the negligence or willful misconduct of Lessee or its employees.

ARTICLE 13 - LIENS

Each Party shall keep the Premises and the property on which the Premises are situated, free from any liens or encumbrances arising out of any work performed, materials furnished or obligations incurred by that Party. If any such lien is filed against the Building, Lessee's leasehold interest or Lessor, the Party responsible for incurring such lien shall cause the same to be discharged within twenty (20) days after the date of filing the same.

ARTICLE 14 - LIABILITY INSURANCE

A. Lessee shall provide and maintain at its sole cost and expense the following minimum insurance coverage throughout the original term of the lease agreement and any extensions thereof:

- Comprehensive or Commercial Form General Liability Insurance (contractual liability included) in the minimum amount of five million dollars (\$5,000,000) per occurrence, and with an aggregate limit of not less than five million dollars (\$5,000,000);
- 2. Workers compensation insurance in accordance with the laws of the state of Idaho;
- 3. Comprehensive business automobile liability insurance, including operation of owned, scheduled, non-owned, and hired automobiles, covering bodily injury and property damage with a combined single limit of not less than five million dollars (\$5,000,000) per occurrence;

B. All insurance required hereunder shall be maintained in full force and effect through a company or companies reasonably satisfactory to Lessor. All insurance required under paragraphs A.1 and A.3, above, shall name (the State of Idaho and the Idaho State Board of Education, in its capacity as

an executive department of state government, and in its capacity as the Board of Regents of the University of Idaho, the Board of Trustees of Boise State University, and the Board of Trustees of Idaho State University, and each of their respective officers, employees, agents, and assigns (all of whom are collectively referred to as the **University Insureds**) as additional insureds), and shall contain a clause requiring written notice to Lessor thirty (30) days in advance of the cancellation, non-renewal, or material modification of said insurance as evidenced by return receipt of United States certified mail; provided, however, that Lessee's insurance shall name the University Insureds as additional insureds solely with regard to claims arising out of the Lessee's use of the Premises under this Lease; and provided further that nothing in this paragraph B shall be construed to extend Lessee's insurance policies to any of the University Insureds with regard to any claims that arise out of or result from the sole actions/inactions of the University Insureds. Coverage on a claims made basis shall survive for a period of not less than three (3) years after termination of this Lease. Certificates of insurance evidencing compliance with this Article shall be supplied contemporaneously to Lessor with the execution and delivery of this Lease. Said certificates shall evidence compliance with all sections of this Article.

ARTICLE 15 - PROPERTY INSURANCE

The Lessor shall provide and maintain "All Risk" property and casualty insurance on the Building, together with rental interruption insurance, insuring the Building against damage or loss resulting from fire or other casualty; provided, however, that Lessee shall pay its pro rata share of such cost of insurance as part of the Service Rent as provided in Article 3.B. Lessee shall be a named insured on such insurance. Lessor shall not provide personal property insurance on any of Lessee's personal property used, stored or otherwise situated within the Building, and Lessee shall bear all responsibility for any damage or loss to said personal property, regardless of the cause. Lessor will provide copies of the insurance policies to Lessee, Battelle, and WG.

ARTICLE 16 - CONDEMNATION

In the event the Premises are condemned for public use, whether the right condemned shall consist of the fee title or interest less than fee simple but of such nature as to render operations by the Lessee impractical or unfeasible, then this Lease shall forthwith terminate, without any further obligation by Lessee, Battelle, or WG under any provision of this Lease. Lessor shall not be obligated in any way to Lessee or Battelle as a result of such condemnation, except (i) to pay to Lessee any sums actually paid to Lessor by the condemning authority for rent paid by Lessee but not yet earned by Lessor, or for leasehold improvements owned by Lessee and (ii) to apply the condemnation proceeds, to the extent available, to pay in full the then-outstanding principal and interest of the Bonds.

ARTICLE 17 - DAMAGE OR DESTRUCTION

A. Except as provided in Article 15, if the Premises are damaged or destroyed by fire or any cause other than an act or omission of Lessee, its employees, agents, invitees, or licensees, Lessor shall restore the Premises, except for such fixtures, improvements and alterations as are installed by Lessee, as nearly as practicable to their condition immediately prior to such damage or destruction.

Lessee, at Lessee's expense, may so restore all such fixtures, improvements, and alterations installed by the Lessee. With regard to losses not covered by insurance, Lessor, at Lessee's expense, shall so restore the Premises with respect to all damage caused by any act or omission of Lessee, its employees, agents, invitees or licenses, and Lessee agrees to reimburse Lessor upon demand for all sums expended for such restoration. The obligations to restore provided in this paragraph shall be subject to Lessor's termination rights provided below. Any restoration shall be promptly commenced and diligently prosecuted, subject to availability of funds (which includes the proceeds of any applicable insurance policy) and to the terms and conditions of any applicable bond purchase or related agreement. Lessor shall not be liable for any consequential damages by reason of any such damage or destruction.

B. Notwithstanding any of the foregoing provisions of this Article, in the event the Premises shall be destroyed or damaged to such an extent that Lessor deems that it is not economically feasible to restore the same, then Lessor may terminate this Lease as of the date of the damage or destruction by giving the Lessee notice to that effect. Upon such termination, neither Lessee, Battelle, nor WG shall have any further obligation to Lessor under any provision of this Lease.

C. If Lessor undertakes to restore the Premises as provided within this Article, then commencing with the date of the damage or destruction and continuing throughout the period of restoration, the Service Rent and Base Rent for the Premises shall be abated for such period in the same proportion as the untenantable portion of the Premises bears to the whole thereof, except that there shall be no abatement to the extent that any such damage or destruction is caused by any act or omission of the Lessee, its employees, agents, invitees or licensees, in which case the Parties expect that rental interruption insurance shall apply.

D. If the damage to or destruction of the Premises (i) renders that portion of the Premises covered by this Lease untenantable in any degree and (ii) such loss or destruction is covered by insurance, the proceeds of such insurance shall be applied as follows: (a) the proceeds from the property and casualty insurance shall be applied to restore the Premises if Lessor determines to restore the Premises; and if Lessor determines not to restore the Premises, such insurance proceeds shall be applied first to payment in full of the then-outstanding principal and interest of the Bonds, with any remainder of the insurance proceeds payable to Lessor, and (b) the proceeds from the rental interruption insurance shall be used to pay the principal and interest coming due on the Bonds.

E. If the damage to or destruction of the Premises renders that portion of the Premises covered by this Lease untenantable in any degree and such loss or destruction is not covered by insurance, Battelle and WG shall be responsible to make payment of the principal and interest on the Bonds as they come due in proportion to their respective obligations described in Article 33.

ARTICLE 18 - CLOSURE AND SURRENDER OF PREMISES

Subject to the covenants and conditions set forth within this Lease, Lessee, at the expiration or sooner termination of this Lease, shall quit and surrender the Premises in good, neat, clean and

sanitary condition, except for reasonable wear and tear, and damage not caused by acts or omissions by Lessee, its employees, agents, invitees or licensees.

ARTICLE 19 - ACCESS TO PREMISES

Lessee will allow Lessor free access at all reasonable times to said Premises for the purpose of inspection and to fulfill any of Lessor's obligations under this Lease. Lessor shall have the right to place and maintain "For Rent" signs in a conspicuous place on said Premises for ninety (90) days prior to expiration of this Lease.

ARTICLE 20 - HOLDOVER

If Lessee lawfully holds over after the expiration of the term of this Lease, such tenancy shall be a month to month tenancy. During such tenancy Lessee agrees to pay Lessor the same rates as the just expired term, and to be bound by all the applicable terms, covenants and conditions herein specified. Such tenancy may be terminated by either Party upon giving thirty (30) days prior written notice to the other Party.

ARTICLE 21 - DISPUTES AND GOVERNING LAW

A. Pending resolution of a disputed matter, the Parties shall continue performance of their respective obligations pursuant to this Lease. Disputes regarding any factual matter relating to this Lease shall be discussed by the Parties' authorized representatives who shall use their reasonable efforts to amicably and promptly resolve the dispute. If the authorized representatives are unable to resolve any controversy or claim arising out of or relating to this Lease, or the breach thereof, the Parties agree that the controversy or claim shall be submitted to non-binding mediation by a mediator satisfactory to both Parties.

B. This Lease shall be governed by the law of the State of Idaho. Any action arising from this Lease shall be filed in State District Court in Bonneville County, Idaho.

ARTICLE 22 - DEFAULT

A. If any rents reserved, or any part thereof, shall be and remain unpaid when the same shall become due, or if Lessee shall violate or default in any of the covenants and agreements herein contained, then the Lessor may terminate this Lease upon giving sixty (60) days prior notice, and re-enter and take possession of said Premises. Notwithstanding such re-entry by Lessor, the liability of Lessee for the Base Rent and Service Rent provided herein shall not be extinguished for a period of 365 days. Lessee shall continue to pay the Base Rent and Service Rent as they become due, and covenants and agrees to make good to the Lessor any deficiency arising after re-entry and re-letting of the Premises at a lesser rental than herein agreed to for a period of no more than 365 days; provided however, to the extent Lessor's duty to mitigate damages is satisfied through the receipt of rentals from another tenant, Lessee's obligation to pay rentals (Base Rent and Service Rent) shall be reduced commensurately. The Lessee shall pay such deficiency each month as the
amount thereof is ascertained by the Lessor. Nothing in this section 22.A shall be construed to affect Battelle's or WG's obligations under Article 33.

B. The rights and remedies of the Lessor in this Article are in addition to any other rights or remedies provided by law and under this Lease.

C. Lessor will not be in default under this Lease unless Lessor fails to perform an obligation required of the Lessor under this Lease within thirty (30) days after written notice by registered mail from Lessee to Lessor and to the Trustee specifying the obligation that the Lessor has failed to perform. If more than thirty (30) days are reasonably required for performance or cure, Lessor shall not be in default if it commences performance within thirty (30) days after written notice. If Lessee's beneficial use of a substantial portion of the Premises is materially impaired by Lessor's default, the opportunity to cure cannot exceed one hundred eighty (180) days. If, at the end of such 180-day cure period, Lessor has not cured the default, this Lease (including all obligations of Lessee, Battelle, and WG to Lessor) may at the Lessee's option (and with notice to the Lessor and the Trustee) be terminated. In the case of damage or destruction of any portion of the Premises, Lessor shall not be considered to be in default hereunder if it is reasonably proceeding with repair or replacement of the damaged or destroyed Premises.

ARTICLE 23 - LIMITATION OF LIABILITIES

A. Neither Party shall be liable to the other Party for indirect, consequential, or special damages whether based on tort, contract, strict liability or other legal or equitable theory or action.

B. Neither Party shall be liable under this Lease for, or be considered to be in material breach or default under this Lease on account of any delay in or failure of performance due to Force Majeure. Force Majeure is defined as any event, cause or condition beyond a party's reasonable control (such events, causes or conditions include but are not limited to: fire, flood, earthquake, volcanic activity, wind, and other acts of the elements; court order and act or failure to act of civil, military or governmental authority; strike, lockout and other labor disputes; riot, insurrection, sabotage and war; breakdown of or damage to facilities or equipment; and any act or omission of any person or entity except an act or omission of such party or of such party's contractors or suppliers of any tier or anyone acting on behalf of such party that is within the reasonable control of such party or of such party's contractors or suppliers of any tier acting on behalf of such party), which prevents or delays the party claiming the Force Majeure from performing its obligations under this Lease; provided, however, that any party claiming Force Majeure shall be entitled to a delay only to the extent, despite the exercise of due diligence, it is unable to overcome the effects of the Force Majeure event. In the event of Force Majeure, the time for performance thereby delayed shall be extended by a period of time reasonably necessary to compensate for such delay. Nothing in this paragraph shall require either party to settle any strike, lockout or other labor dispute. Each party shall give the other party prompt written notice of any event it considers to be a Force Majeure.

Final 10/06/06

ARTICLE 24 - NOTICES

All notices, demands, and requests to be given by either party to the other shall be in writing and served either personally or sent by United States mail, postage pre-paid, to the addresses below or such other addresses as may be designated by the parties from time to time:

TO LESSOR at:	Vice President for Financial Services Idaho State University Campus Box 8219 Pocatello, ID 83209
TO LESSEE at:	Battelle Energy Alliance, LLC ATTN: Director, Supply Chain Management P.O. Box 1625 Idaho Falls, ID 83415
TO TRUSTEE at:	U.S. Bank National Association ATTN: corporate Trust Department P.O. Box 7928 Boise, ID 83707
TO Battelle at:	Russell P. Austin Battelle Memorial Institute 505 King Avenue Columbus, Ohio 43201-2693
TO WG at:	Washington Group International ATTN: Associate General Counsel 106 Newberry Street, SW Aitken, SC 29801

ARTICLE 25 - WAIVER OF RIGHTS

The failure of either Party to insist upon strict performance of any of the covenants and agreements of this Lease, or to exercise any option or right herein conferred, shall not be construed to be a waiver or relinquishment of any such option or right, or any other covenants or agreements, but the same shall be and remain in full force and effect.

ARTICLE 26 - TRANSFER OF OBLIGATION

The covenants and agreements of this Lease shall be binding upon the heirs, legal representatives, successors and agreed assigns of any or all the Parties hereto.

ARTICLE 27 - HAZARDOUS SUBSTANCES TRACKING AND COMMUNICATION

A. Lessee is responsible for tracking, recording, and proper use and disposal of all of the hazardous substances that are received, stored, handled or disposed of by Lessee on or from the Premises, including spills or accidents involving hazardous substances within the Premises, and both planned and unplanned releases to the environment. Lessee shall maintain appropriate inventory and material balance records for their material, accordingly.

B. Lessee shall provide Lessor with current data documenting such tracking and recording required under Paragraph A above, and cooperate with Lessor to integrate such data into a CAES computerized data base system.

ARTICLE 28 - ORDER OF PRECEDENCE

The Contract comprises the following documents in the order of precedence set forth below:

- 1. CAES Lease Agreement and Amendments thereto
- 2. Exhibit C, Base Rental Rate Agreement
- 3. Exhibit A, Description of Premises
- 4. Exhibit B, Commencement Date Agreement
- 5. Exhibit D, Janitorial Services, Maintenance and Repair

The above order of precedence controls in the event of any conflict, inconsistency or ambiguity in the terms and conditions set forth within these documents.

ARTICLE 29 - SHARED USE OF SPACE AND EQUIPMENT

The Parties acknowledge and agree that the cooperation, collaboration, and shared use of space, equipment and personnel for research are important to the successful operation of this facility. In that regard, the provisions of this Lease shall be interpreted in such a manner as to support such purposes.

ARTICLE 30 - ARTICLES INCORPORATED BY REFERENCE

The following Federal Acquisition Regulation (FAR) clauses and Federal Acts are incorporated herein by reference.

- 1. Affirmative Action for Workers With Disabilities, FAR 52.222-36 (Jun 1998)
- 2. Americans with Disabilities Act, 28 CFR Part 36, (as revised July 1, 1994).
- 3. Equal Opportunity, FAR 52.222-26 (Feb 1999)
- 4. Limitations on Payments to Influence Certain Federal Transactions, FAR 52.203-12 (Jun 1997)
- 5. Anti-Kickback Act of 1986 (Jul 1995)
- Equal Opportunity for Special Disabled Veterans, Veterans of the Vietnam Era, and Other Eligible Veterans (Dec 2001), <u>FAR 52.222-35</u>. (Dec 2001)

ARTICLE 31 - EXAMINATION OF RECORDS

Unless exempted by applicable law, Lessor and Lessee agree that the State of Idaho and the Comptroller General of the United States or the DOE Inspector General or any duly authorized representatives shall have access to and the right to examine any books, documents, papers, and records of Lessor or Lessee that are relevant to each such Party's performance under this Lease.

ARTICLE 32 - ASSIGNMENT

Except as otherwise provided in this Lease, the interest of either Party in this Lease shall not be assigned to a third-party without the written consent of the other Party, which consent shall not be unreasonably withheld.

ARTICLE 33 - STATUS OF BATTELLE MEMORIAL INSTITUTE AND WASHINGTON GROUP INTERNATIONAL

- A. Parent Company Assurance:
 - <u>Battelle Memorial Institute</u>. For the purpose of providing partial financing to assist in construction of the Building, Battelle Memorial Institute (**Battelle**) is a party to this Lease for the sole and limited purpose of providing secondary security, as provided in Article 33.D below, for the payment of principal and interest on the Bonds in the principal amount of \$8.88 million (**Battelle Guaranty**), payment of principal and interest on such Bonds being referred to hereafter as **Bond Debt**. The primary security for the Bond Debt is the Base Rent due Lessor under this Lease. Except as provided in Article 33.D, Battelle has no obligation to Lessor under this Lease. The Guaranty is being made by Battelle for good and valuable consideration.
 - 2. <u>Washington Group International</u>. For the purpose of providing partial financing to assist in construction of the Building, Washington Group International (WG) is a party to this Lease for the sole and limited purpose of providing secondary security, as provided in Article 33.D below, for the payment of principal and interest on the Bonds in the principal amount of \$1.12 million (WG Guaranty), payment of principal and interest on such Bonds being referred to hereafter as Bond Debt. The primary security for the Bond Debt is the rentals due Lessor under this Lease. Except as provided in Article 33.D, WG has no obligation to Lessor under this Lease. This Guaranty is being made by WG for good and valuable consideration, including without limitation, satisfaction of WG's obligations to assist in the development of projects such as CAES.
 - 3. <u>Pro-Rata Obligations</u>. The obligations of Battelle and WG to pay the Bond Debt are pro-rated such that any insufficiency in the Base Rent hereunder to pay the Bond Debt shall be satisfied 88.8% by Battelle and 11.2% by WG.

- B. <u>Amortization of Bond Debt</u>. The Parties agree that the Base Rent paid to Lessor under Article 3.A of this Lease shall be applied to pay the Bond Debt of both Battelle and WG in proportion to the Bond Debt each has guarantied as provided in Article 33.A above. Payments made to Lessor under Article 3.B for maintenance of the Building shall not be considered rental payments for purposes of this Article 33.B
- C. <u>Successor Contractor</u>. If (i) the Lessee's contract with DOE (Contract No. De-AC07-05ID14517) expires or is terminated before payment in full of the Bond Debt and (ii) Lessee is succeeded by another contractor, Lessor hereby consents to Lessee's assignment of this Lease to the successor contractor; and Battelle's and/or WG's obligations described in paragraph D below apply to the assigned Lease. The payments of Base Rent made to Lessor by such successor contractor shall be applied as prescribed in Article 33.B above. Lessor agrees that despite assignment of this Lease to any successor contractor, the provisions of this Article 33 with respect to the release of Battelle and/or WG upon payment in full of the Bond Debt is made for the express benefit of Battelle and WG and is enforceable by Battelle and WG with respect to their respective portions of the Bond Debt.

D. Battelle's/WG's Obligations and Rights.

- 1. If the Base Rent payments made under this Lease, as provided in Articles 33.B and 33.C above, are insufficient at any time or from time to time to pay in full the Bond Debt, as and when the same shall come due (in whole or in part), or if Lessee (or its successors or assignees) terminates the Lease pursuant to Article 8, Battelle and WG shall be responsible, in proportion to their respective guaranties of the Bond Debt as provided in Article 33.A above, for making payment to the Trustee (as assignee of Lessee) of amounts sufficient to pay in full the Bond Debt. These assurances of payment by Battelle and WG are guaranties of payment when due and not just collection from Lessee. These guaranties of payment of the Bond Debt shall survive the termination of this Lease, except as otherwise expressly provided herein.
- 2. If Lessee's Base Rent payments are insufficient for Trustee to make any semiannual payment on the Bond Debt when such payment comes due, Lessor shall notify Battelle and WG of (i) the total amount of such deficiency and (ii) the amount of such deficiency that is allocable to Battelle and WG in proportion to their respective guaranties of the Bond Debt as provided in Article 33.A above.

Battelle and WG agree to pay to the Trustee their respective share of such insufficiency within 15 days of their receipt of notice by the Lessor or the Trustee of such insufficiency.

- 3. The Parties agree that the sole obligation of Battelle and WG under Article 33.D.2 shall be to make payments, in their respective proportions as provided in Article 33.A, necessary to cure the deficiency in Lessee's Base Rent payments necessary to make the semi-annual payment on the Bond Debt when the same has come due. In the event that Battelle or WG shall fail for any reason to pay their respective share of the Bond Debt within 60 days of their receipt of notice of insufficiency, the Lessor may, at its sole option, demand payment from Battelle or WG, as the case may be, of the full amount of their respective shares of the Bond Debt then outstanding plus interest thereon (at the Bond interest rate) to the next date upon which the Bonds may be redeemed.
- 4. If Battelle's/WG's guaranty obligations hereunder are triggered by the Lessee's termination of the Lease, Battelle and/or WG, at their sole discretion, reserve the right to assume assignment of the Lease. Upon acceptance of assignment of this Lease, Battelle and/or WG shall succeed to the rights and obligations of Lessee or any successor contractor under this Lease; provided however, that Battelle and/or WG shall accept assignment of this Lease solely for the time necessary to amortize the Bond Debt and once the Bond Debt is amortized, Battelle and/or WG shall be immediately and automatically released from any and all obligations under this Lease.
- E. <u>Release of Battelle/WG</u>. Once the Bond Debt is paid in full as provided under the terms of this Lease, Battelle and WG shall be released from any and all obligations under this Lease.
- F. <u>Sublease</u>. The Parties agree that (i) Lessee or (ii) Battelle and/or WG provided Battelle and/or WG become the Lessee under Article 33.D above, may enter into subleases under this Lease regarding the Leased Premises; provided that any such sublease shall be consistent with the CAES mission as expressed in either this Lease or the BEA prime contract.
- G. <u>Individual Guaranties</u>. The Parties acknowledge and agree that the obligations of Battelle and WG under this Lease, including without limitation their respective guaranties, are

individual and separate as opposed to joint and several. Therefore, (i) in the event Battelle fails to meet its Battelle Guaranty obligation, no Party shall have any right to pursue WG for the same and (ii) in the event WG fails to meet its WG Guaranty obligation, no Party shall have any right to pursue Battelle for the same.

ARTICLE 34 – APPLICATION OF CONSTRUCTION CONTINGENCY FUNDS

If, upon completion of the construction of the Building, Lessor's construction manager has any funds in its account that were contributed by Lessee, the Parties agree that such remaining funds shall be applied to the next-due Bond payment; provided, the Parties agree to make a corresponding adjustment to the next-due Base Rent payment.

ARTICLE 35 - ENTIRE AGREEMENT

This Lease and Exhibits "A", "B", "C" and "D" hereto contain the entire and integrated agreement of the Parties and may not be modified or amended except in writing signed and acknowledged by both Parties.

In the process of securing financing, certain provisions of this Lease may require amendment or conformance with the financing documents. Both Parties agree to cooperate and act in good faith in considering and approving required amendments. In addition, Lessee, Battelle and WG agree to make reasonable disclosures related to and necessary for the process of securing financing.

ARTICLE 36 – BOARD AND CONTRACTING OFFICER APPROVAL

This Lease is executed by the designated representatives of the Parties and entities identified in the signature article below subject to (i) the final approval of the Lease by the State Board of Education and (ii) the Lease satisfying the limiting conditions prescribed in a certain Department of Energy letter dated October 4, 2006 from the contracting officer regarding the Lessee's signature authority for this Lease. Upon the satisfaction of the two foregoing contingencies, this Lease shall be effective immediately and without further consent of or action by the Parties.

ARTICLE 37 – COUNTERPART SIGNATURES

The signatories to this Lease, below, agree that this Lease may be executed in any number of counterparts with the same effect as if all the signatories had signed the same document, and all counterparts shall be construed together and shall constitute one written document.

LESSOR: Idaho State University	President Title	<u>10/23/06</u> Date
LESSEE: Battelle Energy Alliance, LLC	Title	Date
BATTELLE MEMORIAL INSTITUTE	Title	Date
WASHINGTON GROUP INTERNATIONAL ACKNOWLEDGED AND CONSENTED	Title TO:	Date
University of Idaho	Title	Date
Boise State University	Title	Date

Final 10/06/06

ARTICLE 37 - COUNTERPART SIGNATURES

The signatories to this Lease, below, agree that this Lease may be executed in any number of counterparts with the same effect as if all the signatories had signed the same document, and all counterparts shall be construed together and shall constitute one written document.

LESSOR: Idaho State University Title Date 19/06 LESSEE: Battelle Energy Alliance, LLC Title 13/06 President & CEO MEMORIAL INSTITUTE Title R/ Date 0/12/66 EVP & CFO BATTELLE MEMORIAL INSTITUTE Title WASHINGTON GROUP Title Date **INTERNATIONAL** ACKNOWLEDGED AND CONSENTED TO: University of Idaho Title Date Boise State University Title Date Final 10/06/06 18

BAHR - SECTION II

ARTICLE 37 - COUNTERPART SIGNATURES

The signatories to this Lease, below, agree that this Lease may be executed in any number of counterparts with the same effect as if all the signatories had signed the same document, and all counterparts shall be construed together and shall constitute one written document.

LESSOR: Idaho State University	Title	Date
LESSEE: Battelle Energy Alliance, LLC	Title	Date
BATTELLE MEMORIAL INSTITUTE	Title	Date
WASHINGTON GROUP INTERNATIONAL ACKNOWLEDGED AND CONSENTED T	Sr. V.P., Strategic Programs, for <u>President of Energy & Environment</u> Title	<u>Oct. 11, 2006</u> Date
University of Idaho	Title	Date
Boise State University	Title	Date

Final 10/06/06

ARTICLE 37 - COUNTERPART SIGNATURES

The signatories to this Lease, below, agree that this Lease may be executed in any number of counterparts with the same effect as if all the signatories had signed the same document, and all counterparts shall be construed together and shall constitute one written document.

LESSOR: Idaho State University	Title	Date
LESSEE: Battelle Energy Alliance, LLC	Title	Date
BATTELLE MEMORIAL INSTITUTE	Title	Date
WASHINGTON GROUP INTERNATIONAL	Title	Date
ACKNOWLEDGED AND CONSENTED	TO: <u>Vice President, Finance</u> Title and Administration	<u>16 Oct</u> 06 Date
Boise State University	Title	Date

BAHR - SECTION II

Exhibit A

LEASE AGREEMENT NO. 00049377

DESCRIPTION OF PREMISES

The Center for Advanced Energy Studies (CASE) building will be at least 55,000 square feet located on seven acres of State of Idaho property in Idaho Falls, Idaho. The property is part of approximately 19 acres adjacent to land owned by the Idaho State University Foundation. The CAES building is located at 1000 University Boulevard in Idaho Falls, Idaho 83402.

The legal description of the seven acres is: Beginning at the Southeast corner of Lot 1, Block 1, Bureau of Land Management Administration Building Site, Division No. 1, Bonneville County, Idaho; running thence N.10°05'38"W. along the Lot line 707.53 feet; thence N.32°17'32"W. along the Lot line 247.90 feet; thence S.89°52'14"W. 300.96 feet; thence S.10°05'38"E. 693.95 feet to the South line of said Lot 1, thence S.62°58'40"E. along said South line 498.18 feet to the point of beginning.

1



BAHR - SECTION II



Site Location for the Center for Advanced Energy Studies Building

When completed, the CAES building will consist of offices for senior administrators, researchers, and graduate students. It will also have laboratories for research in hydrogen, advanced materials, chemistry, radiochemistry, system modeling, visualization, analytical instrumentation, and imaging to be used collaboratively by members of the CAES team. Space will also be provided for collection and temporary storage of hazardous waste.



The CAES site will provide approximately 168 parking spaces for the building occupants and users.

EXHIBIT B

LEASE AGREEMENT NO. 00049377

COMMENCEMENT DATE AGREEMENT

WHEREAS, by Lease Agreement dated ______, 2006 ("Lease") Lessor leased to Lessee approximately ______ gross square feet of particular space, located in the City of Idaho Falls, Bonneville County, Idaho, (hereinafter called the "Premises") for a period of twenty (20) years ("Term") to commence and expire as provided in Article 2. A of the Lease; and

WHEREAS, Article 2.B of the Lease states that the Commencement Date shall be when the requirements for beneficial occupancy have been met. By fixing a commencement date and termination date and signing this document, Lessor and Lessee agree that those requirements have been met.

NOW, THEREFORE, Lessor and Lessee hereby agree that the Commencement date of the Term shall be ______, 2008, and the Expiration date shall be ______, 2028, unless terminated as otherwise provided in the Lease.

IN WITNESS WHEREOF, the Parties hereunto have executed this Agreement as of the date first above written.

LESSEE:

LESSOR:

BATTELLE ENERGY ALLIANCE, LLC

IDAHO STATE UNIVERSITY

By:	By:	
Title:	Title:	-
Date:	Date:	

EXHIBIT C

LEASE AGREEMENT NO. 00049377

BASE RENT

Agreement made this _____ day of _____ 2006, by and between Idaho State University, an institution of higher education and agency of the State of Idaho, hereinafter called "Lessor" and Battelle Energy Alliance, LLC, a Limited Liability Company organized under the laws of the State of Delaware, hereinafter call "Lessee".

WHEREAS, by Lease Agreement dated ______, 2006 ("Lease") Lessor leased to Lessee approximately ______ gross square feet of particular space, located in the City of Idaho Falls, Bonneville County, Idaho, (hereinafter called the "Premises") for a period of twenty (20) years to commence and expire as provided in Article 2.A of the Lease; and

WHEREAS, Article 3.A of the Lease provides for the Lessor and the Lessee to execute this Exhibit C once the exact monthly rental rate (Base Rent) is known.

Lessor and Lessee hereby further agree that if the Bond Debt is amortized entirely before the termination of the Lease, the Base Rent shall be \$1.00 per square foot. Total square footage leased by Lessee shall be determined by Article 1 of this Lease.

IN WITNESS WHEREOF, the Parties hereunto have executed this Agreement as of the date first above written.

 LESSEE:
 LESSOR:

 BATTELLE ENERGY ALLIANCE, LLC
 IDAHO STATE UNIVERSITY

 By:_____
 By:_____

 Title:_____
 Title:______

 Date:______
 Date:______

BAHR - SECTION II

Final 10/06/06

EXHIBIT "D"

AGREEMENT REGARDING MAINTENANCE AND OPERATIONS SERVICES

BAHR - SECTION 16/06

TAB 4¹Page 79

Table of Contents		
1. Introduction	Page 3	
2. Purpose	4	
3. Organization	4	
3.1 Center for Advanced Energy Studies Steering Committee3.2 Idaho State University3.3 Idaho Falls Center for Higher Education Facilities	4 4	
Maintenance and Management Staff	5	
4. Space Assignments and Utilization	5	
 4.1 Assignment of Space Within the CAES Building 4.1.1 Assigned Space 4.1.2 Multi-use Space 4.1.3 Common Space 	5 5 5 6	
4.2 Service Rent (Occupancy Costs)	6	
4.3 Equipment and Furnishings	6	
5. Facility Maintenance and Operations	6	
5.1 Maintenance 5.1.1 Routine Maintenance 5.1.2 Renovation and Remodeling	6 7 7	
 5.2 Building Operation 5.2.1 Energy Conservation 5.2.2 Telephone and Information Technology (IT) Service 5.2.3 Signage 5.2.4 Parking 	7 7 7 7 8	
6. Public Safety and Security	8	
 6.1 Public Safety 6.1.1 Safety 6.1.2 Security 6.2 Hazardous and Radioactive Materials 	8 8 8	

Signature Block:

1. Introduction

The Center for Advanced Energy Studies (CAES) is a Joint Institute, which brings together the talents, facilities and capabilities of Idaho State University, University of Idaho, Boise State and the Idaho National Laboratory, (including the prime contractor, Battelle Energy Alliance (BEA)). The mission of CAES is to address critical science issues that resolve the many challenges associated with providing an appropriate mix of advanced energy technologies in relation to critical United States and global energy needs. CAES will address a variety of energy sources that are areas critical to ensuring U.S. energy security. Areas of particular concern include, but are not limited to: affordability, limited mission environmental impacts, and leadership in the global energy arena. Advanced energy sources to be researched include nuclear, hydrogen, fossil fuels (coal, oil and gas), and the full spectrum of renewable energy sources.

CAES will advance the education of the next generation of scientists and engineers, engage in long-term university-based research activities, and host a range of national and international events. Activities are being designed to facilitate an informed debate, which will address the questions and issues concerning the best energy technology mix necessary to meet U.S. global needs. This dialogue will present facts about the benefits of various energy sources including nuclear, fossil, renewable, energy in the world energy, and environmental debate, and conduct a wide range of academic and public activities. The Center will serve as a hub for an Idaho universities network and as a conduit for developing collaboration, partnerships and connectivity between researchers.

The CAES facility is located on the Idaho Falls Center for Higher Education campus in Idaho Falls, Idaho. Contributing members of the Center use the building and grounds to collaborate, conduct research and education in support of the Center's mission.

2. Purpose

Pursuant to Article 11 of the Lease Agreement Between Idaho State University and Battelle Energy Alliance, LLC for the Center for Advance Energy Studies, (the "Lease"), the Parties have agreed to set forth in Exhibit "D" of the Lease, the scope and level of maintenance services to be performed by ISU and the parties' pro-rata share of the costs of such services. The purpose of this Exhibit D is to set forth those services and costs and to outline and agree upon basic policies and procedures for the operation and maintenance of the Center for Advanced Energy Studies facility. The CAES Steering Committee (CAESSC) recommends changes to policies and procedures outlined in this Exhibit D, and the Parties hereto must agree in writing, to any amendments.

3. Organization

3.1 Center for Advanced Energy Studies Steering Committee

The CAESSC is responsible for the oversight of the Building's operations and management. The CAESSC consists of one representative from each CAES member, as set forth in the <u>Memorandum of Agreement previously entered into</u> between BSU, ISU, U of I and BEA.

The CAESSC is the first level of institutional partnership and will meet as scheduled by the chair. The CAESSC is responsible for governance and recommendations of:

- 1. Space assignment and use;
- Major renovations or remodeling (changes to interior or exterior structure and organization);
- Policy issues regarding the management and operation of the CAES facility; and
- 4. Resolution of facility use conflicts.

3.2 Idaho State University

Idaho State University (ISU) is the Lessor pursuant to the Lease and is responsible for collecting rents and managing funds for bond repayment and

facility maintenance. ISU will also administer facility maintenance, renovation, and remodeling projects, and will coordinate with the Idaho Division of Public Works and other Idaho state agencies as necessary.

Safety and security will be administered by ISU Public Safety. Control of hazardous materials will be coordinated as set forth in Section 6.2 herein and in compliance with the terms set forth in the Lease.

3.3 Idaho Falls Center for Higher Education Facilities Maintenance and Management Staff

Day-to-day operation and maintenance of the Building is the responsibility of ISU. The facility administrator and the maintenance and operations staff at the Idaho Falls Center for Higher Education will operate, clean, and maintain the Building. Backup support will be provided through contracted service agreements and from the ISU Pocatello campus.

4. Space Assignments, Utilization and Equipment

4.1 Assignment of Space Within the CAES Building

Facilities within the CAES building are classified as assigned, multi-use, and common.

4.1.1 Assigned Space

Use of offices and research laboratory spaces not leased by BEA will be requested from the CAESSC through the CAES Director. The CAES Director will review all such requests and assign space based on need.

4.1.2 Multi-use Space

Multi-use spaces include all laboratory and laboratory-support spaces not leased to BEA or to other CAES Members or to third parties. These spaces may be temporarily assigned by the CAES Director or left open for occasional use. The cost of maintenance and operations will be the responsibility of the agency to which the space is assigned.

4.1.3 Common Space

Subject to the terms of Article 9.A of the Lease, space used by all building occupants (such as restrooms, hallways, mechanical rooms and custodial closets) are common spaces.

4.2 Service Rent (Occupancy Costs)

Service Rent shall be allocated between the Parties as provided in Articles 3.B and 11 of the Lease.

4.3 Equipment & Furnishings

4.3.1 Furnishings

Lessor and Lessee will provide their own office furnishings. Furnishings must be of high quality to match the decor of the CAES building.

4.3.2 Equipment

Lessor and Lessee will provide their own laboratory equipment, as needed. Equipment will be inventoried into the building by ISU. Subject to the terms of Article 9.C of the Lease, the equipment ownership will be retained by the providing agency and will be inventoried into their respective systems. Maintenance and replacement of the equipment will be under the direction of the owners in coordination with the CAES Director. Once placed in the laboratories, the equipment use will be coordinated with the CAES Director.

5. Facility Maintenance and Operations

5.1 Maintenance

5.1.1 Routine Maintenance

Building maintenance, landscape, and custodial services will be accomplished with ISU employees and/or service contractors under the supervision of ISU.

5.1.2 Renovation and Remodeling

Renovations and improvements shall occur as prescribed in Articles 9 and 10 of the Lease.

5.2 Building Operation

5.2.1 Energy Conservation

The Building is constructed to the highest energy conservation standards reasonable under budget restrictions. The programs operating within the Building are involved in advanced CAES studies of energy sources and systems. In keeping with the spirit of the mission, the Building should operate at peak energy efficiency. All reasonable actions will be taken to keep the Building operating within acceptable energy standards and still support the efforts of the occupants.

5.2.2 Telephone Support

Telephone service and support will be provided by ISU with actual costs covered by Lessor and Lessee in accordance with their respective use.

5.2.3 Signage

Except as provided in Article 11.C of the Lease, ISU shall be responsible for constructing and installing signs in and around the Building. Room numbers and laboratory names will be installed during construction. The cost for name plaques and other agency signs will be the responsibility of the users. All signage, other than name plaques, requires the approval of the CAES Director.

5.2.4 Parking

Parking permits will be required for the Building's parking lot. Lessor shall issue a sufficient number of parking permits to Lessee at no cost to Lessee. The standard Idaho Falls Center for Higher Education permit will be used and is valid on the Idaho Falls and Pocatello campuses. Sufficient visitor and disability parking will be identified and the lot controlled during special events.

6. Public Safety and Security

6.1 Public Safety

6.1.1 Safety

Safety is the responsibility of all users of the facility. Consistent with Article 7, all users of the facility shall follow ISU safety policies and procedures found electronically at http://www.isu.edu/departments/pubsafe/. Regarding the leased Premises, Lessee may supplement these procedures, but ISU policies and procedures must be followed as a minimum. When ISU, UI, or BSU employees are using space assigned to BEA, such employees shall follow BEA-established safety requirements.

General laboratory procedures will be the responsibility of the ISU Technical Safety Office and developed under the direction of the CAESSC in coordination with the Director of CAES and consistent with Article 7 of the Lease.

6.1.2 Security

The CAES facility is designed as open-access to facilitate collaboration. Security badges and check points will not be required unless directed by the CAESSC. A receptionist will be strategically located to assist visitors.

For the convenience of Lessee, crime prevention and reporting policies and procedures can be found electronically at http://www.isu.edu/departments/pubsafe/.

6.2 Hazardous and Radioactive Materials

Consistent with Article 27 of the Lease, each Party will be responsible for tracking, recording, and proper use and disposal of all hazardous substances used by that Party, including spills or accidents involving

hazardous substances within the premises, and both planned and unplanned releases to the environment. The Parties shall maintain appropriate inventory and material balance records for their own materials. The Parties shall integrate such data into a CAES computerized database system.

THIS PAGE INTENTIONALLY LEFT BLANK

ATTACHMENT 4



Department of Energy

Idaho Operations Office 1955 Fremont Avenue Idaho Falls, ID 83415

October 4, 2006

Mr. Greg B. Wray, Director Supply Chain Management Battelle Energy Alliance, LLC P. O. Box 1625 Idaho Falls, ID 83415-1303

- SUBJECT: Contract No. DE-AC07-05ID14517 Request for Approval of the Lease Agreement for the Center for Advanced Energy Studies Facility (AS-CMD-P&P-07-004)
- REFERENCE: Letter, G. B. Wray to M. L. Adams, Subject: Contract No. DE-AC07-05ID14517 – Request for Consent of Lease Agreement No. 000493777 for the Center for Advanced Energy Studies Facility – Supplement, dated September 25, 2006

Dear Mr. Wray:

We have received your request for approval of the lease agreement for the Center for Advanced Energy Studies (CAES) facility. Based on our review, we approve the proposed CAES lease with the understanding that the terms of the final lease agreement will result in a facility that will be a minimum of 55,000 square feet, the Idaho National Laboratory will lease 70% of the facility, and the lease rate will not exceed \$24.00 per square foot.

Also, please notify me immediately if you negotiate any changes to the draft lease agreement or to any aspect of the business deal that may affect allowable costs or our conclusion that the lease is an operating lease under OMB Circular A-11.

If you have any questions, please contact the undersigned at 526-5277.

Sincerely,

Michael L. Adams Contracting Officer Contract Management Division

THIS PAGE INTENTIONALLY LEFT BLANK

FIRST AMENDMENT TO AGREEMENT Between the Office of the Governor of the State of Idaho (State) and the Regents of the University of Idaho (Regents) and the Trustees of Idaho State University (Trustees), collectively "the Parties"

Whereas, the Parties entered into an Agreement, effective March 1, 2001, a copy of which is attached hereto as Exhibit A and incorporated herein by reference;

Whereas the original project and project management envisioned at the time of the effective date of the Agreement has changed; and

Whereas the Parties still desire to pursue a project of the same nature and for the same purpose;

Now therefore in consideration of the mutual promises contained herein and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Parties agree as follows:

Effective Date of this Amendment. This Amendment shall be effective as of March 1, 2006.

Amendments. The Parties now desire to extend the term of the Agreement, and to amend the terms of the Agreement, as follows:

A. Purpose and Scope: The purpose and scope of the Agreement in paragraph 1 shall be amended to have the State provide the Funds to the Trustees from the INEEL Settlement Fund for the construction of a Center for Advanced Energy Studies in Idaho Falls, Idaho. The Funds shall be used in accordance with the criteria, terms and conditions set forth in the Agreement as amended.

B. Lead University: The Trustees shall become the lead university and all individual references to the Regents in the Agreement shall change to the Trustees, as the Trustees shall be the recipient of the Funds.

C. Term: The term of the Agreement in paragraph 2 shall be extended from the effective date of this Amendment until such time as the Trustees report that the funds have been expended according to the terms of the Agreement, and the recordkeeping requirements in paragraph 8 have expired.

D. Payments: Paragraph 3 shall be amended to provide as follows. The Trustees shall provide the Office of the Governor, Division of Financial Management, with notice when all of the conditions in subparagraph B have been met, and the State shall promptly transfer to the Trustees the Funds for the construction of a Center for Advanced Energy Studies in Idaho Falls, Idaho.

FIRST AMENDMENT TO AGREEMENT BEWTEEN THE STATE AND UNIVERSITIES -- 1

E. Designated Key Personnel: Paragraph 11 of the Agreement shall be amended to designate the following as key personnel:

Robert A. Wharton Academic Vice President Idaho State University Campus Box 8063 Pocatello, ID 83209

Kenneth R. Prolo Vice President for Financial Services Idaho State University Campus Box 8219 Pocatello, ID 83209

F. Reporting and Auditing Requirements: Paragraph 8 of this Agreement is amended as follows. When all Funds have been expended according to the terms of this Agreement and the Division of Public Works has issued a Certificate of Substantial Completion permitting occupancy of the building, the Trustees shall provide the State with a final report with regard to the expenditure of the Funds. No other reporting and/or auditing requirements shall be required with regard to the Funds. Notwithstanding, the recordkeeping requirements of the Trustees and Universities, respectively, which are described in subparagraphs A through J of paragraph 8, shall continue in effect.

G. Contracting Officer: Paragraph 12 of the Agreement is deleted.

H. Notice: Paragraph 21 of the Agreement shall be amended to reflect the following addresses for providing notice:

If to the State:

Office of the Governor 700 W. Jefferson Boise, ID 83720

If to the Universities:

Kenneth R. Prolo Vice President for Financial Services Idaho State University Campus Box 8219 Pocatello, ID 83209

FIRST AMENDMENT TO AGREEMENT BEWTEEN THE STATE AND UNIVERSITIES - 2

Effect. If there is any conflict between the terms and provisions of this Amendment and the terms and provisions of the Agreement, the terms and provisions of this Amendment shall govern. Except as specifically set forth herein, all other provisions of the Agreement shall remain in full force and effect and be binding upon the Parties in accordance with the terms therein.

STATE_OF IDAHO:

Date: 3/28/06

DIRK KEMPTHORNE Governor

REGENTS OF THE UNIVERSITY OF IDAHO:

Redenil Ferrer

Date: 4/18/06

By: RODERIC W. LEWIS, President

TRUSTEES, IDAHO STATE UNIVERSITY:

By:' RODERIC W. LEWIS, President

Date: __//18/06

FIRST AMENDMENT TO AGREEMENT BEWTEEN THE STATE AND UNIVERSITIES -- 3

EXHIBIT A TO FIRST AMENDMENT TO AGREEMENT:

,

MARCH 1, 2001 AGREEMENT BETWEEN THE STATE, REGENTS & TRUSTEES

.

AGREEMENT

This Agreement (Agreement) is entered into by and between the Office of the Governor of the State of Idaho (State) and the Regents of the University of Idaho, a public corporation, state educational institution, and a body politic and corporate organized and existing under the Constitution and laws of the State of Idaho (Regents) and the Trustees of Idaho State University, a body politic and corporate organized and existing under the Constitution and laws of the State of Idaho (Trustees). The Regents and Trustees are collectively referred to herein as "the Universities," and the Universities and the State are collectively referred to herein as "the Parties."

1. PURPOSE AND SCOPE

To satisfy the intent and terms of a court settlement, dated October 16, 1995, and a subsequent settlement agreement, dated September 23, 1996, (collectively, the "Settlement") between the State and the U.S. Department of Energy ("DOE"), the State will provide funds (the "Funds") to the Regents from the Idaho National Engineering and Environmental Laboratory ("INEEL") Settlement Fund (as defined in Idaho Code §67-806A). The Funds are to be used to create jobs, diversify the economy of southeastern Idaho and induce investment in the technology-based strengths of the region by providing financial assistance for the construction of a Center for Science and Technology ("the Facility") in Idaho Falls, Idaho. The Funds shall be used in accordance with the criteria, terms and conditions set forth in this Agreement.

2. TERM & EFFECTIVE DATE

Upon signature by the Governor of the State of Idaho, this Agreement shall become effective as of March 1, 2001. The Agreement shall remain in effect until all funds have been transferred to the Regents by the State and the Regents and Trustees have expended such funds according to the terms of this Agreement, or until March 1, 2006, whichever is earlier.

3. PAYMENTS

A. The State will provide the Regents with five million dollars (\$5,000,000) subject to the conditions established herein. Of this sum, the State shall provide the Universities with \$100,000 within fourteen (14) calendar days of execution of this Agreement for the purpose of completing the conditions for transfer of the remaining \$4,900,000, as set forth below.

B. The State will provide the Regents with the remaining \$4,900,000 only upon completion of the following actions to the State's satisfaction within the term of this Agreement.

1. Approval of the Facility by the State of Idaho Board of Education and the Regents.

AGREEMENT BETWEEN THE STATE AND UNIVERSITIES PAGE 1 of 11

2. Acquisition of an interest by either the Regents or the State Board of Education in the real property on which the Facility will be located sufficient that a reasonable business person would build a comparable facility on that real property.

3. Development by the Universities of a Project Planning Guide containing the elements listed in Attachment A, or whose contents are otherwise acceptable to the State, and completion of the process described in the Project Planning Guide.

C. Payments shall be in accordance with State of Idaho laws and rules. The Office of the Governor, Division of Financial Management, shall be the financial officer for this Agreement. The Parties may agree upon a means of electronic payment of money to the Regents pursuant to this Agreement.

4. **REFUND OF MONEYS**

To the extent costs for the construction of the Facility are less than the allocated amount, including any interest or income accumulated thereon, or in the event the Universities are unable or choose not to proceed with this project, the Regents shall refund any and all unexpended moneys for this project to the State, within fourteen (14) calendar days of: (1) completion of the project, (2) a decision not to proceed with the project, or (3) March 1, 2006, whichever is earliest.

5. **RESPONSIBILITY FOR COST OVERRUNS**

The Regents shall be solely responsible for all costs of construction of the Facility that exceed the amount of Funds paid to the Regents pursuant to this Agreement and earmarked for those purposes (and any interest or income generated thereon).

6. BANKING OF FUNDS

A. The Regents shall manage the portion of Funds under their control in a manner consistent with state laws and rules.

B. The Regents may only expend interest and investment income accumulated on the portion of Funds under their control in a manner consistent with the terms of this Agreement. The Regents shall account for interest accumulated on the portion of Funds under their control as set forth in Section 8.

7. **REPRESENTATIONS OF THE UNIVERSITIES:**

The Universities represent and warrant that:

- 1. they are Idaho State nonprofit institutions of higher learning;
- 2. they have the power to enter into and perform under this Agreement and have taken all necessary steps to properly exercise such power;

AGREEMENT BETWEEN THE STATE AND UNIVERSITIES

PAGE 2 of 11

3. the persons executing this Agreement on behalf of the Universities are fully authorized to do so and can bind the Universities by their signatures;

4. no person has been employed or retained to solicit or secure this Agreement upon an agreement or understanding for a commission, percentage, brokerage or contingent fee;

5. they will cooperate fully with all governmental agencies as necessary to fulfill the goals and purposes of this Agreement and the use of the Funds; and

6. they will pay, when due, any and all wages, salaries, obligations, and taxes owed by them related to this Agreement.

8. REPORTING AND AUDITING REQUIREMENTS

A. The Regents shall provide the State with semi-annual financial statements of the expenditures and revenues related to the construction of the Facility.

B. The Regents shall provide the State copies of all semi-annual reports submitted to the U.S. Department of Housing and Urban Development related to the Facility.

C. To comply with the provisions of the Settlement between DOE and the State of Idaho, and, to the extent funds are appropriated by the U.S. Congress pursuant to Section 3161 of the National Defense Authorization Act for Fiscal Year 1993, the State must consider the following criteria in determining on what projects and programs moneys from the INEEL Settlement Fund will be expended:

1. projected number of jobs created;

2. quality of jobs created, including but not limited to earning potential and sustainability;

3. ability to induce investment or growth consistent with the identified and potential economic strengths of the region;

4. ability to reduce Idaho's economic dependence on DOE and DOE-related activities;

5. past performance of the applicant or the type of project or program;

6. amount of local participation in the project;

7. integration with other state economic development efforts; AGREEMENT BETWEEN THE STATE AND UNIVERSITIES

PAGE 3 of 11

8 ability of the project to become self-sufficient; and

9. other unique factors, such as innovative features of the project.

D. For the duration of this Agreement, plus an additional period of five (5) years, the Universities shall provide the State with semi-annual performance reports to assist the State in complying with the terms of the Settlement. Reports shall include at a minimum:

1. a narrative summary of the: projected number of jobs created; quality of jobs created, including but not limited to earning potential and sustainability; amount of local participation in the project; integration with other state economic development efforts; and Facility revenues from contracts and grants.

2. a brief status report for the construction and operation of the Facility, including a description of any major problems or concerns;

3. a description of any significant milestones completed during the reporting period; and

4. known or anticipated changes to schedules.

E. For the reporting requirements set forth in paragraphs A and D above, reporting periods shall begin on January 1 and July 1. Financial statements and performance reports shall be due within 30 calendar days of the close of the reporting period.

F. The Universities agree to keep records that are sufficient to permit the preparation of reports required by the State under the terms of the Settlement and to permit the tracing of the Funds to a level of expenditures adequate to ensure that the Funds have been spent lawfully and in accordance with the terms and conditions of this Agreement.

G. The Universities agree to maintain for the duration of this Agreement, plus an additional period of at least five (5) years, all books, records, documents, and other evidence of accounting procedures and practices, which sufficiently and properly reflect all direct and indirect costs of any nature expended in the performance of or related to this Agreement. If any audit or other action involving records is initiated before the five (5) year period has expired, the Universities will retain such records until all issues are resolved, or for three (3) additional years, whichever is longer.

H. The State reserves the right to audit or examine, in such a manner and at all reasonable times as it deems appropriate, all activities of the Universities arising in the course of performance of, or related to, this Agreement.

I. Prior to or upon termination of this Agreement, a final independent audit or examination of costs and expenditures shall be performed This obligation may be met via audits performed under the AGREEMENT BETWEEN THE STATE AND UNIVERSITIES PAGE 4 of 11
Single Audit Act of 1984 on an annual basis. The State shall have the right to recover an appropriate amount after fully considering the recommendations on disallowed costs resulting from the final audit or examination when conducted.

J. Notwithstanding anything contained in this Agreement, the Universities recognize and acknowledge that DOE, or other federal agencies, may also have audit rights with respect to the Funds, which audit rights may or may not be exercised on the same basis or at the same time as the State's audit rights under this Agreement. Any apparent approval by the State with regard to the Funds, or the failure of the State to affirmatively assert a disapproval of the Universities' acts pursuant to the State's audit rights shall not be construed or deemed to be a waiver on the part of the State to exercise any rights under this Agreement, including but not limited to the right to recover the Funds, or any portion thereof, upon a federal audit.

9. TERMINATION

. :

A. This Agreement may be terminated if the State determines that the Universities are in default of their obligations. A determination by the State to terminate under this section shall be a final determination.

B. The Contracting Officer shall provide written notice to the Universities of default by the Universities. If the Universities do not cure such default within ten (10) calendar days after receipt of such written notice, the State may terminate this Agreement and may pursue any and all legal, equitable, and other remedies available to the State.

C. For purposes of this Section, the Universities shall be in "default" if:

1. The Universities fail or refuse to perform under or in accordance with this Agreement; or

2. The Universities fail to abide by, or disregard, any applicable statutes, ordinances, rules, regulations, directives or orders of any governmental entity related to the performance of activities under this Agreement.

10. EFFECT OF TERMINATION

A. Upon termination of this Agreement for any reason, the Regents shall return, within fourteen (14) calendar days of written notice from the State, any unused portion of the Funds.

B. If the State determines that any portion of the Funds was expended in breach of this Agreement, the State has the right to recoup that portion of the Funds from the Regents, their contractor(s) or subcontractor(s). The Regents shall require any individual or entity receiving Funds, or any portion thereof, to acknowledge and abide by the requirements of this section. The requirements of this section shall be included in all agreements, contracts, subcontracts or assignments entered into by the Universities related to this Agreement or the Funds.

AGREEMENT BETWEEN THE STATE AND UNIVERSITIES

PAGE 5 of 11

11. DESIGNATED KEY PERSONNEL

The Universities shall notify the State in writing of any change in key personnel. The following are designated key personnel for the Universities during the term of this Agreement:

Laura Hubbard Director, Capital Planning and Budget University of Idaho P.O. Box 443162 Moscow, Idaho 83844-3162 Robert Stiger Dean, University of Idaho, Idaho Falls Center 1776 Science Center Drive Idaho Falls, ID 83402

12. CONTRACTING OFFICER

The Idaho Department of Commerce shall serve as the Contracting Officer for the State. The State shall notify the Universities in writing of any change in Contracting Officer. The Contracting Officer shall respond to all written requests made by the Universities within fourteen (14) calendar days. Failure to respond within the time allotted shall constitute approval of the request.

13. COMPLIANCE WITH APPLICABLE LAWS

The Universities shall comply with all applicable federal, state, and local laws and regulations in performing all work under this Agreement. The Universities shall require any individual or entity receiving the Funds, or any portion thereof, to acknowledge and comply with this section. The requirements of this section shall be included in any agreements, contracts, subcontracts or assignments entered by the Universities related to this Agreement or the Funds.

14. PROHIBITIONS ON USE OF FUNDS AND ACTIVITIES

A. The Regents shall not use Funds provided by this Agreement to pay administrative costs.

B. The Regents shall not use Funds provided by this Agreement to pay for services rendered or goods or equipment purchased prior to the effective date of this Agreement.

C. The Regents shall not use Funds provided by this Agreement to fund the operations or maintenance of the Facility, or to fund any other project, including but not limited to, training and educational services.

D. Funds provided by this Agreement shall not be used to influence or attempt to influence an officer or employee of any governmental agency, any member, officer or employee of Congress or the Idaho Legislature in connection with the awarding, continuation, renewal, amendment, or modification of any contract, grant, loan, or cooperative agreement.

AGREEMENT BETWEEN THE STATE AND UNIVERSITIES

PAGE 6 of 11

E. The Universities shall not:

1. knowingly hire any ineligible individual(s), embezzle, willfully misapply, steal or obtain by fraud the Funds, or any portion thereof;

2. induce any employee to give up any money or thing of value under threat of dismissal;

3. willfully obstruct or impede any investigation related to this Agreement;

4. directly or indirectly promise any employment, position, compensation, contract, appointment or other benefit made possible in whole or in part by the Funds to any person as consideration, favor or reward for any political activity or for the support of, or opposition to, any candidate or political party in connection with any election, primary or caucus;

5. directly or indirectly knowingly cause or attempt to cause any person to make a contribution of a thing of value (including services) for the benefit of any political cause, candidate or party by means of denial or threat of denial of any employment or other benefit funded by or related to the Funds; or

6. solicit or accept unlawful or illicit gratuities, favors or anything of monetary value under the operation of this Agreement.

F. The Regents and Trustees shall comply with all applicable federal, state and local laws and regulations, including but not limited to those prohibiting discrimination and requiring equal opportunity, in performing all work under this Agreement. The Regents and the Trustees shall require an individual or entity receiving Funds or any portion thereof, to acknowledge and comply with this Section. The requirements of this Section shall be included in any agreements, contracts, subcontracts, or assignments entered by the Regents or the Trustees related to this Agreement,

G. The Universities recognize and acknowledge that criminal penalties may result from theft, embezzlement or other prohibited activities under this Agreement.

H. The Universities shall immediately report to the State any alleged or suspected incident of fraud, abuse or other prohibited activity related to this Agreement or the Funds.

15. PROPERTY

The Facility shall be considered property of the Regents and must be managed in accordance with applicable Idaho laws and rules.

16. INSURANCE REQUIREMENTS

A. The Parties understand that each is insured with respect to tort liability by the Bureau of Risk Management, a statutory system of self insurance established by Idaho Code §67-5776, and subject to AGREEMENT BETWEEN THE STATE AND UNIVERSITIES PAGE 7 of 11 the limits and requirements of the Idaho Tort Claims Act, Idaho Code §6-901 et seq. Each Party agrees to accept that coverage as adequate insurance of the other Party with respect to personal injury and property damage.

B. The Regents shall require any of its contractors during the construction of the Facility to maintain Workers Compensation Insurance as required by statute and Commercial General Liability Insurance and Builder's Risk Insurance in accordance with the guidelines established by the Bureau of Risk Management. The Regents shall require Professional Liability Insurance covering the design of the Facility in an amount of no less than one million dollars (\$1,000,000). If the Professional Liability Insurance required by this section is obtained through a "claims made" policy, this coverage or its replacement shall have a retroactive date of no later than the inception of this Agreement. Such errors and omissions insurance or its replacement shall also provide a minimum of five (5) years' extended reporting coverage, or the maximum under the State of Idaho Statute of Limitations for claims under this coverage, whichever is greater, after the services are last provided under the design of the Facility.

17. LIABILITIES AND LOSSES

The Parties agree that any liability claim, suit, or loss resulting from or arising out of the Parties' performance of activities under this Agreement shall be allocated, as between the state agencies, in accordance with the guidelines adopted by the Department of Administration's Bureau of Risk Management. Each party to this Agreement agrees to notify the Bureau of Risk Management and the other party in the event it receives notice or knowledge of any claims arising out of the performance of activities under this Agreement. To the extent a determination is made that a claim is not within the coverage provided by the insurance policies procured by the Bureau of Risk Management or its self-insurance program, and to the extent allowed by law, the Universities shall be primarily responsible for conducting the defense of and/or payment of such claim.

18. ASSIGNMENT AND SUBCONTRACTING

A. The Regents may contract with other entities to construct the Facility. No other contract or assignment of responsibilities under this Agreement may be made without the State's prior written consent.

B. Any assignment, contract or subcontract shall not relieve the Regents of their obligations under this Agreement, and the Regents shall be responsible for the satisfactory performance by any assignee, contractor or subcontractor. Assignments, contracts or subcontracts permitted under this section will not be deemed to require any additional payment by the State.

19. GOVERNING LAW

This Agreement shall be governed by the laws of the State of Idaho and performed therein.

AGREEMENT BETWEEN THE STATE AND UNIVERSITIES

PAGE 8 of 11

20. APPROPRIATION BY LEGISLATURE

A. This Agreement shall in no way or manner be construed so as to bind or obligate the State beyond the term of any particular appropriation of funds by the Idaho State Legislature or the United States Congress as may exist from time to time. In the event the Legislature of the State of Idaho, or the United States Congress, fails, neglects, or refuses to appropriate such funds as may be required and designated to continue payment of this Agreement, this Agreement shall be at such time automatically terminated. All future rights and liabilities of the Parties shall thereupon cease within thirty (30) calendar days after notice to the Universities.

B. It is the intent of the parties that the Universities will not require the Idaho State Legislature to appropriate any moneys to fund any portion of the construction or continued operation and maintenance of the Facility. It is the intent of the parties that these activities be self-supporting and will be funded from revenues generated by use fees, federal grants and performance contracts.

21. NOTICE

Any notice required under this Agreement shall be in writing and shall be delivered by certified or registered mail, return receipt requested, to the following addresses:

If to the State: CST Contract Officer Idaho Department of Commerce P.O. Box 83720 Boise, ID 83720-0093 If to the Universities: Director, Capital Planning and Budget University of Idaho P.O. Box 443162 Moscow, ID 83844-3162

Either party may modify the above notice information by written notice as provided for in this section.

22. NONWAIVER OF BREACH

The failure of the State to insist upon strict performance of any of the terms and conditions of this Agreement, or to exercise any option herein conferred in any one or all instances, shall not be construed to be a waiver or relinquishment of any such covenant or condition, but the same shall be and remain in full force and effect unless such waiver is evidenced by the prior written consent of the State.

23. SEVERABILITY

If any provision of this Agreement is held invalid, the remainder of the Agreement shall not be affected, and the rights and obligations of the Parties shall be construed and enforced as if the Agreement did not contain the particular provision held to be invalid.

AGREEMENT BETWEEN THE STATE AND UNIVERSITIES

PAGE 9 of 11

24. MERGER

This Agreement and related Attachment constitute the entire agreement between the Parties and shall supersede all previous proposals, oral or written, negotiations, representations, commitments, and all other communications between the Parties concerning the scope of this Agreement. It may not be released, discharged, changed, extended or modified and no claim for additional services not specifically provided herein will be allowed by the State, except to the extent provided by an instrument in writing signed by a duly authorized representative of the Universities and the State.

Date: 6-29-01

DIRK KEMPTHORNE Governor

EN MOGEE

President, Regents of the University of Idaho

Date: 6 21-01

Date: 6-21-01

KAREN McGEE President of the Trustees, Idaho State University

ATTACHMENT A: PROJECT PLANNING GUIDE ELEMENTS

Adoption/Acceptance (Sign Off) by Key Participants/Executives Executive Summary of the PPG Contents Definition of project process, scope, schedule, and budget Project process steps and master calendar of project phases Purpose of the project (mission, stakeholders, desired outcomes) Project Delivery Strategy Site Assessment: Master plan context (district and campus) Real property ownership/agreements/easements Criteria for site selection Site access Existing site features/considerations Zoning and jurisdictional considerations Environmental conditions Goals and objectives for development Infrastructure: Existing conditions **Technical** requirements Jurisdictional considerations Delivery responsibilities Building: Goals and objectives (uses, occupants/tenants) Design Criteria Functional requirements Space relationships Future expansion Technical criteria Maintenance and operational considerations Space requirements (detailed program of needs) Furnishings, fixtures, equipment needs Cost Model: Estimates of all anticipated project expenses Funding and fundraising parameters (revenues) Schedule: Pre-design through Construction phases Other Considerations: (As warranted)

AGREEMENT BETWEEN THE STATE AND UNIVERSITIES - ATTACHMENT A Page 11 of 11

THIS PAGE INTENTIONALLY LEFT BLANK

REFERENCE – APPLICABLE STATUTE, RULE OR POLICY

Idaho State Board of Education GOVERNING POLICIES AND PROCEDURES SECTION: V. FINANCIAL AFFAIRS Subsection: B. Budget Policies

April 2002

B. Budget Policies

8. Major Capital Improvement Project -- Budget Requests

For purposes of Item 8., the community colleges (NIC and CSI), the State Historical Society, and the State Library are included, except as noted in V.B.8.b. (2).

a. Definition

A major capital improvement is defined as the acquisition of an existing building, construction of a new building or an addition to an existing building, or a major renovation of an existing building. A major renovation provides for a substantial change to a building. The change may include a remodeled wing or floor of a building, or the remodeling of the majority of the building's net assignable square feet. An extensive upgrade of one (1) or more of the major building systems is generally considered to be a major renovation.

- b. Preparation and Submission of Major Capital Improvement Requests
 - (1) Permanent Building Fund Requests

Requests for approval of major capital improvement projects to be funded from the Permanent Building Fund are to be submitted to the Office of the State Board of Education on a date and in a format established by the executive director. Only technical revisions may be made to the request for a given fiscal year after the Board has made its recommendation for that fiscal year. Technical revisions must be made prior to November 1.

(2) Other Requests

Requests for approval of major capital improvement projects from other fund sources are to be submitted in a format established by the executive director. Substantive and fiscal revisions to a requested project are resubmitted to the Board for approval. This subsection shall not apply to the community colleges.

c. Submission of Approved Major Capital Budget Requests

The Board is responsible for the submission of major capital budget requests for the institutions, school and agencies under this subsection to the Division of Public Works. Only those budget requests which have been formally approved by the Board will be submitted by the office to the executive and legislative branches.

BUSINESS AFFAIRS AND HUMAN RESOURCES NOVEMBER 30 – DECEMBER 1, 2006

Idaho State Board of Education GOVERNING POLICIES AND PROCEDURES SECTION: V. FINANCIAL AFFAIRS Subsection: K. Construction Projects

April 2002

K. Construction Projects

1. Major Project Approvals - Proposed Plans

Without regard to the source of funding, before any institution, school or agency under the governance of the Board begin formal planning to make capital improvements, either in the form of renovation or addition to or demolition of existing facilities, when the cost of the project is estimated to exceed five hundred thousand dollars (\$500,000), must first be submitted to the Board for its review and approval. All projects identified on the institutions', school's or agencies' six-year capital plan must receive Board approval.

2. Project Approvals

Without regard to the source of funding, proposals by any institution, school or agency under the governance of the Board to make capital improvements, either in the form of renovation or addition to or demolition of existing facilities, when the cost of the project is estimated to be between two hundred fifty thousand dollars (\$250,000) and five hundred thousand dollars (\$500,000), must first be submitted to the executive director for review and approval. Without regard to the source of funding, proposals by any institution, school or agency under the governance of the Board to make capital improvements, either in the form of renovation or addition to or demolition of existing facilities or construction of new facilities, when the cost of the project is estimated to exceed five hundred thousand dollars (\$500,000), must first be submitted to the Board for its review and approval. Project cost must be detailed by major category construction cost, architecture fees, contingency funds, and other). When a project is under the primary supervision of the Board of Regents or the Board and its institutions, school or agencies, a separate budget line for architects, engineers, or construction managers and engineering services must be identified for the project cost. Budgets for maintenance, repair, and upkeep of existing facilities must be submitted for Board review and approval as a part of the annual operating budget of the institution, school or agency.

3. Fiscal Revisions to Previously Approved Projects

Project revisions that substantially alter the use of the project causing changes in project costs between two hundred fifty thousand dollars (\$250,000) and five hundred thousand dollars (\$500,000) must first be submitted to the executive director for review and approval. Changes in project costs of more than five hundred thousand dollars (\$500,000) must first be submitted to the Board for its review and

approval. Requests must be supported by a revised detailed project budget and justification for changes.

4. Project Acceptance

Projects under the supervision of the Department of Administration are accepted by the Department on behalf of the Board and the state of Idaho. Projects under the supervision of an institution, school or agency are accepted by the institution, school or agency and the project architect. Projects under the supervision of the University of Idaho are accepted by the University on behalf of the Board of Regents.

- 5. Statute and Code Compliance
 - a. All projects must be in compliance with Section 504 of the Rehabilitation Act of 1973 and must provide access to all persons. All projects must be in compliance with applicable state and local building and life-safety codes and applicable local land-use regulations as provided in Chapter 41, Title 39, and Section 67-6528, Idaho Code.
 - b. In designing and implementing construction projects, due consideration must be given to energy conservation and long-term maintenance and operation savings versus short-term capital costs.

INSTITUTION/ AGENCY AGENDA IDAHO STATE UNIVERSITY

SUBJECT

Authorization to issue General Revenue Bonds (2006 Bonds) to provide additional financing for the construction of the Center for Advanced Energy Studies (CAES) building.

REFERENCES

April 2005 August 2005 December 2005 December 2005 Overview of CAES at SBOE regular meeting Update information on CAES at SBOE regular meeting Approved motion to issue a Request for Proposal (RFP) through the Division of Public Works (DPW) for selection of an architect and proceed with the pre-design/programming for the CAES Facility.

APPLICABLE STATUTE, RULE, OR POLICY

Idaho State Board of Education Governing Polices & Procedures, Section V.F.1.

BACKGROUND

The Battelle Energy Alliance (BEA) was awarded the contract from the Department of Energy to manage the new Idaho National Laboratory (INL) for nuclear energy research. As part of the contract, Battelle committed to establish CAES which is designed to become a nationally and internationally recognized focal point for the advancement of education in energy science and technology. A Memorandum of Agreement was signed by the Governor, the University Presidents from BSU, ISU and UI, and Battelle Memorial Institute to work towards the establishment of a joint laboratory/university center which would serve as a research center for the INL with the construction of a State-owned building to house the CAES.

DISCUSSION

The CAES facility is planned to open in June 2008 and will have a minimum of 55,000 square feet. The facility is envisioned to be a two-story, structural steel building with a brick façade. The facility will be located on state property north of the ISU/UI Center for Higher Education. When fully occupied, the CAES facility will have a total of up to 175 people, including approximately 100 faculty, researchers, and staff; 50 graduate students; and 25 undergraduate students.

Documents relating to the issuance and sale of the 2006 Bonds are being prepared by bond counsel (Ballard Spahr Andrews & Ingersoll, LLP). The University is meeting with Standard and Poor's Public Finance Ratings and Moody's Investors Service the first week in November to obtain a rating on the 2006 Bonds. The rating will be available at the November 30, 2006, State Board of Education meeting.

INSTITUTION/ AGENCY AGENDA IDAHO STATE UNIVERSITY - continued

The 2006 Bonds would be issued as Additional Bonds under the Board's Resolution adopted September 17, 1992 (Original Resolution), as previously amended and supplemented, and under the Supplemental Resolution authorizing issuance of the 2006 Bonds (Supplemental Resolution). The 2006 Bonds would be issued on parity with all other bonds issued under the Original Resolution, secured by the pledge of the University's Pledge Revenues, including but not limited to the University's Student Facilities Fee/Facilities, the Matriculation Fee, and the Revenues of the Housing System.

DOCUMENTS ATTACHED

Attachment 1 – Preliminary Official StatementPage 5Attachment 2 – Supplemental ResolutionPage 45Attachment 3 – Bond Purchase AgreementPage 73

IMPACT

The design and construction of the facility is estimated to cost approximately \$17 million. Funding for the facility is based upon:

(1) \$4.9 million (\$100,000 previously expended) from the INEEL Settlement Fund, as defined in the Idaho Code 67-806A, for use according to the terms of the agreement for the construction of the Center for Science and Technology in Idaho Falls (dated June 29, 2001), between the Office of the Governor of the State of Idaho and the Regents of the University of Idaho and the Trustees of Idaho State University;

(2) \$1,942,756 from two grants from the U.S. Department of Housing and Urban Development (HUD) to the University of Idaho: HUD Grant B-00-SP-ID-0116 in the amount of \$925,000 and HUD Grant B-01-SP-ID-0172 in the amount of \$1,017,756 for use according to the terms of the grant (approximately \$300,000 has been expended from one of the grants for preliminary design for the previous CST facility);

(3) \$10 million through the issuance of bonds, using ISU's bond capacity, to be retired over 20 years supported by rent paid by BEA and its affiliates for occupancy of approximately 70% of the CAES facility.

STAFF COMMENTS AND RECOMMENDATIONS

As noted in the "Subject", the action requested is for an amount that is greater than has been previously discussed with the Board and staff. Earlier financing discussions have mentioned \$7.0 million bond revenue; the requested action in this motion is for an amount of \$10.0 million. To satisfy the research requirements from the three universities and INL, additional space is required above the space supported by the \$7 million guarantee. INL presented this need

INSTITUTION/ AGENCY AGENDA IDAHO STATE UNIVERSITY - continued

to Battele Memorial (BMI) and Washington Group International (WGI). BMI agreed to increase their guarantee from \$7 million to \$8.8 million and WGI agreed to guarantee \$1.2 million to ensure that the total space would be at least 55,000 square feet.

BOARD ACTION

Action on this agenda items consists of a motion to approve Supplemental Resolution for the 2006 Bonds. The title of the Supplemental Resolution is as follows:

A SUPPLEMENTAL RESOLUTION authorizing the issuance and sale of \$10 million General Revenue Bonds, Series 2006, of the Board of Trustees of Idaho State University; authorizing the execution and delivery of a bond purchase agreement and providing for other matters relating to the authorization, issuance, sale and payment of the Series 2006 Bonds.

Motion:

A motion to approve a Supplemental Resolution authorizing the issuance and sale of \$10 million in General Revenue Bonds, Series 2006, of the Board of Trustees of Idaho State University; authorizing the execution and delivery of a bond purchase agreement and providing for other matters relating to the authorization, issuance, sale, and payment of the Series 2006 Bonds.

Moved by _____ Seconded by _____Carried Yes____ No ____

[Note: motion must be approved by a majority of the members of the Board – the Board's Bylaws provide for approval with a roll-call vote.]

Idaho State Board of Education GOVERNING POLICIES AND PROCEDURES SECTION: V. FINANCIAL AFFAIRS Subsection: F. Bonds and Other Indebtedness

April 2002

F. Bonds and Other Indebtedness

1. General Powers

The University of Idaho, Idaho State University, Lewis-Clark State College, and Boise State University may, by a majority vote of all the members of the Board, borrow money with or without the issuance of bonds pursuant to Chapter 38, Title 33, Idaho Code. The Board must act by formal resolution. Such indebtedness is not an obligation of the state of Idaho but is an obligation solely of the respective institutions and the respective board of trustees. Any indebtedness is to be used to acquire a project, facility, or other asset that may be required by or be convenient for the purposes of the institution. Student fees, rentals, charges for the use of the projected facility, or other revenue may be pledged or otherwise encumbered to pay the indebtedness. Refunding bonds also may be issued.

Eastern Idaho Technical College is not authorized to borrow money under Chapter 38, Title 33, Idaho Code.

PRELIMINARY OFFICIAL STATEMENT DATED _____, 2006

NEW ISSUE – Issued in Book-Entry-Only Form

Insured/Underlying Ratings: ______"___"; "____" ______"; "____" [______ Insured] (See "Ratings" herein)

In the opinion of Ballard Spahr Andrews & Ingersoll, LLP, Bond Counsel to the University, interest on the Series 2006 Bonds is <u>not</u> excludable from gross income for federal income tax purposes. Bond Counsel expresses no opinion regarding any other tax consequences relating to the ownership or disposition of, or the accrual or receipt of interest on, the Series 2006 Bonds. See "TAX MATTERS" herein.



\$10,000,000 IDAHO STATE UNIVERSITY GENERAL REVENUE BONDS (FEDERALLY TAXABLE), SERIES 2006

Dated: Date of Delivery

Due: April 1, as shown on inside cover

The \$10,000,000 Idaho State University General Revenue Bonds (Federally Taxable), Series 2006 (the "Series 2006 Bonds") are issued as fully registered bonds and, when delivered, will be registered in the name of Cede & Co., as nominee of the Depository Trust Company, New York, New York ("DTC"). DTC will act as securities depository for the Series 2006 Bonds. Ownership interests in the Series 2006 Bonds will be in minimum denominations of \$5,000 and multiples thereof. Beneficial Owners of the Series 2006 Bonds will not receive physical bonds, but will receive a credit balance on the books of the nominees of such purchasers. Interest on the Series 2006 Bonds is payable on April 1 and October 1 of each year commencing April 1, 2007. Principal, prepayment premium, if any, and interest due with respect to the Series 2006 Bonds will be payable by U.S. Bank National Association, Salt Lake City, Utah, as Trustee, to DTC, which will, in turn, remit such principal, prepayment premium, if any, and interest due with respect to the Series 2006 Bonds.

The Series 2006 Bonds are subject to optional and mandatory redemption prior to maturity as described herein.

The Series 2006 Bonds are being issued by Idaho State University (the "University") to (i) finance a portion of the costs of the Center for Advanced Energy Studies at the Idaho Falls Center for Higher Education Campus (the "CAES Project"), which will be jointly occupied by a consortium which consists of the three Idaho research universities and Battelle Energy Alliance, LLC a limited liability company organized under the laws of the State of Delaware ("BEA"), and (ii) paying the costs of issuance of the Series 2006 Bonds.

Pursuant to the CAES Lease (defined herein), the University has leased to BEA a portion of the CAES Project and BEA has agreed to pay lease payments (the "CAES Lease Payments") which the University will pledge as part of the Pledged Revenues (described below). Under certain circumstances, payment of CAES Lease Payments are guaranteed by Battelle Memorial Institute ("BMI") and the Washington Group International ("WG"). Because the CAES Lease Payments are expected to be a small portion of the Pledged Revenues, no information is provided in this Official Statement with respect to BEA, BMI or WG and the University makes no representations herein as to the ability of BEA, BMI or WG to make its respective payments (if any) under the CAES Lease and purchasers of the Series 2006 Bonds are advised to look at the Pledged Revenues as a whole, rather than the CAES Lease Payments alone, as the security for the Series 2006 Bonds.

The State Board of Education, acting as the Board of Trustees for Idaho State University (the "Board") serves as the governing body for the University. Under Idaho law, the Board is a body politic and corporate and an independent instrumentality of the State of Idaho. The Series 2006 Bonds are being issued under a supplemental bond resolution (the "Supplemental Resolution") adopted by the Board on November _____, 2006. The Series 2006

Bonds are being issued as "Additional Bonds" pursuant to a Resolution adopted by the Board on September 17, 1992, providing for the issuance of revenue bonds, as amended and restated (the "Original Resolution"). The Original Resolution provided for the issuance of an initial series of revenue bonds and authorized the issuance of additional series of revenue bonds pursuant to Supplemental Resolutions, if certain conditions are met. See "SECURITY FOR THE SERIES 2006 BONDS" herein. The Original Resolution, as previously amended and supplemented and as amended and supplemented by the Supplemental Resolution, is referred to herein as the "Resolution." The revenue bonds issued pursuant to the Resolution, including the Series 2006 Bonds, are collectively referred to herein as the "Bonds." The Bonds are secured by a pledge of the Board to levy and collect certain student fees known as the Student Facilities Fee/Facilities and the Tuition Fee, plus Revenues of the Housing System (as defined in the Resolution), revenues from the CAES Project and certain other revenues. See "SECURITY FOR THE SERIES 2006 BONDS" herein.

The Series 2006 Bonds are limited obligations of the Board and do not constitute a debt or liability of the State of Idaho, its Legislature, or any of its political subdivisions or agencies other than the University and then only to the extent herein described. The Board is not authorized to levy or collect any taxes or assessments other than the revenues and fees described herein to pay the Series 2006 Bonds. The Board has no taxing power.

[The scheduled payment of principal of and interest on the Series 2006 Bonds when due will be guaranteed under an insurance policy to be issued concurrently with the delivery of the Series 2006 Bonds by .]

[LOGO OF BOND INSURER]

The Series 2006 Bonds are offered when, as and if issued and accepted by the Underwriter, subject to prior sale and to the delivery of an approving opinion by Ballard Spahr Andrews & Ingersoll, LLP, as Bond Counsel, and to other conditions. Certain legal matters will be passed upon for the University by University Counsel, Bradley H. Hall, Esq., Pocatello, Idaho. Certain legal matters will be passed on for the Board and the University by the office of the Attorney General of the State. It is expected that the Series 2006 Bonds will be available for delivery on or about ______, 2006.

This cover page contains certain information for quick reference only. It is not a summary of this issue. Investors must read the entire Official Statement to obtain information essential to the making of an informed investment decision.

This Official Statement is dated _____, 2006 and the information contained herein speaks only as of that date.

LEHMAN BROTHERS

\$10,000,000 IDAHO STATE UNIVERSITY GENERAL REVENUE BONDS (FEDERALLY TAXABLE), SERIES 2006

DuePrincipalApril 1AmountInterest RateYield or PriceCUSIP[†]

\$_____% Term Bond due ______at _____

The University is not responsible for the use of CUSIP Numbers, nor is a representation made as to the accuracy of the CUSIP Numbers. The CUSIP Numbers are contained herein solely for the convenience of the readers of this Official Statement.

No dealer, broker, salesperson or other person has been authorized by the Board, the University or by the Underwriter to give any information or to make any representations, other than as contained in this Official Statement, and if given or made, such other information or representations must not be relied upon as having been authorized by the Board, the University or the Underwriter. This Official Statement does not constitute an offer to sell or the solicitation of an offer to buy the Series 2006 Bonds, nor shall there be any sale of the Series 2006 Bonds by any person in any jurisdiction in which it is unlawful for such persons to make such offer, solicitation or sale.

The information set forth herein has been furnished by the Board, [the Insurer] the University, DTC and certain other sources that are believed to be reliable but is not guaranteed as to accuracy or completeness by, and is not to be construed as a representation by, the Underwriter. The information and expressions of opinion contained herein are subject to change without notice. Any statements made in this Official Statement involving matters of opinion or estimates or forecasts, whether or not so expressly stated, are set forth as such and not as representations of fact or representations that estimates will be realized.

IN CONNECTION WITH THIS OFFERING, THE UNDERWRITER MAY ENGAGE IN TRANSACTIONS THAT STABILIZE, MAINTAIN OR OTHERWISE AFFECT THE MARKET PRICES OF THE SERIES 2006 BONDS. SUCH TRANSACTIONS MAY INCLUDE OVERALLOTMENTS IN CONNECTION WITH THE UNDERWRITING, THE PURCHASE OF SERIES 2006 BONDS TO STABILIZE THEIR MARKET PRICES, THE PURCHASE OF SERIES 2006 BONDS TO COVER UNDERWRITER SHORT POSITIONS AND THE IMPOSITION OF PENALTY BIDS. SUCH TRANSACTIONS, IF COMMENCED, MAY BE DISCONTINUED AT ANY TIME.

Neither the delivery of this Official Statement nor any sale made hereunder shall, under any circumstances, create any implication that there has been no change in the affairs of the Board or the University since the date hereof.

This Official Statement is not to be construed as a contract with the purchasers of the Series 2006 Bonds.

The Underwriter has provided the following sentence for inclusion in this Official Statement. The Underwriter has reviewed the information in this Official Statement in accordance with, and as part of, its responsibilities to investors under the federal securities laws as applied to the facts and circumstances of this transaction, but the Underwriter does not guarantee the accuracy or completeness of such information.

The Series 2006 Bonds have not been registered under the Securities Act of 1933, as amended, in reliance upon a specific exemption contained in such act, nor have they been registered under the securities laws of any state.

This Official Statement contains "forward-looking statements" within the meaning of the federal securities laws. These forward-looking statements include, among others, statements concerning expectations, beliefs, opinions, future plans and strategies, anticipated events or trends and similar expressions concerning matters that are not historical facts. The forward-looking statements in this Official Statement are subject to risks and uncertainties that could cause actual results to differ materially from those expressed in or implied by such statements. See "BONDOWNERS' RISKS" herein. Readers are cautioned not to place undue reliance on these forward-looking statements, which speak only as of the date hereof. Forward-looking statements may be found under the headings, ______).

[Other than with respect to information concerning ______ ("_____") contained under the caption "BOND INSURANCE" and "APPENDIX F—SPECIMEN BOND INSURANCE POLICY" herein, none of the information in this Official Statement has been supplied or verified by ______ and ______ makes no representation or warranty, express or implied, as to (i) the accuracy or completeness of such information; or (ii) the validity of the Series 2006 Bonds.]

\$10,000,000 IDAHO STATE UNIVERSITY GENERAL REVENUE BONDS (FEDERALLY TAXABLE), SERIES 2006

THE BOARD OF TRUSTEES OF IDAHO STATE UNIVERSITY

ISU Financial Services 921 South 8th Avenue Stop 8219 Pocatello, Idaho 83209 (208) 282-3000 www.isu.edu/finserve/

THE STATE BOARD OF EDUCATION

Laird Stone	President
Milford Terrell	Vice President
Paul C. Agidius	Secretary
Blake G. Hall	Board Member
Dr. Marilyn Howard	
Roderic W. Lewis	Board Member
Karen McGee	Board Member
Sue Thilo	Board Member

UNIVERSITY OFFICIALS

Arthur C. Vailas	President
Robert A. Wharton	Vice President for Academic Affairs
Kenneth R. Prolo	Vice President for Financial Services and Bursar
Lee Krehbiel	Interim Vice President of Student Affairs
Bradley H. Hall, Esq	University Counsel

BOND COUNSEL

IDAHO STATE UNIVERSITY

Ballard Spahr Andrews & Ingersoll, LLP 201 South Main Street, Suite 600 Salt Lake City, Utah 84111 (801)531-3000 (801) 531-3001 (Fax) Vice President for Financial Services Campus Box 8219 Pocatello, Idaho 83209 (208) 282-2404 (208) 282-4725 (Fax)

TRUSTEE, PAYING AGENT AND REGISTRAR

U.S. Bank National Association 170 South Main Street, Suite 200 Salt Lake City, Utah 84101 (801) 534-6083 (801) 534-6013 (Fax)

Table of Contents

Page

INTRODUCTION
Idaho State University11
Authorization and Purpose of the Official
Statement11
Purpose of the Series 2006 Bonds11
Outstanding Parity Bonds12
Payment and Security for the Series 2006
Bonds12
Bond Insurance12
Redemption12
Bondowners' Risks12
Registration, Manner of Payment13
Conditions of Delivery, Anticipated Date,
Manner and Place of Delivery13
Continuing Disclosure13
Contact Persons13
Other Matters14
THE SERIES 2006 BONDS14
Description of the Series 2006 Bonds14
Optional Redemption14
Mandatory Redemption15
[Notice of Redemption16
Book-Entry-Only System16
Funds and Accounts Created under the
Resolution17
Additional Bonds17
SECURITY FOR THE SERIES 2006 BONDS18
Student Fees
Pledged Fees
Revenues of the Housing System20
CAES Project20
Use of Pledged Revenues and Other Revenues
Not Otherwise Obligated
THE CAES PROJECT
Background of the CAES Project
BOND INSURANCE
ESTIMATED SOURCES AND USES OF
FUNDS
DEBT SERVICE SCHEDULE
HISTORICAL PLEDGED REVENUES AND
DEBT SERVICE
THE UNIVERSITY
Student Body
Five-year Historical Enrollment Summary26
Student Housing
Employees
Employee Retirement Plan27

Insurance	28
Budget Process	28
SOURCES OF FUNDING FOR THE	
UNIVERSITY	28
State Appropriations	
Schedule of State General Account and State	
Endowment Appropriations	29
Fees and Tuition	
Grants and Contracts	
Auxiliary Enterprises	
Sales and Services	
Idaho State University Foundation, Inc.	
Future Plans	
University Debt	
UNIVERSITY GOVERNANCE AND	
ADMINISTRATION	21
	31
The Board of Trustees of Idaho State	
University and The State Board of	- 1
Education	
University Officers	
BONDOWNERS' RISKS	
TAX MATTERS	
UNDERWRITING	
RATINGS	
TRUSTEE	
CONTINUING DISCLOSURE	
LITIGATION	34
LEGAL MATTERS	34
INDEPENDENT AUDITORS	35
NO DEFAULTED BONDS	35
MISCELLANEOUS	35
APPENDIX A GLOSSARY OF CERTAIN	
TERMS USED IN THE RESOLUTION AND	1
OFFICIAL STATEMENT	
APPENDIX B SUMMARY OF THE	
RESOLUTIONE	8-38
APPENDIX C FINANCIAL STATEMENTS OF	
THE UNIVERSITY FOR THE YEARS	
ENDED JUNE 30, 2006 AND 2005 AND	
INDEPENDENT AUDITOR'S REPORT C	7_30
APPENDIX D FORM OF OPINION OF BOND	-57
COUNSEL	10
APPENDIX E PROVISIONS REGARDING	/-40
BOOK-ENTRY-ONLY SYSTEM B	7 4 1
	2-41
APPENDIX F SPECIMEN BOND	7 40
INSURANCE POLICY	-42
APPENDIX G FORM OF CONTINUING	
DISCLOSURE AGREEMENTC	j-43

OFFICIAL STATEMENT

RELATING TO

\$10,000,000 IDAHO STATE UNIVERSITY GENERAL REVENUE BONDS (FEDERALLY TAXABLE), SERIES 2006

INTRODUCTION

The descriptions and summaries of various documents hereinafter set forth do not purport to be comprehensive or definitive, and reference should be made to each document for the complete details of all terms and conditions. All statements herein are qualified in their entirety by reference to each document. See "APPENDIX A" for definitions of certain words and terms used herein. See "APPENDIX B" for a summary of certain provisions of the Resolution (as defined below). The attached Appendices are integral parts of this Official Statement and should be read in their entirety.

Idaho State University

Idaho State University (the "University") is a publicly supported multi disciplinary institution of higher education located in Pocatello, Idaho. The University is a separate and independent legal entity and governmental instrumentality created under the laws of the state of Idaho (the "State"). It has served the citizens of Idaho since 1901, when it was first established as the Academy of Idaho. It was renamed the Idaho Technical Institute in 1915 and reorganized as the Southern Branch of the University of Idaho in 1927. It became Idaho State College in 1947, and was established as Idaho State University in 1963. The University is governed by the State Board of Education, whose members serve as the Board of Trustees for the University (the "Board"). The University operates a satellite campus in the city of Idaho Falls, approximately 50 miles north of Pocatello.

Authorization and Purpose of the Official Statement

This Official Statement, including the cover page and the financial and other information contained in the Appendices hereto, is furnished in connection with the offering of the University's General Revenue Bonds (Federally Taxable), Series 2006 (the "Series 2006 Bonds"). The University is authorized by the Educational Institutions Act of 1935, constituting Section 33 3801, et seq. of the Idaho Code, as amended (the "Act"), to issue bonds for "Projects" (as defined in the Act). The Series 2006 Bonds are being issued pursuant to such statutory authorization and pursuant to the supplemental resolution (the "Supplemental Resolution") adopted by the Board on November ______, 2006. The Series 2006 Bonds are being issued as "Additional Bonds" under a bond resolution adopted September 17, 1992 (the "Original Resolution"). The Original Resolution, together with the Supplemental Resolution and the previous supplemental resolutions authorizing the issuance of Additional Bonds, are referred to collectively herein as the "Resolution."

Purpose of the Series 2006 Bonds

The Series 2006 Bonds are being issued by the University to (i) finance a portion of the costs of the acquisition, construction, equipping and furnishing of a facility that will contain the Center for Advanced Energy Studies at the Idaho Falls Center for Higher Education Campus (the "CAES Project"), which will be jointly occupied by a consortium which consists of the three Idaho research universities (the University, Boise State University, and the University of Idaho) that will collaborate with the Idaho National Laboratory ("INL") in advanced nuclear research, engineering and education and Battelle Energy Alliance, LLC a limited liability company organized under the laws of the State of Delaware ("BEA"), and (ii) paying the costs of issuance of the Series 2006 Bonds.

Pursuant to a Lease Agreement dated as of ______, 2006 between the University and BEA (the "CAES Lease"), the University has leased to BEA approximately ___% of the CAES Project and BEA has agreed to pay lease payments (the "CAES Lease Payments") which are intended to approximate the payments of principal and interest on the Series 2006 Bonds. Under certain circumstances, payment of CAES Lease Payments under the CAES Lease are guaranteed by Battelle Memorial Institute ("BMI") and the Washington Group International ("WG"). The University will pledge the CAES Lease Payments as part of the Pledged Revenues (described below).

The CAES Lease Payments are expected to be less than ___% of the Pledged Revenues (based upon the historical Pledged Revenues for the University's fiscal year ended June 30, 2006). Therefore, no information is provided in this Official Statement with respect to BEA, BMI or WG and the University makes no representations as to the ability of BEA, BMI or WG to make its respective payments (if any) of the CAES Lease Payments and purchasers of the Series 2006 Bonds are advised to look at the Pledged Revenues as a whole, rather than the CAES Lease Payments alone, as the security for the Series 2006 Bonds.

Outstanding Parity Bonds

Pursuant to the Original Resolution, in order to provide funds to finance and refinance certain projects, the University has previously issued [(among others)] its (i) Student Facilities Fee Refunding and Improvement Revenue Bonds, Series 1998 (the "Series 1998 Bonds"), (ii) General Refunding and Improvement Revenue Bonds, Series 2003 (the "Series 2003 Bonds"), (iii) General Revenue Bonds, Series 2004A (the "Series 2004A Bonds"), (iv) General Revenue Bonds, Series 2004C (the "Series 2004B (the "Series 2004B Bonds"), (vi) General Revenue Bonds, the Series 2004C Bonds," and collectively with the Series 1998 Bonds, the Series 2003 Bonds, the Series 2004A Bonds, the Series 2004B Bonds, and the Series 2004C Bonds, the Series 2003 Bonds, the Series 2006 Bonds will be issued on a parity with the Outstanding Parity Bonds and any additional bonds, notes or other obligations that may be issued from time to time under the Original Resolution (the "Additional Bonds"), such that the Series 2006 Bonds, Outstanding Parity Bonds and Additional Bonds will be payable from and secured by an equal lien pledge of the Net Revenues (defined herein). See "SECURITY FOR THE SERIES 2006 BONDS— _______" and "—_____" herein. The Series 2006 Bonds, the Outstanding Parity Bonds and any Additional Bonds are collectively referred to herein as the "Bonds."

Payment and Security for the Series 2006 Bonds

The Series 2006 Bonds are secured on a parity with the Outstanding Parity Bonds by a pledge of the Pledged Revenues, as designated by the Board, including the University's Tuition Fee (the "Tuition Fee"), the Student Facilities Fee/Facilities (the "Student Facilities Fee/Facilities"), Revenues of the Housing System (as defined in the Resolution) and, following the issuance of the Series 2006 Bonds, the CEAS Lease Payments, all investment income derived from the Revenue Fund and the Bond Fund, and certain other revenues. Hereinafter, the Tuition Fee and the Student Facilities Fee/Facilities are referred to collectively as the "Pledged Fees." See "SECURITY FOR THE SERIES 2006 BONDS" herein.

Bond Insurance

The scheduled payment of principal of and interest on the Series 2006 Bonds when due will be guaranteed under an insurance policy to be issued concurrently with the delivery of the Series 2006 Bonds by ______. See "BOND INSURANCE" herein.

Redemption

The Series 2006 Bonds are subject to optional and mandatory redemption as described under the caption "THE SERIES 2006A BONDS—Optional Redemption and —Mandatory Redemption" herein.

Bondowners' Risks

The purchase of the Series 2006 Bonds involves investment risks, certain of which are described in this Official Statement. See "BONDOWNERS' RISKS" herein.

Registration, Manner of Payment

The Series 2006 Bonds are issuable only as fully registered bonds without coupons and, when issued, will be registered in the name of Cede & Co., as nominee for The Depository Trust Company, New York, New York ("DTC"), which will act as initial securities depository of the Series 2006 Bonds. Purchases of Series 2006 Bonds will be made in book-entry form only, through brokers and dealers who are, or who act through, DTC Participants. Beneficial Owners of the Series 2006 Bonds will not be entitled to receive physical delivery of bond certificates so long as DTC or a successor securities depository acts as the securities depository with respect to the Series 2006 Bonds.

Principal and purchase price (as applicable) of, premium, if any, and interest on the Series 2006 Bonds are payable through U.S. Bank National Association, as Paying Agent and Registrar, to DTC, which will in turn be responsible to remit such principal and interest to its Participants, for subsequent disbursements to the Beneficial Owners of the Series 2006 Bonds, as described under the caption "APPENDIX E—PROVISIONS REGARDING BOOK-ENTRY-ONLY SYSTEM" hereto.

Conditions of Delivery, Anticipated Date, Manner and Place of Delivery

The Series 2006 Bonds are offered, subject to prior sale, when, as and if issued and received by the Underwriter subject to the approval of legality by Ballard Spahr Andrews & Ingersoll, LLP, as Bond Counsel to the University, and certain other conditions. Certain legal matters will be passed upon for the University by University Counsel, Bradley H. Hall, Esq., Pocatello, Idaho. It is expected that the Series 2006 Bonds, in book-entry form, will be available for delivery to DTC or its agent on or about ______, 2006.

Continuing Disclosure

The University, for the benefit of the owners and Beneficial Owners of the Series 2006 Bonds, has covenanted to provide certain annual information and notice of the occurrence of certain events in order to enable the Underwriters to make the determinations required by Rule 15c2-12 of the Securities and Exchange Commission (the "Rule"). The University is in compliance with each and every undertaking entered into by it pursuant to the Rule. See "CONTINUING DISCLOSURE" herein and "APPENDIX G—FORM OF CONTINUING DISCLOSURE AGREEMENT" attached hereto.

Contact Persons

The chief contact person for the University concerning the Series 2006 Bonds is:

Mr. Kenneth R. Prolo Vice President for Financial Services and Bursar Idaho State University Campus Box 8219 Pocatello, Idaho 83209 Telephone: (208) 282-2404 Fax: (208) 282-4725 E-mail: _____

The chief contact person for the Underwriter concerning the Series 2006 Bonds is:

Mr. Richard B. King Senior Vice President Lehman Brothers Inc. 701 Fifth Avenue, Suite 1701 Seattle, Washington 98104 Telephone: (206) 344-5838 Fax: (212) 520-0837 E-mail: rking@lehman.com

Other Matters

The descriptions and summaries of the Resolution, the Series 2006 Bonds and various other documents herein set forth do not purport to be comprehensive or definitive, and reference is made to each document for the complete details of its terms and conditions. All statements herein are qualified in their entirety by reference to such documents. Capitalized terms used, but not otherwise defined, herein have the same meaning as ascribed to them in the Resolution. Descriptions of the Resolution and the Series 2006 Bonds are qualified by reference to bankruptcy laws affecting the remedies for the enforcement of the rights and security provided therein and the effect of the exercise of the police power by any entity having jurisdiction. See "APPENDIX B—SUMMARY OF THE RESOLUTION" herein.

THE SERIES 2006 BONDS

Description of the Series 2006 Bonds

The Series 2006 Bonds will be dated the date of their delivery and are issuable in fully registered form, book-entry-only, in the aggregate principal amount of \$10,000,000. The Series 2006 Bonds bear interest from the date of issuance, payable semiannually on April 1 and October 1 of each year, commencing April 1, 2007. Payment will be made to Beneficial Owners through the Book-Entry-Only System described below. For a further description of certain terms of the Resolution authorizing the issuance of the Series 2006 Bonds, please refer to "APPENDIX B—SUMMARY OF THE RESOLUTION."

Optional Redemption

The Series 2006 Bonds are subject to redemption, in whole or in part, at the option of the University at any time or from time to time upon notice as provided in the Resolution, on any date prior to their maturity at a Redemption Price equal to 100% of the principal amount of such Series 2006 Bonds plus the Make-Whole Premium, if any, as described below, plus the accrued interest, if any, thereon to the redemption date.

The amount of the Make-Whole Premium with respect to any Series 2006 Bond to be redeemed will be equal to the excess, if any, of:

1. the sum of the present values, calculated as of the redemption date, of:

(a) each interest payment that, but for such redemption, would have been payable on the Series 2006 Bond or portion thereof being redeemed on each interest payment date occurring after the redemption date (excluding any accrued interest for the period prior to the redemption date); and

(b) the principal amount that, but for such redemption, would have been payable at the final maturity of the Series 2006 Bond being redeemed; over

2. the principal amount of the Series 2006 Bond being redeemed.

The present values of interest and principal payments referred to in clause (1) above shall be determined in accordance with generally accepted principles of financial analysis. These present values will be calculated by discounting the amount of each payment of interest or principal from the date that each such payment would have been payable, but for the redemption, to the redemption date at a discount rate equal to the "comparable treasury yield" (as defined below) plus 12.5 basis points. The Make-Whole Premium will be calculated by an independent investment banking institution of national standing appointed by the University (which may be the Underwriter). If the University fails to appoint an independent investment banker at least 45 days prior to the redemption date, or if the independent investment banker appointed by the University is unwilling or unable to make the calculation, the calculation will be made by an independent investment banking institution of national standing appointed by the University fails to make the calculation, the rate of the university is unwilling or unable to make the calculation, the rate calculation will be made by an independent investment banking institution of national standing appointed by the University is unwilling or unable to make the calculation, the calculation will be made by an independent investment banking institution of national standing appointed by the University is unwilling or unable to make the calculation, the calculation will be made by an independent investment banking institution of national standing appointed by the University is unwilling or unable to make the calculation, the calculation will be made by an independent investment banking institution of national standing appointed by the University is university is university is university is university in the university is university is university in the university is united by the University is united by

For purposes of determining the Make-Whole Premium, "comparable treasury yield" means a rate of interest per annum equal to the weekly average yield to maturity of United States Treasury Securities that have a constant maturity that corresponds to the remaining term to maturity of the Series 2006 Bonds, calculated to the nearest 1/12th of a year. The comparable treasury yield will be determined as of the third Business Day immediately preceding the applicable redemption date. [Discuss]

The weekly average yields of United States Treasury Securities will be determined by reference to the most recent statistical release published by the Federal Reserve Bank of New York and designated "H.15(519) Selected Interest Rates" or any successor release. If the H.15 statistical release sets forth a weekly average yield for United States Treasury Securities having a constant maturity that is the same as the remaining term calculated as set forth above, then the comparable treasury yield will be equal to such weekly average yield. In all other cases, the comparable treasury yield will be calculated by interpolation on a straight-line basis, between the weekly average yields on the United States Treasury Securities that have a constant maturity closest to and less than the remaining term (in each case as set forth in the H.15 statistical release or any successor release). Any weekly average yields calculated by interpolation will be rounded to the nearest 1/100th of 1%, with any figure of 1/200th of 1% or above being rounded upward. If weekly average yields for United States Treasury Securities are not available in the H.15 statistical release or otherwise, then the comparable treasury yield will be calculated by interpolation of comparable rates selected by an independent investment banker selected in the manner described above. [Discuss]

Mandatory Redemption

If not previously redeemed as described above, the Series 2006 Bonds maturing on ______, _____ (the "Series 2006 Term Bonds") will be redeemed in part by operation of sinking fund installments upon notice as provided in the Resolution at a redemption price equal to 100% of the portion of the principal amount thereof to be redeemed, together with accrued interest to the Mandatory Redemption Date. The principal amounts and due dates of the Mandatory Redemption Amounts for the Series 2006 Term Bonds are set forth in the following table:

April 1 of the year Mandatory Redemption Amount

* Final maturity

Each of the Series 2006 Term Bonds shall be subject to redemption in part, pro rata and not by lot, on each Mandatory Redemption Date. The portion of the principal amount of each Series 2006 Term Bond to be redeemed (per \$5,000 original principal amount) on each Mandatory Redemption Date and the resulting principal amount of

each Series 2006 Term Bond remaining outstanding (per \$5,000 original principal amount) upon each such redemption are shown in the following table:

Mandatory Redemption Date April 1 Principal Amount to Be Redeemed (Per \$5,000 original <u>Mandatory Redemption Amount)</u> Remaining Principal Amount (Per \$5,000 original Mandatory Redemption Amount)

* Final maturity

[Notice of Redemption

When the Series 2006 Bonds are called for redemption, notice must be sent by the Trustee, postage prepaid, by first class mail not less than thirty-five (35) nor more than sixty (60) days prior to the redemption date to the registered owners of the Series 2006 Bonds to be redeemed at the address shown on the Bond Register. As provided in the Resolution, the Trustee may give further notice of redemption at least thirty-five (35) days before the redemption date by registered or certified mail or overnight delivery service to certain registered national securities depositories and national information services; provided, however that no defect in such further notice or failure to give all or any portion of such further notice will in any manner defeat the effectiveness of a call for redemption.]

Each notice of redemption may further state, in the case of optional redemption, that such redemption shall be conditioned upon the receipt by the Trustee on or prior to the date fixed for such redemption of moneys sufficient to pay the principal of and interest on such Series 2006 Bonds to be redeemed and that if such moneys shall not have been so received said notice shall be of no force and effect and such Series 2006 Bonds shall not be required to be redeemed. In the event that such notice of redemption contains such a condition and such moneys are not so received, the redemption shall not be made and the Trustee shall within a reasonable time thereafter give notice, one time, in the same manner in which the notice of redemption was given, that such moneys were not so received.

Book-Entry-Only System

The Series 2006 Bonds will be available only in book-entry-only form in the principal amounts shown on the inside cover page of this Official Statement. The Depository Trust Company ("DTC") will act as Securities

Depository for the Series 2006 Bonds. The ownership of one fully registered Series 2006 Bond for maturity as set forth on the inside cover page of this Official Statement, in the aggregate principal amount of each maturity of the Series 2006 Bonds, will be registered in the name of Cede & Co., as nominee for DTC. See "APPENDIX E— PROVISION REGARDING BOOK-ENTRY-ONLY SYSTEM."

Funds and Accounts Created under the Resolution

The Revenue Fund. The Resolution creates the Revenue Fund to be held by the University into which Pledged Revenues shall be deposited. Monies in the Revenue Fund shall be disbursed in the following order of priority:

1. To transfer to the Trustee for deposit in the Debt Service Account of the Bond Fund for payment of any interest, principal, or redemption premium, if any, coming due on the Bonds;

2. Amounts remaining in the Revenue Fund in excess of the amounts necessary to make payments required by subsection (1) above may be applied by the University, free and clear of the lien of the Resolution, for any other lawful purpose of the University.

The Bond Fund. The Resolution creates a Bond Fund consisting of a Debt Service Account held by the Trustee to be used for paying the principal of, premium, if any, and interest on the Bonds.

The Construction Fund. The Resolution provides for a Construction Fund to be held, disbursed and invested by the University, into which proceeds from any series of Bonds issued to fund Projects being built or acquired with a series of Bonds issued under the Resolution are deposited. Upon completion of a Project, any unexpended monies held in the Construction Fund will be transferred to the Debt Service Account of the Bond Fund.

The Series 2006 Bonds Cost of Issuance Fund. The Supplemental Resolution creates the Series 2006 Bonds Cost of Issuance Fund in the Construction Fund to be held by the University for costs of issuance of the Series 2006 Bonds to be paid from.

The Rebate Fund. The Resolution creates a Rebate Fund to be held and administered by the University, separate and apart from other funds and accounts of the University. The University shall make deposits into the Rebate Fund of all amounts necessary to make payments to the United States required under the Code. The Resolution permits the University to consolidate all Rebate Funds for bonds issued by the Board for filing a single, consolidated report to the U.S. Treasury under the Code.

Additional Bonds

The Resolution currently provides that Additional Bonds (such as the Series 2006 Bonds) secured by Pledged Revenues may be issued by the University upon the satisfaction of various conditions specified therein. The amount of Additional Bonds that may be issued is limited neither by law nor by the Resolution.

The Resolution provides for the issuance of Additional Bonds to finance Projects or to refund the Bonds or Additional Bonds issued hereafter under the Resolution. In connection with the issuance of Additional Bonds, the University is required to file, among other things, the following documents with the Trustee:

1. copy of the Supplemental Resolution authorizing such Additional Bonds;

2. a certificate of the University to the effect that, upon the delivery of the Additional Bonds, the University will not be in default in the performance of any of the covenants, conditions, agreements, terms or provisions of the Resolution or any of the Bonds;

3. a Written Certificate of the University signed by an Authorized Officer of the University, setting forth the then estimated completion date and the then estimated cost of construction of the Project(s) being financed by the Additional Bonds;

4. either (a) an Accountant's Certificate which demonstrates that, for any twelve-month period in the preceding twenty-four months, Revenues Available for Debt Service shall have equaled at least 110% of the Maximum Annual Debt Service for all Bonds then Outstanding and any Additional Bonds proposed to be issued; or

(b) a Written Certificate of the University showing that Estimated Revenues Available for Debt Service (assuming completion of the proposed Project on its then estimated Completion Date) will equal at least 110% of the Maximum Annual Debt Service on all Bonds then Outstanding and the Additional Bonds proposed to be issued for (i) each of the Fiscal Years of the University during which any of the Bonds will be Outstanding following the estimated completion date of the Project being financed by the Additional Bonds, if interest during construction of the Project being financed by the Additional Bonds is capitalized; or (ii) the University's current Fiscal Year and any succeeding Fiscal Year during which any of the Bonds will be Outstanding, if interest during construction of the Project being financed by the Additional Bonds is not capitalized.

SECURITY FOR THE SERIES 2006 BONDS

The Series 2006 Bonds are secured by Pledged Revenues pursuant to the Resolution on a parity with all Bonds issued under the Resolution. Pledged Revenues include: (i) Pledged Fees, (ii) Revenues of the Housing System; (iii) CAES Lease Payments; (iv) other revenues of other University enterprises or sources of funds as shall be designated by the Board; (v) any investment income derived from the Revenue Fund and the Bond Fund; and (vi) proceeds from the sale of a series of Bonds and money and investment earnings thereon. In addition, the University has covenanted that Revenues Available for Debt Service will equal 110% of Annual Debt Service on a year by year basis. See also "HISTORICAL AND PROFORMA PLEDGED REVENUES AND DEBT SERVICE" herein.

Student Fees

The Board is exclusively empowered to establish and collect tuition charges for students attending the University and to establish and collect student fees from both resident and non-resident students. Student fees and tuition charges are not subject to a referendum by students or approval by any other governmental entity. The Board has established a policy that the University may not request more than a 10% annual increase in the total full time student fees unless otherwise authorized by the Board. Although Board policy provides that fee changes will be considered when appropriate or necessary, the Board has traditionally adjusted fees annually, with fee adjustments effective for the subsequent fall term each year. The 2006-2007 fee schedule, which was approved by the Board at the April 2006 meeting, follows. See "SOURCES OF FUNDING FOR THE UNIVERSITY—Fees and Tuition" for a comparison of full-time fees over the Fiscal Years 2003 through 2007.

The 2006-2007 fee schedule reflects an approximately 4.8% overall increase to student fees from the 2005-2006 academic year, including an approximately 4.0% increase to Tuition. The University bases the Estimated Annual Revenue to be collected from each of the fees on budgeting assumptions of the student fees approved for the current academic year (2006-2007 with the exception of fees from the summer session which are based on the 2005-2006 fee schedule), and the number of full time and part time students for the previous academic year (2005-2006). The number of students obtained by dividing the Estimated Annual Revenue line items for full-time students on the fee schedules is less than the full time equivalents and fall semester full time enrollees for Fall 2006 shown under the heading "THE UNIVERSITY-Five Year Historical Enrollment Summary." This is consistent with historic budgeting assumptions, including consideration of the University's policy to provide fee waivers or discounts to certain scholarship recipients and to certain employees and spouses of certain employees. The University's estimates include certain assumptions concerning refunds, late fees and other variabilities in individual fees, such that the annual estimated revenues of each fee are not the numerical product of the fee rates times a constant number for students paying such fees, but nonetheless represent the University's best estimate of fee revenues. As more fully discussed under the heading "THE UNIVERSITY-Five-year Historical Enrollment Summary", the University's enrollment has declined each year for the past several years. While the University is empowered to set fees and tuition, it cannot control the number of students enrolled in any year and a continued decline in enrollment could impact the ability of the University to collect sufficient Pledged Revenues to pay principal and interest on the Bonds. Pledged Fees are shown in **bold** on the following table.

Student Fee and Tuition Schedule⁽¹⁾

ACADEMIC YEAR 2006 - 2007

	Part-Time/Per			Total	
	<u>Full-Time</u>		Credit Hour		Estimated
	Amount Per	Estimated	Amount Per	Estimated	Annual
	Semester	Annual Revenue	Semester	Annual Revenue	Revenue
FACILITY FEES					
Student Facilities Fee	\$192.00	\$3,276,000	_	_	\$3,276,000
Student Housing System	25.00	450,000	-	_	450,000
Campus Technology	75.00	<u>1,350,000</u>	<u>\$9.00</u>	<u>\$298,998</u>	<u>1,648,998</u>
Subtotal Facility Fees	<u>292.00</u>	5,076,000	<u>9.00</u>	<u>298,998</u>	<u>5,374,998</u>
GENERAL EDUCATION					
Tuition	<u>1,344.56</u>	24,193,061	<u>161.60</u>	5,313,100	29,506,161
DEDICATED ACTIVITY FEES					
Intercollegiate Athletics	\$103.40	\$1,861,200	\$3.00	\$62,929	\$1,924,129
Student Health Center	55.50	999,000	2.00	85,126	1,084,126
Student Union Operations	131.50	2,367,000	9.50	316,458	2,683,458
Associated Student Body/Student					
Programming (Summer Only)	63.15	1,136,700	7.00	233,179	1,369,979
Student ID Card	2.35	42,300	1.25	41,640	83,940
Childcare Services	12.55	225,900	2.00	66,622	292,522
Gender Resource Center	5.70	102,600	1.20	39,974	142,574
Leadership and Counselor Training	3.30	59,400	1.00	33,311	92,711
Marching Band	6.75	121,500	1.65	34,611	156,111
Debate Team	4.35	78,300	_	_	78,300
Rodeo Team	4.89	88,020	_	_	88,020
Recreation Facility Operation	21.00	378,000	3.75	124,918	502,918
Intramurals	4.70	84,600	1.00	20,976	105,576
Student Band/Choir	5.00	90,000	_	_	90,000
Student Support Service	5.00	90,000	1.00	33,311	123,311
Alumni	2.50	45,000	_	_	45,000
Scholarships	19.50	351,000	_	_	351,000
Stadium Operations	_	_	6.00	199,869	199,869
Outreach Program	_	_	2.00	48,120	48,120
Wellness Program	4.10	73,800	1.05	34,977	108,777
CW HOG (Handicap Outdoor Group)	3.20	57,600			57,600
Subtotal Dedicated Activity Fees	458.44	8,251,920	43.40	1,376,021	9,627,941
Subtotal Student Fees	<u>2,095.00</u>	37,520,981	214.00	<u>6,988,119</u>	44,509,100
OTHER FEES / TUITION					
Graduate/Professional	\$370.00	\$573,500	\$37.00	\$242,350	\$815,850
Non-Resident Tuition	4,135.00	984,130	115.00	272,050	1,256,180
Inservice	_		82.00	509,559	<u>509,559</u>
Subtotal Other Fees/Tuition	_	<u>\$1,557,630</u>	_	<u>\$1,023,959</u>	<u>\$2,581,589</u>
	Grand Total, Full and Part Time		<u>\$47,090,689</u>		
	Total Estimated Pledged				
	Revenues from P	Pledged Fees	<u>\$33.</u>	,232,161	
		ll and Part Time Pledged	_ <u>\$47.</u>	<u>\$1,023,959</u> ,090,689	

⁽¹⁾ The information presented in this table is based upon the budget of the University adopted in April 2006. Since that time, enrollment at the University has declined below expectations at the time this budget was established. See heading "THE UNIVERSITY—Five-year Historical Enrollment Summary" for a discussion of the decline in enrollment over the past several years.

Pledged Fees

Pursuant to the Resolution, the University's two largest fees, Tuition and the Student Facilities Fee/Facilities, are pledged to the Bonds. Below is a description of each fee and the estimated revenue generated.

Student Facilities Fee/Facilities. The Student Facilities Fee/Facilities is an existing student fee charged to full time students established by the Board and is currently pledged under the Resolution. The Student Facilities Fee/Facilities constitutes a portion of Pledged Revenues under the Resolution, is \$217.00 per student, per semester for full time students, and produced \$3,876,981 for the Fiscal Year ending June 30, 2006 and is expected to produce \$3,726,000 for the Fiscal Year ending June 30, 2007.

Tuition. Tuition is an existing fee charged to full-time and part-time students attending the University and is pledged under the Resolution. This fee is used to provide general operating revenues for the University. The Tuition fee for the 2006-2007 academic year is established at \$1,344.56 per semester for each full-time student and \$161.60 per semester-hour for part-time and summer-session students. The University received approximately \$______ from Revenues from Tuition in the Fiscal Year ended June 30, 2006 and expects to receive approximately \$29,506,161 in revenues from the Tuition fee during the Fiscal Year ending June 30, 2007.

Revenues of the Housing System

The University owns and operates all student housing facilities on the Pocatello campus. See "THE UNIVERSITY—Student Housing" herein. Housing fees for residence hall residents for the 2006-2007 academic year range from \$2,350 to \$2,825 per semester per student and include the cost of meal plans, in which housing residents are required to participate. Housing fees for suite-style residence hall rooms for the 2006-2007 academic year range from \$350 to \$400 per month, which does not include the cost of meal plans. Housing fees for residents of University apartments for the 2006-2007 academic year will range from \$450 to \$580 per month, which does not include the costs of meal plans. Revenues of the Housing System for the Fiscal Years ending June 30, 2005 and June 30, 2006 were \$3,370,820 and \$3,635,382 respectively, and are expected to be approximately \$3,670,300 for the Fiscal Year ending June 30, 2007. As the Resolution provides for a pledge of the Revenues of the Housing System for payment of debt service prior to the payment of operation and maintenance costs of the Housing System, these numbers reflect gross revenues of the Housing System. As shown below under "HISTORICAL PLEDGED REVENUES AND DEBT SERVICE", the Housing System has lost money on a net revenues basis for each of the past 3 Fiscal Years and had only slight net revenues for each of the two prior Fiscal Years.

CAES Project

As more fully discussed herein, the CAES Project will consist of a facility that will contain the Center for Advanced Energy Studies at the Idaho Falls Center for Higher Education Campus and will be jointly occupied by the Consortium and BEA. The CAES Lease Payments are intended to approximate principal and interest on the Series 2006 Bonds and will be included in Pledged Revenues and used to support the principal and interest payments on the Bonds.

Use of Pledged Revenues and Other Revenues Not Otherwise Obligated

After the University has made the payments and deposits required under the Resolution, amounts remaining in the Revenue Fund in excess of the amounts necessary to make the required payments thereunder may be used for any legal purpose of the University, including the redemption or purchase of the Bonds, subject to policies adopted by the Board.

THE CAES PROJECT

Background of the CAES Project

BEA was awarded a contract from the Department of Energy to manage the new Idaho National Laboratory ("INL") for nuclear energy research. As part of the contract, BEA committed to establish the Center for Advanced

Energy Studies ("CAES"), which is designed to become a nationally and internationally recognized focal point for the advancement of education in energy science and technology. A Memorandum of Agreement was signed by the Governor of the State of Idaho, the Presidents of each of the Consortium Universities, and BMI to work towards the establishment of a joint laboratory/university center which would serve as a research center for the INL with the construction of a building to house CAES.

The CAES Project is planned to open in June 2008 and is expected to be approximately 55,000 square feet. The CAES project will be located on state property north of the University/University of Idaho Center for Higher Education. Pursuant to the CAES Lease, BEA will have use of approximately ____% of the building space and will be responsible for an equivalent portion of the occupancy costs of the building. While the CAES Lease Payments received from BEA are expected to be used to pay principal and interest on the Series 2006 Bonds, such payments will be included in Pledged Revenues under the Resolution and the holders of the Series 2006 Bonds will be secured by all of the Pledged Revenues on a parity with the holders of all Bonds issued under the Resolution. The Consortium will share the remaining building space. When fully occupied, the CAES project will have a total of up to 175 people, including approximately 100 faculty, researchers, and staff; 50 graduate students; and 25 undergraduate students.

Only the CAES Lease Payments are included in Pledged Revenues and no other revenues from this building will be pledged. In addition, no lien on or mortgage of the CAES Project or this building will be given to secure holders of the Bonds.

BOND INSURANCE

[To be provided by the bond insurer.]

ESTIMATED SOURCES AND USES OF FUNDS

The estimated sources and uses of funds for the Series 2006 Bonds are as follows:

Sources

Par Amount of Series 2006 Bonds	\$
Net Re-Offering Premium/Discount	
Total	_
	-

Uses

Deposit to the Series 2006 Project Account in the Construction Fund	\$
Costs of Issuance ⁽¹⁾	
Total	

⁽¹⁾ Including underwriters' discount, [bond insurance] legal fees, trustee fees, rating agency fees, printing fees and other miscellaneous costs of issuance.

In addition to the proceeds of the Series 2006 Bonds, funding for the remaining costs of the facility containing the CAES Project will include: (i) approximately \$4.9 million from a settlement fund intended for use according to the terms of an agreement for the construction of [the Center for Science and Technology in Idaho Falls between the Office of the Governor of the State of Idaho and the Regents of the University of Idaho and the Trustees of Idaho State University,] and (ii) \$1,942,756 from two grants from the U.S. Department of Housing and Urban Development ("HUD") to the University of Idaho, for use according to the terms of the grant (approximately \$300,000 has been expended from one of the grants for preliminary designs for the CST facility).] [is CST CAES?]

DEBT SERVICE SCHEDULE

The following table sets forth the Annual Debt Service Requirements for the University's currently Outstanding Bonds and the Series 2006 Bonds:

Fiscal Year	Principal*	Interest*	Outstanding Bonds ⁽¹⁾	Total Debt Service
2007			\$4,643,473	
2008			4,638,112	
2009			4,644,732	
2010			4,642,651	
2011			4,644,825	
2012			4,644,138	
2013			4,646,883	
2014			4,641,857	
2015			4,545,501	
2016			4,545,759	
2017			4,543,894	
2018			4,544,655	
2019			4,547,153	
2020			4,016,307	
2021			4,027,911	
2022			4,023,801	
2023			4,023,456	
2024			361,588	
2025			361,413	
2026			360,775	
2027			359,675	
2028			358,425	
2029			356,725	
2030			359,575	
2031			361,038	
2032			361,788	
2033			356,825	
2034			<u>361,388</u>	
	<u>\$</u>	<u>\$</u>	<u>\$79,924,323</u>	

Series 2006 Bonds*

⁽¹⁾ See "THE UNIVERSITY—University Debt."

^{*} Preliminary; subject to change.

HISTORICAL PLEDGED REVENUES AND DEBT SERVICE

The following table shows the amounts of revenues pledged under the Resolution on a historical basis. In 2003 the University expanded the definition of Pledged Revenues to include the Tuition Fee and the Revenues of the Housing System. The table also presents Debt Service on all Bonds secured with such revenues for the Fiscal Years ended June 30, 2002-2006. [The revenues in the Fiscal Years ended June 30, 2002-2006 are derived from the University's internal records used to prepare its financial statements, and the revenues for Fiscal Year 2006 represent the University's budgeted amounts adjusted to reflect the actual enrollment results for the Fall 2005 and Spring 2006 semesters.]

			Actual		
<u>Revenues</u> ⁽¹⁾	2002	2003	2004	2005	2006
Revenues of the Housing System	\$3,371,914	\$3,605,430	\$3,524,086	\$3,370,821	3,635,383
Expenses of the Housing System ⁽²⁾	<u>3,352,684</u>	<u>3,543,729</u>	<u>3,820,351</u>	<u>3,993,345</u>	<u>3,984,913</u>
Net Revenues of the Housing System	<u>\$19,230</u>	<u>\$61,701</u>	<u>\$(296,265)</u>	<u>\$(622,524)</u>	<u>\$(349,530)</u>
Pledged Fees					
Tuition	17,717,958	21,849,100	25,566,700	26,925,319	28,289,953
Student Facilities Fee/Facilities	3,284,488	3,382,505	3,455,263	3,850,589	3,876,981
Investment Income	42,187	<u>-0-</u>	30,154	<u>-0-</u>	<u>-0-</u>
Total Revenues Available for Debt Service	<u>\$21,063,863</u>	<u>\$25,293,306</u>	<u>\$28,755,852</u>	<u>\$30,153,384</u>	<u>\$31,817,404</u>
Total Debt Service Requirements	\$2,191,947	\$2,194,537	3,726,538	3,768,028	4,648,321

(1) Amounts rounded to the nearest dollar. (2) The Energy of the Hamiltonian Sector

The Expenses of the Housing System for Fiscal Year 2004 as shown include in excess of \$200,000 in expenses for a number of capital improvement and deferred maintenance programs.

THE UNIVERSITY

The University is located in the city of Pocatello, Idaho, which serves as an economic center for the southeastern part of the State. The University campus in Pocatello, Idaho includes approximately 980 acres, has approximately 100 buildings, and currently serves approximately 10,151 full and part time students. In addition to University Place Idaho Falls, the University operates outreach centers in Boise, Twin Falls, Coeur d'Alene, American Falls, Blackfoot, Preston and Soda Springs.

The following map shows all locations where the University has either a campus or an outreach center.

ISU's Video Classroom Network


The University serves a diverse population that includes traditional students entering the University directly from high school, non-traditional students who have delayed their university education, working professionals and senior citizens.

The University provides both general education and specialized programs in the arts, humanities, sciences, the professions and technologies. Bachelor and master degrees are awarded in a variety of fields by the Colleges of Arts and Sciences, Business, Education, Engineering, Health Professions, Pharmacy, Technology and the Graduate School. Terminal degrees offered include: Master of Business Administration, Master of Fine Arts; Doctor of Pharmacy; Doctor of Philosophy; Doctor of Arts and Doctor of Education. Through its programs in pharmacy, health professionals and the Family Practice Medical Residency, the University is a center for education in the health professions. The University has responsibility for Idaho's dental education program.

Student Body

The University admits all Idaho residents who graduate from accredited high schools in the state with an overall grade point average of at least 2.0 or who have an enhanced ACT composite score of at least 18 and who have successfully met the statewide admission standards established by the Board. Approximately 91% of the University's Fall 2006 student body were residents of Idaho. The table on the following page sets out certain statistics concerning the University's enrollment for the Fall terms of the years indicated.

(The remainder of this page intentionally left blank.)

Five-year Historical Enrollment Summary

	2002	<u>2003</u>	<u>2004</u>	<u>2005</u>	<u>2006</u>
Students Fall Semester, 10th Day of Class					
Full Time Equivalents (FTE)	10,074	10,287	10,180	10,052	9,500
Head Count	13,352	13,623	13,802	13,977	12,676
Undergraduate Students	Academic Head	Count 10th Day	of Class, Fall Sem	iester	
Full-time:					
Residents	7,453	7,672	7,386	7,220	6,620
Non-residents	<u>593</u>	<u>626</u>	<u>633</u>	<u>616</u>	<u>607</u>
Subtotal	8,046	8,298	8,019	7,836	7,227
Part-time:					
Residents	3,329	3,202	3,691	3,860	3,282
Non-residents	<u>69</u>	<u>65</u>	<u>93</u>	<u>75</u>	<u>128</u>
Subtotal	3,398	3,267	3,784	3,935	3,410
Graduate Students					
Full-time:					
Residents	579	625	636	666	790
Non-residents	221	<u>241</u>	<u>252</u>	<u>282</u>	<u>285</u>
Subtotal	800	866	888	948	1,075
Part-time:					
Residents	1,010	1,085	1,000	1,131	839
Non-residents	<u>98</u>	107	<u>111</u>	127	125
Subtotal	1,108	1,192	1,111	1,258	964
Total Undergraduate	11,444	11,565	11,803	11,771	10,637
Total Graduate Students	<u>1,908</u>	<u>2,058</u>	<u>1,999</u>	2,206	<u>2,039</u>
Grand Total	<u>13,352</u>	<u>13,623</u>	<u>13,802</u>	<u>13,977</u>	<u>12,676</u>
No. of Freshmen Freshman Class Statistics, Fall Semester					
Applying	3,769	3,573	3,750	3,566	3,396
Accepted	2,967	2,638	2,677	2,731	2,602
Enrolled	2,212	2,107	2,526	2,490	2,147
Resident	1,716	1,692	2,134	2,127	1,724
Average ACT Score	21	21	21	21	21
Average High School GPA	3.08	3.31	3.20	3.30	3.15
Percentage graduating in the top 25% of their high school class:	35%	35%	33%	32%	35%

Source: Idaho State University

As the preceding information illustrates, full-time undergraduate enrollment at the University has decreased over the past three years. This is attributed to several factors. Changing demographics in the region has resulted in fewer high school students accessing higher education, Idaho's economy is thriving and students who may have been in school are choosing to work. Increased competition from BYU-Idaho, now a 4-year institution and out-of-state universities such as Utah State has also led to a drop in overall full-time undergraduate student enrollment. Although ISU has had steady enrollment growth since 1990, the landscape for recruiting students to ISU has changed over the last three years. Further, many of ISU's high demand health science programs such as nursing and dental hygiene have a limited number of seats available (accreditation standards). Thus, some students not gaining admission to these programs have chosen not to continue their education at ISU. At the graduate level, an attractive array of new health science programs on campus and at outreach sites across the state has helped offset part of this decline. Full-time graduate student enrollment has increased by 15% over three years from 751 students to 866. Currently, ISU is identifying new potential student markets and working to invest in bringing those students to ISU in order to maintain modest enrollment growth.

Student Housing

The University operates a dormitory system accommodating up to 540 students in five traditional residence halls, most of whom are undergraduates. The University's dormitory room charges are adjusted annually to an amount deemed necessary by University officials to pay operation, maintenance, debt amortization, and depreciation expenses.

The housing portion of the Rendezvous Project is scheduled to be completed in Fall 2007 and will add approximately 79 suite style rooms, which will house up to 302 additional students.

Purchasing a meal plan is mandatory for residence hall students, and recommended for suite residents. Resident meal service at the University is provided through a central dining hall located in Turner Hall. Meals may be taken at residence hall dining facilities and the food court at the Pond Student Union Building. The food service operations are provided through a management contract with the Compass Group North America, Inc. [Upon completion of the Rendezvous Project, meal service will also be available in the Rendezvous building.]

The University currently has 425 apartments, consisting of a mix of efficiency, one and two bedroom rental units. Rental charges are collected monthly and continuing students are allowed to remain in the apartments during the summer term even if the student does not attend summer sessions. Rental rates are reviewed and adjusted at the end of each fiscal year.

For the past five years, the average occupancy rate for the Housing System is 90%. The occupancy rate for the Housing System for the 2006-2007 academic year is approximately 87%.

All of the dormitory and apartment facilities of the University are professionally maintained and kept in a sound state of repair. The University has no current plans to construct additional dormitory housing; however, because apartment demand remains strong, the University may decide to construct additional housing within the next five years.

Employees

During the 2006-2007 academic term, the University had approximately 1,070 professional and 660 classified full time employees. The University is not a party to any collective bargaining agreement, although there are employee associations that bring any issues and concerns to the attention of the University. The University considers its relations with its employees to be good.

Employee Retirement Plan

All benefit eligible employees of the University are eligible for one of two retirement plans: the State of Idaho's "Public Employees Retirement System of Idaho" ("PERSI"), and the "Optional Retirement Plan" ("ORP") which has been offered to non classified staff since 1990.

The University's employees, including its faculty hired prior to July 1, 1990, are covered under the PERSI. PERSI covers eligible personnel who work 20 hours or more per week. The membership of PERSI includes employees of the State of Idaho, teachers, firemen, police and employees of political subdivisions, local school districts, colleges and universities.

Faculty and exempt staff hired on or after July 1, 1990, have been enrolled in the University's ORP and faculty and staff hired before that date were offered a one-time opportunity in 1990 to withdraw from PERSI and join ORP. This is a portable, defined contribution retirement plan with options offered by Teachers' Insurance and Annuity Association/College Retirement Equities Fund (TIAA/CREF) and Variable Annuity Life Insurance Company (VALIC). The total contribution rate will be the same for all employees, with a portion of the employer's contribution for ORP members being credited to the employee's account and a portion to the PERSI unfunded liability until 2015.

Insurance

The University maintains liability, property, and employee fidelity insurance in amounts deemed adequate by University officials. The University has a risk management staff that administers insurance coverage and claims, and reviews the adequacy of such policies and verifies the University's compliance with requirements imposed by agreements, such as the Resolution. As of July 1, 2006, the total insured replacement value of the University's buildings, contents and improvements was approximately \$630,000,000.

Budget Process

The University operates on an annual budget system. Its Fiscal Year begins July 1 of each year. The budget process, as well as the administration of the expenditures authorized through the process, is administered through the office of the President and Vice President of Financial Services, in collaboration with the departmental faculty and administrative officers. The internal budget process begins with a general budget proposal for the following Fiscal Year being submitted in consolidated form by the University Administration to the Board in August of each year.

The University's operating budget is approved by the Board prior to the commencement of the Fiscal Year, usually at its June meeting. At that meeting, the Board, serving also as the governing boards for the other institutions of higher education, approves the annual budgets for those institutions, as well.

SOURCES OF FUNDING FOR THE UNIVERSITY

The University relies on a number of sources of funding for the achievement of its educational and research missions. The principal sources of revenues are: direct appropriation of State revenues by the Idaho Legislature, the fees and nonresident tuition it charges its students, federal government appropriations, grants and contracts, gifts to the University, revenues derived from investments and property holdings of the University, and the revenues derived from the sale of certain products and services managed or owned by the University. These revenue sources are more fully discussed below.

State Appropriations

Legislatively approved State general account and State endowment appropriations represent slightly more than thirty percent of the total University budget. The table below sets forth the legislative appropriation from the State General Fund for all higher education institutions and for the University net of one-time funding, reversions and holdbacks. The appropriations for higher education have increased Fiscal Years 2004 through 2007.

If in the course of a fiscal year prior to the commencement of the legislative session, the Governor determines that the expenditures authorized by the Legislature for the current fiscal year exceed anticipated revenues expected to be available to meet those expenditures, the Governor by executive order may reduce ("holdback") the spending authority on file in the office of the Division of Financial Management for any department, agency or institution of the State. No holdbacks occurred during Fiscal Years 2004 through 2006.

Schedule of State General Account and State Endowment Appropriations

Fiscal Year	<u>2003</u>	<u>2004</u>	<u>2005</u>	<u>2006</u>	2007
All Higher Education	\$217,481,200	\$226,874,700	\$230,296,700	\$235,363,700	\$248,236,200
Idaho State University	58,385,200	60,654,600	61,818,600	63,496,400	67,051,200
University's percentage increase/ decrease over prior year	N/A	3.9%	1.9%	2.7%	5.6%

Fees and Tuition

The major component of this funding source is student fees, which, for the academic year 2006-2007 are \$2,095 per semester for full-time undergraduate students. Non-resident students pay per-semester tuition of \$4,135 in addition to student fees. Additional dedicated fees are charged to students enrolled in graduate programs, pharmacy, physical/occupational therapy, graduate level nursing and the Idaho Dental Education program.

The University expects fees for all students and tuition for non-resident students to increase in the future in response to generally higher costs of operating the University.

The following table shows per semester full-time fees from Fiscal Year 2003 through Fiscal Year 2007. Pledged Fees are shown in bold.

Full-Time Fees	2003	<u>2004</u>	2005	<u>2006</u>	2007
Student Facilities Fee/Facilities	\$180.00	\$180.00	\$207.00	\$217.00	\$217.00
Campus Technology Fee	<u>36.00</u>	36.00	41.00	<u>50.00</u>	75.00
Subtotal Facility Fees	216.00	216.00	248.00	267.00	292.00
Tuition	948.60	1,099.60	1,184.30	1,292.85	1,344.56
Dedicated Activity Fees	403.40	408.40	417.70	440.15	458.44
Subtotal Student Fees	<u>1,568.00</u>	1,724.00	<u>1,850.00</u>	2,000.00	2,095.00
Graduate/Professional	316.00	330.00	340.00	370.00	370.00
Non-Resident Tuition	3,120.00	3,300.00	3,540.00	3,850.00	4,135.00

Financial aid, primarily in the form of student loans, scholarships, grants, student employment, awards, tuition waivers, fee reductions and waivers, and deferred payments is available to students. The University believes that the amount of available financial aid, which totaled \$91,010,163 in Fiscal Year ended June 30, 2005 is adequate. [During the 2004-2005 academic year, the direct financial aid to students in the form of scholarships and grants was \$24,097,105.] [More recent?]

Grants and Contracts

The United States government and various other public and private sponsoring agencies, through various grant and contract programs, provide a substantial percentage of the University's current fund revenues. The use of such funds is usually restricted to specific projects and is not included in the budget for the University. Such revenues include grants and contracts for research, public service, instruction and training programs, fellowships, scholarships, endowment scholarship programs, student aid programs, and grants for construction projects. The University believes it has complied with all material conditions and requirements of these various grants and contracts.

Auxiliary Enterprises

This amount represents income earned by the University on its income-producing operations such as the University's bookstore, housing, student health center, food service, athletic facility, student union and certain other operations.

Sales and Services

Various University departments provide services and products to the student body and, in some instances, to the community for which payment is received.

Idaho State University Foundation, Inc.

The ISU Foundation is a nonprofit corporation organized under Idaho law in 1967. Its purpose is to receive, manage and otherwise deal in property and apply the income, principal and proceeds of such property for the benefit of the University. A 25-member board of directors manages the ISU Foundation. Joseph C. Jensen, President, Retired-Pfizer, Inc., Animal Health Group, serves as President of the ISU Foundation.

Financial information concerning the ISU Foundation is contained in Note 13 to the University's financial statements included in "APPENDIX A" hereto. Permanent Endowments of the ISU Foundation at June 30, 2006 were \$28,306,055.

[The ISU Foundation recently completed a five-year Centennial campaign. It surpassed its \$102 million goal with over \$152 million in gifts and pledges. A significant portion of the fundraising campaign was directed toward obtaining donations for construction of the L.E. and Thelma E. Stephens Performing Arts Center (the "Performing Arts Center"). In May 2001, the ISU Foundation issued variable rate bonds in the amount of \$21,170,000 payable from and secured by the donations and pledges identified for the Performing Arts Center. Such bonds are backed by a letter of credit issued by Wells Fargo Bank, N.A. The total cost of the Performing Arts Center is estimated at approximately \$34,000,000, with the balance of the construction cost being provided by donations already received by the ISU Foundation and other ISU Foundation sources. The ISU Foundation will lease the Performing Arts Center to the University, and the University has received authorization to use state-appropriated funds for operating expenses once occupied.]

Future Plans

The University has several capital projects in various stages of planning. At its April 2005 meeting, the State Board of Education (SBOE) approved an aquatic center addition to Reed Gym, at an estimated project cost of \$6.5 million. Current discussions could result in a revised project scope, which could increase the cost of the project to as much as \$22.2 million (with SBOE approval of the project and additional student fees to support bonding & operations). In October 2005, the SBOE approved the University's request to develop a Request for Proposal to replace the hardware and software of the University's information system. A portion of the estimated \$15 million implementation cost will require financing. The University is exploring options related to a permanent instructional facility in Boise to consolidate currently leased space. Bonding needs are estimated to be less than \$15 million. It is difficult to estimate the timing of any of these projects, but the University expects that [none of them would be financed within the next 6-12 months.

University Debt

Prior to the issuance of the Series 2006 Bonds, the University had \$53,180,000 of Bonds outstanding secured by Pledged Revenues. Set forth below is the University's schedule of outstanding indebtedness as of the end of its most recently completed Fiscal Year.

Outstanding Indebtedness Prior to the Issuance of the Series 2006 Bonds	Date Incurred	Final Maturity Date	Amount of Original <u>Indebtedness</u>	Amount of Debt Outstanding as of June 30, 2006
Student Facilities Fee Refunding and Improvement Revenue Bonds, Series 1998	1998	2022	\$12,400,000	\$9,535,000
General Refunding and Improvement Revenue Bonds, Series 2003	2003	2023	35,895,000	33,360,000
General Revenue Bonds, Series 2004A	2004	2023	4,980,000	4,770,000
General Revenue Bonds, Series 2004B	2004	2034	3,305,000	3,305,000
General Revenue Bonds (Taxable), Series 2004C	2004	2034	<u>2,305,000</u>	2,210,000
Subtotal:			<u>\$58,885,000</u>	<u>\$53,180,000</u>
General System Revenue Bonds (Federally Taxable), Series 2006	2006	[]	\$10,000,000	\$10,000,000
Total Bonded Indebtedness After Issuance of Series 2006 Bonds			<u>\$68,885,000</u>	<u>\$63,180,000</u>

Source: Idaho State University

UNIVERSITY GOVERNANCE AND ADMINISTRATION

The responsibility for overall management and determination of University policy and standards is vested with the Board of Trustees of Idaho State University who also serve as the Idaho State Board of Education and simultaneously, among other duties, as the Regents of the University of Idaho, the Trustees for Boise State University and Lewis-Clark State College in Lewiston and as the State Board for Professional-Technical Education. The combined boards are appointed by the Governor for five-year terms. The membership, terms, residences and occupations are listed below.

The Board of Trustees of Idaho State University and The State Board of Education

Name	Residence	Occupation	Term Expires
Laird Stone (President)	Twin Falls	Attorney	March 1, 2008
Milford Terrell (Vice President)	Boise	Owner/President of DeBest Plumbing	March 1, 2007
Paul C. Agidius (Secretary)	Moscow	Attorney	March 1, 2011
Blake G. Hall	Idaho Falls	Attorney	March 1, 2010
Dr. Marilyn Howard*	Boise	State Superintendent of Public Instruction	January 1, 2007
Roderic W. Lewis	Boise	General Counsel, Micron Technology, Inc.	March 1, 2010
Karen McGee	Pocatello	Audiologist and a former member of Pocatello City Council	March 1, 2009
Sue Thilo	Coeur d'Alene	Community Leader	March 1, 2009

^{*} Dr. Howard serves ex-officio to the State Board of Education in her capacity as State Superintendent of Public Instruction, which is a statewide elective office. Her current term on the Board expires on January 1, 2007.

The State Board of Education has a full time professional staff headed by Dwight Johnson, Executive Director. He was appointed Interim Executive Director in October 2005 and Executive Director effective March 28, 2006.

University Officers

The affairs of the University are managed by the President of the University and the staff. The President is appointed by, reports to, and serves at the pleasure of the Board. Following is a brief biographical resume of President Vailas and his cabinet:

Arthur C. Vailas, President. Dr. Vailas assumed the position of President of Idaho State University on July 1, 2006. Dr. Vailas previously was vice chancellor of all five University of Houston (UH) System campuses, and vice president for research and intellectual property management at the UH main campus. He joined the University of Houston in 1995 as vice provost for graduate studies, and professor and distinguished chair in biology and biochemistry. From 1988 to 1994, he held numerous positions at the University of Wisconsin – Madison. They included associate dean for research and development in the School of Education; professor of surgery, division of orthopedic surgery, College of Medicine; professor of kinesiology, School of Education; professor, department of poultry science, College of Agriculture; and professor and director of the Biodynamics Laboratory. His Ph.D. degree is from the University of Iowa with an area of emphasis in connective tissue physiology.

Robert A. Wharton, Vice President for Academic Affairs. Dr. Wharton oversees all academic aspects of the University. He works with University leadership and the State Board of Education to advance campus initiatives-leading to excellence in pursuing the University's mission. Prior to joining the University, Dr. Wharton was a member of the federal government's Senior Executive Service and served as executive officer for the National Science Foundation's Office of Polar Programs.

Kenneth R. Prolo, Vice President for Financial Services and Bursar. Mr. Prolo was appointed Vice President in 1999 after serving as the Director of Human Resources at the University since 1994. Prior to joining the University he was with the IBM Corporation for over 25 years serving in various human resource and education related staff and management positions. For over twenty years Mr. Prolo has served as an adjunct professor at San Jose State University, Santa Clara University, and the University. While in the private sector he was active on various community boards, including serving on the advisory board for continuing education for San Jose State University. His bachelor's degree is in Business Administration and his master's degree in Finance, both from San Jose State University.

Lee Krehbiel, Interim Vice President of Student Affairs. Dr. Krehbiel was appointed to the position of Interim Vice President for Student Affairs in September 2006. Originally coming to ISU in 1996 to serve as Assistant Dean of Students, he was promoted to Associate Dean of Students in 2005. Prior to joining the University, Dr. Krehbiel was a Research Associate at the Indiana Education Policy Center, Indiana University-Bloomington. He previously served as Director of Alumni Building (Union) and Campus Activities at Berea College. He received his Ph.D. in Educational Leadership and Policy Studies from Indiana University.

BONDOWNERS' RISKS

The purchase of the Series 2006 Bonds involves certain risks that are discussed throughout this Official Statement. Each prospective purchaser of the Series 2006 Bonds should make an independent evaluation of all of the information in this Official Statement in order to make an informed investment decision.

Limited Obligation. The Series 2006 Bonds are limited obligations of the Board and do not constitute a debt or liability of the State of Idaho, its Legislature, or any of its political subdivisions or agencies other than the University and then only to the extent herein described. The Board is not authorized to levy or collect any taxes or assessments other than the revenues and fees described herein to pay the Series 2006 Bonds. The Board has no taxing power.

<u>Enrollment Decline</u>. The University is empowered to set fees and tuition, but the University cannot control the number of students enrolled in any year and a continued decline in enrollment could impact the ability of the University to collect sufficient Pledged Revenues to pay principal and interest on the Bonds.

TAX MATTERS

In the opinion of Ballard Spahr Andrews & Ingersoll, LLP, Bond Counsel to the University, interest on the Series 2006 Bonds is <u>not</u> excludable from gross income for federal income tax purposes. Bond Counsel expresses no opinion regarding any other tax consequences relating to the ownership or disposition of, or the accrual or receipt of interest on, the Series 2006 Bonds.

UNDERWRITING

Lehman Brothers Inc. (the "Underwriter"), has agreed, subject to certain conditions, to purchase all of the Series 2006 Bonds from the Board at a purchase price of \$______ (representing the par amount of the Series 2006 Bonds plus/less a net reoffering premium/discount of \$______ and less an underwriting discount of \$______ and less an underwriting discount of certain dealers (including dealers depositing the Series 2006 Bonds into investment trusts) at prices lower than the initial public offering prices reflected on the cover page of this Official Statement and that such public offering prices may be changed from time to time.

RATINGS

[_________") and ______" and "_____" respectively, to the Series 2006 Bonds with the understanding that upon the delivery of the Series 2006 Bonds, a policy insuring the payment when due of the principal of and interest on the Series 2006 Bonds will be issued by ______.]

[______ and ______ have assigned underlying ratings of "____" and "____," respectively, to the Series 2006 Bonds.]

Any explanation of the significance of such ratings may only be obtained from the rating agency furnishing the same. The above ratings are not recommendations to buy, sell or hold the Series 2006 Bonds.

There is no assurance such ratings will continue for any given period of time or that such ratings will not be revised downward or withdrawn entirely by the rating agencies, if in the judgment of such rating agencies, circumstances so warrant. Any such downward revision or withdrawal of such ratings may have an adverse effect on the market price of the Series 2006 Bonds.

TRUSTEE

By appointment of the Authority, U.S. Bank National Association, Salt Lake City, Utah, shall act as the trustee, bond registrar, authenticating agent, paying agent and transfer agent with respect to the Series 2006 Bonds.

The Trustee is to carry out those duties assignable to it under the Resolution. Except for the contents of this section, the Trustee has not reviewed or participated in the preparation of this Official Statement and assumes no responsibility for the nature, content, accuracy or completeness of the information set forth in this Official Statement or for the recitals contained in the Resolution or the Series 2006 Bonds, or for the validity, sufficiency, or legal effect of any of such documents.

Furthermore, the Trustee has no oversight responsibility, and is not accountable, for the use or application by the Authority of any of the Series 2006 Bonds authenticated or delivered pursuant to the Resolution or for the use or application of the proceeds of such Series 2006 Bonds by the Authority. The Trustee has not evaluated the risks, benefits, or propriety of any investment in the Series 2006 Bonds and makes no representation, and has reached no conclusions, regarding the value or condition of any assets or revenues pledged or assigned as security for the Series

2006 Bonds or the investment quality of the Series 2006 Bonds, about all of which the Trustee expresses no opinion and expressly disclaims the expertise to evaluate.

CONTINUING DISCLOSURE

The University and the Trustee have entered into a "Disclosure Agreement" pursuant to which the University will provide to the Trustee within 180 days following the end of its fiscal year a copy of its annual audited financial statements and such other financial, statistical and operating data for such fiscal year in form and scope similar to the financial, statistical and operating data included in this Official Statement. The University also has agreed to deliver to the Trustee notice of the events described in paragraph (b)(5)(i)(C) of Rule 15c2-12 as promulgated by the Securities and Exchange Commission under the Securities Exchange Act of 1934, as amended (the "Rule"). The Trustee has agreed to deliver the information and the notices described in the preceding two sentences upon receipt thereof from the University to each nationally recognized municipal securities information repository and to the State of Idaho information depository, and to deliver any notice of an event described in paragraph (b)(5)(i)(C) of Rule 15c2-12 to the Municipal Securities Rulemaking Board. The Trustee also agrees that if it has knowledge that the University has not delivered its annual audited financial statements or has not provided the financial, statistical and operating data as described above or if it has knowledge of the occurrence of an event described in clauses (1), (3) or (4) of paragraph (b)(5)(i)(C) of Rule 15c2-12, it will directly notify such nationally recognized municipal securities information repository and the Municipal Securities Rulemaking Board of the University's failure to deliver such information or the occurrence of such event.

The University has not failed to perform any obligation with respect to any existing undertaking to provide continuous disclosure under the Rule. A failure by the University to comply with the Disclosure Agreement will constitute an event of default under the Disclosure Agreement and will entitle any Bondholder (including any Beneficial Owner) to bring an action for specific performance and to take such other remedies as are provided in the Disclosure Agreement.

A failure by the University to comply with the Disclosure Agreement must be reported in accordance with the Rule and must be considered by any broker, dealer or municipal securities dealer before recommending the purchase or sale of the Series 2006 Bonds in the secondary market. Consequently, such a failure may adversely affect the transferability and liquidity of the Series 2006 Bonds and their market price.

LITIGATION

The University has reported as of the date hereof that there is no litigation pending or threatened that, if decided adversely to the interests of the University, would have a materially adverse effect on the operations or financial position of the University. There is no litigation of any nature now pending or threatened restraining or enjoining the issuance, sale, execution or delivery of the Series 2006 Bonds or in any way contesting or affecting the validity of, or having a material adverse effect on, the Series 2006 Bonds, the pledge and application of Pledged Revenues or the existence or powers of the University.

LEGAL MATTERS

All legal matters incident to the authorization and issuance of the Series 2006 Bonds are subject to the approval of Ballard Spahr Andrews & Ingersoll, LLP, as Bond Counsel to the University whose approving opinions will be delivered with the Series 2006 Bonds. Certain legal matters will be passed upon for the University by its University Counsel, Bradley H. Hall, Esq., Pocatello, Idaho. Certain legal matters will be passed on for the Board and the University by the office of the Attorney General of the State. A copy of the opinion of Bond Counsel in substantially the form set forth in APPENDIX D of this Official Statement will be available from the University upon request.

Other than the form of such opinion, Bond Counsel has not assumed responsibility for any of the remaining material in the Official Statement and has not undertaken to review or independently verify the information set out therein. In addition, Bond Counsel has not assumed responsibility for any agreement, representation, offering

circulars or other material of any kind not mentioned in this paragraph, relating to the offering of the Series 2006 Bonds for sale.

INDEPENDENT AUDITORS

The financial statements of the University for the years ended June 30, 2005 and 2004 and Independent Auditor's Report, included in this Official Statement as "APPENDIX C," have been audited by Moss Adams LLP ("Moss Adams"), independent auditors, as stated in their report appearing therein.

NO DEFAULTED BONDS

The University has never failed to pay principal and interest when due on their respective outstanding bonded indebtedness or other obligations.

MISCELLANEOUS

The Appendices are integral parts of this Official Statement and must be read together with all other parts of this Official Statement. The references herein to the Resolution, the Series 2006 Bonds and the Act, are brief outlines of certain provisions thereof. Such outlines do not purport to be complete and reference is made to such documents and Act for full and complete statements of their provisions. Copies of these documents and Act are available for inspection at the principal corporate trust office of the Trustee in Salt Lake City, Utah and during the offering period for the Series 2006 Bonds from the Underwriter.

Any statements in this Official Statement involving matters of opinion or of estimates, whether or not so expressly stated, are set forth as such and not as representations of fact, and no representation is made that any estimates will be realized.

This Preliminary Official Statement is in a form "deemed final" by the University for purposes of Rule 15c2-12 of the Securities and Exchange Commission.

This Official Statement and its distribution and use by the Underwriter has been duly authorized by the Board and the University.

IDAHO STATE UNIVERSITY

By:_____ Vice President for Financial Services and Bursar

APPENDIX A

GLOSSARY OF CERTAIN TERMS USED IN THE RESOLUTION AND OFFICIAL STATEMENT

APPENDIX B

SUMMARY OF THE RESOLUTION

BAHR - SECTION II

APPENDIX C

FINANCIAL STATEMENTS OF THE UNIVERSITY FOR THE YEARS ENDED JUNE 30, 2006 AND 2005 AND INDEPENDENT AUDITOR'S REPORT

APPENDIX D

FORM OF OPINION OF BOND COUNSEL

[To be provided.]

APPENDIX E

PROVISIONS REGARDING BOOK-ENTRY-ONLY SYSTEM

APPENDIX F

SPECIMEN BOND INSURANCE POLICY

APPENDIX G

FORM OF CONTINUING DISCLOSURE AGREEMENT

THIS PAGE INTENTIONALLY LEFT BLANK

Attachment 2

THE BOARD OF TRUSTEES OF IDAHO STATE UNIVERSITY

Supplemental Resolution Authorizing the Issuance and Sale of \$10,000,000 General Revenue Bonds (Federally Taxable), Series 2006 of The Board of Trustees of Idaho State University

Adopted November 30, 2006

Table of Contents

		Page
ARTICLE I DEFINI	TIONS	49
Section 1.1	Definitions	49
Section 1.1 Section 1.2	Authority for Supplemental Resolution	
Section 1.2	<u>runonty for Supprementar Resonation</u>	
ARTICLE II AUTHO	ORIZATION, TERMS AND ISSUANCE OF SERIES 2006	
BONDS		52
Section 2.1	Authorization of Series 2006 Bonds, Principal Amount,	
~	Designation and Series.	52
Section 2.2	Addition to Facilities, Addition to Pledged Revenues and	
~	Confirmation of Pledged Revenues	
Section 2.3	Finding and Purpose	
Section 2.4	Issue Date	
Section 2.5	Series 2006 Bonds	
Section 2.6	Sale of Series 2006 Bonds	
Section 2.7	Delivery of Series 2006 Bonds	
Section 2.8	Redemption of Series 2006 Bonds Prior to Maturity	
Section 2.9	Form of Series 2006 Bond	
Section 2.10	Submittal to Attorney General	58
Section 2.11	Book-Entry Only System.	58
Section 2.12	Successor Securities Depository	59
	TION OF A COOLINTS ADDLICATION OF SEDIES 2006	
	TION OF ACCOUNTS, APPLICATION OF SERIES 2006	(0)
BOND PROC	CEEDS	60
Section 3.1	Creation of Accounts	60
Section 3.2	Application of Proceeds of Series 2006 Bonds	60
Section 3.3	CAES Cost of Issuance Fund	60
ARTICLE IV MISCI	ELLANEOUS	62
Section 4.1	Coverning Low	62
	Governing Law	
Section 4.2	Partial Invalidity	
Section 4.3	Savings Clause	
Section 4.4	Conflicting Resolutions; Effective Date	62
EXHIBIT A DESCR	RIPTION OF REAL PROPERTY – CAES PROJECT	A-64
EXHIBIT B FORM	OF SERIES 2006 BOND	B-65

SUPPLEMENTAL RESOLUTION

A SUPPLEMENTAL RESOLUTION authorizing the issuance and sale of \$10,000,000 General Revenue Bonds (Federally Taxable), Series 2006 of the Board of Trustees of Idaho State University; making certain findings under Title 33, Chapter 38 of the Idaho Code, and by the Resolution for the issuance of additional bonds; setting forth the issue date, interest payment dates, interest rates, maturity dates, maturity amounts and redemption amounts of the Series 2006 Bonds; authorizing the execution and delivery of a bond purchase agreement; and providing for other matters relating to the Series 2006 Bonds.

WHEREAS, Idaho State University (the "University") is a state institution of higher education and body politic and corporate organized and existing under and pursuant to the Constitution and laws of the State of Idaho; and

WHEREAS, the Board of Trustees of the University (the "Board") is authorized, pursuant to the Educational Institutions Act of 1935, the same being Title 33, chapter 38, Idaho Code, and the Constitution of the State of Idaho, to issue bonds for "projects" as defined in said Act; and

WHEREAS, on September 17, 1992, the Board adopted a Resolution (the "1992 Resolution") relating to the issuance and sale of \$10,000,000 Student Facilities Fee Revenue Bonds, Series 1992 (the "Series 1992 Bonds"), and providing among other things for the issuance of additional Student Facilities Fee Revenue Bonds for future projects ("Additional Bonds"); and

WHEREAS, the University is authorized under the provisions of Article VII of the Resolution to issue Additional Bonds upon compliance with the requirements of Section 7.2 of the Resolution; and

WHEREAS, pursuant to a Supplemental Resolution adopted May 14, 1993 which amended and supplemented the 1992 Resolution, the Board authorized the issuance of the University's \$3,000,000 Student Facilities Fee Revenue Bonds, Series 1993, to acquire certain facilities on and improvements to the campus of the University at Pocatello, Idaho (the "Series 1993 Bonds"); and

WHEREAS, pursuant to a Supplemental Resolution adopted February 19, 1998 which amended and supplemented the 1992 Resolution, the Board authorized the issuance of Additional Bonds thereunder, namely the University's \$12,400,000 Student Facilities Fee Refunding and Improvement Revenue Bonds, Series 1998, to acquire certain facilities on and improvements to the University's campus and to advance refund certain of the Series 1992 Bonds; and

WHEREAS, pursuant to a Supplemental Resolution adopted June 26, 2003, which amended and supplemented the 1992 Resolution, the Board authorized the issuance of Additional Bonds thereunder, namely the University's \$35,895,000 General Refunding and Improvement Revenue Bonds, Series 2003, to currently refund the

Outstanding Series 1993 Bonds, refinance or pay off certain other indebtedness of the University, and acquire certain housing and student services facilities of the Rendezvous Center and energy-saving improvements; and

WHEREAS, pursuant to a Supplemental Resolution adopted August 12, 2004, amending and supplementing the 1992 Resolution, the Board authorized the issuance of Additional Bonds thereunder, namely, the University's \$4,980,000 General Revenue Bonds, Series 2004A (the "Series 2004A Bonds"), to finance additional project costs of the Rendezvous Center; and

WHEREAS, pursuant to a Supplemental Resolution adopted October 24, 2004, amending and supplementing the 1992 Resolution, the Board authorized the issuance of Additional Bonds thereunder, namely, the University's \$3,305,000 General Revenue Bonds, Series 2004B (the "Series 2004B Bonds") and \$2,305,000 General Revenue Bonds (Taxable), Series 2004C (the "Series 2004C Bonds"), to acquire certain property situated in Idaho Falls, Idaho; and

WHEREAS, the Board has determined that it is both necessary for the proper operation of the University and economically feasible for the University (i) to construct a facility that will contain the Center for Advanced Energy Studies at the Idaho Falls Center for Higher Education Campus (the "CAES Project"), which will be jointly occupied by a consortium of the Idaho State research universities and Battelle Energy Alliance, LLC ("BEA"); and (ii) to provide funds to finance the CAES Project by the issuance of its Series 2006 Bonds (hereinafter defined) pursuant to the Resolution in the manner provided by Title 33, chapter 38, Idaho Code; and

WHEREAS, the Board has at the meeting at which this Supplemental Resolution is adopted, approved the lease of the CAES Project to BEA pursuant to the CAES Lease (as defined herein) and desires to include the Base Rent (as defined in the CAES Lease) payments in Pledged Revenues under the Resolution.

NOW, THEREFORE, BE IT RESOLVED BY THE BOARD OF TRUSTEES OF IDAHO STATE UNIVERSITY AS FOLLOWS:

ARTICLE I

DEFINITIONS

Section 1.1 <u>Definitions</u>.

(a) Except as provided in subparagraph (b) of this Section 1.1, all defined terms contained in this Supplemental Resolution shall have the same meanings as set forth in the Resolution.

(b) As used in this Supplemental Resolution, unless the context shall otherwise require, the following terms shall have the following meanings:

"Acquisition Proceeds" means the portion of the proceeds of the Series 2006 Bonds deposited into the CAES Project Account in the Construction Fund pursuant to Section 3.2(b) of this Supplemental Resolution for purposes of acquiring and constructing the CAES Project.

"Battelle" means Battelle Memorial Institute.

"BEA" means Battelle Energy Alliance, LLC a Delaware limited liability company.

"Bond Purchase Agreement" means the Bond Purchase Agreement dated November 30, 2006, between the Board and the Underwriter pursuant to which the Series 2006 Bonds are to be sold.

"Bond Register" means the registration records of the University, maintained by the Trustee, on which shall appear the names and addresses of the Registered Owners of the Series 2006 Bonds.

"Book-Entry System" means the book-entry system of registration of the Series 2006 Bonds described in Section 2.11 of this Supplemental Resolution.

"CAES Cost of Issuance Fund" means the special account created by Section 3.1 of this Supplemental Resolution, from which the Costs of Issuance of the Series 2006 Bonds shall be paid.

"CAES Lease" means Lease Agreement No. 00049377 between the University and BEA.

"CAES Base Rent" means the base rent (as referred to in the CAES Lease) payable by BEA under the CAES Lease.

"CAES Project" means the design and construction of a facility, as more particularly described on <u>Exhibit A</u> attached to this Supplemental Resolution and incorporated herein by this reference, that will contain the Center for Advanced Energy Studies at the Idaho Falls Center for Higher Education Campus, to be owned by the University and jointly occupied by a consortium of the three Idaho research universities and BEA.

"CAES Project Account" means the special account created by Section 3.1 of this Supplemental Resolution, from which a portion of the Cost of Acquisition and Construction of the CAES Project shall be paid.

"Cede & Co." means Cede & Co., as nominee of The Depository Trust Company, New York, New York.

"DTC Participants" means those financial institutions for whom the Securities Depository effects book-entry transfers and pledges of securities deposited with the Securities Depository, as such listing of Participants exists at the time of such reference.

"Information Reporting Agreement" means the agreement with respect to the Series 2006 Bonds, between the University and the Trustee and authorized by Section 2.6 of this Supplemental Resolution.

"Representation Letter" means the Blanket Representation Letter [dated June 26, 2003, from the University to DTC].

"Resolution" means the 1992 Resolution, as amended and supplemented by Supplemental Resolutions thereto, including the Restatement.

"Restatement" means the restatement of the 1992 Resolution, as amended and supplemented by Supplemental Resolutions, which restatement was adopted by the Board on August 12, 2004.

"Securities Depository" means The Depository Trust Company, New York, New York, or any successor Securities Depository appointed pursuant to Section 2.10.

"Series 2006 Bondholder" means the Holder of any Series 2006 Bond.

"Series 2006 Bonds" means the \$10,000,000 principal amount of General Revenue Bonds (Taxable), Series 2006 authorized to be issued pursuant to this Supplemental Resolution.

"Series 2006 Term Bonds" means ______.

"Supplemental Resolution" means this Supplemental Resolution of the University adopted on November 30, 2006, authorizing the Series 2006 Bonds.

"Trustee and Paying Agent" means U.S. Bank National Association, and its successors and permitted assigns under the Resolution.

"Underwriter" means Lehman Brothers.

"WG" means the Washington Group International.

BAHR – SECTION II

The terms "hereby," "hereof," "hereto," "herein," "hereunder," and any similar terms as used in this Supplemental Resolution refer to this Supplemental Resolution.

Section 1.2 <u>Authority for Supplemental Resolution</u>. This Supplemental Resolution is adopted pursuant to the provisions of the Act and the Resolution.

Section 1.3 <u>Amendment of Definition of Pledged Revenues</u>. As contemplated by the Resolution with regard to the addition of additional revenues, the definitions of Pledged Revenues and Revenues Available for Debt Service are hereby amended to read as follows:

<u>Pledged Revenues</u> shall include (i) the Student Facilities Fee/Facilities, (ii) the Matriculation Fee (also referred to as the Tuition Fee) and other fees as shall be designated by the Board as Pledged Revenues; (iii) Revenues of the Housing System and CAES Base Rent, (iv) other revenues of other University enterprises or sources of funds as shall be designated by the Board as Pledged Revenues, (v) any investment income deposited from the Revenue Fund and the Debt Service Fund; and (vi) proceeds from the sale of a series of Bonds and moneys and investment earnings thereon, except as otherwise provided in this Bond Resolution or a Supplemental Resolution. Upon approval of the annual budget by the Board, the amounts of fees and other revenues so approved by the Board shall become Pledged Revenues and, when deposited into the Revenue Fund, shall become available for payment into the Bond Fund for payment of Debt Service in accordance with this Bond Resolution.

<u>Revenues Available for Debt Service</u> shall mean revenues in clauses (i), (ii), (v) and (vi) of the definition of Pledged Revenues, plus Net Revenues of the Housing System, plus CAES Base Rent, plus revenues described in clause (iv) of the definition of Pledged Revenues less Operation and Maintenance Expenses of any University enterprises the revenues of which have been included in Pledged Revenues by virtue of such clause (iv).

ARTICLE II

AUTHORIZATION, TERMS AND ISSUANCE OF SERIES 2006 BONDS

Section 2.1 <u>Authorization of Series 2006 Bonds, Principal Amount,</u> <u>Designation and Series</u>. In order to provide sufficient funds to finance a portion of the costs of the CAES Project, and in accordance with and subject to the terms, conditions and limitations established in the Resolution, as previously amended, a series of Additional Bonds is hereby authorized to be issued in the aggregate principal amount of \$10,000,000, shall be designated "Idaho State University General Revenue Bonds (Federally Taxable), Series 2006." The Series 2006 Bonds shall be issued only in fully registered form, without coupons.

Section 2.2 <u>Addition to Facilities and Confirmation of Pledged Revenues</u>. As contemplated by the Resolution, the term "Facilities" as defined therein shall include the CAES Project. The Series 2006 Bonds are secured by the pledge of Pledged Revenues under Section 5.1 of the Resolution, as previously amended, equally and ratably with all Outstanding Bonds issued under the Resolution

Section 2.3 <u>Finding and Purpose</u>. The Board hereby finds, determines and declares:

(a) pursuant to Section 33-3805, Idaho Code, the CAES Project is desirable and necessary for the proper operation of the University and is economically feasible;

(b) pursuant to Section 33-3805A, Idaho Code, the CAES Project will not require state general account appropriated funds for construction, operation or maintenance;

(c) pursuant to Section 33-3806, Idaho Code, fees, rentals and other charges from those that are served by the CAES Project shall be the same as those applicable to any existing project similar in nature and purpose, provided that there may be allowed reasonable differentials based on the condition, type, location and relative convenience of such other project, but the differentials shall be uniform as to all those similarly accommodated;

(d) pursuant to Section 33-3809, Idaho Code, this Supplemental Resolution does not contract a debt on behalf of, or in any way obligate the State of Idaho, or pledge, assign or encumber in any way, or permit the pledging, assigning or encumbering in any way of, appropriations made by the Legislature, or revenue derived from the investment of the proceeds of the sale, and from the rental of such lands as have been set aside by the Idaho Admission Bill approved July 3, 1890, or other legislative enactments of the United States, for the use and benefit of the respective state educational institutions;

(e) pursuant to Section 33-3810, Idaho Code, the Series 2006 Bonds shall be exclusively obligations of the University, payable only in accordance

with the terms thereof and shall not be obligations general, special or otherwise of the State of Idaho; and

(f) the requirements of Section 7.2 of the Resolution, as amended, will have been complied with upon the delivery of the Series 2006 Bonds.

Section 2.4 <u>Issue Date</u>. The Series 2006 Bonds shall be dated as of the date of their initial delivery.

Section 2.5 Series 2006 Bonds.

(a) The Series 2006 Bonds shall bear interest at the rates and mature on the dates and in the principal amounts in each year as shown below:

April 1 of the Year

Amount Maturing

Interest Rate

* Final Maturity

(b) The Series 2006 Bonds shall bear interest from the date of delivery thereof, payable on April 1, 2007 and semiannually thereafter on each April 1 and October 1.

Section 2.6 <u>Sale of Series 2006 Bonds</u>. The Series 2006 Bonds authorized to be issued herein are hereby sold to the Underwriter at an aggregate purchase price equal to the par amount of the Series 2006 Bonds, plus net premium of \$______, on the terms and conditions set forth in the Bond Purchase Agreement. [Under the Bond Purchase Agreement, the Underwriter is to be paid a fee of \$______ for purchasing the Series 2006 Bonds.]

To evidence the acceptance of the Bond Purchase Agreement, the Bursar is hereby authorized to execute and deliver, and the Secretary to attest, the Bond Purchase Agreement in the form presented at this meeting.

The final Official Statement of the University for the sale of the Series 2006 Bonds, in substantially the form presented at this meeting, with such changes, omissions, insertions and revisions as the Bursar shall approve, is hereby authorized, and the Bursar shall sign such final Official Statement and deliver such final Official Statement to the Underwriter for distribution to prospective purchasers of the Series 2006 Bonds and other interested persons, which signature shall evidence such approval. The use of the Preliminary Official Statement dated ______, 2006, by the Underwriter and the actions of the University, including the certification by the Bursar as to the "deemed finality" of the Preliminary Official Statement pursuant to SEC Rule 15c2-12 in connection with the offering of the Series 2006 Bonds, are hereby acknowledged, approved and ratified.

In order to comply with subsection (b)(5) of SEC Rule 15c2-12, the Underwriter has provided in the Bond Purchase Agreement that it is a condition to delivery of the Series 2006 Bonds that the University and the Trustee shall have executed and delivered the Information Reporting Agreement. The Information Reporting Agreement is hereby ratified and approved in all respects and the President, Vice President for Financial Services or the Bursar are hereby authorized to execute and deliver the Information Reporting Agreement in the form of Exhibit B to the Bond Purchase Agreement.

The Vice President for Financial Services of the University and the President and Secretary of the Board are, and each of them is, hereby authorized to do or perform all such acts as may be necessary or advisable to comply with the Bond Purchase Agreement and to carry the same into effect.

Section 2.7 <u>Delivery of Series 2006 Bonds</u>. The Series 2006 Bonds shall be delivered to the Underwriter upon compliance with the provisions of Section 3.2 of the Resolution, at such time and place as provided in, and subject to, the provisions of the Bond Purchase Agreement. The Series 2006 Bonds shall be executed as provided in Section 3.2 of the Resolution.

Section 2.8 <u>Redemption of Series 2006 Bonds Prior to Maturity</u>.

(a) <u>Optional Redemption</u>. The Series 2006 Bonds are subject to redemption, in whole or in part, at the option of the University at any time or from time to time upon notice as hereinafter provided, on any date prior to their maturity at a redemption price equal to 100% of the principal amount of such Series 2006 Bonds plus the "Make-Whole Premium," if any, as described below, plus the accrued interest, if any, thereon to the redemption date.

The amount of the Make-Whole Premium with respect to any Series 2006 Bond to be redeemed will be equal to the excess, if any, of:

(i) the sum of the present values, calculated as of the redemption date, of:

(A) each interest payment that, but for such redemption, would have been payable on the Series 2006 Bond or portion thereof being redeemed on each interest payment date occurring after the redemption date (excluding any accrued interest for the period prior to the redemption date); and (B) the principal amount that, but for such redemption, would have been payable at the final maturity of the Series 2006 Bond being redeemed; over

(ii) the principal amount of the Series 2006 Bond being redeemed.

The present values of interest and principal payments referred to in clause (1) above shall be determined in accordance with generally accepted principles of financial analysis. These present values will be calculated by discounting the amount of each payment of interest or principal from the date that each such payment would have been payable, but for the redemption, to the redemption date at a discount rate equal to the "comparable treasury yield" (as defined below) plus 12.5 basis points. The Make-Whole Premium will be calculated by an independent investment banking institution of national standing appointed by the University (which may be the Underwriter). If the University fails to appoint an independent investment banker at least 45 days prior to the redemption date, or if the independent investment banker appointed by the University is unwilling or unable to make the calculation, the calculation will be made by an independent investment banking institution of national standing appointed by the Calculation of national standing appointed by the calculation, the calculation will be made by an independent investment banker appointed by the University is unwilling or unable to make the calculation, the calculation will be made by an independent investment banking institution of national standing appointed by the Trustee.

For purposes of determining the Make-Whole Premium, "comparable treasury yield" means a rate of interest per annum equal to the weekly average yield to maturity of United States Treasury Securities that have a constant maturity that corresponds to the remaining term to maturity of the Series 2006 Bonds, calculated to the nearest 1/12th of a year. The comparable treasury yield will be determined as of the third Business Day immediately preceding the applicable redemption date.

The weekly average yields of United States Treasury Securities will be determined by reference to the most recent statistical release published by the Federal Reserve Bank of New York and designated "H.15(519) Selected Interest Rates" or any successor release. If the H.15 statistical release sets forth a weekly average yield for United States Treasury Securities having a constant maturity that is the same as the remaining term calculated as set forth above, then the comparable treasury yield will be equal to such weekly average yield. In all other cases, the comparable treasury yield will be calculated by interpolation on a straight-line basis, between the weekly average yields on the United States Treasury Securities that have a constant maturity closest to and less than the remaining term (in each case as set forth in the H.15 statistical release or any successor release). Any weekly average yields calculated by interpolation will be rounded to the nearest 1/100th of 1%, with any figure of 1/200th of 1% or above being rounded upward. If weekly average yields for United States Treasury Securities are not available in the H.15 statistical release or otherwise, then the comparable treasury yield will be calculated by interpolation of comparable rates selected by an independent investment banker selected in the manner described above.

(b) <u>Mandatory Redemption</u>. If not previously redeemed as described above, the Series 2006 Bonds maturing on ______ (the "Series 2006 Term Bonds") will be redeemed in part by operation of sinking fund installments upon notice as provided in the Resolution at a redemption price equal to 100% of the portion of the principal amount thereof to be redeemed, together with accrued interest to the Mandatory Redemption Date. The principal amounts and due dates of the Mandatory Redemption Amounts for the Series 2006 Term Bonds are set forth in the following table:

> April 1 of the Year

Mandatory Redemption Amount

^{*} Final Maturity

Each of the Series 2006 Term Bonds shall be subject to redemption in part, pro rata and not by lot, on each Mandatory Redemption Date. The portion of the principal amount of each Series 2006 Term Bond to be redeemed (per \$5,000 original principal amount) on each Mandatory Redemption Date and the resulting principal amount of each Series 2006 Term Bond remaining outstanding (per \$5,000 original principal amount) upon each such redemption are shown in the following table:

	Principal Amount to Be	Remaining Principal
Mandatory	Redeemed (Per \$5,000	Amount (Per \$5,000
Redemption Date	original Mandatory	original Mandatory
<u>April 1</u>	Redemption Amount)	Redemption Amount)

* Final Maturity

(c) <u>Conditional Notice of Redemption</u>. In addition to the items required by the Resolution, each notice of redemption may further state, in the case of optional redemption, that such redemption shall be conditioned upon the receipt by the Trustee on or prior to the date fixed for such redemption of moneys sufficient to pay the principal of and interest on such Series 2006 Bonds to be redeemed and that if such moneys shall not have been so received said notice shall be of no force and effect and such Series 2006 Bonds shall not be required to be redeemed. In the event that such notice of redemption contains such a condition and such moneys are not so received, the redemption shall not be made and the Trustee shall within a reasonable time thereafter give notice, one time, in the same manner in which the notice of redemption was given, that such moneys were not so received.

Section 2.9 Form of Series 2006 Bond. The form of Series 2006 Bond is attached to this Supplemental Resolution as Exhibit B, and is incorporated herein by this reference.

Section 2.10 <u>Submittal to Attorney General</u>. There shall promptly be submitted to the Attorney General of the State of Idaho by the Secretary of the Board a certified copy of this Supplemental Resolution, together with the proceedings had in their adoption, in order that the Attorney General may examine into and pass upon the validity of the Series 2006 Bonds and the regularity of such proceedings, in the manner and with the effect specified in chapter 38 of Title 33, Idaho Code, as amended.

Section 2.11 Book-Entry Only System.

(a) The Series 2006 Bonds shall initially be registered on the Bond Register in the name of Cede & Co., the nominee for the Securities Depository, and no Beneficial Owner will receive certificates representing their respective interests in the Series 2006 Bonds, except in the event the Trustee issues Replacement Bonds as provided below. It is anticipated that during the term of the Series 2006 Bonds, the Securities Depository will make book-entry transfers among the DTC Participants and receive and transmit payments of principal of and interest on the Series 2006 Bonds to the Beneficial Owners as described below. So long as any of the Series 2006 Bonds are registered in the name of Cede & Co., as nominee of the DTC, all payments with respect to principal of, premium, if any, and interest on the Series 2006 Bonds and all notices with respect to the Series 2006 Bonds shall be made and given in the manner provided in the Representation Letter.

(b) If the Securities Depository determines to discontinue providing its services with respect to the Series 2006 Bonds and the University cannot obtain a qualified successor Securities Depository, or if the University determines not to use the book-entry system of the Securities Depository, the University shall execute and the Trustee shall authenticate and deliver one or more Series 2006 Bond certificates (the "Replacement Bonds") to the DTC Participants in principal amounts and maturities corresponding to the identifiable Beneficial Owners' interests in the Series 2006 Bonds, with such adjustments as the Trustee may find necessary or appropriate as to accrued interest and previous calls for redemption, if any. In such event, all references to the Securities Depository herein shall relate to the period of time when the Securities Depository has possession of at least one Series 2006 Bond. Upon the issuance of Replacement Bonds, all references herein to obligations imposed upon or to be performed by the Securities Depository shall be deemed to be imposed upon and performed by the Trustee, to the extent applicable with respect to such Replacement Bonds.

(c) With respect to Series 2006 Bonds registered in the name of Cede & Co. as nominee for the Securities Depository, neither the University nor the Trustee shall have any responsibility to any Beneficial Owner with respect to:

(i) the sending of transaction statements, or maintenance, supervision, or review of records of the Securities Depository;

(ii) the accuracy of the records of the Securities Depository or Cede & Co. with respect to any ownership interest in the Series 2006 Bonds;

(iii) the payment to any Beneficial Owner, or any person other than the Securities Depository, of any amount with respect to principal of, interest on, or redemption premium, if any, on the Series 2006 Bonds; or

(iv) any consent given or other action taken by the Securities Depository or Cede & Co. as owner of the Series 2006 Bonds.

(d) The University has previously executed and delivered to DTC the Representation Letter in connection with the issuance of its Bonds, including the Series 2006 Bonds. Such Representation Letter is for the purpose of effectuating the initial Book-Entry System for the Series 2006 Bonds through DTC as Securities Depository and shall not be deemed to amend, supersede or supplement the terms of this Supplemental Resolution which are intended to be complete without reference to the Representation Letter. In the event of any conflict between the terms of the Representation Letter and the terms of this Supplemental Resolution, the terms of this Supplemental Resolution shall control. The Securities Depository may exercise the rights of a Registered Owner hereunder only in accordance with the terms hereof applicable to the exercise of such rights.

Section 2.12 <u>Successor Securities Depository</u>. In the event the Securities Depository resigns, is unable to properly discharge its responsibilities or is no longer qualified to act as a securities depository and registered clearing agency under the Securities and Exchange Act of 1934, as amended, or other applicable state or federal statute or regulation, the University, may appoint a successor Securities Depository, and shall deliver notice of such appointment to the Trustee. Any such successor Securities Depository shall be a securities depository which is a registered clearing agency under the Securities and Exchange Act of 1934, as amended, or other applicable state or federal statute or regulation. Upon the appointment of a successor Securities Depository, the former Securities Depository shall surrender the Series 2006 Bonds to the Trustee for transfer to the successor Securities Depository, and the Trustee shall cause the authentication and delivery of Series 2006 Bonds to the successor Securities Depository in appropriate denominations and form as provided herein.

ARTICLE III

CREATION OF ACCOUNTS, APPLICATION OF SERIES 2006 BOND PROCEEDS, ACQUISITION OF CAES PROJECT AND ENFORCEMENT OF CAES LEASE

Section 3.1 <u>Creation of Accounts</u>. In accordance with Section 5.4B of the Resolution, there is hereby established a "CAES Project Account" in the Construction Fund and a "CAES Cost of Issuance Fund," all to be held by the University. The University shall invest the monies on deposit in the CAES Project Account in Investment Securities.

Section 3.2 <u>Application of Proceeds of Series 2006 Bonds</u>. The proceeds of the sale of the Series 2006 Bonds (less the bond insurance premium paid by the Underwriter on behalf of the University) shall be paid or deposited as follows:

(a) Bond proceeds in the amount of \$______ to the University for deposit into the CAES Cost of Issuance Fund, for payment of the Costs of Issuance of the Series 2006 Bonds, including the Underwriting fees, such costs either paid by the University or paid by the Trustee on behalf of the University; and

(b) Acquisition Proceeds in the amount of \$_____ for deposit into the CAES Project Account to acquire and construct a portion of the CAES Project.

(c) Pursuant to Section 5.4 of the Resolution, before any payment is made from the CAES Project Account, the University shall execute a Written Certificate showing with respect to each payment to be made the name of the person to whom payment is due and the amount to be paid and certifying that the obligation to be paid was incurred and is a proper charge against the CAES Project Account in the Construction Fund and in a reasonable amount against the respective account and has not been theretofore included in a prior Written Certificate.

Section 3.3 <u>CAES Cost of Issuance Fund</u>. The CAES Cost of Issuance Fund shall be held by the University. There shall be deposited so much of the Bond proceeds of the Series 2006 Bonds as shall be required to pay the Costs of Issuance and from which all Costs of Issuance shall be paid. Before any such payment shall be made, the University shall execute its Written Certificate, signed by an authorized officer of the University, stating, in respect of each payment to be made, (a) the name and address of the person, firm, or corporation to whom payment is due, (b) the amount to be paid, (c) the particular item of the Cost of Issuance to be paid, and (d) that the cost or obligation in the stated amount is a proper item of the Costs of Issuance and has not been paid. Any balance remaining in the CAES Cost of Issuance Fund after payment of the Costs of Issuance, and no later than February 1, 2007, shall be transferred to the Bond Fund, and the CAES Cost of Issuance Fund shall be closed.
Section 3.4 <u>Construction of CAES Project and Enforcement of CAES Lease.</u> The University covenants and agrees that it shall take such actions as are within its control to cause the CAES Project to be constructed and made available for use as contemplated by the CAES Lease in an expeditious manner, including seeking application of the additional sources of funding intended for such project. Pursuant to the CAES Lease, BEA is directed to make CAES Base Rent directly to the Trustee and the Trustee shall deposit such payments to the Debt Service Account of the Bond Fund. The CAES Lease has been previously approved for execution by the University and the University covenants and agrees to take such action as shall be necessary to enforce the terms and provisions of the CAES Lease for the benefit of the holders of the Bonds, including all obligations of BEA, Battelle and WG.

ARTICLE IV

MISCELLANEOUS

Section 4.1 <u>Governing Law</u>. By the acceptance of the Series 2006 Bonds, the Series 2006 Bondholders shall be deemed to agree that their respective rights shall be governed by the laws of the State of Idaho.

Section 4.2 <u>Partial Invalidity</u>. If any one or more of the covenants or agreements, or portions thereof, provided in this Supplemental Resolution on the part of the University (or of the Trustee or of any Paying Agent) to be performed should be contrary to law, such covenant or covenants, such agreement or agreements, or such portions thereof shall be null and void and shall be deemed separable from the remaining covenants and agreements or portions thereof and shall in no way affect the validity of this Supplemental Resolution or of the Series 2006 Bonds; but the Series 2006 Bondholders shall retain all the rights and benefits accorded to them under the Act or any other applicable provisions of law.

Section 4.3 <u>Savings Clause</u>. Except as amended by this Supplemental Resolution, the Resolution shall remain in full force and effect.

Section 4.4 <u>Conflicting Resolutions; Effective Date</u>. All resolutions or parts thereof in conflict herewith are, to the extent of such conflict, hereby repealed.

Attachment 2

ADOPTED AND APPROVED this _____ day of _____, 2006.

THE BOARD OF TRUSTEES OF IDAHO STATE UNIVERSITY

President

ATTEST:

Secretary

EXHIBIT A

DESCRIPTION OF CAES PROJECT

BEA was awarded a contract from the Department of Energy to manage the new Idaho National Laboratory ("INL") for nuclear energy research. As part of the contract, BEA committed to establish the Center for Advanced Energy Studies ("CAES"), which is designed to become a nationally and internationally recognized focal point for the advancement of education in energy science and technology. A Memorandum of Agreement was signed by the Governor of the State of Idaho, the Presidents of each of the Consortium Universities, and BMI to work towards the establishment of a joint laboratory/university center which would serve as a research center for the INL with the construction of a building to house CAES.

The CAES Project is planned to open in June 2008 and is expected to be approximately 55,000 square feet. The CAES project will be located on state property north of the University/University of Idaho Center for Higher Education. Pursuant to the CAES Lease, BEA will have use of approximately ____% of the building space and will be responsible for an equivalent portion of the occupancy costs of the building. While the CAES Base Rent received from BEA are expected to be used to pay principal and interest on the Series 2006 Bonds, such payments will be included in Pledged Revenues under the Resolution and the holders of the Series 2006 Bonds will be secured by all of the Pledged Revenues on a parity with the holders of all Bonds issued under the Resolution. The Consortium will share the remaining building space. When fully occupied, the CAES project will have a total of up to 175 people, including approximately 100 faculty, researchers, and staff 50 graduate students; and 25 undergraduate students.

EXHIBIT B

FORM OF SERIES 2006 BOND

R			\$		
UNITED STATES OF AMERICA STATE OF IDAHO IDAHO STATE UNIVERSITY GENERAL REVENUE BONDS (FEDERALLY TAXABLE), SERIES 2006					
Interest Rate	Maturity Date	Dated Date	CUSIP		
Registered Owner:	CEDE & CO.				
Principal Amount:		DOLLARS*****	*****		

KNOW ALL MEN BY THESE PRESENTS that Idaho State University, a body politic and corporate and an institution of higher education of the State of Idaho (the "University"), for value received, hereby promises to pay, from the Bond Fund hereinafter defined, to the registered owner identified above, or registered assigns, on the maturity date specified above, the principal sum indicated above, and to pay interest thereon from the Bond Fund from the dated date hereof, or the most recent date to which interest has been paid or duly provided for, at the rate per annum specified above, payable on April 1, 2007, and semiannually on each April 1 and October 1 thereafter, until the date of maturity or prior redemption of this Bond. Interest shall be calculated on the basis of a 360-day year and twelve 30-day months.

This Bond is an obligation of the University payable solely in accordance with the terms hereof and is not an obligation, general, special, or otherwise of the State of Idaho, does not constitute a debt, legal, moral, or otherwise, of the State of Idaho, and is not enforceable against the State, nor shall payment hereof be enforceable out of any funds of the University other than the revenues, fees, and charges pledged thereto in the Bond Resolution (defined herein). Pursuant to the Bond Resolution, sufficient revenues have been pledged and will be set aside into the Bond Fund (as defined in the Bond Resolution) to provide for the prompt payment of the principal of, interest on, and redemption price of the Bonds of which this Bond is a part. For a more particular description of the security afforded thereby, reference is made to the provisions of the Bond Resolution.

Both principal of and interest on this Bond are payable in lawful money of the United States of America to the registered owner hereof whose name and address shall appear on the registration books of the University (the "Bond Register") maintained by the Corporate Trust Department of U.S. Bank National Association (the "Trustee"), in St. Paul, Minnesota. Interest shall be paid to the registered owner whose name appears on the Bond Register on the fifteenth day of the calendar month next preceding the interest payment date, at the address appearing on the Bond Register, and shall be paid by check or draft of the Trustee mailed to such registered owner on the due date at the address appearing on the Bond Register or at such other address as may be furnished in writing by such registered owner to the Trustee. Principal shall be paid to the registered owner upon presentation and surrender of this Bond at the principal corporate trust office of the Trustee on or after the date of maturity or prior redemption.

This Bond is one of a duly authorized issue of Bonds of like date, tenor, and effect, except for variations required to state numbers, denominations, rates of interest, and dates of maturity, aggregating \$10,000,000 in principal amount. The Bonds are issued pursuant to and in full compliance with the Constitution and statutes of the State of Idaho, particularly chapter 38 of Title 33, Idaho Code, and proceedings duly adopted and authorized by the Board on behalf of the University, more particularly the Resolution adopted by the Board on September 17, 1992, as previously amended, supplemented and restated from time to time, including with respect to the Bonds by a Supplemental Resolution by the Board on November 30, 2006, authorizing the issuance of the Bonds (collectively, the "Bond Resolution").

This Bond is one of the General Revenue Bonds (Federally Taxable), Series 2006, of the University issued under the provisions of Title 33, chapter 38, Idaho Code, for the purpose of providing funds with which to (i) construct a facility that will contain the Center for Advanced Energy Studies at the Idaho Falls Center for Higher Education Campus (the "CAES Project"), and (ii) pay expenses properly incident thereto and to the issuance of the Bonds. The principal of, interest on, and redemption price of the Bonds is payable from revenues and funds of the University pledged therefor and consisting generally of the University's tuition, Student Facilities Fee/Facilities, Net Revenues of the Housing System, certain revenues from leasing the CAES Project and certain other fees and revenues, as more particularly set forth in the Bond Resolution.

The Series 2006 Bonds are subject to redemption at the time, in the amounts and at the prices and with such notice all as provided in the Resolution.

The Bonds are initially issued in the form of a separate single certificated fully registered Bond for each maturity, and registered in the name of Cede & Co., as nominee of The Depository Trust Company, New York, New York ("DTC").

Unless this Bond is presented by an authorized representative of DTC to the University or its agent for registration of transfer, exchange, or payment, and any certificate issued is registered in the name of Cede & Co. or in such other name as is requested by an authorized representative of DTC (and any payment is made to Cede & Co. or to such other entity as is requested by an authorized representative of DTC), ANY TRANSFER, PLEDGE, OR OTHER USE HEREOF FOR VALUE OR OTHERWISE BY OR TO ANY PERSON IS WRONGFUL inasmuch as the registered owner hereof, Cede & Co., has an interest herein.

Upon any partial redemption of this Bond, Cede & Co., in its discretion, may request the Trustee to authenticate a new Series 2006 Bond or shall make an appropriate notation on this Bond indicating the date and amount of prepayment, except in the case of final maturity, in which case this Bond must be presented to the Trustee prior to payment.

The Series 2006 Bonds shall not be transferable or exchangeable except as set forth in the Resolution.

[This Bond is transferable by the registered owner hereof in person or by his attorney duly authorized in writing upon presentation and surrender of this Bond at the principal corporate trust office of the Trustee. Upon such transfer, a new Bond, of the same denomination, maturity, and interest rate will be issued to the transferee in exchange therefor.]

Reference is hereby made to the Bond Resolution for the covenants and declarations of the University and other terms and conditions under which this Bond and the bonds of this issue have been issued. The covenants contained herein and in the Bond Resolution may be discharged by making provisions at any time for the payment of the principal of and interest on this Bond in the manner provided in the Bond Resolution.

This Bond shall not be valid or become obligatory for any purpose or be entitled to any security or benefit under the Bond Resolution until the Certificate of Authentication hereon shall have been manually signed by the Trustee.

IT IS HEREBY CERTIFIED AND DECLARED that all acts, conditions, and things required by the Constitution and statutes of the State of Idaho to exist, to have happened, been done, and performed precedent to and in the issuance of this Bond have happened, been done, and performed, and that the issuance of this Bond and the Bonds of this issue does not violate any Constitutional, statutory, or other limitation upon the amount of bonded indebtedness that the University may incur.

IN WITNESS WHEREOF, the Board of Trustees of Idaho State University (the "Board"), has caused this Bond to be executed by the manual or facsimile signature of the President of the Board and of the Bursar of the University and attested by the manual or facsimile signature of the Secretary of the Board, and a facsimile or original of the official seal of the University to be imprinted hereon, as of the dated date set forth above.

IDAHO STATE UNIVERSITY

By:_____ President of the Board of Trustees of Idaho State University

COUNTERSIGNED:

(SEAL)

By:_____

Bursar

ATTEST:

By:______Secretary of the Board of Trustees of Idaho State University

CERTIFICATE OF AUTHENTICATION

This Bond is one of the General Revenue Bonds (Federally Taxable), Series 2006, of Idaho State University, described in the within-mentioned Bond Resolution.

> U.S. BANK NATIONAL ASSOCIATION, as Trustee

By:_____Authorized Signature

Date of Authentication:

BAHR – SECTION II

TAB 5 Page 68

VALIDATION CERTIFICATE

I hereby certify that I have examined a certified copy of the record of proceedings taken preliminary to and in the issuance of the within bond; that such proceedings and such bond conform to and show lawful authority for the issuance thereof in accordance with the provisions of Title 33, chapter 38, Idaho Code, as amended. Such bond has been issued in accordance with the Constitution and laws of the State of Idaho and shall in any suit, action or proceeding involving its validity be conclusively deemed to be fully authorized by Title 33, chapter 38, Idaho Code, and to have been issued, sold, executed, and delivered in conformity with the Constitution and laws of the State of Idaho and to be valid and binding and enforceable in accordance with its terms, and such bond is incontestable for any cause.

Hon. Lawrence G. Wasden Attorney General

ASSIGNMENT

(Social Security or Other Identifying Number of Assignee)

(Please Print or Typewrite Name and Address of Assignee)

the within Bond and hereby irrevocably constitutes and appoints _________ of _________ to transfer the said bond on the books kept for registration thereof with full power of substitution in the premises.

Dated:_____

Signature:_____

NOTICE: The signature on this assignment must correspond with the name(s) of the Registered owner as it appears upon the face of the within Bond in every particular without alteration or enlargement or any change whatsoever.

SIGNATURE GUARANTEED:

NOTICE: Signature(s) must be guaranteed by a member firm of the New York Stock Exchange or a commercial bank or trust company and must correspond with the name as it appears upon the face of the within bond in every particular, without alteration or enlargement or any change whatever.

PREPAYMENT PANEL

The following installments of principal (or portions thereof) of this Bond have been prepaid in accordance with the terms of the Resolution.

Date of Prepayment	Principal Prepaid	Signature of Authorized <u>Representative of DTC</u>

** Include when Bonds registered with DTC.**[Bracketed text deleted when Bonds DTC-registered.]

(End of Form of Series 2006 Bond)

THIS PAGE INTENTIONALLY LEFT BLANK

BOND PURCHASE AGREEMENT

Idaho State University \$10,000,000 General Revenue Bonds (Federally Taxable), Series 2006

November 30, 2006

The Board of Trustees of Idaho State University Attn: Ken Prolo, Vice President for Financial Services Building #10, 2nd Floor Campus Box 8219 Pocatello, ID 83209

Ladies and Gentlemen:

The undersigned, Lehman Brothers Inc., as underwriter (the "Underwriter"), hereby offers to enter into this Bond Purchase Agreement with Idaho State University (the "University") which, upon the acceptance by the University of this offer, shall be in full force and effect in accordance with its terms and shall be binding upon the University and the Underwriter.

This offer is made subject to your acceptance and approval on or before 5:00 p.m. Mountain Time, on November 30, 2006. Terms not otherwise defined herein shall have the same meaning as are set forth in the Official Statement referred to hereinafter.

ARTICLE I

1.1 <u>Purchase and Sale</u>. Upon the terms and conditions and upon the basis of the representations, warranties and covenants herein set forth, the Underwriter hereby agrees to purchase from the University, and the University hereby agrees to sell to the Underwriter, all but not less than all of the University's General Revenue Bonds (Federally Taxable), Series 2006 (the "Series 2006 Bonds" or the "Bonds") for a purchase price equal to \$_____, representing \$_____ par amount of the Series 2006 Bonds, plus [net] bond issuance premium of \$_____. [Upon the issuance of the Bonds, the University shall pay the Underwriter, by wire transfer, an underwriting fee of \$_____.]

1.2 <u>The Bonds</u>. The Bonds are being issued to (i) construct a facility that will contain the Center for Advanced Energy Studies at the Idaho Falls Center for Higher

Education Campus; and (ii) pay a portion of the costs relating to the issuance and sale of the Bonds.

The Bonds shall be dated as of the date of delivery, bear interest at the rates, mature in the amounts and on the dates set forth in <u>Exhibit A</u> hereto and be otherwise as described in the Preliminary Official Statement dated _______, 2006. The Bonds shall be further described in and shall be issued under and pursuant to the Resolution of the Board of Trustees of the University (the "Board") adopted September 17, 1992 as supplemented, amended and restated and a Supplemental Resolution of the Board dated November 30, 2006 (collectively, the "Resolution"), substantially in the form heretofore delivered to the Underwriter, with only such changes therein as shall be mutually agreed upon among us.

1.3 <u>Preliminary Official Statement</u>. The University has previously delivered to the Underwriter the Preliminary Official Statement. The University represents and warrants that the Preliminary Official Statement has been deemed final by the University for purposes of Rule 15c2-12(b)(1) under the Securities Exchange Act of 1934 (the "1934 Act"), except for the omission of no more than offering prices, interest rates, Underwriter's fee, aggregate principal amount and principal amounts per maturity, delivery dates, ratings and other terms of the Bonds depending on such matters. The University hereby ratifies, approves and confirms the distribution and use of the Preliminary Official Statement by the Underwriter in connection with the public offering and sale of the Bonds prior to the availability of the final Official Statement.

1.4 Official Statement. Within seven business days after the date hereof and in sufficient time to accompany any confirmation that requests payment for the Bonds from any customer of the Underwriter, the University shall deliver to the Underwriter at the expense of the University two copies of the final Official Statement executed on behalf of the University by the President or Vice President for Financial Services of the University and a sufficient quantity of conformed copies thereof to enable the Underwriter to comply with paragraph b(4) of Rule 15c2-12 under the 1934 Act and the rules of the Securities and Exchange Commission (the "SEC") and the Municipal Securities Rulemaking Board and to meet the requirements of Section 1.6 hereof, containing only such changes from the Preliminary Official Statement to finalize the terms of the Series 2006 Bonds or as are previously consented to [in writing] by the Underwriter (such final Official Statement, including all appendices thereto, and with such supplements and amendments as are consented to in writing by the Underwriter, being herein called the "Official Statement"). The University hereby authorizes the distribution and use by the Underwriter of the Official Statement in connection with the public offering and sale of the Bonds.

1.5 <u>Amendments or Supplements</u>. If at any time from the date hereof until the Closing (defined below), and for a period of 25 days following the "end of the underwriting period," the beginning of which the University may assume is the Closing Date unless notified otherwise by the Underwriter, any event known to the University relating to or affecting the University or the Series 2006 Bonds or any agreement related to the Series 2006 Bonds shall occur which might affect the accuracy or completeness of

any statement of a material fact contained in the Official Statement or any document incorporated by reference therein, the University shall promptly notify the Underwriter in writing of the circumstances and details of such event. The University will cooperate with the Underwriter in the preparation of such amendments and supplements to the Official Statement as may be advisable, in the reasonable judgment of the Underwriter, to assure that the Official Statement as amended or supplemented will at no time include any untrue statement of a material fact or omit to state any material fact necessary in order to make the statements made in the Official Statement, in light of the circumstances under which they are made, not misleading.

1.6 <u>Delivery to NRMSIR</u>. The University authorizes the Underwriter, and the Underwriter agrees, to deliver three copies of the Official Statement to each nationally recognized municipal securities information repository ("NRMSIR") not later than seven days after the date of the Official Statement. The Underwriter shall fulfill all other responsibilities imposed upon underwriters by Rule 15c2-12 of the SEC.

1.7 Public Offering. It shall be a condition to the University's obligations to sell and to deliver the Bonds to the Underwriter and to the Underwriter's obligations to purchase, to accept delivery of and to pay for the Bonds that the entire principal amount of the Bonds shall be issued, sold and delivered by the University and purchased, accepted and paid for by the Underwriter at Closing. [The Underwriter agrees to make a bona fide public offering of all the Bonds at a price which does not exceed the initial public offering prices or the yields as set forth on the inside cover page of the Official Statement; however, the Underwriter reserves the right to make concessions to dealers and to change such initial offering prices as the Underwriter shall deem necessary in connection with the marketing of the Bonds.] The Underwriter may reoffer and sell the Bonds to certain dealers (including dealers depositing the Bonds into investment trusts) at a price lower than the offering prices set forth in the Official Statement. In connection with the offering of the Bonds, the Underwriter may over-allot or effect transactions that stabilize or maintain the market price of the Bonds at a level above that which might otherwise prevail in the open market. Such stabilizing, if commenced, may be discontinued at any time by the Underwriter without prior notice.

1.8 <u>Closing</u>. At 8:00 a.m., Mountain Daylight Time on ______, 2006, with respect to the Bonds, or at such other time or on such earlier or later date upon which we mutually agree (herein called the "Closing"), the University will deliver or cause to be delivered to the Underwriter the executed Bonds in book entry form and the other documents herein mentioned; and the Underwriter will accept such delivery and pay the purchase price for the Bonds in immediately available funds to the order of U.S. Bank National Association, Salt Lake City, Utah, as Trustee for the Bonds. Such delivery of Bonds shall occur take place at the offices of Ballard Spahr Andrews & Ingersoll, LLP, in Salt Lake City, Utah, or at such other place as the Underwriter may designate.

1.9 In order to assist the Underwriter in complying with paragraph (b)(5) of Rule 15c2-12, the University will undertake, pursuant to the Information Reporting Agreement in the form authorized by the University and attached hereto as Exhibit B, to

send annually to each nationally recognized municipal securities information repository ("NRMSIR's") and to the repository, if any, designated by the State of Idaho, as the state information depository (the "SID") and recognized as such by the Securities and Exchange Commission (the "Commission") for purposes of Rule 15c2-12, and to provide notice of certain material events to the NRMSIR's, the SID and the Municipal Securities Rulemaking Board pursuant to the requirements of Section (b)(5) of Rule 15c2-12.

ARTICLE II

REPRESENTATIONS AND WARRANTIES OF THE UNIVERSITY

To induce the Underwriter to enter into this Bond Purchase Agreement and to sell the Bonds with full realization and appreciation of the fact that the investment value of the Bonds and the ability of the University to sell and the Underwriter to resell the Bonds are in part dependent upon the credit standing of the University, and in consideration of the foregoing and execution and delivery of this Bond Purchase Agreement, the University represents and warrants to the Underwriter as follows:

2.1 The University has been duly organized and is validly existing under the Constitution and laws of the State of Idaho and has all power and authority to consummate the transactions contemplated by this Bond Purchase Agreement and the Official Statement, including the execution, delivery and/or approval of all documents and agreements referred to herein or therein.

2.2 The adoption, execution and delivery of the Resolution, the approval by the University of the Bond Purchase Agreement and the Bonds, and the application of the proceeds of the Bonds for the purposes described in the Official Statement do not and will not conflict with or result in the breach of any of the terms, conditions or provisions of, or constitute a default under, any existing law, court or administrative regulation, decree or order, or any other agreement, indenture, mortgage, lease or instrument by which the University or any of its respective property is or may be bound.

2.3 The University has duly authorized all necessary action to be taken by it for the issuance and sale of the Bonds by the University upon the terms and conditions set forth herein, in the Official Statement, in the Resolution, and in the approval of the Official Statement and the Bonds.

2.4 Except as disclosed in the Official Statement, there is no action, suit, proceeding, inquiry or investigation at law or in equity or before or by any court or administrative body pending or, to the knowledge of the University, threatened against or affecting the University, or to the knowledge of the University, any meritorious basis therefor, wherein an unfavorable decision, ruling or finding would have a material adverse effect on the financial condition of the University or the transactions contemplated by this Bond Purchase Agreement and the Official Statement, or would have an adverse effect on the validity or enforceability of the Bonds, the Resolution or

any agreement or instrument by which the University is or may be bound, or would in any way adversely affect the existence or powers of the University.

2.5 The University is not in breach of or in default under any existing law, court or administrative regulation, decree or order, or any other agreement, indenture, mortgage, lease, sublease or other instrument to which the University is a party or by which it or its property is or may be bound, and no event has occurred or is continuing which, with the passage of time or the giving of notice, or both, would constitute a default or an event of default thereunder, in either case in any manner or to enough extent which could have a material adverse effect on the financial condition of the University or the transactions contemplated by this Bond Purchase Agreement and the Official Statement, or would have an adverse effect on the validity or enforceability in accordance with their respective terms of the Bonds, or the Resolution, or would in any way adversely affect the existence or powers of the University.

2.6 The information contained in the Preliminary Official Statement was, as of its date, and the information contained in the Official Statement was and will be, as of the Closing Date, true and correct in all material respects. The Preliminary Official Statement does not contain, and the Official Statement, as of its date and as of the Closing Date, will not contain any untrue statement of a material fact, and the Preliminary Official Statement does not omit and the Official Statement, as of its date and as of the Closing Date, will not omit to state a material fact required to be stated therein or necessary to make the statements made therein, in the light of the circumstances under which they were made, not misleading; provided, however, that this representation and warranty shall not be deemed to cover or apply to (x) information provided to the University in writing by the Underwriter and included on the cover page of the Preliminary Official Statement or the Official Statement regarding the principal amount, interest rates, maturities and initial public offering prices of the Series 2006 Bonds or (y) statements in the Preliminary Official Statement or the Official Statement under the captions "THE SERIES 2006 BONDS-Book-Entry-Only System", "BOND INSURANCE", or "UNDERWRITER" and in Appendix E and Appendix F thereto.

2.7 Each representation, warranty or agreement stated in any certificate signed by any officer of the University and delivered to the Underwriter at or before the Closing shall constitute a representation, warranty or agreement by the University, as the case may be, upon which the Underwriter shall be entitled to rely.

ARTICLE III

UNDERWRITER'S CONDITIONS

3.1 The Underwriter has entered into this Bond Purchase Agreement in reliance upon the performance by the University of its obligations hereunder, both as of the date hereof and as of the date of Closing. The Underwriter's obligations under this Bond Purchase Agreement are and shall be subject to the following further conditions:

(a) At the time of Closing (i) the Official Statement, the Resolution and this Bond Purchase Agreement shall be in full force and effect and shall not have been amended, modified or supplemented, except as therein permitted or as may have been agreed to in writing by the Underwriter, and (ii) the proceeds of sale of the Bonds shall be paid to the Trustee of the Bonds for deposit or use as described in the Official Statement.

(b) At the time of Closing and in compliance with subsection (b)(5) of SEC Rule 15c2-12, the University and the Trustee shall have executed and delivered the Information Reporting Agreement in the form authorized by the University and attached hereto as <u>Exhibit B</u>.

(c) The Underwriter shall have the right to cancel its obligation to purchase the Bonds if, between the date hereof and the Closing, (i) there shall exist any event that, in the reasonable judgment of the Underwriter, either (A) makes untrue or incorrect in any material respect any statement or information in the Official Statement or (B) is not reflected in the Official Statement but should be reflected therein in order to make the statements and information therein not misleading in any material respect, or (ii) legislation shall be enacted, or actively considered for enactment by the Congress, with an effective date prior to the date of Closing, or a decision by a court of the United States shall be rendered, or a ruling or regulation by the Securities and Exchange Commission or other governmental agency having jurisdiction of the subject matter shall be made, the effect of which is that (A) the Bonds are not exempt from the registration, qualification or other requirements of the Securities Act of 1933, as amended and as then in effect, or the Securities Exchange Act of 1934, as amended and as then in effect, or (B) the Resolution is not exempt from the registration, qualification or other requirements of the Trust Indenture Act of 1939, as amended and as then in effect, or (iii) a stop order, ruling or regulation by the Securities and Exchange Commission shall be issued or made, the effect of which is that the issuance, offering or sale of the Bonds, as contemplated herein or in the Preliminary Official Statement or the Official Statement, is in violation of any provision of the Securities Act of 1933, as amended and as then in effect, the Securities Exchange Act of 1934, as amended and as then in effect, or the Trust Indenture Act of 1939, as amended and as then in effect, or (iv) there shall have occurred any outbreak or escalation of hostilities or mobilization in anticipation thereof, any terrorist activities, or any national or international calamity or crisis, financial or otherwise, which, in the sole opinion of the Underwriter, materially and adversely affects the market price or marketability of the Bonds or the ability of the Underwriter to enforce contracts for sale of such Bonds, or (v) there shall be in force a general suspension of trading on the New York Stock Exchange or minimum or maximum prices for trading substantially all listed securities thereon shall have been fixed and be in force, or maximum ranges for prices for such securities shall have been required to be in force on the New York Stock Exchange, whether by virtue of a determination by that Exchange or by order of the Securities and Exchange Commission or any other governmental authority having jurisdiction, or (vi) a general banking moratorium shall have been declared

by either federal or New York authorities having jurisdiction and be in force or (vii) any order, decree or injunction of any court of competent jurisdiction or any judicial proceeding or any order, ruling, regulation or administrative proceeding by any governmental body or agency shall have been enacted, the purpose or effect of which is to prohibit the issuance, offering or sale of the Bonds as contemplated hereby or by the Official Statement, or to prohibit the performance by the University of its obligations under the Resolution or this Bond Purchase Agreement in accordance with their respective terms; and

(d) At or prior to the Closing for the Bonds, the Underwriter shall receive the following relating to the series of Bonds being delivered at such Closing:

(1) The approving opinion of Ballard Spahr Andrews & Ingersoll, LLP ("Bond Counsel"), dated the date of Closing, in substantially the form included as Appendix __ to the Official Statement;

(2) The supplemental opinion of Ballard Spahr Andrews & Ingersoll, LLP, dated the date of Closing and addressed to the Underwriter in substantially the form attached hereto as <u>Exhibit C</u>;

(3) The opinion of Bradley H. Hall, counsel to the University, in substantially the form attached hereto as <u>Exhibit D</u>;

(4) The University's certificate or certificates signed by its Vice President for Financial Services dated the date of the Closing to the effect that (A) no litigation is pending or, to its knowledge, threatened (i) to restrain or enjoin the collection of revenues pledged or to be pledged under the Resolution (ii) in any way contesting or affecting any authority for the issuance of the Bonds, the validity of the Bonds or the Resolution, this Bond Purchase Agreement, or (iii) in any way contesting the powers or operation of the University; (B) to the best of his knowledge, the descriptions and information contained in the Official Statement are correct in all material respects except that no view need be expressed about the information under the captions "THE SERIES 2006 BONDS-Book-Entry-Only System", **"BOND** INSURANCE". or "UNDERWRITER" and in Appendix E and Appendix F thereto; (C) such descriptions and information, as of the date of the Official Statement, did not and, as of the date of Closing, do not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements made therein, in light of the circumstances under which they were made, not misleading; (D) at the time of the Closing, no default or event of default has occurred and is continuing which, with the lapse of time or the giving of notice, or both, would constitute a default or an event of default under the Resolution, this Bond Purchase Agreement or any other material agreement or material instrument to which the University is a party or by which it is or may be bound or to which any of its property or other assets is or may be subject; (E) the Resolution authorizing or approving the execution of this Bond Purchase Agreement, the Information Reporting Agreement, the Official Statement, and the form of the Bonds has been duly adopted by the Board and have not been modified, amended or repealed; (F) no event affecting the University has occurred since the date of the Official Statement that either makes untrue or incorrect, as of the date of Closing, any statement or information relating to the same and contained in the Official Statement or that should be disclosed therein in order to make the statements and information therein, in light of the circumstances under which they were made, not misleading; (G) the representations of the University herein are true and correct in all material respects as of the date of Closing;

(5) Two copies of the Resolution;

(6) Satisfactory evidence that the Bonds are rated "Aaa" by Moody's Investors Service, Inc., and "AAA" by Standard & Poor's and have underlying ratings of ["A2" and "A," respectively;]

(7) Ten copies of the Official Statement related to the Bonds executed on behalf of the University;

(8) An original executed copy of the CAES Lease Agreement (as defined in the Resolution);

(9) Two specimens of the Bonds;

(10) Evidence that a copy of the Official Statement has been filed with the Treasurer of the State of Idaho, pursuant to Idaho Code Section 67-1222;

(11) Such additional legal opinions, certificates, proceedings, instruments and other documents as the Underwriter or Bond Counsel may reasonably request.

ARTICLE IV

FEES AND EXPENSES

All expenses and costs of the University incident to the performance of its obligations in connection with the authorization, issuance and sale of the Bonds to the Underwriter, including the costs of printing of the Bonds, advertising costs, the costs of posting, printing, duplicating and mailing the Preliminary Official Statement, the Official Statement, the fees of consultants and rating agencies, the initial fee of the Trustee in connection with the issuance of the Bonds, and the fees and expenses of Bond Counsel and counsel for the University shall be paid out of funds made available by the University. All out-of-pocket expenses of the Underwriter, except for expenses of the University advanced by the Underwriter for which the Underwriter will be reimbursed by

BAHR - SECTION II

the University, including travel and other expenses and the fees and expenses of any counsel employed by it, shall be paid by the Underwriter.

ARTICLE V

GENERAL PROVISIONS

Any notice or other communication to be given to the University under this Bond Purchase Agreement may be given by delivering the same in writing, at the University's address set forth above, and any such notice or other communication to be given to the Underwriter may be given by delivering the same in writing to Lehman Brothers Inc., 701 Fifth Avenue, Suite 7101, Seattle, WA 98104.

This Bond Purchase Agreement is made solely for the benefit of the University and the Underwriter (including the successors or assigns of the Underwriter), and no other person, partnership, association or corporation shall acquire or have any right hereunder or by virtue hereof. All agreements, representations and warranties of the University in this Bond Purchase Agreement shall remain operative and in full force and effect regardless of any investigation made by or on behalf of the Underwriter and shall survive the delivery of and payment for the Bonds.

The approval of the Underwriter when required hereunder or the determination of its satisfaction with respect to any document referred to herein shall be in writing signed by the Underwriter and delivered to the University. This Bond Purchase Agreement shall become legally effective upon mutual acceptance and approval.

This Bond Purchase Agreement may be executed in several counterparts, each of which shall be regarded as an original and all of which shall constitute one and the same document.

LEHMAN BROTHERS INC.

By:_____ Richard B. King Senior Vice President

ACCEPTED:

IDAHO STATE UNIVERSITY

By:_____ Kenneth R. Prolo Vice President of Financial Services and Bursar

EXHIBIT A

IDAHO STATE UNIVERSITY \$10,000,000 GENERAL REVENUE BONDS (FEDERALLY TAXABLE), SERIES 2006

Purchase Price:

Principal Amount	\$
Plus Net Premium	
Purchase Price	\$

Interest on the Idaho State University General Revenue Bonds (Federally Taxable), Series 2006 (the "Series 2006 Bonds") is payable on April 1, 2007 and semiannually on each April 1 and October 1 thereafter. Principal on the Series 2006 Bonds is payable on the dates, and the Series 2006 Bonds will bear interest at the rates, set forth below:

Maturity (April 1)	Principal Amount	Interest <u>Rate</u>	Yield
2007			
2008			
2009			
2010			
2011			
2012			
2013			
2014			
2015			
2016			
2017			
2018			
2019			
2020			
2021			
2022			
2023			
2024			
2025			
2026			
2027			
2028			
2029			
2030			

Attachment 3

Maturity	Principal	Interest	Yield
(April 1)	<u>Amount</u>	<u>Rate</u>	
2031 2032 2033 2034			

Redemption Provisions:

Optional Redemption of the Series 2006 Bonds

The Series 2006 Bonds are subject to redemption, in whole or in part, at the option of the University at any time or from time to time upon notice as hereinafter provided, on any date prior to their maturity at a redemption price equal to 100% of the principal amount of such Series 2006 Bonds plus the "Make-Whole Premium," if any, as described below, plus the accrued interest, if any, thereon to the redemption date.

The amount of the Make-Whole Premium with respect to any Series 2006 Bond to be redeemed will be equal to the excess, if any, of:

(1) the sum of the present values, calculated as of the redemption date, of:

(a) each interest payment that, but for such redemption, would have been payable on the Series 2006 Bond or portion thereof being redeemed on each interest payment date occurring after the redemption date (excluding any accrued interest for the period prior to the redemption date); and

(b) the principal amount that, but for such redemption, would have been payable at the final maturity of the Series 2006 Bond being redeemed; over

(2) the principal amount of the Series 2006 Bond being redeemed.

The present values of interest and principal payments referred to in clause (1) above shall be determined in accordance with generally accepted principles of financial analysis. These present values will be calculated by discounting the amount of each payment of interest or principal from the date that each such payment would have been payable, but for the redemption, to the redemption date at a discount rate equal to the "comparable treasury yield" (as defined below) plus 25 basis points. The Make-Whole Premium will be calculated by an independent investment banking institution of national standing appointed by the University (which may be the Underwriter). If the University fails to appoint an independent investment banker at least 45 days prior to the redemption date, or if the independent investment banker appointed by the University is unwilling or unable to make the calculation, the calculation will be made by an independent investment banking institution of national standing appointed by the calculation, the calculation will be made by an independent investment banker appointed by the University is unwilling or unable to make the calculation, the calculation will be made by an independent investment banking institution of national standing appointed by the Trustee.

For purposes of determining the Make-Whole Premium, "comparable treasury yield" means a rate of interest per annum equal to the weekly average yield to maturity of

United States Treasury Securities that have a constant maturity that corresponds to the remaining term to maturity of the Series 2006 Bonds, calculated to the nearest 1/12th of a year. The comparable treasury yield will be determined as of the third Business Day immediately preceding the applicable redemption date.

The weekly average yields of United States Treasury Securities will be determined by reference to the most recent statistical release published by the Federal Reserve Bank of New York and designated "H.15(519) Selected Interest Rates" or any successor release. If the H.15 statistical release sets forth a weekly average yield for United States Treasury Securities having a constant maturity that is the same as the remaining term calculated as set forth above, then the comparable treasury yield will be equal to such weekly average yield. In all other cases, the comparable treasury yield will be calculated by interpolation on a straight-line basis, between the weekly average yields on the United States Treasury Securities that have a constant maturity closest to and less than the remaining term (in each case as set forth in the H.15 statistical release or any successor release). Any weekly average yields calculated by interpolation will be rounded to the nearest 1/100th of 1%, with any figure of 1/200th of 1% or above being rounded upward. If weekly average yields for United States Treasury Securities are not available in the H.15 statistical release or otherwise, then the comparable treasury yield will be calculated by interpolation of comparable rates selected by an independent investment banker selected in the manner described above.

Mandatory Redemption of the Series 2006 Bonds

The Series 2006 Bonds maturing on April 1, 2007 shall be subject to mandatory redemption and retirement prior to maturity, in part, pro rata and not by lot, on the date of payment of each Mandatory Redemption Amount. The portion of the principal amount of each Series 2006 Bond to be redeemed (per \$5,000 original principal amount) on each Mandatory Redemption Date and the resulting principal amount of each Series 2006 Bond remaining outstanding (per \$5,000 original principal amount) upon each such redemption are shown in the following table:

Mandatory Redemption Date (April 1) Principal Amount to be Redeemed (Per \$5,000 original Mandatory <u>Redemption Amount</u>) Remaining Principal Amount (Per \$5,000 original Mandatory Redemption Amount)

* Final maturity

EXHIBIT B

INFORMATION REPORTING AGREEMENT

Re: \$______ principal amount of General Revenue Bonds (Federally Taxable), Series 2006 (the "Series 2006 Bonds" or the "Bonds"), dated as of the date of delivery, of Idaho State University (the "Issuer") and issued pursuant to a Supplemental Resolution, adopted November 30, 2006 (the "Resolution") which names U.S. Bank National Association, as Trustee (the "Trustee")

THIS INFORMATION REPORTING AGREEMENT (the "Agreement") is executed and delivered by the Issuer and the Trustee, as of the date set forth below, in order for the Issuer to authorize and direct the Trustee, as the agent of the Issuer, to make certain information available to the public, as herein provided.

WITNESSETH:

1. <u>Background</u>. The Issuer has resolved to issue the Bonds pursuant to the Resolution. The CUSIP number assigned to the final maturity of the Bonds is

2. <u>Appointment of Trustee</u>. The Issuer hereby appoints the Trustee and any successor Trustee acting as such under the Resolution to be the Issuer's agent under this Agreement to disseminate the information furnished by the Issuer hereunder in the manner and at the times provided herein and to discharge the other duties assigned.

3. <u>Information to be Furnished by the Issuer</u>. The Issuer hereby covenants for the benefit of the registered and beneficial owners of the Bonds that, as long as the Bonds are outstanding under the Resolution, the Issuer will deliver the following information to the Trustee:

(a) Within 180 days after the end of the Issuer's fiscal year, the audited financial statements of the Issuer, together with the report thereon of the Issuer's independent auditors, beginning with Fiscal Year ended June 30, 2007.

(b) The other financial, statistical and operating data for said fiscal year of the Issuer in the form and scope similar to the financial, statistical and operating data contained in "SECURITY FOR THE SERIES 2006 BONDS," "HISTORICAL PLEDGED REVENUES AND DEBT SERVICE" and, in the event the Bonds are called, the financial data contained in "DEBT SERVICE REQUIREMENTS" of the Issuer's Official Statement.

(c) In a timely manner, notice of any of the following events with respect to the Bonds, if material:

1. Principal and interest payment delinquencies on the Bonds;

- 2. Nonpayment related defaults;
- 3. Unscheduled draws on debt service reserves reflecting financial difficulties;
- 4. Unscheduled draws on credit enhancements reflecting financial difficulties;
- 5. Substitution of credit or liquidity providers, or their failure to perform;
- 6. Modifications to rights of Bondholders;
- 7. Bond calls;
- 8. Defeasances;
- 9. Release, substitution or sale of property securing repayment of the Bonds;
- 10. Rating changes; and
- 11. Notice of a failure of the Issuer to provide the annual financial information specified in Section 3(a) and 3(b) hereof.

4. <u>Manner and Time by Which Information is to be made Public by the</u> <u>Information Agent</u>.

(a) The information required to be delivered to the Trustee pursuant to Sections 3(a) and 3(b) shall be referred to as the Continuous Disclosure Information (the "Continuous Disclosure Information") and the notices required to be delivered to the Trustee pursuant to Section 3(c) shall be referred to as the Event Information (the "Event Information").

(b) Not later than five business days after the receipt of any Continuous Disclosure Information or any Event Information, the Trustee will deliver the information as provided in the following subsection (c).

- (c) It shall be the Trustee's duty to:
 - 1. deliver the Continuous Disclosure Information to the Repositories once it is received from the Issuer;
 - 2. deliver the Event Information to the Repositories and the MSRB once it is received from the Issuer; and

3. determine the identity of all Repositories to which Continuous Disclosure Information and Event Information must be sent under the rules and regulations promulgated by the MSRB or by the SEC and to determine the existence of any SID by calling the SEC's Public Reference Room at (202) 942-8090.

(d) The Trustee shall have no duty or obligation to disclose any information other than (i) Continuous Disclosure Information that the Trustee actually has received from the Issuer and (ii) Event Information about which the Trustee has received notice from the Issuer. Any such disclosures shall be required to be made only as and when specified in this Agreement and only to those entities specified in this Section 4. The Trustee's duties and obligations are only those specifically set forth in this Agreement and the Resolution, and the Trustee shall have no implied duties or obligations.

(e) On or before the 31st day of January of each year, the Trustee shall provide to the Issuer written notice of the names of the Repositories to which the Trustee has sent Continuous Disclosure Information during the preceding calendar year and the names of the Repositories to which the Trustee proposes to send Continuous Disclosure Information during the current calendar year.

5. <u>Indemnification</u>.

(a) The Trustee shall have no obligation to examine or review the Continuous Disclosure Information and shall have no liability or responsibility for the accurateness or completeness of the Continuous Disclosure Information disseminated by the Trustee hereunder. The Continuous Disclosure Information shall contain a legend to such effect.

(b) The Issuer hereby agrees to hold harmless and to indemnify the Trustee, its employees, officers, directors, agents and attorneys from and against any and all claims, damages, losses, liabilities, reasonable costs and expenses whatsoever (including attorneys' fees and expenses, whether incurred before trial, at trial, or on appeal, or in any bankruptcy or arbitration proceedings), which may be incurred by the Trustee by reason of or in connection with the disclosure of information in accordance with this Agreement, except to the extent such claims, damages, losses, liabilities, costs or expenses result directly from the willful misconduct or gross negligence of the Trustee in the performance of its duties under this Agreement.

6. <u>Compensation</u>. The Issuer hereby agrees to compensate the Trustee for the services provided and the expenses incurred pursuant to this Agreement in an amount to be agreed upon from time to time hereunder. Such compensation shall be in addition to any fees previously agreed upon with respect to the fiduciary services of the Trustee in its capacity as the Trustee.

7. <u>Enforcement</u>. The obligations under this Agreement of the Issuer shall be for the benefit of the registered and beneficial holders of the Bonds. Any holder of the Bonds then outstanding, including any Beneficial Owner of the Bonds (as defined in the Resolution), may enforce specific performance of such obligations by any judicial proceeding available, provided however, that a default under this Agreement will not be a default under the Resolution.

8. <u>Definitions</u>. As used herein, the following terms shall have the following meanings:

"MSRB" shall mean the Municipal Securities Rulemaking Board.

"Official Statement" shall mean the final official statement relating to the Bonds dated ______, 2006.

"Repository" shall mean (i) each nationally recognized municipal securities information repository recognized by the SEC from time to time pursuant to the Rule and (ii) any SID.

"Rule 1 5c2-12" shall mean Rule 15c2-12, as amended, promulgated by the SEC under the Securities Exchange Act of 1934, as amended.

"SEC" shall mean the Securities and Exchange Commission.

"SID" means any state information depository of the State of Idaho.

9. <u>Amendments and Termination</u>. This Agreement may be amended with the mutual agreement of the Issuer and the Trustee and without the consent of any registered or beneficial holders of the Bonds under the following conditions:

(a) the amendment is made in connection with a change in circumstances that arises from a change in legal requirements, change in law, or change in the identity, nature, or status of the obligated person or type of business conducted;

(b) this Agreement, as amended, would have complied with the requirements of Rule 15c2-12 at the time of the primary offering, after taking into account any amendments or interpretations of Rule 15c2-12, as well as any change in circumstances; and

(c) the amendment does not materially impair the interests of holders of the Bonds, as determined by parties unaffiliated with the Issuer (such as the Trustee or nationally recognized bond counsel).

Any party to this Agreement may terminate this Agreement by giving written notice of its intent to terminate to the other parties at least thirty (30) days prior to such termination, provided that no such termination shall relieve the obligation of the Issuer to comply with Rule 15c2-12(b)(5) either through a successor agent or otherwise.

BAHR – SECTION II

The Issuer's next annual report must explain, in narrative form, the reasons for any such amendment or termination and the impact of the change in the type of operating data or financial information being provided or, in the case of accounting principles, the presentation of such operating data or financial information.

The undertaking contained in this Agreement shall be in effect from and after the issuance and delivery of the Bonds and shall extend to the earlier of (i) the date all principal and interest on the Bonds shall have been paid pursuant to the terms of the Resolution; (ii) the date that the Issuer shall no longer constitute an "obligated person" within the meaning of Rule 15c2-12; or (iii) the date on which those portions of Rule 15c2-12 that require this written undertaking (a) are held to be invalid by a court of competent jurisdiction in a nonappealable action, (b) have been repealed retroactively, or (c) in the opinion of counsel who is an expert in federal securities laws, acceptable to the Issuer or the Trustee, otherwise, do not apply to the Bonds. The Issuer shall notify the Repositories if this Agreement is terminated pursuant to (iii), above.

10. <u>Successor Trustee</u>. Upon the transfer of the duties created under the Resolution from the current Trustee to a successor Trustee, such successor Trustee shall succeed to the duties under this Agreement without any further action on the part of any party, and the then current Trustee shall have no further duties or obligations upon the transfer to a successor Trustee. Such Successor Trustee may terminate this Agreement or cause it to be amended as provided in paragraph 9.

11. <u>Notices</u>. Notices and the required information under this Agreement shall be given to the parties at their addresses set forth below under their signatures or at such places as the parties to this Agreement may designate from time to time.

12. <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, and each such instrument shall constitute an original counterpart of this Agreement.

13. <u>Governing Law</u>. This Agreement shall be governed by the laws of the State of Idaho.

IN WITNESS WHEREOF, the Issuer and the Trustee have caused this Agreement to be executed and delivered by a duly authorized officer of each of them, all as of this _____ day of ______, 2006.

IDAHO STATE UNIVERSITY

By_____ Kenneth R. Prolo Vice President of Financial Services

U.S. BANK NATIONAL ASSOCIATION, as Trustee

By_____ Kim R. Galbraith Vice President

EXHIBIT C

SUPPLEMENTAL OPINION OF BALLARD SPAHR ANDREWS & INGERSOLL, LLP,

EXHIBIT D

OPINION OF COUNSEL TO IDAHO STATE UNIVERSITY

[Idaho State University letterhead]

December ____, 2006

The Board of Trustees of Idaho State University Campus Box 8219 Pocatello, ID 83209

Ballard Spahr Andrews & Ingersoll, LLP 201 South Main Street, Suite 600 Salt Lake City, UT 84111

Lehman Brothers Inc. 701 Fifth Avenue, Suite 7101 Seattle, WA 98104

Re: General Revenue Bonds (Federally Taxable), Series 2006 in the Principal Amount of \$10,000,000

Ladies and Gentlemen:

As General Counsel to Idaho State University (the "University"), I have reviewed certain documents in connection with the issuance and sale by the University of the above-captioned bonds (the "Bonds"), including the Resolution of the Board of Trustees of the University (the "Board") adopted on September 17, 1992 as previously amended, supplemented and restated (the "Original Resolution") and as further supplemented and amended dated November 30, 2006, authorizing the issuance and sale of the Bonds (the "Supplemental Resolution," and, together with the Original Resolution as previously "Resolution"), the Preliminary Official Statement dated supplemented. the ____, 2006, and the Official Statement dated ______, 2006 (the "Official Statement"), and such other documents as I deemed necessary to render this opinion. Capitalized terms used as defined terms in this opinion have the meaning assigned to such terms in the Resolution. This opinion is rendered under the Bond Purchase Agreement dated November 30, 2006, between the University and Lehman Brothers Inc. (the "Bond Purchase Agreement").

BAHR – SECTION II

Based upon my examination, it is my opinion that:

1. The University is an institution of higher education and a body politic of the State of Idaho, duly and validly created and existing pursuant to the laws of the State of Idaho, with full legal right, power, and authority (i) to issue bonds of the University pursuant to the Resolution; (ii) to adopt the Resolution; (iii) to enter into the Bond Purchase Agreement; (iv) to pledge the Pledged Revenues (as defined in the Resolution) to secure the payment of the principal of and interest on the Bonds; (v) to execute and deliver the CAES Lease Agreement (as defined in the Resolution); and (vi) to carry out and consummate the transactions contemplated by the Resolution, the CAES Lease Agreement and Bond Purchase Agreement.

2. The provisions of the Resolution, the execution and delivery of the Bond Purchase Agreement and the CAES Lease Agreement, and the performance by the University of the transactions contemplated thereby will not conflict with or constitute a breach of, or default under, any commitment, note, agreement or other instrument to which the University is a party or by which it or any of its property is bound, or any provision of the Idaho Constitution or laws or any existing law, rule, regulation, ordinance, judgment, order or decree to which the University or its Board is subject.

3. Based upon conferences with and representations of officials of the University, the statements in the Official Statement under the subheading "Idaho State University" under the caption "INTRODUCTORY STATEMENT" and under the captions "SECURITY FOR THE SERIES 2006 BONDS," "THE CAES PROJECT," "THE UNIVERSITY," "SOURCES OF FUNDING FOR THE UNIVERSITY," "UNIVERSITY GOVERNANCE AND ADMINISTRATION" and "LITIGATION" contained in the Official Statement are true and correct in all material respects and do not contain an untrue statement or omission of a material fact, it being understood that, in rendering this opinion, I am not expressing an opinion with respect to financial, statistical or operating data contained under these captions of the Official Statement. I hereby consent to the inclusion of my name as counsel to the University in the section of the Official Statement entitled "LEGAL MATTERS" and on the cover page thereof.

4. Except as described in the Official Statement, there is no action, suit, proceeding, official inquiry or investigation, at law or in equity, pending which (i) questions the existence or powers of the Board or the University or the title to office of any present official of the University; (ii) seeks to prohibit, restrain or enjoin the sale, issuance or delivery of any of the Bonds or the execution and delivery of the Bond Purchase Agreement or the CAES Lease Agreement; (iii) affects the collection of the Pledged Revenues pledged or to be pledged to pay the principal of and interest on the Bonds, or the pledge of the revenues and other funds and accounts under the Resolution; (iv) contests the completeness or accuracy of the Official Statement; or (v) contests any authority for the issuance of the Bond Purchase Agreement or the CAES Lease Agreement or the CAES Lease Agreement or the caes and the adoption of the Resolution, or the execution and delivery of the Bond Purchase Agreement or the CAES Lease Agreement or the CAES Lease Agreement or the caes and other funds and accounts under the Resolution; or the execution and delivery of the Bond Purchase Agreement or the CAES Lease Agreement or the caes and the adoption of the Resolution, or the execution and delivery of the Bond Purchase Agreement or the CAES Lease Agreement or the validity of any proceedings taken by the University in connection with the issuance or sale of the Bonds.

Very truly yours,

Bradley H. Hall University Counsel

THIS PAGE INTENTIONALLY LEFT BLANK
INSTITUTION / AGENCY AGENDA UNIVERSITY OF IDAHO

SUBJECT

Amendments to University of Idaho Faculty-Staff Handbook Sections 5300 "Copyrights, Protectable Discoveries and Other Intellectual Property Rights," and 5400 "Employment Agreement Concerning Intellectual Property."

APPLICABLE STATUTE, RULE, OR POLICY

Idaho State Board of Education Governing Policies and Procedures, Section V.M.2.c.

BACKGROUND

Section 5300 of the University of Idaho Faculty Staff Handbook sets out UI policy concerning copyrights, maskworks, and patents as they arise from the work of university employees and students and from the use of university resources, including the assignment of ownership and development of such intellectual property. This section was part of the 1979 Handbook but was revised in a significant way 1) in July of 1992 to reflect changes in applicable federal law, and 2) in January of 1995 to reflect the then change in the Regents' intellectual property and the then conflict of interest rule (IDAPA 08.01.09.101.03.c).,

Section 5400 of the Handbook sets out the agreement concerning patents and copyrights that all employees must sign. This section was added to the Handbook in June of 1988 and revised in July of 1992. Temporary revisions of this form occurred in 2003 and 2005, pending internal approval through the institution's process of final revisions.

DISCUSSION

The University of Idaho encourages the creation of scholarly works as an integral part of its mission. University participation in the development, marketing, and dissemination of educational materials has as its aim the improvement of the quality, effectiveness, and efficiency of student learning and of faculty and staff development. The University recognizes its obligation to transfer technology and useful discoveries to society. With respect to all types of intellectual property, the rights and obligations of the University of Idaho, its employees and students and other third parties are governed by sections 5300 and 5400.

The proposed amendments to Faculty-Staff Handbook sections 5300 and 5400 update the policy and employee agreement, and clarify the rights and obligations of employees and students related to all types of intellectual property. The amendments are the result of the work of an ad-hoc committee of the Faculty Council, which included staff, students and faculty, and have been approved by

INSTITUTION / AGENCY AGENDA UNIVERSITY OF IDAHO - continued

the Faculty Council and the general faculty. They are presented to the Board for approval pursuant to Board policy section V.M.2.c., which requires State Board approval of institutional policies and amendments thereto that address intellectual property and the division and use of royalties from patents.

Faculty-Staff Handbook Sections 5300 and 5400, including tracked changes, are attached.

IMPACT

The intellectual property section of the University's Faculty-Staff Handbook has not been updated since 1995. Terminology in this area has evolved such that the handbook needs updating. In addition, some parts of the current handbook lack clarity which has led to instances of confusion over the rights and obligations of faculty, staff and students with respect to intellectual property created or used by them.

The proposed amendments update terminology and add clarity to the rights and obligations of the University and of its employees and students in dealing with intellectual property, all in line with the University's renewed focus on an environment of openness and transparency. There is no financial impact, nor is there any material change in the University's rights to intellectual property.

ATTACHMENTS.

Attachment 1 – Proposed Section 5300	Page 3
Attachment 2 – Redline version – Section 5300	Page 9
Attachment 3 – Proposed Section 5400	Page 17
Attachment 4 – Redline version – Section 5400	Page 21

STAFF COMMENTS AND RECOMMENDATIONS

These proposed changes have been reviewed by Board staff and have been discussed extensively with University counsel.

Staff recommends approval of the amendments.

BOARD ACTION

Motion to approve amendments to University of Idaho Faculty-Staff Handbook Sections 5300 "Copyrights, Protectable Discoveries and Other Intellectual Property Rights," and 5400 "Employment Agreement Concerning Intellectual Property."

Moved by	Seconded by	Carried Yes	No	
----------	-------------	-------------	----	--

BAHR - SECTION II

Chapter 5: RESEARCH POLICIES Section 5300: Copyrights, Protectable Discoveries and Other Intellectual Property Rights July 2006

Attachment 1

UI FACULTY-STAFF HANDBOOK CHAPTER FIVE: RESEARCH POLICIES

July 2006

5300

COPYRIGHTS, PROTECTABLE DISCOVERIES AND OTHER INTELLECTUAL PROPERTY RIGHTS

PREAMBLE: This section outlines UI policy concerning copyrights, maskworks, and patents as they arise from university research. Particularly this section discusses the assignment of ownership to such copyrights, maskworks, and patents. This section was part of the 1979 Handbook but was revised in a significant way 1) in July of 1992 to reflect changes in applicable federal law, 2) in January of 1995 by the addition of subsection C-5 to reflect the change in the Regent' intellectual property and conflict of interest rule (former IDAPA 08.01.09.101.03c), and 3) in 2006 to update terminology and add clarity to the rights and obligations of the University and of its employees and students in dealing with intellectual property . For more information, contact the Research Office (208-885-6651). [ed. 7-98]

CONTENTS:

- A. Introduction
- B. Copyrights
- C. Protectable Discoveries
- D. Dispute Resolution
- E. Special Arrangements for Federal, State, and Private Grants
- F. Record-Keeping

A. INTRODUCTION. The UI encourages the creation of scholarly works as an integral part of its mission. UI participation in the development, marketing, and dissemination of educational materials has as its aim the improvement of the quality, effectiveness, and efficiency of student learning and of faculty and staff development. The UI recognizes its obligation to transfer technology and useful discoveries to society. With respect to all types of intellectual property, the rights and obligations of UI, its employees and students and other third parties shall be governed by this policy. To the extent permitted by this policy, individuals may enter into contracts with UI to address intellectual property, in which case the contract terms shall control, provided that the contract was entered into in a manner consistent with this policy.

A-1. DEFINITIONS. For purposes of this Section 5300 and Section 5400, the following terms shall have the following meanings:

a. "electronic" shall mean relating to technology having electrical, digital, magnetic, wireless, optical, electromagnetic, or similar capabilities.

b. "written" or "in writing" shall include information created, generated, sent, communicated, received, or stored by electronic means, including without limitation email, telecopy, and facsimile transmissions. c. "natural person or persons" means natural person or persons involved in the creation or development of intellectual property.

B. COPYRIGHTS. UI participation in the development of copyrightable works raises questions concerning the ownership and use of materials in which UI has become an active and intentional partner through substantial investment of resources. This policy is established to clarify the rights of the natural person or persons and the UI regarding ownership and use of copyrightable materials in the absence of a valid written agreement between the natural person or persons and UI. The UI acknowledges the right of faculty and staff members and students to prepare and publish materials that are copyrightable in the name of the natural person or persons? is to be construed broadly as including producers of creative works in the arts and sciences and creators of literary or **BAHR – SECTION II**

Chapter 5: RESEARCH POLICIES

Section 5300: Copyrights, Protectable Discoveries and Other Intellectual Property Rights

July 2006

scholarly writing.)

B-1. Coverage. The types of materials to which this policy applies include:

- **a.** Study guides, tests, syllabi, bibliographies, texts, books, and articles.
- b. Films, filmstrips, photographs, slides, charts, transparencies, illustrations, and other visual aids.
- c. Programmed instructional materials.
- d. Audio and video recordings.
- e. Simultaneously recorded live audio and video broadcasts.
- f. Dramatic, choreographic, and musical compositions.
- g. Pictorial, graphic, and sculptural works.

h. Computer software, including computer programs, procedural design documents, program documents, and databases as defined below: *[ed. 7-00]*

(1) "Computer program" means a set of instructions that direct a computer to perform a sequence of tasks.

(2) "Procedural design document" refers to material that describes the procedural steps involved in the creation of a computer program.

(3) "Program document" refers to material created for the purpose of aiding the use, maintenance, or other interaction with a computer program.

(4) "Data base" means a collection of data elements grouped together in an accessible format.

i. Other copyrightable materials, including materials generated in the production of any of the above works.

B-2. Assignment of Ownership. Faculty, staff members, and students retain all rights in the copyrightable materials they create except in the cases of "UI-Sponsored Materials" as defined in Subsection B-2-b below, materials covered by a Grant or Contract as discussed in Subsection E below, and materials covered by a valid written agreement between the natural person or persons and the UI as discussed in Subsection B-5 below. Faculty members, staff members, and students shall co-operate with reasonable requests from UI for the creation of any documents and records needed to vest and memorialize UI's rights, if any.

a. Retention of Rights. Except as otherwise provided in Subsection B-2-b, the natural person or persons involved in the creation or development of intellectual property retain the rights to: (1) copyrightable works produced while on sabbatical leave; (2) study guides and similar materials; and (3) works prepared as part of the general obligation to produce scholarly or other creative works of the natural person or persons, such as, but not limited to articles, books, musical compositions, and works of art.

b. UI-Sponsored Materials. Materials are "UI-Sponsored Materials" within the meaning of this policy if the natural person or persons: (1) was commissioned specifically in writing by UI or one of its distinct units to develop the material as part of his or her employment duties and the writing states that the resulting works would be considered "UI Sponsored"; (2) received extra pay from UI to prepare the specific materials pursuant to a valid written agreement providing that the extra pay is consideration for the preparation of the specific materials; (3) received release time from regular duties to produce the specific materials; or (4) made "substantial use" of UI resources in the creation or development of the specific materials, provided however that the use of UI resources regularly and customarily available to him/her as part of his/her regular employment or as part of

his/her regular academic enterprise, shall not be considered "substantial use" of UI resources.

B-3. Registration of Copyrightable Materials. Absent a valid written agreement otherwise, UI Sponsored Materials are to be registered in the name of the Regents of the University of Idaho or its' assignee. UI or its designee has the right to file registrations of UI Sponsored copyrightable works.

B-4. Royalties and Income.

a. Out of the gross receipts from royalties and other income from sale or rental of UI Sponsored Materials, the UI, college, department, other unit, or UI's designated agent may recover reasonable expenses that it incurred in the development, marketing, or dissemination of the materials.

b. Absent a valid written agreement to the contrary, the net proceeds are distributed as follows: 40 percent to the natural person or persons, 40 percent to UI or its designated agent, and 20 percent to the college or service unit of the natural person or persons. At least half of the share allocated to the college or other unit is given to the department of the natural person or persons for use in furtherance of its goals.

c. UI retains a right to royalty-free internal use of any materials designated UI Sponsored under this policy.

B-5. Written Agreements.

a. The provost represents UI in negotiating agreements with the natural person or persons pursuant to this policy. The natural person or persons of copyrightable material may negotiate with the provost and arrive at a mutually agreeable contract. The provost consults with the dean or departmental administrator of the department of the natural person or persons in drafting these agreements. (For purposes of this policy, "dean" includes persons with equivalent administrative capacities.)

b. Valid written agreements concerning copyright ownership, use of copyrighted materials, and distribution of royalties and income from copyrightable works which are entered into by one or more natural person or persons and the provost supersede the provisions of this Section 5300. To be valid, such agreements must (1) comply with the terms of any relevant Grants or Contracts as discussed in Subsection E below, (2) comply with the policies of the UI Board of Regents, (3) comply with UI agreements with the Idaho Research Foundation (IRF), and (4) comply with Idaho state and federal law.

B-6. Use of UI-Sponsored Materials. Use of UI Sponsored Materials under this policy is subject to the following conditions:

a. Internal Use. Internal use is use by anyone employed by UI, or attending the UI as a student, while acting within the scope of his or her employ or academic enterprise, or any agent of UI acting within the scope of his or her agency, either directly or through a grant or contract, or by any UI unit. Internal use of UI Sponsored Materials for the same general purpose for which they were developed, and revision of such materials, do not require the prior approval or notification of any of the natural person or persons. However, for as long as any natural person or persons involved in the creation or development of UI Sponsored Materials remains a UI employee or student, such natural person or persons may, in a professionally appropriate manner, propose revisions of the material.

b. External Use. External use is any use other than that defined in <u>Subsection B-6-a</u>, above. Licensing or sale of UI Sponsored Materials for external use must be preceded by a valid written agreement between the natural person or persons and UI or the UI's designated agent specifying the conditions of use, and including provisions concerning updating or revision of the materials.

B-7. Protection.

a. Allegations of unauthorized use or copyright infringement of UI Sponsored Materials should be made to the **BAHR – SECTION II TAB 6 Page 5**

Chapter 5: RESEARCH POLICIES Section 5300: Copyrights, Protectable Discoveries and Other Intellectual Property Rights July 2006

Intellectual Property Committee for investigation. The committee will recommend appropriate action to the provost.

b. If such action is initiated by UI alone or in concert with the natural person or persons, the costs are borne by UI or UI's agent. Proceeds from the action in excess of costs are shared as provided in Subsection B-4-b.

c. If the natural person or persons involved in the creation or development of the allegedly infringed intellectual property desires to institute a suit and UI decides not to act, UI will co-operate either by assigning to the natural person or persons such rights as are necessary for the natural person or persons to pursue redress or by some other reasonable method acceptable to UI. The costs of the suit will be born by the natural person or persons desiring to sue, who will also obtain any monetary relief obtained from the alleged infringer due to the prosecution of the suit.

B-8. Liability. When either UI or the natural person or persons involved in the creation or development of materials copyrighted by UI or its assignee is alleged to have violated personal or property rights, UI or its designated agent assumes responsibility for the defense against such allegation and the satisfaction of any judgment rendered against UI or the natural person or persons except insofar as liability of governmental entities is limited by Idaho Code 6-903 as currently written or later amended.

B-9. Waiver. Any natural person or persons governed by Section 5300 B waives any claim that otherwise legal use of the material by UI, its agents, employees, distinct units, or IRF creates legal liability by UI, its agents, employees, distinct units, or IRF on any theory of indirect liability for allegedly infringing actions of third parties.

C. PROTECTABLE DISCOVERIES. "Protectable Discoveries," for purposes of this Section 5300 is defined to include anything which might be protected by utility patent, plant patent, design patent, plant protection certificate, maskwork, or trade secret. All Protectable Discoveries made by UI employees at any of its facilities in the course of programs carried on by UI or made by persons in the course of working on such programs or projects under contracts or agreements with UI belong to UI. The natural person or persons involved in the creation or development of such Protectable Discoveries shall assign to UI all such (1) Protectable Discoveries, (2) applications for legal protection of such Protectable Discoveries, and (3) utility patents, plant patents, design patents, and plant protection certificates resulting from such Protectable Discoveries. Absent an valid written agreement to the contrary, any Protectable Discoveries made by UI employees or such other natural person or persons identified above with the use of facilities (other than library resources, normal office use, incidental use of the UI internet network consistent with UI internet use policy, and other facilities for which the person has paid use fees) owned by UI or made available to it for project or research purposes are deemed to have been made in the course of working on a research program or project of UI.

C-1. Ownership By Other Than UI. A Protectable Discovery made by a natural person or persons wholly on his or her own time outside of his or her duties at UI and without the use of UI facilities (other than library resources, normal office use, incidental use of the UI internet network consistent with UI internet use policy, and other facilities for which the person has paid use fees) belongs to that natural person or persons, even though it falls within the field of competence relating to the person's UI position. This provision also allows any Protectable Discovery made by a natural person or persons in the course of private consulting services carried out by the person in conformance with the UI's policy on professional consulting and additional workload [see 3260] to be assigned to the consulting sponsor.

C-2. IRF and UI Processes. UI and the Idaho Research Foundation (IRF) agree that all Protectable Discoveries made by natural person or persons in the course of working on a UI research program or project must be submitted to IRF for acceptance. If a Protectable Discovery is accepted by IRF for development, management, marketing, licensing, or assignment in any manner for the purposes of this policy, UI must cause such property to be conveyed, assigned, or transferred to IRF. IRF has full power to manage such rights and to enter into contracts and licensing concerning such rights, including the right to join in agreements with other nonprofit intellectual property-management entities. [rev. 7-97, 7-06]

Chapter 5: RESEARCH POLICIES Section 5300: Copyrights, Protectable Discoveries and Other Intellectual Property Rights July 2006

a. Upon submission of intellectual property to IRF, IRF must make a formal written decision to pursue commercialization for that property within three months or return the rights to UI. If IRF does not file for protection of the intellectual property within eighteen months of the date the disclosure was submitted, the rights are returned to UI. If IRF submits a provisional patent application for intellectual property protection, a "full" and non-provisional patent application must be submitted within nine months of the date of the submission of the provisional patent or the rights to the property are returned to UI. The property may remain with IRF for a second eighteen-month period if both UI and IRF agree. *[add. 7-97; ed. 7-98]*

b. The IRF shall submit semi-annual reports, as long as it owns the property, to both the inventor/natural person or persons and UI on (1) the status of the application until such time that protection is granted, (2) the marketing activities for the property being serviced, and (3) an accounting for funds received from the property. In the event that IRF has been unsuccessful in transferring a property or filing a patent application within three years after its first acceptance, IRF must notify UI in writing and the property shall be transferred to UI. [add. 7-97, rev. 7-06]

c. If IRF determines not to pursue commercialization of a Protectable Discovery that it has accepted it shall reconvey, assign, and transfer the Protectable Discovery back to the University. The University may elect to pursue commercialization of the Protectable Discovery or, subject to controlling federal law, including but not limited 37 CFR 401 "Bayh-Dole"), reconvey, assign and transfer the Protectable Discovery to the natural person or persons involved in the creation of the intellectual property.

C-3. Proceeds. IRF will make provision to share the net proceeds, management, and licensing of any Protectable Discovery assigned to IRF as follows:

a. Legal and development expenses incurred by IRF constitute a lien on the net proceeds until paid.

b. Absent a valid written agreement to the contrary, the net proceeds in excess of such expenses shall be distributed as follows: 40 percent to the natural person or persons; 40 percent to IRF for tax-exempt purposes; and 20 percent to the college or service unit of the natural person or persons. At least half of the share allocated to the college or other unit is given to the department of the natural person or persons for use in furthering its goals.

C-4. Ownership Questions. Questions as to the ownership of a Protectable Discovery or division of proceeds between persons involved in development of such discoveries and departments are referred in the first instance to the Intellectual Property Committee. The disputes will be decided in accordance with Section 5300(D).

Note: rule is gone and Regent's policy does not require this.

D. DISPUTE RESOLUTION. From time to time, disputes will inevitably occur concerning ownership of the intellectual property (copyrights and Protectable Discoveries) contemplated in this Section 5300. Resolution of such disputes shall be achieved by the following procedure:

D-1. Intellectual Property Committee. The Intellectual Property Committee (IP Committee) shall be an Ad Hoc Committee formed when necessary by appointments made by the Provost, in consultation with the Chair of Faculty Council and the President of the Graduate and Professional Student Association (GPSA). Normally the IP Committee shall be composed of five faculty members and two graduate students. The Provost shall appoint the chair from among the faculty members. In the event the GPSA shall fail to appoint one or more student members, the IP Committee may nonetheless be formed by the Provost and conduct business without the GPSA student representatives.

D-2. Recommendation by the Intellectual Property Committee. The IP Committee considers, investigates, and makes recommendations toward resolution of disputes concerning (1) ownership of copyrightable materials and Protectable Discoveries, and (2) allegations or unauthorized use or copyright infringement of UI Sponsored Materials. It reviews all relevant evidence submitted to it before making its recommendation to the provost. The IP Committee's recommendation is to be made no later than 60 days after receiving the matter for consideration. The IP

Chapter 5: RESEARCH POLICIES Section 5300: Copyrights, Protectable Discoveries and Other Intellectual Property Rights July 2006

Committee's recommendation is determined by a majority of all its members voting by secret ballot at a meeting at which over one-half its appointed members are present. No member may participate in any matter in which his or her ownership rights are being determined.

D-3. Decision by the Provost. After receiving the recommendation of the IP Committee, the provost makes a decision concerning ownership or infringement. The provost's decision is made no later than 30 days after receiving the IP Committee's recommendation. That decision is transmitted in writing to the natural person or persons and to his or her departmental administrator and dean.

D-4. Appeal of the Decision of the Provost. The decision of the Provost may be appealed to the President of the University. Further appeals shall be made as from any other decision of an administrative body under the laws of the State of Idaho in effect from time to time.

E. SPECIAL ARRANGEMENTS FOR FEDERAL, STATE, AND PRIVATE GRANTS. Nothing in this policy shall prevent UI from accepting research grants from, and conducting research for, agencies of the United States upon terms and conditions under applicable provisions of federal law or regulations that require a different disposition of rights in any form of intellectual property. Moreover, nothing herein shall prevent cooperative arrangements with other agencies of the state of Idaho for research. Where receipt of a grant in support of research from any nonprofit agency or group may be dependent upon acceptance of terms and conditions of the established intellectual property policy of the grantor that differ from those stated herein, UI may specifically authorize acceptance of such grant upon such terms and conditions. UI may also specifically authorize contractual arrangements with an industrial sponsor for different disposition of rights in any form of intellectual property resulting from its sponsored research.

F. RECORD-KEEPING. See Section 5500 for record-keeping procedures that are recommended in order to safeguard the property rights of UI or the faculty member in research and potentially patentable results.

[For form of employment agreement concerning patents, see 5400.]

Chapter 5: RESEARCH POLICIES

Section 5300: Copyrights, Maskworks, and PatentsProtectable Discoveries and Other Intellectual Property Rights

July 2006

Attachment 2

UI FACULTY-STAFF HANDBOOK CHAPTER FIVE: RESEARCH POLICIES

July 20002006

5300

COPYRIGHTS, <u>PROTECTABLE DISCOVERIES AND OTHER INTELLECTUAL PROPERTY</u> <u>RIGHTS</u>MASKWORKS, AND PATENTS

PREAMBLE: This section outlines UI policy concerning copyrights, maskworks, and patents as they arise from university research. Particularly this section discusses the assignment of ownership to such copyrights, maskworks, and patents. This section was part of the 1979 Handbook but was revised in a <u>significantmajor</u> way <u>1</u>) in July of 1992 to reflect changes in applicable federal law. <u>2</u>) and in January of 1995 by the addition of subsection C-5 to reflect the change in the Regents= intellectual property and conflict of intent rule (former IDAPA 08.01.09.101.03c), and 3) in 2006 to. Unless otherwise noted, the text is as of July 1996. For more information, contact the Research Office (208-885-6651). [ed. 7-98]

CONTENTS:

- A. Introduction
- B. Copyrights and Maskworks
- C. PatentsProtectable Discoveries
- D. Dispute Resolution
- E. Special Arrangements for Federal, State, and Private Grants
- F. Record-Keeping

A. INTRODUCTION. The university <u>UI</u> encourages the creation of scholarly works as an integral part of its mission. UI participation in the development, marketing, and dissemination of educational materials has as its aim the improvement of the quality, effectiveness, and efficiency of student learning and of faculty and staff development. The university <u>UI</u> recognizes its obligation to transfer technology and useful discoveries to society. <u>With respect to all types</u> of intellectual property, the rights and obligations of UI, its employees and students and other third parties shall be governed by this policy. To the extent permitted by this policy, individuals may enter into contracts with UI to address intellectual property, in which case the contract terms shall control, provided that the contract was entered into in a manner consistent with this policy.

A-1. DEFINITIONS. For purposes of this Section 5300 and Section 5400, the following terms shall have the following meanings:

a. "electronic" shall mean relating to technology having electrical, digital, magnetic, wireless, optical, electromagnetic, or similar capabilities.

b. "written" or "in writing" shall include information created, generated, sent, communicated, received, or stored by electronic means, including without limitation email, telecopy, and facsimile transmissions. c. "natural person or persons" means natural person or persons involved in the creation or development of intellectual property..

B. COPYRIGHTS-AND MASKWORKS. UI participation in the development of copyrightable works and maskworks raises questions concerning the ownership and use of materials in which UI has become an active and intentional partner through substantial investment of resources. This policy is established to clarify the rights of the natural person or persons authors and the UI university regarding ownership and use of copyrightable materials and maskworks in the absence of a valid written agreement between the natural person or persons author and UI. The

BAHR – SECTION II

Chapter 5: RESEARCH POLICIES

Section 5300: Copyrights, Maskworks, and PatentsProtectable Discoveries and Other Intellectual Property Rights

July 2006

<u>university-UI</u> acknowledges the right of faculty and staff members and students to prepare and publish materials that are copyrightable in the <u>author's-name of the natural person or persons</u> and that may generate royalty income for the <u>authornatural person or persons</u>. (In this policy, "<u>author</u>"-"<u>natural person or persons</u>" is to be construed broadly as including producers of creative works in the arts and sciences and <u>creators</u> of literary or scholarly writing.)

B-1. Coverage. The types of materials to which this policy applies include:

a. Study guides, tests, syllabi, bibliographies, texts, books, and articles.

b. Films, filmstrips, photographs, slides, charts, transparencies, illustrations, and other visual aids.

c. Programmed instructional materials.

d. Audio and video recordings.

e. <u>Simultaneously recorded Ll</u>ive audio and video broadcasts.

f. Dramatic, choreographic, and musical compositions.

g. Pictorial, graphic, and sculptural works.

h. Computer software, including computer programs, procedural design documents, program documents, maskworks, and data–bases as defined below: *[ed. 7-00]*

(1) "Computer program" means a set of instructions that direct a computer to perform a sequence of tasks.

(2) "Procedural design document" refers to material that describes the procedural steps involved in the creation of a computer program.

(3) "Program document" refers to material created for the purpose of aiding the use, maintenance, or other interaction with a computer program.

<u>(4)</u> "Maskwork" refers to a series of related images, however fixed or encoded, including but not limited to pattern generation tapes, (a) having or representing the predetermined three dimensional pattern of metallic, insulating, or semiconductor material present or removed from the layers of a semiconductor chip product and (b) in which series the relation of the images to one another is that each image has the pattern of the surface of one form of the semiconductor chip product.

(45) "Data base" means a collection of data elements grouped together in an accessible format.

i. Other copyrightable materials, including materials generated in the production of any of the above works.

B-2. Assignment of Ownership. Faculty, and staff members, and students retain all rights in the copyrightable materials and maskworks-they create except in the cases of "university<u>UI-Ssponsored" mMaterials" as defined in Subsection B-2-b below[see B 2 b]</u>, materials covered by the terms of a <u>gG</u>rant or e<u>C</u>ontract as discussed in <u>Subsection E below</u>, and materials covered by a valid written agreement between the <u>natural person or persons author</u> and the university<u>UI</u> as discussed in <u>Subsection B-5 below</u>. Faculty members, staff members, and students shall cooperate with reasonable requests from UI for the creation of any documents and records needed to vest and memorialize <u>UI's rights, if any.</u>

a. <u>Retention of Rights.</u> Except as otherwise provided in Subsection B-2-b, the natural person or persons involved in the creation or development of intellectual property. In particular, authors retain the rights to: (1) copyrightable works produced while on sabbatical leave, unless there is a valid written agreement to the contrary; (2) study guides and similar materials not specifically commissioned by UI and the preparation of

Chapter 5: RESEARCH POLICIES

Section 5300: Copyrights, Maskworks, and PatentsProtectable Discoveries and Other Intellectual Property Rights

July 2006

which did not involve substantial use of UI resources [see B 2_b(4)]; and (3) works prepared as part of the author's general obligation to produce scholarly or other creative works of the natural person or persons, such as, as well as works that customarily have been considered the property of the author, including but not limited to articles, books, musical compositions, and works of art.

b. University UI-Sponsored Materials. Materials are "UI-sSponsored Materials" within the meaning of this policy if the <u>natural person or persons author</u>: (1) <u>waswere specifically</u> commissioned <u>specifically</u> in writing by UI or one of its distinct units to develop the materials as part of his or her <u>regular employment</u> duties <u>and the</u> writing states that the resulting works would be considered "UI Sponsored."; (2) received extra pay from UI to prepare the <u>specific</u> materials <u>pursuant to a valid written agreement providing that the extra pay is consideration</u> for the preparation of the specific materials; (3) received release time from regular duties to produce the <u>specific</u> materials; or (4) made "substantial use" of UI resources (such as computer centers, audiovisual service units, or other similar department service centers) in the creation or their development of the specific materials, provided however that the use of unless UI accepts or waives reimbursement for such use; limited use of secretarial support, use of the library for which special charges are not normally assessed, and other routine use of UI resources are resources regularly and customarily available to him/her as part of his/her regular employment or as part of his/her regular academic enterprise, shall not <u>be</u> considered to <u>be</u> "substantial" use" of UI resources.

B-3. Registration of Copyrightable Materials and Maskworks. UI or its designee is responsible for applying to the U.S. Copyright Office for registration of copyrightable materials when under this policy materials have been finally designated UI sponsored. Such application must be made no later than 60 days after the materials have been designated UI sponsored. Absent a valid written agreement <u>otherwise</u>to the contrary, UI-<u>S</u>sponsored <u>mM</u>aterials are to be copyrighted and maskworks</u>-registered in the name of the Regents of the University of Idaho or <u>itsUI's</u> assignee. <u>UI or its designee has the right to file registrations of UI Sponsored copyrightable works</u>.

B-4. Royalties and Income.

a. Out of the gross receipts from royalties and other income from sale or rental of $UI-\underline{sS}$ ponsored <u>M</u>materials, the <u>UIuniversity</u>, college, department, other unit, or UI's designated agent may recover reasonable expenses that it incurred in the development, marketing, or dissemination of the materials.

b. Absent a valid written agreement to the contrary, the net proceeds are distributed as follows: 40 percent to the <u>natural person or personsauthor(s)</u>, 40 percent to UI or its designated agent, and 20 percent to the author's college or service unit <u>of the natural person or persons</u>. At least half of the share allocated to the college or other unit is given to the author's department <u>of the natural person or persons</u> for use in furtherance of its goals.

c. UI retains a right to royalty-free internal use of any materials designated UI_<u>-sS</u>ponsored under this policy.

B-5. Written Agreements.

a. The provost represents UI in negotiating agreements with <u>the natural person or persons authors</u>-pursuant to this policy. The <u>natural person or persons author</u> of copyrightable material may negotiate with the provost and arrive at a mutually agreeable contract. The provost consults with the <u>author's dean or</u> departmental administrator <u>of the department of the natural person or persons and dean</u> in drafting these agreements. (For purposes of this policy, "dean" includes persons with equivalent administrative capacities.)

b. Valid written agreements concerning copyright ownership, use of copyrighted materials, and distribution of royalties and income from copyrightable works which are and income entered into by one or more an authornatural person or persons and the provost may supersede the provisions of this Section 5300. this policy and are binding, provided they are in compliance with the regents' rules, UI agreements with the Idaho Research Foundation (IRF), and other UI contractual agreements. To be valid, such agreements must (1) comply with the terms of any relevant Grants or Contracts as discussed in Subsection E below, (2) comply with the rulespolicies of the UI Board of Regents, (3) comply with UI agreements with the Idaho Research Foundation (IRF), and 4) comply with Idaho state and federal law.

Chapter 5: RESEARCH POLICIES

Section 5300: Copyrights, Maskworks, and PatentsProtectable Discoveries and Other Intellectual Property Rights

July 2006

B-6. Use of UniversityUI-Sponsored Materials. Use of UI-<u>sS</u>ponsored copyrightable mMaterials under this policy is subject to the following conditions:

a. Internal Use. Internal use is use by anyone employed by UI, <u>or attending the UI as a student</u>, while acting within the scope of his or her employ <u>or academic enterprise</u>, or any agent of UI acting within the scope of his or her agency, either directly or through a grant or contract, or by any UI unit.

(1) Internal use of UI_<u>-sS</u>ponsored <u>M</u>materials for the same <u>general</u> purpose for which they were developed, <u>and revision of such materials</u>, does not require <u>the</u> prior approval <u>or notification of any of the natural person</u> <u>or persons</u>. by the author unless there is a valid written agreement to the contrary between the author and the provost or designee. HoweverEven so, for as long as any the author natural person or persons involved in the creation or development of UI Sponsored Materials remains a UI employee or student, <u>such natural person or persons may</u>, in a professionally he or she may propose reasonable revisions of the materials. If the author and the departmental administrator cannot agree on appropriate <u>mannerrevisions</u>, proposed revisions of the material the question will be referred to the author's dean for final determination.

(2) If the use is to be for a purpose other than that for which the materials were developed, the author, as long as he or she remains a UI employee or student, may propose reasonable revisions before each instance of such use. If the author and the departmental administrator cannot agree on appropriate revisions, the questions will be referred to the author's dean. If the author and the dean cannot agree on appropriate revisions, the author may withdraw the materials from use.

(3) If withdrawn materials have been copyrighted, UI may assign its rights to the author, subject to a valid written agreement between UI and the author regarding further internal or external use of the materials and division of income from any subsequent use.

(4) When the author ceases to be a UI employee or student, UI retains the right to make internal use of the materials without the author's consent unless the author and UI have agreed in writing on different conditions for subsequent internal use of the materials and the procedures for their revision.

b. External Use. External use is any use other than that defined in <u>a-Subsection B-6-a.</u> above. Licensing or sale of UI_<u>-sS</u>ponsored <u>mM</u>aterials for external use must be preceded by a valid written agreement between the <u>natural person or persons author</u> and UI or the <u>UI's university's</u>-designated agent specifying the conditions of use, and including provisions concerning updating or revision of the materials.

B-7. Protection.

a. Allegations of unauthorized use or copyright infringement of UI<u>-sS</u>ponsored \underline{mM} aterials should be made to the Intellectual Property Committee for investigation. The committee will recommend appropriate action to the provost.

b. If such action is initiated by UI alone or in concert with the <u>authornatural person or persons</u>, the costs are borne by UI or UI's agent. Proceeds from the action in excess of costs are shared as provided in <u>Subsection</u>B-4-b.

c. If the action is initiated by the author and UI decides not to act, UI assigns to the author such rights as are necessary for the author to pursue redress. The costs are borne by the author, and the proceeds go to the author. If the natural person or persons involved in the creation or development of the allegedly infringed intellectual property desires to institute a suit and UI decides not to act, UI will co-operate either by assigning to the natural person or persons such rights as are necessary for the natural person or persons to pursue redress or by some other reasonable method acceptable to UI. The costs of the suit will be born by the natural person or persons desiring to sue, who will also obtain any monetary relief obtained from the alleged infringer due to the

BAHR - SECTION II

TAB 6 Page 12

Chapter 5: RESEARCH POLICIES

Section 5300: Copyrights, Maskworks, and PatentsProtectable Discoveries and Other Intellectual Property Rights

July 2006

prosecution of the suit.

B-8. Liability. When either UI or the <u>author natural person or persons involved in the creation or development</u> of materials copyrighted by UI or its assignee is alleged to have violated personal or property rights, UI or its designated agent assumes responsibility for the defense against such allegation and the satisfaction of any judgment rendered against UI or the <u>natural person or persons</u> <u>author</u> except insofar as liability of governmental entities is limited by Idaho Code 6-903 as currently written or later amended.

B-9. Waiver. Any natural person or persons governed by Section 5300 B waives any claim that otherwise legal use of the material by UI, its agents, employees, distinct units, or IRF creates legal liability by UI, its agents, employees, distinct units, or IRF on any theory of indirect liability for allegedly infringing actions of third parties.

C. PATENTSPROTECTABLE DISCOVERIES. "Protectable Discoveries," for purposes of this Section 5300 is defined to include anything which might be protected by utility patent, plant patent, design patent, plant protection certificate, maskwork, or trade secret. All Protectable Discoveries made by UI employees at any of its facilities in the course of programs carried on by UI or made by persons in the course of working on such programs or projects under contracts or agreements with UI belong to UI. The natural person or persons involved in the creation or development of such Protectable Discoveries shall assign to UI all such (1) Protectable Discoveries, (2) applications for legal protection of such Protectable Discoveries, and (3) utility patents, plant patents, design patents, and plant protection certificates resulting from such Protectable Discoveries. Absent an valid written agreement to the contrary, any Protectable Discoveries made by UI employees or such other- natural person or persons identified above with the use of facilities (other than library resources, normal office use, incidental use of the UI internet network consistent with UI internet use policy, and other facilities for which the person has paid use fees) All patentable inventions made by UI employees at any of its facilities in the course of projects or research programs carried on by UI or made by persons in the course of working on such programs or projects under contracts or agreements with UI belong to UI. The inventor(s) assign all such inventions, patent applications, and resulting patents as directed by UI. Any such inventions made by UI employees or such other persons in the course of their employment by or for UI or with the use of facilities (other than library resources and normal office use) owned by UI or made available to it for project or research purposes are deemed to have been made in the course of working on a research program or project of UI.

C-1. <u>Ownership By Other Than UI.</u> A Protectable Discovery <u>An invention</u> made by a <u>natural</u> person <u>or persons</u> wholly on his or her own time outside of his or her duties at UI and without the use of UI facilities (other than library resources, <u>and</u> normal office use, <u>incidental use of the UI internet network consistent with UI internet use policy</u>, and <u>other facilities for which the person has paid use fees</u>) belongs to that <u>natural person or persons</u>, even though it falls within the field of competence relating to the person's UI position. This provision also allows any <u>Protectable</u> <u>Discovery invention</u> made by a <u>natural person or personsperson</u> in the course of private consulting services carried out by the person in conformance with the UI's policy on professional consulting and additional workload [see 3260] to be assigned to the consulting sponsor.

C-2. IRF and UI Processes. UI and the Idaho Research Foundation (IRF) agree that all <u>Protectable Discoveries</u> inventions made by <u>natural person or personspersons</u> in the course of working on a UI research program or project must be submitted to IRF for acceptance. If <u>a Protectable Discovery such property</u> is accepted by IRF for development, management, marketing, licensing, or assignment in any manner for the purposes of this policy, UI must cause such property to be conveyed, assigned, or transferred to IRF. IRF has full power to manage such patent rights and to enter into contracts and licensing concerning such rights, including the right to join in agreements with other nonprofit <u>intellectual propertypatent</u> management entities. *[rev. 7-97, <u>7-06</u>]*

a. Upon submission of intellectual property to IRF, IRF must make a formal written decision to pursue commercialization for that property within three months or return the rights to UI. If IRF does not file for protection of the intellectual property within eighteen months of the date the disclosure was submitted, the rights are returned to UI. If IRF submits a provisional patent application for intellectual property protection, a "full" and non-provisional patent application must be submitted within nine months of the date of the submission of the provisional patent or the rights to the property are returned to UI. The property may remain with IRF for a second

Chapter 5: RESEARCH POLICIES

Section 5300: Copyrights, Maskworks, and PatentsProtectable Discoveries and Other Intellectual Property Rights

July 2006

eighteen_-month period if both UI and IRF agree. [add. 7-97; ed. 7-98]

b. The IRF shall submit semi-annual reports, as long as it owns the property, to both the inventor/author the <u>natural person or persons</u> and UI on 1) the status of the application until such time that protection is granted, 2) the marketing activities for the property being serviced, and 3) an accounting for funds received from the property. In the event that IRF has been unsuccessful in transferring a property or filing a patent application within three years after its first acceptance, IRF must notify UI in writing and the property shall be transferred to UI. [add. 7-97, <u>rev. 7-06</u>]

c. If IRF determines not to pursue commercialization of a Protectable Discovery that it has accepted it shall reconvey, assign, and transfer the Protectable Discovery back to the University. The University may elect to pursue commercialization of the Protectable Discovery or, subject to controlling federal law, including but not limited 37 CFR 401 "Bayh-Dole"), reconvey, assign and transfer the Protectable Discovery to the natural person or personsperson involved in the creation of the intellectual property.

C-3. <u>Proceeds.</u> IRF will make provision to share the net proceeds, management, and licensing of any <u>patent</u> <u>Protectable Discovery</u> assigned to IRF as follows:

a. Legal and development expenses incurred by IRF constitute a lien on the net proceeds until paid.

b. Absent a valid written agreement to the contrary, the net proceeds in excess of such expenses shall be distributed as follows: 40 percent to the <u>inventor(s)natural person or persons</u>; 40 percent to IRF for tax-exempt purposes; and 20 percent to the college or service unit of the <u>natural person or persons inventor(s)</u>. At least half of the share allocated to the college or other unit is given to the <u>inventor's department of the natural person or persons</u> for use in furthering its goals.

C-4. <u>Ownership Questions.</u> Questions as to <u>the</u> ownership of an <u>Protectable Discoveryinvention</u> or division of proceeds between persons involved in development of such discoveries and departments are referred in the first instance inventors and departments are referred to the Intellectual Property Committee. The <u>disputes will be decided</u> in accordance with Section 5300(D). final decision is made by the provost after a review of the findings and conclusions of the committee and is binding.

C-5. <u>Provost Reports.</u> As required by rule promulgated by the Regents (IDAPA 08.01.09.03c), the provost shall report two weeks in advance of the state board meeting on patent, copyright, and technology transfer activities that have occurred at the institution since the prior meeting of the board. With respect to patents, that report will also indicate whether employees of the institution or of its research foundation have a financial interest in the company to which the intellectual property was transferred. Terms of any licensee or technology transfer contract will be made available in confidence upon request for inspection by the state board. Note: rule is gone and Regent's policy does not require this.

D. DISPUTE RESOLUTION. From time to time, disputes will inevitably occur concerning ownership of the intellectual property (copyrights, maskworks, and patentsProtectable Discoveries) contemplated in this <u>Section</u> <u>5300policy</u>. Resolution of such disputes shall be achieved by the following procedure:

D-1. Intellectual Property Committee. The Intellectual Property Committee (IP Committee) shall be an Ad Hoc Committee formed when necessary by appointments made by the Provost, in consultation with the Chair of Faculty Council and the President of the Graduate and Professional Student Association (GPSA). Normally the IP Committee shall be composed of -five faculty members and two graduate students. The Provost shall appoint the chair from among the faculty members. In the event the GPSA shall fail to appoint one or more student members, the IP Committee may nonetheless be formed by the Provost and conduct business without the GPSA student representatives. is composed of five faculty members appointed each year by the Committee on Committee on Committee on the panel is three years, with initial terms staggered to form a rotation pattern. The Committee on Committees selects the chair.

D-2. Recommendation by the Intellectual Property Committee. The IPC committee considers, investigates, and

BAHR – SECTION II

Chapter 5: RESEARCH POLICIES

Section 5300: Copyrights, Maskworks, and PatentsProtectable Discoveries and Other Intellectual Property Rights

July 2006

makes recommendations toward resolution of disputes concerning (1) ownership of maskworks and copyrightable and patentable materials and Protectable Discoveries, and (2) allegations or unauthorized use or copyright infringement of UI <u>sSponsored mM</u>aterials. It reviews all relevant evidence submitted to it before making its recommendation to the provost. The <u>IP Ceommittee's</u> recommendation is to be made no later than 60 days after receiving the matter for consideration. The <u>IP eCommittee's</u> recommendation is determined by a majority of all its members voting by secret ballot at a meeting at which over one-half its appointed members are present. No member may participates in any matter in which his or her ownership rights are being determined.

D-3. Decision by the Provost. After receiving the recommendation of the <u>IP_eC</u>ommittee, the provost makes a decision concerning ownership or infringement. The provost's decision is made no later than 30 days after receiving the <u>IP_eC</u>ommittee's recommendation. That decision is transmitted in writing to the <u>author_natural person or persons</u> or inventor <u>Protectable Discovery</u> and to his or her departmental administrator and dean.

D-4. Appeal of the Decision of the Provost. The decision of the Provost may be appealed to the President of the University. Further appeals shall be made as from any other decision of an administrative body under the laws of the State of Idaho in effect from time to time.

E. SPECIAL ARRANGEMENTS FOR FEDERAL, STATE, AND PRIVATE GRANTS. Nothing in this policy shall prevent UI from accepting research grants from, and conducting research for, agencies of the United States upon terms and conditions under applicable provisions of federal law or regulations that require a different disposition of patent rights in any form of intellectual property. Moreover, nothing herein shall prevent cooperative arrangements with other agencies of the state of Idaho for research. Where receipt of a grant in support of research from any nonprofit agency or group may be dependent upon acceptance of terms and conditions of the established patent-intellectual property policy of the grantor that differ from those stated herein, UI may specifically authorize acceptance of such grant upon such terms and conditions. UI may also specifically authorize contractual arrangements with an industrial sponsor for different disposition of patent-rights in any form of intellectual property resulting from its sponsored research.

F. RECORD-KEEPING. See <u>Section</u> 5500 for record-keeping procedures that are recommended in order to safeguard the property rights of UI or the faculty member in research and potentially patentable results.

[For form of employment agreement concerning patents, see 5400.]

Chapter 5: RESEARCH POLICIES

Section 5300: Copyrights, Maskworks, and PatentsProtectable Discoveries and Other Intellectual Property Rights

July 2006

THIS PAGE INTENTIONALLY LEFT BLANK

July 2006

5400

EMPLOYMENT AGREEMENT CONCERNING INTELLECTUAL PROPERTY

PREAMBLE: UI uses the following form of employment agreement concerning patents and copyrights. This section was added to the Handbook in June of 1988 and revised in July of 1992, given a substantial but temporary revision in April 2003 and given a second temporary revision in April 2005. For further information, contact the Technology Transfer Office (208-885-4630) or the Office of the Faculty Secretary (208-885-6151).

ADDITIONAL NOTICE: The UI uses the Memorandum of Understanding form of agreement concerning patents and copyrights with non-employee students and visitors participating in university research activities. This allows the non-employee student to participate in the university's patent/copyright income distribution program (FSH 5300) while protecting the interests of the faculty, staff, student, and university. This section was added to the Handbook in May 2003. For further information, contact the Research Office (208-885-6651).

Employment Agreement Regarding Intellectual Property Between The University of Idaho and

As an employee of the University of Idaho (the university), I acknowledge that I am subject to the policies and rules of the Regents of the University of Idaho (Regents) published at the Idaho State Board of Education's website <u>http://www.idahoboardofed.org/policies/index.asp</u> and to the policies and procedures of the university as published in the university's *Faculty-Staff Handbook* and *Administrative Procedures Manual*.

Pursuant to those policies, I hereby agree to the following:

A. With regard to protectable discoveries (which include discoveries potentially protectable as a utility patent, plant patent, design patent, plant variety protection certificate, maskwork, and trade secret):

A-1. I understand that under UI policy FSH 5300, the university owns all protectable discoveries made by UI employees at any of its facilities in the course of projects or research programs carried on by UI or made by persons in the course of working on such programs or projects under contracts or agreements with UI. I will exercise my best effort in notifying the university's Research Office of potentially pprotectable discoveries conceived or first reduced to practice in whole or in part in the course of my university responsibilities. If in doubt about the protectability of adiscovery, I will confer with the university's Research Office.

A-2. I will exercise my best effort in notifying the university's Research Office of potentially protectable discoveries conceived or first reduced to practice in whole or in part through the use of university resources when that use is more than incidental (FSH 5300). Again, if in doubt as to what is incidental use I will confer with the university's Research Office.

A-3. I agree to collaborate with the university in the assignment of rights, title and interests in such protectable discoveries, as required by the policies of the Regents and the university.

A-4. I will exercise my best efforts in providing relevant documentation and participate in actions to complete the assignment of rights, title and interests in such protectable discoveries.

A-5. I will refrain from actions which jeopardize the university's potential rights, including any action which might create a statutory bar preventing grant of patent on an otherwise patentable invention. I recognize that publication, public use, sale or offering for sale of such protectable discovery may create a statutory bar. When in doubt, I will consult with the university's Research Office.

B. I acknowledge that some of the copyrightable materials that I create may be the property of the university as

UI FACULTY-STAFF HANDBOOK Chapter V: RESEARCH POLICIES Section 5400: Patent and Copyright Agreement for University of Idaho Employees July 2006

explained in FSH 5300 B. I acknowledge that I have read these provisions and agree to them. I will cooperate with reasonable requirements of the university to promptly assign or confirm in writing any possible right I might otherwise have in any copyrightable work when such right belongs to university according to FSH 5300 B.

C. I agree to inform all students and visiting scholars wishing to participate in my university research programs, about the university "Memorandum of Understanding Regarding Research Participation and the University Intellectual Property Rights" available from the Technology Transfer Office and about the FSH 2300 - Article II.2 Student Code of Conduct. I will not allow any student or visiting scholar to participate in my university research program who has not signed a copy of the "Memorandum of Understanding Regarding Research Participation and the University Intellectual Property Rights" or an alternative document negotiated between the university's Research Office and the individual.

D. Attached to this agreement are 1. a Disclosure of Invention Work in Progress, and 2. a Disclosure of Prior Contracts (together the "Disclosures"). The Disclosures set forth inventions and/or work with prior employers or firms with which I currently consult that may be protectable discoveries. The work referenced in the Disclosures is excluded specifically from the university's ownership claims so long as no university facilities (other than library resources, normal office use, incidental use of the university's internet network consistent with university's internet use policy, and other facilities for which the person has paid use fees) are used in further development of the works referenced in the Disclosures

E. I acknowledge that I am under no consulting or other obligation to any third person, organization or corporation that is in conflict with this Intellectual Property Agreement with respect to rights to protectable discoveries or copyrightable materials. [NOTE - Any individual who believes that s/he cannot comply with this provision must contact the University Research Office and Idaho Research Foundation]

Signature

Title

Printed Name

Date

Chapter V: RESEARCH POLICIES Section 5400: Patent and Copyright Agreement for University of Idaho Employees July 2006

Memorandum of Understanding **Regarding Research Participation** And **University Intellectual Property Rights**

This memorandum of understanding is entered into by _____ _____, a student at the University of Idaho ("participant"), ____ _____, a professor/researcher at University of Idaho ("faculty"), and the Regents of the University of Idaho ("Regents").

The participant is involved in research activities or enrolled in ______, which may involve working on research or design projects. These activities or projects may or may not result in the development of intellectual property in which the University of Idaho ("UI") and/or a sponsor may have a proprietary interest.

Therefore, it is important that the participant, faculty, and the UI have a full understanding of the participant's rights and obligations regarding these proprietary interests, and intellectual property. This memorandum sets forth the understanding of the parties.

a. The participant acknowledges receipt of copies of the relevant intellectual property policies of the Regents and the UI.

b. The participant agrees to promptly disclose any discoveries he/she makes that may be protectable under any intellectual property theory, including patent and copyright.

c. The participant has the right to submit any thesis, dissertation, or other academic product based upon or resulting from his/her work as part of the fulfillment of the requirements for obtaining an undergraduate, masters, or doctoral degree from the UI resulting from collaboration with the UI provided that such submission is done in a manner that does not crate a statutory bar to the later grant of patent rights in an otherwise protectable discovery.

d. In exchange for the opportunity to participate in these projects and the right to receive royalties, the participant agrees to assign his/her right, title, and interest in any research or other project outcome, including intellectual property rights, derived from the participant's work in this class or research activities to the UI. This assignment vests rights in the UI as provided for in the UI's intellectual property policies and is subject to the participant's right to share in royalties in the same manner as employees of the UI.

Participant	Date	
1 -		

Supervising Faculty _____ Date

Chapter V: RESEARCH POLICIES Section 5400: Patent and Copyright Agreement for University of Idaho Employees

July 2006

DISCLOSURE OF INVENTION WORK IN PROGRESS

This disclosure is made this _____ day of ______, 20___, as part of that Employment Agreement Regarding Intellectual Property between The University of Idaho (UI) and _______, a student or employee of UI (the "Inventor"), dated this _____ day of ______, 20 ___. This Disclosure lists all inventions and developments of the Inventor made prior to employment by, or matriculation as a student at, UI. Subject to UI legal review and verification by UI's Technology Transfer Officer, the UI acknowledges that the inventions and developments listed below constitute the property of the Inventor or the party with whom the Inventor has contracted. A brief description of each invention is provided.

INVENTION

DATES OF WORK RIGHTS OWNED BY

DISCLOSURE OF PRIOR CONTRACTS

This disclosure is made this ____ day of _____, 20__, as part of that Employment Agreement Regarding Intellectual Property between The University of Idaho (UI) and ______, a student or employee of UI (the "Inventor"), dated this ____ day of _____, 20 __. This Disclosure lists all contractual obligations of the Inventor entered into prior to employment by, or matriculation at, UI. Subject to UI legal review and verification by UI's Technology Transfer Officer, the UI acknowledges that prior contracts that remain in effect may be honored by the Inventor. A brief description of each contract is provided below. The types of contracts listed below include, but are not limited to, employment, non-disclosure, non-compete, and fiduciary obligations.

<u>COMPANY OR PERSON</u> <u>TYPE OF CONTRACT</u>

<u>TYPE OF CONTRACT</u> <u>RELEVANT TERMS</u>

July 2006

5400

EMPLOYMENT AGREEMENT CONCERNING INTELLECTUAL PROPERTY

PREAMBLE: UI uses the following form of employment agreement concerning patents and copyrights. This section was added to the Handbook in June of 1988 and revised in July of 1992, given a substantial but temporary revision in April 2003 and given a second temporary revision in April 2005. For further information, contact the Technology Transfer Office (208-885-4630) or the Office of the Faculty Secretary (208-885-6151).

SPECIAL NOTICE: The April 2005 revision is only temporary and will expire July 1, 2006 unless superseded earlier.

ADDITIONAL NOTICE: The UI uses the Memorandum of Understanding form of agreement concerning patents and copyrights with non-employee students and visitors participating in university research activities. This allows the non-employee student to participate in the university's patent/copyright income distribution program (FSH 5300) while protecting the interests of the faculty, staff, student, and university. This section was added to the Handbook in May 2003. For further information, contact the Research Office (208-885-6651).

Employment Agreement Regarding Intellectual Property	
Between	
The University of Idaho and	

As an employee of the University of Idaho (the uUniversity), I acknowledge that I am subject to the policies and rules of the Regents of the University of Idaho (Regents) published at the Idaho State Board of Education's website <u>http://www.idahoboardofed.org/policies/index.asp</u> and to the policies and procedures of the <u>uUniversity as published in the University's University Faculty-Staff Handbook</u> and the University Administrative Procedures Manual.

Pursuant to those policies, I hereby agree to the following:

A. With regard to <u>inventionsprotectable discoveries</u> (which include discoveries potentially protectable as a utility patent, plant patent, design patent, plant variety protection certificate, maskwork, and trade secret):

A-1. I understand that under UI policy FSH 5300, the university owns all <u>patentable_protectable</u>inventions <u>discoveries</u> made by UI employees at any of its facilities in the course of projects or research programs carried on by UI or made by persons in the course of working on such programs or projects under contracts or agreements with UI. I will exercise my best effort in notifying the <u>u</u>University's Research Office of potentially patentable inventionsprotectable discoveries conceived or first reduced to practice in whole or in part in the course of my university responsibilities. If in doubt about the <u>patentability-protectability</u> of an inventiondiscovery, I will confer with the <u>u</u>University's Research Office.

A-2. I will exercise my best effort in notifying the <u>uUniversity's</u> Research Office of potentially <u>patentable inventions</u> <u>protectable discoveries</u> conceived or first reduced to practice in whole or in part through the use of <u>uUniversity</u> resources when that use is more than incidental (FSH 5300). Again, if in doubt as to what is incidental use I will confer with the <u>uUniversity's Research Office</u>.

A-3. I agree to collaborate with the university in the assignment <u>of rights</u>, <u>title and interests in such protectable</u> <u>discoveries</u>, as required by the policies of the <u>Regents</u>-Board of and the university., of rights, title and interests in such patentable inventions.

A-4. I will exercise my best efforts in providing relevant documentation and participate in actions to complete the assignment of rights, title and interests in such patentable inventions protectable discoveries.

A-5. I will refrain from actions which jeopardize the university's potential rights, including any action which might create a statutory bar preventing grant of patent on an otherwise patentable invention. I

recognize that publication, public use, sale or offering for sale of such protectable discovery may create a statutory bar. When in doubt, I will consult with the university's Research Office.

B. I acknowledge that some of the copyrightable materials that I create may be the property of the university as explained in FSH 5300 B. I acknowledge that I have read these provisions and agree to them. I will cooperate with reasonable requirements of the university to promptly assign or confirm in writing any possible right I might otherwise have in any copyrightable work when such right belongs to university according to FSH 5300 B.in the State Board of Education Policies and Procedures Section V.M.3.b., the Board of Regents and the University of Idaho claim no ownership interest in works submitted for publication by faculty, students, and contractors. Employees, students, and contractors retain the right to copyright and publish their own works. That is, as a general rule, all rights to copyrightable material are the property of the creator. The distribution of royalties, if any, is a matter between the creator and his/her publishers and licensees. However, the university may elect, by contract or policy, to claim an interest in the copyrightable material produced, in whole or in part, by their employees and contractors. For example, different treatment may be accorded by the university in cases of specific contracts providing for an exception where the university may employ personnel for the purpose of producing a specific work. Different treatment may be deemed necessary to reflect the contribution of the university to the work as in the case of software, mask works for microcircuit chips, or audiovisual material, or where a sponsored agreement requires otherwise. Specifically, the university policy, except as provided by FSH 5300, is that all rights in copyright shall remain with the creator.

C. I agree to inform all students and visiting scholars wishing to participate in my university research programs, about the university "Memorandum of Understanding Regarding Research Participation and the University Intellectual Property RightsPatent Rights and Copyright" available from the Technology Transfer Office and about the FSH 2300 - Article II.2 Student Code of Conduct. I will not allow any student or visiting scholar to participate in my university research program who has not signed a copy of the "Memorandum of Understanding **F**Regarding **F**Research Participation and the University Intellectual Property RightsPatent Rights and Copyrights" or an alternative document negotiated by between the university's Research Office and the individual.

D. I certify that I am under no consulting or other obligation to any third person, organization or corporation that is, or could be reasonably construed to be, in conflict with this agreement with respect to rights to inventions or copyrightable materials. Attached to this agreement are 1. a Disclosure of Invention Work in Progress, and 2. a Disclosure of Prior Contracts (together the "Disclosures"). The Disclosures set forth inventions and/or work with prior employers or firms with which I currently consult that may be protectable discoveries. The work referenced in the Disclosures is excluded specifically from the university's ownership claims so long as no university facilities (other than library resources, normal office use, incidental use of the university's internet network consistent with university's internet use policy, and other facilities for which the person has paid use fees) are used in further development of the works referenced in the Disclosures

E. I will not enter into any agreement creating copyright or patent obligations in conflict with this agreement. I acknowledge that I am under no consulting or other obligation to any third person, organization or corporation that is in conflict with this Intellectual Property Agreement with respect to rights to protectable discoveries or copyrightable materials. [NOTE - Any individual who believes that s/he cannot comply with this provision must contact the University Research Office and Idaho Research Foundation]

Signature

Title

Printed Name

Date

BAHR – SECTION II

Chapter V: RESEARCH POLICIES Section 5400: Patent and Copyright Agreement for University of Idaho Employees July 20065

Memorandum of Understanding Regarding Research Participation

And

University Patent Rights and CopyrightsIntellectual Property Rights

This memorandum of understanding is entered into by ______, a student at the University of Idaho ("participant"), ______, a professor/researcher at University of Idaho ("faculty"), and the Regents of the University of Idaho ("universityRegents").

The participant is involved in research activities or enrolled in ______, which may involve working on research or design projects. These activities or projects may or may not result in the development of patentable subject matter and/or copyrightable work productsintellectual property in which the <u>University of Idaho ("UI")</u> university and/or a sponsor may have a proprietary interest.

Therefore, it is important that the participant, faculty, and the <u>university UI</u> have a full understanding of the participant's rights and obligations regarding these proprietary interests, <u>and intellectual property</u> copyright, and patent rights. This memorandum sets forth the understanding of the parties.

a. The participant acknowledges receipt of copies of the relevant <u>patent/copyrightintellectual property</u> policies of the Regents of the University of Idaho and the U<u>Iniversity of Idaho</u>.

b. The participant agrees to make promptly disclose any discoveries he/she makes that may be protectable under any intellectual property theory, including patent and copyright. his/her best effort to disclose to the Technology Transfer Office any of_his/her discoveries_he/she makes that may be patentable _____.

c. The participant has the right to submit any thesis, dissertation, or other academic product based upon or resulting from his/her work as part of the fulfillment of the requirements for obtaining an undergraduate, master²s, or doctoral degree from the <u>university-UI or</u>-resulting from collaboration with the <u>UI university provided that such submission is done in a</u> manner that does not crate a statutory bar to the later grant of patent rights in an otherwise protectable discovery.

d. In exchange for the opportunity to participate in these projects and the right to receive royalties, the participant agrees to assign his/her right, title, and interest in any research or other project outcome, including <u>eopyright or patentintellectual</u> <u>property</u> rights, derived from the participant's work in this class or research activities to the <u>Uluniversity</u>. This assignment vests rights in the <u>Uluniversity</u> as provided for in the <u>Uluniversity</u>'s <u>eopyright and patent-intellectual</u> property policies attached hereto and is subject to the participant's right to share in royalties in the same manner as employees of the <u>Uluniversity</u>.

Participant	Date	

Supervising Faculty _____ Date _____

Chapter V: RESEARCH POLICIES Section 5400: Patent and Copyright Agreement for University of Idaho Employees July 20065

DISCLOSURE OF INVENTION WORK IN PROGRESS

This disclosure is made this ____ day of _____, 20__, as part of that Employment Agreement Regarding Intellectual Property between The University of Idaho (UI) and , a student or employee of UI (the "Inventor"), dated this day of , 20 . This Disclosure lists all inventions and developments of the Inventor made prior to employment by, or matriculation as a student at, UI. Subject to UI legal review and verification by UI's Technology Transfer Officer, tThe UI acknowledges that the inventions and developments listed below constitute the property of the Inventor or the party with whom the Inventor has contracted. A brief description of each invention is provided.

INVENTION DATES OF WORK RIGHTS OWNED BY

DISCLOSURE OF PRIOR CONTRACTS

, 20, as part of that Employment Agreement Regarding This disclosure is made this ____ day of _____ Intellectual Property between The University of Idaho (UI) and , a student or employee of UI (the "Inventor"), dated this day of , 20 . This Disclosure lists all contractual obligations of the Inventor entered into prior to employment by, or matriculation at, UI. Subject to UI legal review and verification by UI's Technology Transfer Officer, tThe UI acknowledges that prior contracts that remain in effect may be honored by the Inventor. A brief description of each contract is provided below. The types of contracts listed below include, but are not limited to, -employment, non-disclosure, non-compete, and fiduciary obligations.

COMPANY OR PERSON TYPE OF CONTRACT RELEVANT TERMS

REFERENCE: APPLICABLE STATUTE, RULE, OR POLICY

Idaho State Board of Education GOVERNING POLICIES AND PROCEDURES SECTION: V. FINANCIAL AFFAIRS M. Intellectual Property

April 2002

2. Intellectual Property

- a. Claim of ownership interest. The State Board of Education, on behalf of the state of Idaho, and the Board of Regents, on behalf of the University of Idaho, claim ownership of any invention or patentable discovery developed under any of the following circumstances:
 - (1) Arising from any work performed by an employee of the State Board during the course of his duties to the agency, school or institution;
 - (2) Arising from any work performed by an employee of the State Board using state resources not similarly available to members of the general public; or
 - (3) Arising from any work performed by an employee of the State Board under contract in a program or project sponsored by a State Board agency, school or institutions or a closely related research foundation.
- b. Disclaimer of ownership interest. The State Board of Education and the Board of Regents of the University of Idaho claim no ownership interest in any invention or patentable discovery developed by the employees or contractors under the following circumstances:
 - (1) When the work is performed outside their assigned duties; and
 - (2) When the employee/contractor is without benefit of State facilities except those available to members of the general public, such as libraries.
- c. Policy review Agencies, institutions and the school under the governance of the State Board must secure to the state of Idaho their ownership interest in inventions and patentable discoveries. Agency, school and institutional policies setting out patent administration, including evaluating, financing, assignment, marketing, protection, and the division and use of royalties, as well as amendments thereto, must be submitted to the State Board for its review and approval.

THIS PAGE INTENTIONALLY LEFT BLANK

INSTITUTION / AGENCY AGENDA UNIVERSITY OF IDAHO

SUBJECT

Site lease of two University of Idaho parking lots.

APPLICABLE STATUTE, RULE, OR POLICY

Idaho State Board of Education Policies and Procedures, Section V.I.5.b.1.

BACKGROUND

In 1990, the Board of Regents of the University of Idaho acquired 24,772 sq. ft. (less than one acre) of land located northeast of the UI Student Union Building. This acquired land and some additional property that UI has leased from the adjoining railroad was developed and has been used as a remote parking area for about 90 vehicles. UI Parking Services estimates revenue from parking permits and fines attributable to this site at about \$8,000 per year. Annual maintenance and operations expenses for this site are approximately \$2,673.

DISCUSSION

A local developer, Johnson & Stephens, LLC, is interested in constructing a mixed use development on land they recently acquired, which is between the parking lots and other UI property. The developer has inquired about the availability of these two small UI parking lots. UI staff is supportive of this mixed use development of neighboring property at a major entrance to campus and is proposing a site lease of these remote parking lots at a competitive market rate.

The proposed lease will substantially increase current revenue from the property, eliminate operating expenses and guarantee an income stream for so long as the property is leased. The proposed lease provides a term to 2068 and establishes a rent schedule with escalations that average 2.4% annually for the term of the lease. The annual income stream is unrestricted and may be applied to any needs identified by UI administration. The property and any beneficial improvements constructed by the tenant will revert to UI at the end of the term.

IMPACT

Approval of this site lease will impose no direct costs on UI. The UI is providing land and existing parking lots only. There is adequate available parking of the same grade (blue permits) to make up for the loss of 90 spaces without constructing additional parking spaces. Under the lease, UI will not be responsible for any ongoing maintenance costs and the property will be returned in substantially the same or improved condition upon termination of the lease. The rent schedule attached to the lease provides a fixed (though escalating) payment for the duration of the lease which starts at \$16,800 per year for the first five years.

INSTITUTION / AGENCY AGENDA UNIVERSITY OF IDAHO - continued

ATTACHMENTS

Attachment 1 Attachment 2 Attachment 3 Site Lease Exhibit A - Site Map Exhibit B – Rent Schedule

STAFF COMMENTS AND RECOMMENDATIONS

Parking is an important dynamic of an educational institution. Staff would like a more detailed explanation from the UI regarding the additional parking to be provided and how this proposal does not negatively affect the overall parking situation on the campus.

With the previous comment in mind, and assuming the University does have sufficient additional parking available, staff recommends approval.

BOARD ACTION

A motion to authorize the Vice President for Finance and Administration at the University of Idaho to execute a site lease with Johnson & Stephens, LLC, in substantial conformance to the form submitted to the Board.

Moved by _____ Seconded by _____ Carried Yes ____ No ____

SITE LEASE

THIS SITE LEASE is entered into as of the ____day of December, 2006, by and between THE BOARD OF REGENTS OF THE UNIVERSITY OF IDAHO, a state educational institution and body politic and corporate organized and existing pursuant to the Constitution and laws of the State of Idaho ("Landlord") and Johnson & Stephens, LLC ("Tenant"). This agreement is hereinafter referred to as the "Lease".

RECITALS

A. Landlord is the owner of certain real property shown in Exhibit A ("Premises"). The Landlord desires to enter into a ground lease for the Premises with Tenant. Said Lease to commence as of the date entered into and with occupation of Premises by Tenant to commence on July 1, 2008; and

B. Premises are currently improved as paved parking lots and all Premises parking improvements are owned by Landlord; and

C. Landlord and Tenant agree that Premises shall continue to be maintained and utilized for such parking purposes for the term of this Lease unless and until such a time as Tenant deems, and Landlord consents pursuant to paragraph 9, other site improvements will provide a higher and better use of the Premises to the Tenant and are beneficial to Tenant's project on this and/or close sites in the Neighborhood. Pursuant to such determination, any and all improvements of the Premises shall be the sole cost, expense and responsibility of the Tenant, its assigns, heirs or successors in interest to the Lease, including, but not limited to the construction, maintenance, insurance, and taxation of said improvements. At the end of the term of the Lease, said additional improvements will revert to the Landlord in functional use as specified in paragraph 6 below; and

NOW, THEREFORE, THE PARTIES AGREE AS FOLLOWS:

1. <u>Recitals:</u> The parties acknowledge the foregoing Recitals are true and are incorporated into this Lease as if set forth in full.

2. <u>**Premises:**</u> Landlord hereby leases to Tenant, subject to the conditions expressed herein, certain real property located in the City of Moscow in Latah County, Idaho, specifically shown in Exhibit A attached hereto and incorporated herein. As used herein the term "Premises" refers solely to the real property, including the existing parking improvements (including but not limited to asphalt paving, striping, lights, curbing, and landscaping).

3. <u>**Term:**</u> Except as early termination is provided in paragraph 20 of this Lease, this Lease shall be effective from the date entered into and shall expire on June 30, 2068. Tenant's

occupation of Premises, and rights and duties associated with such occupation shall commence on July 1, 2008. Prior to July 1, 2008, Landlord shall remain responsible for the Premises and shall have all rights and obligations to Premises.

4. **<u>Rent:</u>** The Premises are leased to the Tenant for the amount shown in Exhibit B "Rent Schedule". The amounts are payable on or before July 1 of each year commencing July 1, 2008. Premises rent shall be made payable to Bursar, University of Idaho, and mailed to the attention of "General Accounting, University of Idaho, Moscow ID 83844-3166."

5. <u>Maintenance, Repair and Replacement:</u> Through June 30, 2008, Landlord shall maintain and repair Premises in the usual and customary manner in which it manages all of its other paved parking facilities. On and after July 1, 2008, Tenant shall maintain, repair and replace the Premises over the Term of this Lease. Tenant shall not be permitted to construct new buildings without the written consent of Landlord and in accordance with the provisions of paragraph 9 of this Lease.

6. **Surrender of Premises:** If the Tenant has made improvements to the Premises (as provided by paragraph 9 of this Lease), then Tenant agrees to return the improved Premises to the Landlord with said improvements in a clean, operational, and functional condition. Otherwise Tenant shall surrender Premises to Landlord in a condition that ensures the immediate functional use of no less than fifty (50) parking spaces. All such repair or replacement of the parking facilities to render the parking lot functional shall be completed at Tenant's sole cost and expense. Tenant agrees to restore Premises to a condition satisfactory to Tenant shall not be entitled to compensation for any value attributed to the Landlord. improvements surrendered under the terms of this Lease. In the event the improved Premises are not returned in a clean, operational, and functional condition or Premises are not economically viable, or are physically obsolete for Landlord, Landlord shall notify Tenant of such inadequacies within thirty (30) days of termination. Functional, economic, and physical obsolescence shall be determined by an independent appraisal performed by at least one but not more than three qualified appraisers agreed to by both parties after Landlord's thirty (30) day notification. Upon a determination of obsolescence by selected appraiser(s), Landlord shall provide Tenant with an additional ninety (90) days to correct the situation. In the event Tenant fails to correct the inadequacies within this time period, Landlord may take reasonable actions to cure such obsolescence of the improvements and make the Premises clean, operational, and functional, for use as a parking lot (as described above in this section) and invoice the itemized charges to Tenant for prompt payment.

7. <u>**Taxes:**</u> Tenant shall pay all applicable taxes, license fees, special assessments or other charges (if any) which may become due or which may be lawfully assessed against Premises, against Tenant, against the business conducted on Premises and against any and all improvements thereon during or for the period of the term of this Lease even if such charges are not due and payable until after termination of this Lease.

8. **<u>Right to Assign:</u>** Tenant shall not assign this Lease or sublet the Premises, or any part or portion thereof, without the prior written consent of the Landlord. Landlord shall not unduly withhold consent of assignment of this Lease or of any subletting when such

subtenancy is exclusively for the permitted uses. Notwithstanding any assignment, Tenant shall continue to be bound and obligated by the terms, conditions, covenants and provisions of this Lease until the assignee shall execute and deliver to Landlord an instrument by the terms and provisions of which such assignee shall assume and agree to be bound by and to perform all of the terms, conditions, covenants and obligations of Tenant under this Lease. Upon the execution and delivery of such instrument, Tenant shall be relieved and discharged of and from all obligations under this Lease accruing from and after the date of execution of the instrument. Unless otherwise assumed by the assignee and such assumption is enforceable by Landlord, no assignment of this Lease shall release, waive or discharge Tenant from any liability or obligation arising from or accruing prior to the date any assignee assumes and agrees to be bound by and to perform all of the terms, conditions, covenants and obligations of Tenant under this Lease. Landlord's consent to one assignment shall not waive Tenant's obligation to obtain Landlord's consent or Landlord's right to object to any future assignment.

In the event of sale, transfer or assignment by Landlord of Landlord's interest in the property on which the Premises are located, Landlord shall cause such successor in interest to expressly assume in writing all of Landlord's duties and obligations pursuant to this Lease. Upon receipt of such assumption by the successor in interest, Landlord shall be released from any and all obligations or duties arising under this Lease.

9. **Permitted Uses, Improvements and Prohibited Uses:** Tenant shall use and occupy Premises for the sole and exclusive purpose of maintaining and operating a vehicle parking lot. Tenant or Tenant's assignee may request permission from Landlord to change the permitted uses and make improvements to the property. Landlord's approval of such request shall not be unreasonably withheld, so long as the improvements and corresponding change in use may be readily used by the Landlord for residential, office, and/or retail/commercial uses immediately upon surrender to Landlord as provided in paragraph 6 of this Lease. Tenant shall not use Premises for any of the following uses: a gun range; the sale of guns as a primary use; cemetery; mortuary; any establishment engaged in the business of selling, exhibiting or delivering pornographic or obscene materials, books, magazines or videos; junk yard; recycling facility or stockyard; automotive maintenance or repair facility.

10. <u>Utilities:</u> Tenant agrees to pay all utilities required by Tenant for the use of Premises.

11. **Indemnification and Insurance:** Upon occupation as provided in paragraph 3 of this Lease, Tenant is required to carry the types and limits of insurance shown in this insurance clause, and to provide Landlord with a Certificate of Insurance. At any time, Landlord reserves the right to amend insurance requirements or require a security bond if circumstances warrant such action. Additionally and at its option, Landlord may request certified copies of required policies and endorsements. Such copies shall be provided within ten (10) days of Landlord's request. All insurers shall have a Best's rating of AV or better and be licensed and admitted in Idaho. Tenant shall furnish Landlord with a certificate of insurance executed by a duly authorized representative of each insurer, showing compliance with the insurance

requirements set forth below. All policies required shall be written as primary policies and not contributing to nor in excess of any coverage Landlord may choose to maintain. All certificates shall provide for thirty (30) days' written notice to Landlord prior to cancellation or material change of any insurance referred to therein. All policies shall name State of Idaho and the Regents of the University of Idaho as an additional insured. Certificates shall be mailed to: P.O. Box 443162, Moscow, ID 83844-3162, Attn: Risk Management. All policies shall contain waiver of subrogation coverage or endorsements.

Failure of Landlord to demand such certificate or other evidence of full compliance with these insurance requirements or failure of Landlord to identify a deficiency from evidence that is provided shall not be construed as a waiver of Tenant's obligation to maintain such insurance. Failure to maintain the required insurance may result in termination of this Agreement at Landlord's option. By requiring insurance herein, Landlord does not represent that coverage and limits will necessarily be adequate to protect Tenant, and such coverage and limits shall not be deemed as a limitation on Tenant's liability under the indemnities granted to Landlord in this Site Lease. Tenant shall require any subtenants to comply with the insurance provisions of this Site Lease. In the event Tenant is not occupying space and is enforcing the subtenant's compliance with these insurance provisions, Tenant shall not be responsible, as a requirement of this Site Lease, to maintain the specific insurance required below.

Tenant shall obtain or require subtenant's to obtain insurance of the types and in the amounts described below.

A) Commercial General and Umbrella Liability Insurance. Tenant (or subtenant) shall maintain Commercial General Liability (CGL) and, if necessary, commercial umbrella insurance with a limit of not less than **\$1,000,000** each occurrence and in the aggregate. If such CGL insurance contains a general aggregate limit, it shall apply separately to the Demised Premises and shall not be less than **\$1,000,000**. CGL insurance shall be written on standard ISO occurrence form (or a substitute form providing equivalent coverage) and shall cover liability arising from premises, operations, independent contractors, products-completed operations, personal injury and advertising injury, and liability assumed under an insured contract including the tort liability of another assumed in a business contract.

B) Commercial Auto Insurance. Tenant (or subtenant) shall maintain a Commercial Auto policy with a Combined Single Limit of \$1,000,000; Underinsured and Uninsured Motorists limit of \$1,000,000; Comprehensive; Collision; and a Medical Payments limit of \$10,000. Coverage shall include Non-Owned and Hired Car coverage.

C) Personal property. Tenant (or subtenant) shall purchase insurance to cover Tenant's (or subtenant's) personal property. In no event shall Landlord be liable for any damage to or loss of personal property sustained by Tenant (or subtenant), whether or not insured, even if such loss is caused by the negligence of Landlord, its employees, officers or agents. D) Workers' Compensation. Where required by law, Tenant (or subtenant) shall maintain all statutorily required coverages including Employer's Liability.

12. <u>Property Encumbrances:</u> This Lease is subject to all applicable restrictions, reservations, limitations, and other rights of record, and is subject to any and all easements for public utilities of record.

13. **Hazardous Materials:** Tenant shall not, nor shall it allow others to, accumulate, use, or store on the Premises materials classified as hazardous, biomedical or toxic waste except in compliance with environmental laws and other applicable state, federal, or local laws, rules or regulations. Tenant shall comply and require subtenants to comply with any lawful order by an entity with authority to regulate the use, accumulation, storage or disposal of hazardous waste. As used herein, the term "environmental laws" shall mean the Comprehensive Environmental Response Compensation and Liability Act of 1980, as amended (CERCLA), the Resource Conservation Recovery Act, as amended (RCRA), the Federal Water Pollution Control Act, the Clean Air Act and any similar local, state or federal law, rule, ordinance or regulation. As used herein, the term "hazardous waste or toxic substances defined in any environmental laws including, without limitation, petroleum and petroleum products, asbestos and asbestos containing materials, PCBs and urea-formaldehyde.

Tenant hereby agrees to indemnify, defend, and hold Landlord harmless from and against any and all claims, damages, liabilities, costs, expenses (including reasonable attorneys' fees), causes of action and judgments arising out of or related to hazardous materials existing in, or under the Premises subsequent to Tenant's occupation of the Premises. Tenant shall not be liable for any pre-existing conditions on the Premises or conditions occurring during Landlord's use of the Premises.

14. <u>Waste and Nuisance Prohibited:</u> Tenant shall comply, during the term of this Lease, with all applicable laws affecting the Premises, the violation of which might result in any penalty assessed upon the Landlord or forfeiture of the Landlord's title to the Premises. Tenant shall not commit, or suffer to be committed, any waste on the Premises or improvements, or any nuisance.

15. **<u>Remedies and Forbearance/Waivers:</u>** No delay or omission on the part of the Landlord or Tenant to exercise any right or power granted herein shall impair any such right or power nor shall be construed as a waiver thereof, and every such right or power may nevertheless be exercised.

16. <u>Officials, Agents. and Employees Not Personally Liable:</u> It is agreed that in no event shall any official, officer, employee or agent of the Landlord, nor any official, officer, employee or agent of the Landlord be in any way personally liable or responsible for any covenant or agreement herein contained, whether expressed or implied, nor for any statement, representation or warranty made herein or in any way connected with this Lease.

17. **Quiet Enjoyment:** Landlord covenants that the Tenant shall have the peaceful and quiet enjoyment of the Premises for the term of the Lease.

18. **<u>Right of Entry:</u>** Tenant shall permit the Landlord and the agents and employees of the Landlord to enter into and upon the Premises at all reasonable times for the purpose of inspecting the same for compliance with the terms of this Lease; provided, however, that Landlord shall first give twenty four (24) hours written notice of its desire to inspect Premises. Such notice for right of entry and inspection procedures shall not apply during an emergency in which such notice is impractical and Landlord's access to Premises is necessary for preservation of life and/or property.

19. **<u>Right of First Refusal:</u>** In the event that at any time during the Term of this Lease Landlord shall receive from any third party a bona fide offer to purchase all or any portion of the Premises at a price and terms acceptable to Landlord, Landlord shall give written notice of such purchaser's price and terms to Tenant ("**Notice**") and Tenant shall have thereafter a thirty (30) day exclusive right and option to purchase Premises at the same, or better price and terms as given in the Notice. Tenant may exercise such option if at all, by timely giving Landlord written notice of its exercise within the thirty (30) day period. If Tenant shall fail to exercise such option within such thirty (30) day period, Landlord shall thereafter be free to sell the Premises to such third party on the same terms and conditions as set forth in the Notice. If Premises is not sold by Landlord within one (1) year thereafter to the same third party (or affiliate of such third party) and on the same price and terms as given in the Notice, then the right of first refusal herein shall be restored.

20. <u>Tenant's Right to Early Termination:</u> Except as provided by this paragraph, Tenant shall have the right to terminate this Lease at any time with a one (1) year written notice to Landlord. Said right to terminate Lease early shall continue in effect throughout the term of this Lease so long as the Premises are being used exclusively as a parking lot. Upon initiation of any building construction or once maintenance for exclusive parking use has ceased on Premises, the right to early termination is extinguished until such time as parking is restored to the extent described in paragraph 6.

21. **Default:** In the event Landlord shall at any time deem Tenant or any subtenant in breach of this Lease, Landlord shall promptly notify Tenant, in writing, stating specifically the nature of any such alleged breach. Tenant shall not be deemed to be in default hereunder unless Tenant fails to cure any such default within ninety (90) calendar days after its receipt of such written notice. In the event of default, Landlord shall have the right to terminate Lease in addition to all rights and remedies provided by law.

22. <u>Attorney Fees and Costs:</u> In the event that either party to this Lease shall enforce any of the provisions hereof in any action at law or in equity the prevailing party to such litigation shall be entitled to recover from the other party or parties all costs and expenses, including reasonable attorney fees, incurred therein.

23. <u>Integration:</u> This Lease embodies the entire agreement regarding the disposition of the rights associated with the Premises and represents the understanding of the parties relating to the subject matter herein and supersedes all prior understandings relating thereto. This Lease shall not be modified except in writing signed by all parties to be bound.

24. <u>Execution of Documents</u>: The parties agree that they shall sign or cause to be signed all documents necessary to the effectuation of this Lease or any of the provisions herein.

25. <u>Authority to Enter Lease</u>: Landlord has the authority to enter into this Lease and that the execution, delivery of this Lease and the performance of the contractual obligations set forth herein are not in violation of any federal, state, or local statute, ordinance, rule or regulation and that no consents not already obtained are required. Individuals signing on behalf of Landlord and Tenant have the delegated authority to obligate their respective entity as provided by this Lease.

24. <u>Notices:</u> All notices under this Lease shall be in writing and shall be deemed to have been duly given on the date of service if served personally on the party to whom notice is to be given, or on the date of mailing if mailed to the party to whom notice is to be given by registered or certified United States mail, postage prepaid, and properly addressed as follows:

If to the Landlord:	Regents of the University of Idaho Attn: Vice President, Finance and Administration University of Idaho Moscow ID 83844-3168
If to the Tenant:	Johnson & Stephens, LLC 11245 N Eastshore Dr. Hayden ID 83835-7505

The addresses provided above may be changed and additional addresses or notices may be specified from time to time by notice given in writing in accordance with this Section.

25. **<u>Binding Effect:</u>** This Lease shall be binding upon and shall inure to the benefit of the heirs, personal representatives, successors and assigns of the parties.

26. <u>Severability:</u> If any term or provision of this Lease or the application of it to any person or entity or circumstances shall to any extent be invalid or unenforceable, the remainder of this Lease or the application of such term or provision to persons, entities or circumstances, other than those as to which it is invalid or unenforceable, shall not be affected thereby, and each term and provision of this Lease shall be valid and shall be enforced to the extent permitted by law.

27. <u>Headings:</u> Section headings contained herein are for convenience and reference and are not intended to define or limit the scope of any provisions of this Lease.

28. <u>Counterparts:</u> This Lease may be simultaneously executed in several counterparts, each of which shall be an original and all of which shall constitute but one and the same instrument.

29. <u>**Time of the Essence:**</u> Time is of the essence in this Lease, and of each and every covenant, term, condition, and provisions thereof.

30. **Nondiscrimination and Affirmative Action:** Landlord and Tenant shall not discriminate against any employee or applicant for employment in the performance of this Lease, with respect to tenure, terms, conditions or privileges of employment, or any matter directly or indirectly related to employment, because of race, sex, color, religion, age, status as disabled or a veteran, or physical or mental handicaps, national origin or ancestry. Breach of this covenant may be regarded as a material breach of this Site Lease. Landlord and Tenant certify that they do not, and will not maintain segregated facilities or accommodations on the basis of race, color, religion or national origin. Regarding any position for which an employee or an applicant is qualified, the Landlord and Tenant agree to take affirmative action to employ, train, advance in employment, and retain individuals in accordance with applicable laws and regulations including:

A) For nondiscrimination based on race, color, religion, sex or national origin, this includes, but is not limited to, the U.S. Constitution, and Parts II and IV of Executive Order 11246, September 24, 1965 (30 FR 12319). Grantee disputes related to compliance with its obligations shall be handled according to the rules, regulations, and relevant orders of the Secretary of Labor (See 41 CFR 60-1.1).

B) For nondiscrimination based on Disabled or Vietnam Veterans this includes, but is not limited to, the Vietnam Era Veterans Readjustment Assistance Act of 1972, as amended (38 U.S.C. 4012)(the Act); Executive Order 11701, January 24, 1973 (38 CFR 2675, January 29, 1973); and the regulations of the Secretary of Labor (41 CFR Part 60-250).

C) For nondiscrimination based on the Handicapped this includes, but is not limited to, Section 503 of the Rehabilitation Act of 1973, as amended (29 U.S.C. 793)(the Act); Executive Order 11758, January 15, 1974; and the regulations of the Secretary of Labor (41 FR Part 60- 741).

D) For nondiscrimination based on Age this includes, but is not limited to, executive Order 11141, February 12, 1964 (29 CFR 2477).

E) Landlord and Tenant shall include the terms of this clause in every subcontract or purchase order exceeding \$50,000 and shall act as specified by the Department of Labor to enforce the terms and implement remedies.

31. <u>Venue, Governing Law:</u> Any legal proceeding instituted between the parties shall be in the courts of the County of Latah, State of Idaho, and each of the parties agrees to submit to the jurisdiction of such courts. It is further agreed that this Agreement shall be governed by the laws of the State of Idaho.
IN WITNESS WHEREOF, the parties have caused this Lease to be executed effective as of the day and year first above written.

LANDLORD:

TENANT:

BOARD OF REGENTS OF THE UNIVERSITY OF IDAHO

JOHNSON & STEPHENS, LLC

By_____ Lloyd E Mues, Vice President Finance and Administration

By_____ Russell L Johnson

By_____ James L Stephens

Date_____

Date

THIS PAGE INTENTIONALLY LEFT BLANK



ATTACHMENT 2

Exhibit B Rent Schedule

Annual Rent

July 2008 through June 2013 July 2013 through June 2018 July 2018 through June 2023 July 2023 through June 2028 July 2028 through June 2033 July 2038 through June 2038 July 2043 through June 2043 July 2043 through June 2048 July 2048 through June 2053 July 2053 through June 2058 July 2058 through June 2063 July 2063 through June 2068 \$16,800/yr \$18,816/yr \$21,074/yr \$23,603/yr \$26,435/yr \$29,607/yr \$33,160/yr \$37,139/yr \$41,596/yr \$46,588/yr \$52,179/yr \$58,440/yr

REFERENCE: APPLICABLE STATUTE, RULE, OR POLICY

Idaho State Board of Education GOVERNING POLICIES AND PROCEDURES SECTION: V. FINANCIAL AFFAIRS I. Real and Personal Property Services

April 2002

5. Disposal of Real Property

a. Temporary Permits

Permits to make a temporary and limited use of real property under the control of an institution, school or agency may be issued by the institution, school or agency without prior Board approval.

- b. Board approval of other transfers
 - (1) Leases to use real property under the control of an institution, school or agency require prior Board approval - if the term of the lease exceeds five (5) years or if the lease revenue exceeds two hundred fifty thousand dollars (\$250,000).
 - (2) Easements to make a permanent use of real property under the control of an institution, school or agency require prior Board approval - unless easements are to public entities for utilities.
 - (3) The transfer by an institution, school or agency of any other interest in real property requires prior Board approval.

THIS PAGE INTENTIONALLY LEFT BLANK

INSTITUTION / AGENCY AGENDA UNIVERSITY OF IDAHO

SUBJECT

The University of Idaho (UI) requests approval to proceed with a contract for pouring and vending rights.

APPLICABLE STATUTE, RULE, OR POLICY

Idaho State Board of Education Policies and Procedures Section V.I.6.b.

BACKGROUND

The UI issued a Request for Proposal (RFP) for pouring and vending rights on May 5, 2006. Following the evaluation process, UI determined that a proposal presented by Idaho Beverages Inc., which provides Pepsi products, was in the best interest of the University of Idaho.

DISCUSSION

UI requests permission to grant Idaho Beverages Inc., exclusive pouring and vending rights.

IMPACT

The agreement is for a five (5) year term with options to renew annually for an additional five (5) years. The vendor shall compensate the University of Idaho \$212,000 annually for rights granted.

STAFF COMMENTS AND RECOMMENDATIONS

This contract is for soft drink and other products from the proposed vendor of Pepsi products. The Cover Sheet and attached agreement, Attachment 1, mention 'syrup' and 'bottled water'. Also attached is the Request for Proposal related to this item, Attachment 2.

Staff recommends approval of the contract with the provisions noted in the proposed motion.

BOARD ACTION

A motion to authorize the Vice President for Finance and Administration at the University of Idaho to execute a contract with Idaho Beverages, Inc., in substantial consistency with the draft submitted as part of this agenda item.

Moved by _____ Seconded by _____ Carried Yes ____ No ____

THIS PAGE INTENTIONALLY LEFT BLANK

REFERENCE: APPLICABLE STATUTE, RULE, OR POLICY

Idaho State Board of Education GOVERNING POLICIES AND PROCEDURES SECTION: V. FINANCIAL AFFAIRS I. Real and Personal Property Services

April 2002

6. Disposal of Personal Property

Sale, surplus disposal, trade-in, or exchange of property with a value greater than two hundred fifty thousand dollars (\$250,000) requires prior Board approval.

a. First Refusal

When the property has a value greater than five thousand dollars (\$5,000), the institution, school or agency must first make a good faith effort to give other institutions, school and agencies under Board governance the opportunity of first refusal to the property before it turns the property over to the Department of Administration or otherwise disposes of the property.

b. Sale of Services

The sale of any services or rights (broadcast or other) of any institution, school or agency requires prior approval of the Board when it is reasonably expected that the proceeds of such action may exceed two hundred fifty thousand dollars (\$250,000). Any sale of such services or rights must be conducted via an open bidding process or other means that maximizes the returns in revenues, assets, or benefits to the institution, school or agency.

c. Inter-agency Transfer

Transfer of property from one Board institution, school or agency to another institution, school or agency under Board governance may be made without participation by the State Board of Examiners or the Department of Administration, but such transfers of property with a value greater than two hundred fifty thousand dollars (\$250,000) require prior Board approval.

THIS PAGE INTENTIONALLY LEFT BLANK



Purchasing Services 645 West Pullman Road P.O. Box 441202 Moscow, Idaho 83844-4350 208-885-6123 208-885-6060 FAX juliam@uidaho.edu

October 18, 2006

Gary Prasil, President Idaho Beverages 2108 First Avenue, North Lewiston, Idaho 83501

Subject: University of Idaho Agreement Number UI-605

Dear Gary,

Enclosed please find two original agreements in accordance with the University of Idaho's Request for Proposals Number 06-58M for Pouring and Vending Rights.

Please sign both original copies and return one to me:

Julia R. McIlroy University of Idaho Purchasing Services PO Box 441202 Moscow, Idaho 83844-1202

If you have any questions, please call me at 885.6123, or e-mail juliam@uidaho.edu.

Thank you for doing business with the University of Idaho.

Best regards,

Julia R. McJlroy

Julia R. McIlroy Senior Buyer

UNIVERSITY OF IDAHO AGREEMENT NUMBER UI-605

The University of Idaho (hereinafter called the University) hereby awards to Idaho Beverages, Inc. (hereinafter called the Contractor) agreement number UI-605 to furnish Pouring and Vending services to the University, as specified in University of Idaho Request for Proposals Number

06-58M, in accordance with the terms and conditions of the Request for Proposals.

This agreement is supplemented by a) University of Idaho Request for Proposals Number 06-58M; b) Idaho Beverages' response dated May 31, 2006; and c) University of Idaho General Terms and Conditions, which have been agreed to by the parties and by this reference are made a part hereof as though fully set forth herein. To the extent such terms, conditions, or provisions may be in conflict or be inconsistent, their order of authority shall be as follows: 1) University of Idaho Agreement Number UI-605; 2) University of Idaho Request for Proposals Number 06-58M; 3) Idaho Beverages' response dated May 31, 2006; and 4) University of Idaho General Terms and Conditions.

1.1 NOTICES

Any notice under this Agreement shall be in writing and be delivered in person or by public or private courier service (including U.S. Postal Service Express Mail) or certified mail with return receipt requested or by facsimile. All notices shall be addressed to the parties at the following addresses or at such other addresses as the parties may from time to time direct in writing:

the University:	University of Idaho Purchasing Services PO Box 441202 Moscow, Idaho 83844-1202 Attn.: Julia McIlroy Phone: (208) 885-6123 Fax: (208) 885-6060	
	Copy to: Assistant Vice President Auxiliary Services PO Box 442014 Moscow, Idaho 83844-2014 Phone: Fax:	
the Contractor:	Idaho Beverages 2108 First Avenue North Lewiston, Idaho 83501 Attn: Gary Prasil	

Phone: (208) 743-6535 Fax: (208) 746-2273

Any notice shall be deemed to have been given on the earlier of: (a) actual delivery or refusal to accept delivery, (b) the date of mailing by certified mail, or (c) the day facsimile delivery is verified. Actual notice, however and from whomever received, shall always be effective.1.2 SEVERABILITY

The terms and conditions of this Agreement are declared severable if any term or condition of this Agreement or the application thereof to any person(s) or circumstance(s) is held invalid. Such invalidity shall not affect other terms, conditions, or applications which can be given effect without the invalid term, condition, or application.

1.3 BID PRICE

The bid price shall include everything necessary for the prosecution and completion of this Agreement, including, but not limited to, furnishing all materials, equipment, management, superintendence, labor, and service, except as otherwise provided in this Agreement. Prices quoted on the Bid Form shall include all freight and/or delivery charges. In the event of a discrepancy between the unit price and the total price, the unit price will govern and the total price will be adjusted accordingly.

1.4 TERM OF AGREEMENT

The initial term of this agreement shall be five (5) years, commencing upon the date of execution by the University. The term of this agreement may, if mutually agreed in writing, be extended by one (1) year increments for a total of five (5) additional years, provided the Contractor receives written notice of each extension at least thirty (30) days prior to the expiration date of such term or extension. During extension periods, all terms and conditions of this Agreement shall remain in effect.

1.5 EXCLUSIVITY AND FINANCIAL INCENTIVE

Contractor shall have 100% exclusive pouring and vending services to the University for the term of this agreement. Contractor shall provide the following financial incentive to the University : \$212,000.00 per year throughout the term of the agreement as outlined in 1.4. Contractor shall submit quarterly payments due by the last business day of each quarter, with final payment per fiscal year due by June 30. All payments shall be sent to:

Assistant Vice President Auxiliary Services PO Box 442014

Moscow, Idaho 83844-2014

1.6 ANNUAL BUSINESS REVIEW

The Parties shall meet annually in person to conduct a business review. The meeting will be be scheduled between June 15th and August 15,th, and will include, but not be limited to, discussion of past year performance and recommended changes for the upcoming year. Contractor shall bring sufficient financial data to the meeting to show all elements of the past year's performance.

1.7 VENDING REPORTS

Contractor shall provide monthly vending reports to University that verify the number of machines on campus, location, machine type, features and the monthly revenue.

1.8 FISCAL OPERATING YEAR

Due to the University annual fiscal schedule, the operating year for vending reports will begin June 1st and terminate May 31st each year.

1.9 SYRUP SUPPLY SYSTEM

Contractor shall provide the "Bag in a Box" syrup delivery system. If Contractor wishes to install a different system, such installation shall be subject to University review and written approval. If the syrup delivery system changes, it shall be changed throughout the entire campus so that there is a consistent delivery system across campus.

1.10 BOTTLED WATER

Contractor shall provide an off-brand water with a retail price-point below \$1.00 per bottle.

1.11 CONTRACT MANAGEMENT

The contract will be managed by the University's Auxiliary Services Administration Office. This management includes but is not limited to the collection of all payments, machine locations, scheduling of upgrades, vandal card coordination, any reconciliation and coordination of any meetings. All initial communications regarding contract administration and management shall be made through the Auxilary Services Administration Office. Contractor shall not initiate conversations with other University departments or divisions without first notifying and obtaining the permission of the Assistant Vice President for Auxiliary Services. Formal notices under this contract shall be distributed as provided in section 1.1.

1.12 UNIVERSITY PROGRAM SPONSORSHIPS

The Parties will discuss programs that are determined by the University to be sponsorship opportunities for the Contractor, within the scope of this Contract, with the exception of the Lionel Hampton Jazz Festival and Vandal Athletics, which are agreed uponsponsorship opportunities. The distribution of income from this Contract to sponsored programs shall be determined solely by the University. Contractor, at the request of the Assistant Vice President for Auxiliary Services, will work directly with each individual program for the specific marketing solutions for each sponsored program, which shall correspond to the dollar amount allocated to that program from Contract income for the sponsorship.

1.13 VANDAL CARD READERS

Contractor is responsible for purchase and maintenance of the Vandal Card readers on all vending machines. The University will work cooperatively with the Contractor as needed for repairs and installation. The University will maintain the Vandal Card infrastructure.

1.14 INDEMNITY AND INSURANCE COVERAGE

INDEMNITY CLAUSE

Contractor shall indemnify, defend and hold the State of Idaho, and University and its governing board, employees, agents, and assigns harmless from and against any and all claims, losses, damages, injuries, liabilities and all costs, including attorneys fees, court costs and expenses and liabilities incurred in or from any such claim, arising from any breach or default in the performance of any obligation on Contractor's part to be performed under the terms of this Agreement, or arising from any act, negligence or the failure to act of Contractor, or any of its agents, contractors, employees, invitees or guests.

This clause shall survive the termination of this Agreement.

INSURANCE CLAUSE

General Requirements

CONTRACTOR is required to carry the types and limits of insurance shown in this insurance clause, and to provide UNIVERSITY with a Certificate of Insurance. Certificates shall be provided (7) seven days prior to CONTRACTOR'S use of UNIVERSITY premises.

Additionally and at its option, UNIVERSITY may request certified copies of required policies and endorsements. Such copies shall be provided within (10) ten days of the UNIVERSITY'S request.

All insurers shall have a Best's rating of AV or better and be licensed and admitted in Idaho. Prior to use of premises, CONTRACTOR shall furnish UNIVERSITY with a certificate of insurance executed by a duly authorized representative of each insurer, showing compliance with the insurance requirements set forth below. All policies required shall be written as primary policies and not contributing to nor in excess of any coverage UNIVERSITY may choose to maintain. All certificates shall provide for thirty (30) days' written notice to UNIVERSITY prior to cancellation or material change of any insurance referred to therein.

All policies shall name State of Idaho and the Regents of the University of Idaho as an additional insured. Certificates shall be mailed to: P.O. Box 443162, Moscow, ID 83844-3162, Attn: Risk Management.

All policies shall contain waiver of subrogation coverage or endorsements.

Failure of UNIVERSITY to demand such certificate or other evidence of full compliance with these insurance requirements or failure of UNIVERSITY to identify a deficiency from evidence that is provided shall not be construed as a waiver of CONTRACTOR's obligation to maintain such insurance.

Failure to maintain the required insurance may result in termination of this Agreement at UNIVERSITY's option.

CONTRACTOR shall provide certified copies of all insurance polices required above within then (10) days of UNIVERSITY's written request for said copies.

No Representation of Coverage Adequacy. By requiring insurance herein, UNIVERSITY does not represent that coverage and limits will necessarily be adequate to protect CONTRACTOR, and such coverage and limits shall not be deemed as a limitation on CONTRACTOR's liability under the indemnities granted to UNIVERSITY in this Lease.

Required Insurance Coverage.

CONTRACTOR shall obtain insurance of the types and in the amounts described below.

Commercial General and Umbrella Liability Insurance. CONTRACTOR shall maintain commercial general liability (CGL) and, if necessary, commercial umbrella insurance with a limit of not less than \$5,000,000 each occurrence and in the aggregate. If such CGL insurance contains a general aggregate limit, it shall apply separately to the Demised Premises and shall not be less than \$5,000,000.

CGL insurance shall be written on standard ISO occurrence form (or a substitute form providing equivalent coverage) and shall cover liability arising from premises, operations, independent contractors, products-completed operations, personal injury and advertising injury, food borne illness and contamination, and liability assumed under an insured contract including the tort liability of another assumed in a business contract.

Commercial Auto Insurance. CONTRACTOR shall maintain a Commercial Auto policy with a Combined Single Limit of \$1,000,000; Underinsured and Uninsured Motorists limit of \$1,000,000; Comprehensive; Collison; and a Medical Payments limit of \$10,000. Coverage shall include Non-Owned and Hired Car coverage.

Personal property. CONTRACTOR shall purchase insurance to cover CONTRACTOR's personal property. In no event shall UNIVERSITY be liable for any damage to or loss of personal property sustained by CONTRACTOR, whether or not insured, even if such loss is caused by the negligence of UNIVERSITY, its employees, officers or agents.

Workers' Compensation. Where required by law, CONTRACTOR shall maintain all statutorily required coverages including Employer's Liability at \$1,000,000 limit.

1.15 CONTINUATION DURING DISPUTES

Notwithstanding the existence of any dispute between the Parties regarding this Agreement, each party will continue to perform the obligations required of it under this Agreement during the continuation of any such dispute, unless enjoined or prohibited by any court with jurisdiction over the matter.

1.16 INVOICES

All invoices must contain the name of the University department, purchase order number, itemization of materials and services, and correct Agreement pricing. A packing slip referencing current pricing must accompany each order.

Invoices for payment must be submitted by the Contractor to:

University of Idaho Accounts Payable PO Box 444244 Moscow, ID 83844-4244

1.17 ENTIRE AGREEMENT

This Agreement constitutes the entire Agreement between the parties. No change thereto shall be valid unless agreed upon in writing by the Parties and signed by the individuals with authority to bind each party.

The effective date of this Agreement is September 1, 2006.

UNIVERSITY OF IDAHO	IDAHO BEVERAGES
SIGN	SIGN
PRINT	PRINT

TITLE	 TITLE	
DATE	DATE	

Request for Certificate of Insurance for Certificate Holder (State of Idaho and the Regents of the University of Idaho)

RETURN A COPY OF THESE INSTRUCTIONS WITH YOUR CERTIFICATE.

The organization or individual ("Insured") seeking to [negotiate an Agreement or use facilities] with the University of Idaho ("Certificate Holder") is required to carry the types and limits of insurance shown in this Request, and to provide Certificate Holder with a Certificate of Insurance.

All insurers shall have a Best's rating (or equivalent) of AV or better and be licensed and admitted in Idaho. Prior to inception of grant or contract, Insured shall furnish Certificate Holder with a certificate of insurance executed by a duly authorized representative of each insurer, showing compliance with the insurance requirements set forth below. All policies required shall be written as primary policies and not contributing to nor in excess of any coverage Certificate Holder may choose to maintain.

• Certificate Holder and Additional Insured shall read:

State of Idaho and the Regents of the University of Idaho Attn: Risk Management P.O. Box 443162 Moscow, ID 83844-3162

- Description area of certificate shall read: Attn: [Dept for type of Agreement or facility]
- All certificates shall provide for thirty (30) days' written notice to Certificate Holder prior to cancellation or material change of any insurance referred to in the certificate.
- All policies, other than Workers Compensation, shall name Certificate Holder as an additional insured.
- Failure of Certificate Holder to demand a certificate or other evidence of full compliance with these insurance requirements or failure of Certificate Holder to identify a deficiency from evidence that is provided shall not be construed as a waiver of Insured's obligation to maintain such insurance.
- Failure to maintain the required insurance may result in termination of this grant or contract at the Certificate Holder's option.
- By requiring this insurance, Certificate Holder does not represent that coverage and limits will necessarily be adequate to protect Insured, and such coverage and limits shall not be deemed as a limitation on Insured's liability under the terms of the grant or contract.

Required Insurance Coverage. Insured shall obtain insurance of the types and in the amounts described below.

- Commercial General and Umbrella Liability Insurance. Insured shall maintain commercial general liability (CGL) and, if necessary, commercial umbrella insurance with a limit of not less than \$1,000,000 each occurrence and in the aggregate.
- CGL insurance shall be written on standard ISO occurrence form (or a substitute form providing equivalent coverage) and shall cover liability arising from premises, operations, independent contractors, products-completed operations, personal injury and advertising injury, and liability assumed under an insured contract including the tort liability of another assumed in a business contract.
- Insured shall maintain Automobile Liability in the amount of \$1,000,000 Combined Single Limit.

Coverage shall include Non-Owned and Hired Car coverage.

- If liquor is served, Insured must comply with all policies and procedures established by the Regents of the University of Idaho, including policies regarding a valid liquor license and service procedures. Liquor Legal Liability is required, at a limit of \$1,000,000. It is acceptable to have this provided by the caterer or other liquor supplier, as long as State of Idaho and the Regents of the University of Idaho are named as an additional insured on the Liquor Legal Liability policy.
- Personal property. Insured shall purchase insurance to cover Insured's personal property. In no event shall Certificate Holder be liable for any damage to or loss of personal property sustained by Insured, whether or not insured, even if such loss is caused by the negligence of Certificate Holder, its employees, officers or agents.
- Workers' Compensation. Where required by law, Insured shall maintain all statutorily required Workers Compensation coverages. Coverage shall include Employer's Liability, at minimum limits of \$100,000 / \$500,000 / \$100,000.

If you have additional questions, please contact:

Nancy Spink, Risk Management Officer, University of Idaho. PH (208) 885-6177. FAX (208) 885-9490 nspink@uidaho.edu



Purchasing Services 645 West Pullman Road P.O. Box 441202 Moscow, Idaho 83844-1202

REQUEST FOR PROPOSALS NO. 06-58M

FOR

POURING AND VENDING SERVICES

For Additional Information, Please Contact: Julia R. McIlroy, Senior Buyer Phone (208) 885-6123 Fax (208) 885-6060 juliam@uidaho.edu www.purchasing.uidaho.edu

Date Issued: May 5, 2006

Proposals Due: May 31, 2006 @ 4:00 p.m.

BAHR – SECTION II

SECTION 1 - INSTRUCTIONS TO PROPOSERS

1-1 SCOPE OF WORK

The University of Idaho (herein referred to as the University) is soliciting proposals for a campus-wide pouring and vending services contract.

1-2 PROPOSAL SUBMISSION

Proposal must be **SEALED and CLEARLY IDENTIFIED** with the Request for Proposals' number, due date and time, Proposer's name and address, and submitted no later than 4:00 p.m., Pacific Time, on May 31, 2006 to University of Idaho Purchasing Services, 645 West Pullman Road, P.O. Box 441202, Moscow, ID 83844-1202.

A facsimile response or an electronic response to this Request for Proposals does not meet the requirement of a sealed proposal and will not be accepted.

The proposal must be signed by such individual or individuals who have full authority from the Proposer to enter into a binding Agreement on behalf of the Proposer so that an Agreement may be established as a result of acceptance of the proposal submitted. By reference, the terms and conditions set forth in the Request for Proposals shall serve as the Agreement terms and conditions. In addition, the laws of the State of Idaho shall apply. No other terms and conditions will apply unless submitted as a part of the proposal response and accepted by the University.

Proposals received after the exact time specified for receipt will not be considered.

1-3 REQUEST FOR PROPOSAL SCHEDULE

- 5/5/06 Request for Proposals issued
- 5/31/06 Proposals Due @ 4:00 p.m.

1-4 INQUIRIES

All inquiries concerning this request shall be submitted in writing and received by the University's office of Purchasing Services no later than 4:00 p.m., Pacific Time, on May 24, 2006 to:

Julia R. McIlroy, Senior Buyer University of Idaho Purchasing Services 645 West Pullman Road P.O. Box 441202 Moscow, ID 83844-1202 Telephone: (208) 885-6123 Fax: (208) 885-6060 E-mail: juliam@uidaho.edu

BAHR – SECTION II

TAB 8 Page 16

Proposers should consider Purchasing Services as the first and prime point of contact on all matters related to the procedures associated with this RFP. If additional information is needed from any source, Purchasing Services will work with the Proposer and with the various offices of the University to gather that information.

1-5 INTERPRETATION, CORRECTIONS, OR CHANGES IN RFP

Any interpretation, correction, or change in the RFP will be made by addendum by the University. Interpretations, corrections, or changes to the RFP made in any other manner will not be binding, and no Proposer may rely upon any such interpretation, correction, or change.

1-6 MODIFICATION OR WITHDRAWAL OF PROPOSALS

A Proposer may modify or withdraw a proposal at any time prior to the specified time and date set for the proposal closing. Such a request for modification or withdrawal must be in writing, and executed by a person with authority as set forth under paragraph 1-2 above, or by facsimile notice subsequently confirmed in writing.

1-7 ERASURES AND INTERLINEATIONS

Erasures, interlineations, or other changes in the proposal must be initialed by the person(s) signing the proposal.

1-8 ACKNOWLEDGMENT OF ADDENDUMS TO RFP

Receipt of an addendum to this RFP must be acknowledged by a Proposer on the Proposal Response Certification (Attachment A).

1-9 PROPOSAL COPIES

Seven (7) complete copies of the proposal shall be submitted to the University.

1-10 OFFER ACCEPTANCE PERIOD

A proposal shall constitute an offer to contract on the terms and conditions contained in this RFP and the proposal. Said proposal shall constitute an irrevocable offer for ninety (90) calendar days from the proposal opening date, even if the University makes one or more counter offers.

1-11 REJECTION OF PROPOSALS

The University in its sole discretion, expressly reserves the right to reject any or all proposals or portions thereof, to reissue a Request for Proposal, and to waive informalities, minor irregularities, discrepancies, and any other matter or shortcoming.

1-12 PROPOSAL PRICE

The prices submitted in the proposal shall include everything necessary for the prosecution and completion of the Agreement including, but not limited to, furnishing all materials and all management, supervision, labor

BAHR – SECTION II

Attachment 2

and service, except as may be provided otherwise in the Agreement Documents. In the event of discrepancy between the unit prices and their extensions, the total price will be adjusted accordingly. In the event of discrepancy between the sum of the extended total prices, the Total Proposal Price will be adjusted accordingly. The proposal price shall not include any allowance for Idaho State sales/use tax.

The University will evaluate the total price for the basic requirements with any options(s) exercised at the time of award. Evaluation of options will not obligate the University to exercise the option(s).

The University may reject an offer if it is materially unbalanced as to process for the basic requirements and the option quantities. An offer is unbalanced when it is based on prices significantly less than cost for some work and prices that are significantly overstated for other work.

1-13 TERM OF AGREEMENT

The initial term of this agreement shall be five (5) years, commencing upon the date of execution by the university. The term of this agreement may, if mutually agreed upon in writing, be extended by one (1) year increments for a total of five (5) additional years, provided written notice of each extension is given to the bidder at least thirty (30) days prior to the expiration date of such term or extension. In the event funding approval is not obtained by the University, this Agreement shall become null and void effective the date of renewal. During extension periods, all terms and conditions of this Agreement shall remain in effect.

1-14 AWARD OF AGREEMENT

The University shall make the award to the responsible Proposer whose proposal will be most advantageous to the University with respect to price, conformance to the specifications, quality, and other factors as evaluated by the University. The University is not required or constrained to award the Agreement to the Proposer proposing the lowest price.

The University may award an Agreement on the basis of initial offers received, without discussion; therefore, each initial offer should contain the offerer's best terms from a cost and technical standpoint.

1-15 PUBLIC AGENCY

The Contractor has agreed to extend contract usage to other public agencies, such as any city or political Subdivision of this state, including, but not limited to counties; school districts; highway districts; port authorities; instrumentalities of counties, cities or any political subdivision created under the laws of the State of Idaho; any agency of the state government; or any city or political subdivision of another state.

1-16 PROPOSAL CONFIDENTIALITY

Each Proposer agrees that the contents of each proposal submitted in response to this RFP is confidential, proprietary, and constitutes trade secret information as to all technical and financial data, and waives any right of access to such proposals, except as provided for by law. Except as determined by the University's Office of Purchasing Services, in its sole discretion, no information will be given regarding any proposals or evaluation progress until after an award is made, except as provided by law.

1-17 RECORD OF PURCHASES

Contractor will provide Purchasing Services a detailed usage report of items/services ordered, quantities, and pricing under this Agreement upon request.

1-18 APPEAL OF AWARD

BAHR – SECTION II

A Proposer aggrieved by the award of an Agreement may file an appeal by writing to the Manager of Purchasing Services. The appeal must be received by the Manager of Purchasing Services within five working

days after the award is made, must describe the basis for the appeal, and must include all argument and evidence the Proposer wishes the Manager of Purchasing Services to consider. Keeping track of the date an award is made is the responsibility of the Proposer(s).

SECTION 2 - INSTRUCTIONS FOR PREPARING PROPOSALS

2-1 <u>GENERAL</u>

To aid in the evaluation process, it is required that all responses comply with the items and sequence as presented in paragraph 2-2, RFP Response Outline. Paragraph 2-2 outlines the minimum requirements and packaging for the preparation and presentation of a response. Failure to comply may result in rejection of the response. The proposal should be specific and complete in every detail, prepared in a simple and straightforward manner.

Proposers are expected to examine the entire Request for Proposals, including all specifications, standard provisions, and instructions. Failure to do so will be at the Proposer's risk. Each Proposer shall furnish the information required by the invitation. It is required that proposal entries be typewritten. Periods of time, stated in number of days, in this request or in the Proposer's response, shall be in calendar days. Propose your best price on each item.

2-2 <u>RFP RESPONSE OUTLINE</u>

- A. <u>Response Sheet</u>: The proposal Response Certification (page 5) shall be attached to the front of the proposal and shall contain the Proposer's certification of the submission. An official who has full authority to enter into an Agreement shall sign it.
- B. <u>Background and History</u>: Describe the company, organization, officers or partners, number of employees, and operating policies that would affect this Agreement. State the number of years your organization has been continuously engaged in business.
- C. <u>References</u>: The Proposer shall provide a minimum of three (3) references including names of persons who may be contacted, title of person, addresses, phone numbers, and e-mail, where products or services similar in scope to the requirements of this RFP have been provided.
- D. <u>Experience and Support</u>: Describe Proposer's experience in performing the requested services.
- E. <u>Costs</u>: Include itemized costs for all components and features to be delivered. Costs should be identified as one-time or continuing. Purchase prices, lease prices, installation charges, and maintenance charges must be identified. All equipment prices must be stated as FOB: Moscow, ID.
- F. <u>Warranties</u>: Describe warranties provided by the Proposer. Include discussions of any additional support provided after the sale.
- G. <u>Proposer Exceptions</u>: Describe any exceptions to the terms and conditions contained within this document.

BAHR – SECTION II

SECTION 3 - TECHNICAL SPECIFICATIONS & BID FORM

This Request for Proposal ("RFP") represents an opportunity for each respondent to provide examples of its management, marketing and operational strategies, innovation and creativity, and understanding of the University of Idaho communities and objectives.

Proposals should define the market size and potential market capture, identify specific sales and revenue objectives of the University, and include assumptions, expectations and commitments required to obtain these financial and operational goals. Each proposal should also include management, service and reporting commitments, together with specific suggestions regarding communication, planning and performance review. The expectations, rights, and responsibilities of both the University and the supplier will be identified and reviewed at the outset of the contract.

Proposals should create a comprehensive, integrated beverage program that will increase customer satisfaction by providing quality products, easier access, state-of-the-art equipment with top selections, technology, and improved distribution and locations throughout the campus. It should target aggressive sales and profitability goals for achieving substantial growth in beverage sales campus-wide. As importantly, the University is interested in collaborative and innovative strategies and programs which support and advance its broad objectives. The University specifically solicits your creative ideas and recommendations.

The University anticipates a contract that provides 90/10 rights for the selected beverage supplier to sell and promote the sale of beverages throughout the University, and food operations. Beverages may include, without limitation, carbonated and non-carbonated artificially flavored drinks, packaged waters, fruit and/or vegetable juices, fruit and/or juice containing drinks, coffee and tea products, and drink and beverage bases in the form of syrups, whether powders, crystals, concentrates or otherwise, from which such drinks and beverages can be prepared.

SECTION 4 - PROCUREMENT PROCESS

4-1 PROPOSER LIST AND QUALIFICATION EVALUATION

After the established date for receipt of proposals, a listing of Proposers submitting proposals will be prepared, and will be available for public inspection.

Qualifications and proposals submitted by interested Proposers will be reviewed and evaluated based on the evaluation factors set forth in the RFP.

4-2 PROPOSAL CLASSIFICATION

For the purpose of conducting discussions with individual offerers, if required, proposals will initially be classified as:

- A. Potentially Acceptable
- B. Unacceptable

Discussions may be conducted with any or all of the Proposers whose proposals are found acceptable or potentially acceptable. Offerers whose proposals are unacceptable will be notified promptly. The Manager of Purchasing will establish procedures and schedules for conducting oral and/or written discussions.

Proposers are advised that the University may award an Agreement on the basis of initial offers received, without discussions; therefore, each initial offer should contain the offerer's best terms from a cost and technical standpoint.

4-3 PROPOSER INVESTIGATION

The University will make such investigations as it considers necessary to obtain full information on the Proposers selected for discussions, and each Proposer shall cooperate fully in such investigations.

4-4 FINAL OFFERS AND AWARD OF AGREEMENT

Following any discussions with Proposers regarding their technical proposals, alternative approaches, or optional features, a number of the firms may be requested to submit best and final offers. The committee will rank the final Proposers for the project, giving due consideration to the established evaluation criteria. The committee will propose award to the proposal which is found to be most advantageous to the University, based on the factors set forth in the Request for Proposals.

The University reserves the right to reject any or all proposals, or portions thereof. The selection of a successful Proposer, if any, will be made based upon which proposal the University determines would best meet its requirements and needs.

5-1 EVALUATION CRITERIA

The evaluation criteria are listed below:

- Beverage Operations and Management Plan

Innovative and creative management, marketing and operational strategies and programs

- Contract Administration, Communication and Reporting

Process to assure continual, optimal performance Accurate and user-friendly reporting tools and controls Highly flexible and inclusive contract administration procedures and operations management

- Financial Structure and Total Economic Value

Non-volume related initiative commitments Volume related commitments Price support for products purchased from supplier Performance incentives Commissions

- Team Experience, Commitment and References

Qualifications of respondent and personnel committed to the contract Past experience in maximizing similar beverage-related opportunities Resources available to assure meeting an aggressive transition and implementation schedule

SECTION 6 - GENERAL CONTRACTUAL TERMS AND CONDITIONS

6-1 AGREEMENT TERMS AND CONDITIONS

The submission of a proposal herein constitutes the agreement of any Proposer that any Agreement to be drawn as the result of an award herein shall be prepared by the University and shall include at a minimum, all terms and conditions set forth in this RFP. The submission of a proposal shall further constitute the agreement of each Proposer that it will not insist on the use of standard contract agreements, documents, or forms, and that it waives any demand for the use of its standard agreements.

6-2 ASSIGNMENT

No assignment of this Agreement or of any right accruing under this Agreement shall be made, in part or in whole, by Contractor without the written consent of the University. Notwithstanding any assignment, Contractor shall remain fully liable on this Agreement and shall not be released from performing any of the terms, covenants, and conditions of this Agreement.

6-3 TERMINATION FOR CONVENIENCE

The University may terminate this Agreement, in whole or in part, at any time by written notice to the Contractor. The Contractor shall be paid its reasonable costs, including reasonable close-out costs and a reasonable profit on work performed up to the time of termination. The Contractor shall promptly submit its termination claim for payment. If the Contractor has any property in its possession belonging to the University, the Contractor will account for the same and dispose of it in the manner the University directs.

6-4 TERMINATION FOR DEFAULT

If the Contractor does not deliver the materials in accordance with the Contract delivery schedule, or if the Contract is for services and the Contractor fails to perform in the manner called for in the Contract, or if the Contractor fails to comply with any other provisions of the Contract, the University may terminate this Contract for default. Termination shall be effected by serving on the Contractor a notice of termination setting forth the manner in which the Contractor is in default. The Contractor will be paid a reasonable price for materials delivered and accepted, or services performed in accordance with the manner of performance set forth in the Contract.

6-5 APPLICABLE LAW AND FORUM

This Agreement shall be construed in accordance with, and governed by the laws of the State of Idaho. Any legal proceeding related to this Agreement shall be instituted in the courts of the county of Latah, state of Idaho, and Contractor agrees to submit to the jurisdiction of such courts.

6-6 LAWS, REGULATIONS AND PERMITS

The Contractor shall give all notices required by law and comply with all applicable Federal, State, and local laws, ordinances, rules and regulations relating to the conduct of the work. The Contractor shall be liable for all violations of the law in connection with work furnished by the Contractor, including the Contractor's subcontractors.

6-7 <u>GENERAL QUALITY</u>

All of the Contractor's work shall be performed with the highest degree of skill and completed in accordance with the Agreement Documents.

6-8 PROOF OF COMPLIANCE WITH AGREEMENT

In order that the University may determine whether the Contractor has complied with the requirements of the Agreement Documents, the Contractor shall, at any time when requested, submit to the University properly authenticated documents or other satisfactory proofs as to compliance with such requirements.

6-9 PAYMENT AND ACCEPTANCE

Except as otherwise provided herein, payments shall be due and payable within (30) days after acceptance of such goods or services or after receipt of properly completed invoice, whichever is later. No advance payment shall be made for goods or services furnished pursuant to this Agreement.

6-10 CONTINUATION DURING DISPUTES

The Contractor agrees that notwithstanding the existence of any dispute between the parties, insofar as possible under the terms of the Agreement to be entered into, each party will continue to perform the obligations required of it during the continuation of any such dispute, unless enjoined or prohibited by any court.

6-11 <u>SEVERABILITY</u>

If any term or condition of this Agreement or the application thereof to any person(s) or circumstances is held invalid, such invalidity shall not affect other terms, conditions or applications which can be given effect without the invalid term, condition or application; to this end the terms and conditions of this Agreement are declared severable.

6-12 INTEGRATION

This Agreement constitutes the entire Agreement between the parties. No change thereto shall be valid unless in writing communicated in the stipulated manner, and signed by the University and the Contractor.

6-13 BINDING EFFECT

This Agreement is for the benefit only of the parties hereto and shall inure to the benefit of and bind the parties hereto and their respective heirs, legal representatives, successors, and assigns.

6-14 APPROPRIATIONS CLAUSE

If the term of this Agreement is longer than one year, the University's obligations and liabilities hereunder are subject to the appropriation of funds from the State of Idaho, which appropriation shall be in the State of Idaho's sole discretion, from revenues legally available to the University for the ensuing fiscal year for the purposes of this Agreement. If the State of Idaho does not appropriate the funds for the purpose of this Agreement, the Agreement shall terminate and neither party shall have any further obligations hereunder.

6-15 IRS SECTION 501(C)(3) AND SECTION 115 CONSIDERATIONS

If any provision of this Agreement may cause the University to lose its status as an Internal Revenue Code Section 501(c)(3) corporation, this Agreement shall be voidable. In the alternative, at the sole option of the University, the offending provision(s) shall be modifiable such that the provision(s) will no longer cause the University to lose its status as a 501(c)(3) corporation. The terms of the modification shall be subject to agreement in writing by all parties.

6-16 COMPLIANCE WITH GOVERNOR'S EXECUTIVE ORDER

BAHR - SECTION II

In the event any provision of this Agreement shall cause the University to be in violation of any of the Governor of Idaho's Executive Orders, then this Agreement shall be voidable at the sole option of the University.

6-17 NON-USE OF NAMES AND TRADEMARKS

Contractor shall not use the name, trade name, trademark, or other designation of the University, or any contraction, abbreviation, or simulation any of the foregoing, in any advertisement or for any commercial or promotional purpose (other than in performing under this Agreement) without the University's prior written consent in each case.

6-18 CONTRACTOR REPRESENTATIONS

Contractor represents and warrants the following: (a) that it is financially solvent, able to pay its debts as they mature, and possessed of sufficient working capital to provide the equipment and goods, complete the services, and perform its obligations required hereunder; (b) that it is able to furnish any of the plant, tools, materials, supplies, equipment, and labor required to complete the services required hereunder and perform all of its obligations hereunder and has sufficient experience and competence to do so; (c) that it is authorized to do business in Idaho, properly licensed by all necessary governmental and public and quasi-public authorities having jurisdiction over it and the services, equipment, and goods required hereunder, and has or will obtain all licenses and permits required by law; and (d) that it has visited the site of the project and familiarized itself with the local conditions under which this Agreement is to be performed.

6-19 REGENTS' APPROVAL

This Agreement may be subject to approval by the Regents of the University of Idaho, and if it is and if such approval is not granted this Agreement shall be void and neither party shall have any further obligations or liabilities hereunder.

6-20 SURVIVAL OF TERMS

The terms and provisions hereof, and all documents being executed hereunder, if any, including, without limitation, the representations and warranties, shall survive this Agreement and shall remain in full force and effect thereafter.

6-21 HEADINGS

The headings contained in this Agreement are for reference purposes only and shall not in any way affect the meaning or interpretation hereof.

6-22 ADDITIONAL ACTS

Except as otherwise provided herein, in addition to the acts and deeds recited herein and contemplated to be performed, executed and/or delivered by the parties, the parties hereby agree to perform, execute and/or deliver or cause to be performed, executed and/or delivered any and all such further acts, deeds and assurances as any party hereto may reasonably require to consummate the transaction contemplated hereunder.

6-23 TIME OF ESSENCE

All times provided for in this Agreement, or in any other document executed hereunder, for the performance of any act will be strictly construed, time being of the essence.

6-24 WAIVER

No covenant, term or condition or the breach thereof shall be deemed waived, except by written consent of the party against whom the waiver is claimed, and any waiver of the breach of any covenant, term or condition shall not be deemed to be a waiver of any other covenant, term or condition herein. Acceptance by a party of any performance

BAHR - SECTION II

by another party after the time the same shall have become due shall not constitute a waiver by the first party of the breach or default of any such covenant, term or condition unless otherwise expressly agreed to by the first party in writing.

6-25 FORCE MAJEURE

Any prevention, delay or stoppage due to strikes, lockouts, labor disputes, acts of God, inability to obtain labor or materials or reasonable substitutes therefore, governmental restrictions, governmental regulations, governmental controls, enemy or hostile governmental action, civil commotion, fire or other casualty, and other causes beyond the reasonable control of the party obligated to perform (except for financial ability), shall excuse the performance, except for the payment of money, by such party for a period equal to any such prevention, delay or stoppage.

6-26 NO JOINT VENTURE

Nothing contained in this Agreement shall be construed as creating a joint venture, partnership, or agency relationship between the parties.

6-27 INFORMATION TRUE AND CORRECT

All documents, agreements and other information provided to the University by Contractor or which Contractor has caused to be provided to the University are true and correct in all respects and do not omit to state any material fact or condition required to be stated, necessary to make the statement or information not misleading, and there are no other agreements or conditions with respect thereto.

6-28 EQUAL OPPORTUNITY

Contractor represents and agrees that it will not discriminate in the performance of this Agreement or in any matter directly or indirectly related to this Agreement on the basis of race, sex, color, religion, national origin, disability, ancestry, or status as a Vietnam veteran. This non-discrimination requirement includes, but is not limited to, any matter directly or indirectly related to employment. Breach of this covenant may be regarded as a material breach of Agreement.

6-29 <u>CONFIDENTIALITY</u>

The parties hereto agree that the terms and conditions of this Agreement shall be held in confidence except as required by or for applicable disclosure laws, financing sources, enforcement of the Agreement, mergers and acquisitions, or as otherwise mutually agreed by the Parties, and such agreement shall not be withheld unreasonably.

6-30 UNIVERSITY'S RULES, REGULATIONS, AND INSTRUCTIONS

Contractor will follow and comply with all rules and regulations of the University and the reasonable instructions of University personnel. The University reserves the right to require the removal of any worker it deems unsatisfactory for any reason.

SECTION 7 - INDEMNITY, RISKS OF LOSS, INSURANCE

7-1 RISK OF LOSS

Until all improvements, equipment, or goods to be provided under this Agreement are installed on property owned or controlled by University and working properly, Contractor and its subcontractors of any tier shall bear all risks of all loss or damage to the improvements, equipment, or goods, excluding loss or damage caused by acts, omissions, or negligence of the University. Once all improvements, equipment, or goods to be provided under this Agreement are installed on property owned or controlled by University and working properly, the risk of all loss or damage shall be borne by University, excluding loss or damage caused by acts, omissions, or negligence of the Contractor. Contractors shall require its subcontractors of any tier to bear the same risk of loss and .

7-2 INDEMNIFICATION

Contractor shall indemnify, defend and hold the University and the State of Idaho harmless from and against any and all claims, losses, damages, injuries, liabilities and all costs, including attorneys fees, court costs and expenses and liabilities incurred in or from any such claim, arising from any breach or default in the performance of any obligation on Contractor's part to be performed under the terms of this Agreement, or arising from any act, negligence or the failure to act of Contractor, or any of its agents, subcontractors, employees, invitees or guests. Contractor, upon notice from the University, shall defend the University at Contractor's expense by counsel reasonably satisfactory to the University. Contractor, as a material part of the consideration of the University, hereby waives all claims in respect thereof against the University.

Contractor shall: (a) notify the University in writing as soon as practicable after notice of an injury or a claim is received; (b) cooperate completely with the University and/or the University's insurers in the defense of such injury or claim; and (c) take no steps such as admission of liability which would prejudice the defense or otherwise prevent the University from protecting the University's interests.

7-3 Insurance

7.3.1 General Requirements

7.3.1.1 Contractor and its subcontractor(s) of any tier are required to carry the types and limits of insurance shown in this insurance clause, section 8.0, and to provide University with a Certificate of Insurance ("certificate"). All certificates shall be coordinated by the Contractor and provided to the University within seven (7) days of the signing of the contract by the Contractor. Certificates shall be executed by a duly authorized representative of each insurer, showing compliance with the insurance requirements set forth below. All certificates shall provide for thirty (30) days' written notice to University prior to cancellation, non-renewal, or other material change of any insurance referred to therein as evidenced by return receipt of United States certified mail. Said certificates shall evidence compliance with all provisions of this section 8.0. Exhibit A of this Agreement contains a Request for Certificate of Insurance which shall be given to the insurance broker or agent of the Contractor and its subcontractor(s) of any tier, upon award of bid to Contractor.

7.3.1.2 Additionally and at its option, Institution may request certified copies of required policies and endorsements. Such copies shall be provided within (10) ten days of the Institution's request.

7.3.1.3 All insurance required hereunder shall be maintained in full force and effect with insurers with Best's rating of AV or better and be licensed and admitted in Idaho. All policies required shall be written as primary policies and not contributing to nor in excess of any coverage University may choose to maintain. Failure to maintain the required insurance may result in termination of this Agreement at University's option.

BAHR – SECTION II

TAB 8Page 28

7.3.1.4 All policies except Workers Compensation and Professional Liability shall name University as Additional Insured. The Additional Insured shall be stated as: "State of Idaho and The Regents of the University of Idaho". Certificate Holder shall read: "University of Idaho." Certificates shall be mailed to: University of Idaho, Risk Management, P.O. Box 443162, Moscow, ID 83844-3162.

7.3.1.5 Failure of University to demand such certificate or other evidence of full compliance with these insurance requirements or failure of Institution to identify a deficiency from evidence that is provided shall not be construed as a waiver of the obligation of Contractor and its subcontractor(s) of any tier to maintain such insurance.

7.3.1.6 No Representation of Coverage Adequacy. By requiring insurance herein, University does not represent that coverage and limits will necessarily be adequate to protect Contractor and its subcontractor(s) of any tier, and such coverage and limits shall not be deemed as a limitation on the liability of the Contractor and its subcontractor(s) of any tier under the indemnities granted to University in this Agreement.

8.1.7 Contractor is responsible for coordinating the reporting of claims and for the following: (a) notifying the Institution in writing as soon as practicable after notice of an injury or a claim is received; (b) cooperating completely with University in the defense of such injury or claim; and (c) taking no steps (such as admission of liability) which will prejudice the defense or otherwise prevent the University from protecting its interests.

7.3.2 Required Insurance Coverage.

Contractor and its subcontractor(s) of any tier shall at its own expense obtain and maintain:

7.3.2.1 Commercial General and Umbrella / Excess Liability Insurance. Contractor and its subcontractor(s) of any tier shall maintain Commercial General Liability ("CGL") written on an occurrence basis and with a limit of not less than \$1,000,000 each occurrence and in the aggregate. If such CGL insurance contains a general aggregate limit, it shall apply separately by location and shall not be less than \$1,000,000. CGL insurance shall be written on standard ISO occurrence form (or a substitute form providing equivalent coverage) and shall cover liability arising from premises, operations, independent contractors, products-completed operations, personal injury and advertising injury, and liability assumed under a contract including the tort liability of another assumed in a business contract. Waiver of subrogation language shall be included. If necessary to provide the required limits, the Commercial General Liability policy's limits may be layered with a Commercial Umbrella or Excess Liability policy.

7.3.2.2 Commercial Auto Insurance. Contractor and its subcontractor(s) of any tier shall maintain a Commercial Auto policy with a Combined Single Limit of not less than \$1,000,000; Underinsured and Uninsured Motorists limit of not less than \$1,000,000; Comprehensive; Collision; and a Medical Payments limit of not less than \$10,000. Coverage shall include Non-Owned and Hired Car coverage. Waiver of subrogation language shall be included.

7.3.2.3 Business Personal Property. Contractor and its subcontractor(s) of any tier shall purchase insurance to cover Business Personal Property of Contractor and its subcontractor(s) of any tier. In no event shall University be liable for any damage to or loss of personal property sustained by Contractor, even if such loss is caused by the negligence of Institution, its employees, officers or agents. Waiver of subrogation language shall be included.

7.3.2.4 Workers' Compensation. Contractor and its subcontractor(s) of any tier shall maintain all coverage statutorily required of the Contractor and its subcontractor(s) of any tier, and coverage shall be in accordance with the laws of Idaho. Contractor and its subcontractor(s) of any tier shall maintain Employer's Liability with limits of not less than \$100,000 / \$500,000 / \$100,000.

7.3.2.4 Professional Liability. If professional services are supplied to Institution, Contractor and its subcontractor(s) of any tier, Contractor and its subcontractor(s) of any tier shall maintain Professional

BAHR – SECTION II

TAB 8 Page 29

Liability (Errors & Omissions) insurance on a claims made basis, covering claims made during the policy period and reported within three years of the date of occurrence. Limits of liability shall be not less than one million dollars (\$1,000,000).
UNIVERSITY OF IDAHO GENERAL TERMS AND CONDITIONS

- 1. THIS ORDER EXPRESSLY LIMITS ACCEPTANCE TO THE TERMS AND CONDITIONS STATED HEREIN. ALL ADDITIONAL OR DIFFERENT TERMS PROPOSED BY CONTRACTOR ARE OBJECTED TO AND ARE HEREBY REJECTED, UNLESS OTHERWISE PROVIDED FOR IN WRITING BY THE PURCHASING MANAGER, UNIVERSITY OF IDAHO.
- 2. CHANGES: No alteration in any of the terms, conditions, delivery, price, quality, quantity or specifications of this order will be effective without the written consent of the University of Idaho Department of Purchasing Services.
- 3. PACKING: No charges will be allowed for special handling, packing, wrapping, bags, containers, etc., unless otherwise specified.
- 4. DELIVERY: For any exceptions to the delivery date as specified on the order, Contractor shall give prior notification and obtain approval thereto from the University of Idaho Department of Purchasing Services. With respect to delivery under this order, time is of the essence and order is subject to termination for failure to deliver within the timeframe specified in this order.
- 5. SHIPPING INSTRUCTIONS: Unless otherwise instructed, all goods are to be shipped prepaid and allowed, FOB Destination.
- 6. ORDER NUMBERS: Agreement order numbers or purchase order numbers shall be clearly shown on all acknowledgments, shipping labels, packing slips, invoices, and on all correspondence.
- 7. REJECTION: All goods, materials, or services purchased herein are subject to approval by the University of Idaho. Any rejection of goods, materials, or services resulting from nonconformity to the terms, conditions or specifications of this order, whether the goods are held by the University of Idaho or returned, will be at Contractor's risk and expense.
- 8. QUALITY STANDARDS: Brand names, models, and specifications referenced in herein are meant to establish a minimum standard of quality, performance, or use required by the University. No substitutions will be permitted without written authorization of the University of Idaho Department of Purchasing Services.
- 9. WARRANTIES: Contractor warrants that all products delivered under this order shall be new, unless otherwise specified, free from defects in material and workmanship, and shall be fit for the intended purpose. All products found defective shall be replaced by the Contractor upon notification by the University of Idaho. All costs of replacement, including shipping charges, are to be borne by the Contractor.
- 10. PAYMENT, CASH DISCOUNT: Invoices will not be processed for payment nor will the period of computation for cash discount commence until receipt of a properly completed invoice or invoiced items are received and accepted, whichever is later. If an adjustment in payment is necessary due to damage or dispute, the cash discount period shall commence on the date final approval for payment is authorized. Payment shall not be considered late if a check or warrant is available or mailed within the time specified.
- 11. LIENS, CLAIMS AND ENCUMBRANCES: Contractor warrants and represents that all the goods and materials delivered herein are free and clear of all liens, claims or encumbrances of any kind.
- 12. TERMINATION: In the event of a breach by Contractor of any of the provisions of this Agreement, the University of Idaho reserves the right to cancel and terminate this Agreement forthwith upon giving written notice to the Contractor. Contractor shall be liable for damages suffered by the University of Idaho resulting from Contractor's breach of Agreement.
- 13. TRADEMARKS: Contractor shall not use the name, trade name, trademark, or any other designation of the University, or any contraction, abbreviation, adaptation, or simulation of any of the foregoing, in any advertisement or for any commercial or promotional purpose (other than in performing under this Agreement) without the University's prior written consent in each case.

BAHR – SECTION II

- 14. OSHA REGULATIONS: Contractor guarantees all items, or services, meet or exceed those requirements and guidelines established by the Occupational Safety and Health Act.
- 15. TAXES: The University of Idaho is exempt from payment of Idaho State Sales and Use Tax. In addition, the University is generally exempt from payment of Federal Excise Tax under a permanent authority from the District Director of the Internal Revenue Service. Exemption certificates will be furnished as required upon written request by Contractor. If Contractor is required to pay any taxes incurred as a result of doing business with the University of Idaho, it shall be solely responsible for the payment of those taxes. If Contractor is performing public works construction, it shall be responsible for payment of all sales and use taxes.
- 16. BINDING EFFECT: This Agreement is for the benefit only of the parties hereto and shall inure to the benefit of and bind the parties and their respective heirs, legal representatives, successors and assigns.
- 17. ASSIGNMENTS: No Agreement, order, or any interest therein shall be transferred by Contractor to any other party without the approval in writing of the Purchasing Manager, University of Idaho. Transfer of an Agreement without approval may cause the recission of the transferred Agreement at the option of the University of Idaho.
- 18. WAIVER: No covenant, term or condition, or the breach thereof, shall be deemed waived, except by written consent of the party against whom the waiver is claimed, and any waiver of the breach of any covenant, term, or condition herein. Acceptance by a party of any performance by another party after the time the same shall have become due shall not constitute a waiver by the first party of the breach or default unless otherwise expressly agreed to in writing.
- 19. FORCE MAJEURE: Any prevention, delay or stoppage due to strikes, lockouts, labor disputes, acts of God, inability to obtain labor or materials or reasonable substitutes thereof, governmental restrictions, governmental regulations, governmental controls, enemy or hostile governmental action, civil commotion, fire or other casualty, and other causes beyond the reasonable control of the party obligated to perform (except for financial ability), shall excuse the performance by such party for a period equal to any such prevention, delay or stoppage.
- 20. NO JOINT VENTURE: Nothing contained in this Agreement shall be construed as creating a joint venture, partnership, or employment or agency relationship between the parties.
- 21. PRICE WARRANTY FOR COMMERCIAL ITEMS: Contractor warrants that prices charged to the University of Idaho are based on Contractor's current catalog or market prices of commercial items sold in substantial quantities to the general public and prices charged do not exceed those charged by Contractor to other customers purchasing the same item in like or comparable quantities.
- 22. NONDISCRIMINATION: Contractor represents and agrees that it will not discriminate in the performance of this Agreement or in any matter directly or indirectly related to this Agreement on the basis of race, sex, color, religion, national origin, disability, ancestry, or status as a Vietnam veteran. This non-discrimination requirement includes, but is not limited to, any matter directly or indirectly related to employment. Breach of this covenant may be regarded as a material breach of Agreement.
- 23. UNIVERSITY REGULATIONS: Contractor shall follow and comply with all rules and regulations of the University and the reasonable instructions of University personnel.
- 24. GOVERNING LAW: This Agreement shall be construed in accordance with, and governed by the laws of the State of Idaho. Any legal proceeding related to this Agreement shall be instituted in the courts of the county of Latah, state of Idaho, and Contractor agrees to submit to the jurisdiction of such courts.

UNIVERSITY OF IDAHO - REQUEST FOR PROPOSAL

Exhibit A – Request for Certificate of Insurance from Contractor* *If bid is awarded to Contractor Page 1 of 2

Give this form to your insurance agent / broker

Agents/ Brokers: RETURN A COPY OF THESE INSTRUCTIONS WITH YOUR CERTIFICATE.

Certificates without a copy of these instructions will not be accepted.

Contractor and its subcontractors of any tier ("Insured") are required to carry the types and limits of insurance shown in this Request, and to provide University of Idaho ("Certificate Holder") with a Certificate of Insurance within seven (7) days of the signing of this Contract.

• Certificate Holder shall read:

State of Idaho and the Regents of the University of Idaho Attn: Risk Management P.O. Box 443162 Moscow, ID 83844-3162

- Description area of certificate shall read: Attn: Contract for Services
- All certificates shall provide for thirty (30) days' written notice to Certificate Holder prior to cancellation or material change of any insurance referred to in the certificate.
- All insurers shall have a Best's rating of AV or better and be licensed and admitted in Idaho.
- All policies required shall be written as primary policies and not contributing to nor in excess of any coverage Certificate Holder may choose to maintain.
- All policies (except Workers Compensation and Professional Liability) shall name the following as Additional Insured: The Regents of the University of Idaho, a public corporation, state educational institution, and a body politic and corporate organized and existing under the Constitution and laws of the state of Idaho.
- Failure of Certificate Holder to demand a certificate or other evidence of full compliance with these insurance requirements or failure of Certificate Holder to identify a deficiency from evidence that is provided shall not be construed as a waiver of Insured's obligation to maintain such insurance.
- Failure to maintain the required insurance may result in termination of this grant or contract at the Certificate Holder's option.
- By requiring this insurance, Certificate Holder does not represent that coverage and limits will necessarily be adequate to protect Insured, and such coverage and limits shall not be deemed as a limitation on Insured's liability under the terms of the grant or contract.
- A copy of this certificate request must be sent with the Certificate. UNIVERSITY OF IDAHO - REQUEST FOR PROPOSAL

BAHR – SECTION II

Exhibit A – Request for Certificate of Insurance from Contractor* *If bid is awarded to Contractor Page 2 of 2

Required Insurance Coverage. Insured shall obtain insurance of the types and in the amounts described below.

- <u>Commercial General and Umbrella Liability Insurance</u>. Insured shall maintain commercial general liability (CGL) and, if necessary, commercial umbrella insurance with a limit of not less than \$1,000,000 each occurrence and in the aggregate. If such CGL insurance contains a general aggregate limit, it shall apply separately by location and shall not be less than \$1,000,000. CGL insurance shall be written on standard ISO occurrence form (or a substitute form providing equivalent coverage) and shall cover liability arising from premises, operations, independent contractors, products-completed operations, personal injury and advertising injury, and liability assumed under an insured contract including the tort liability of another assumed in a business contract. Waiver of subrogation language shall be included. If necessary to provide the required limits, the Commercial General Liability policy's limits may be layered with a Commercial Umbrella or Excess Liability policy.
- <u>Commercial Auto Insurance</u>. Insured shall maintain a Commercial Automobile Policy with a Combined Single Limit of not less than \$1,000,000; Underinsured and Uninsured Motorists limit of not less than \$1,000,000; Comprehensive; Collision; and a Medical Payments limit of not less than \$5,000. Coverage shall include Non-Owned and Hired Car coverage. Waiver of subrogation language shall be included.
- <u>Business Personal Property and/or Personal Property</u>. Insured shall purchase insurance to cover Insured's personal property. In no event shall Certificate Holder be liable for any damage to or loss of personal property sustained by Insured, whether or not insured, even if such loss is caused by the negligence of Certificate Holder, its employees, officers or agents.
- <u>Workers' Compensation</u>. Where required by law, Insured shall maintain all statutorily required Workers Compensation coverages. Coverage shall include Employer's Liability, at minimum limits of \$100,000 / \$500,000 / \$100,000.
- <u>Professional Liability</u>. If professional services are supplied to the Institution, Insured shall maintain Professional Liability (Errors & Omissions) insurance on a claims made basis, covering claims made during the policy period and reported within three years of the date of occurrence. Limits of liability shall be not less than one million dollars (\$1,000,000).

If you have additional questions, please contact:

Carry Salonen, Risk Specialist, University of Idaho. PH (208) 885-7177. FAX (208) 885-9490 csalonen@uidaho.edu



Purchasing Services 645 West Pullman Road Moscow, Idaho 83844-1202 208-885-6123 208-885-6060 juliam@uidaho.edu

Date: May 23, 2006

To: All Interested Bidders

From: Julia R. McIlroy, Senior Buyer

Subject: Request for Proposals No. 06-58M ~ Campus-Wide Pouring & Vending Rights

This letter will serve as Addendum Number One to the above referenced RFP. The following change will be made:

Proposals due date & time: June 14, 2006 @ 2:00 p.m. (local time)

All terms and conditions of the RFP remain the same.

Failure to acknowledge this addendum may result in rejection of your proposal. Acknowledgement should accompany your offer.

If you need additional information, please call (208) 885-6123, or e-mail juliam@uidaho.edu.

Thank you for your interest in the University of Idaho.

(Company)

(Signature)

THIS PAGE INTENTIONALLY LEFT BLANK

INSTITUTION / AGENCY AGENDA UNIVERSITY OF IDAHO

SUBJECT

Adoption of a resolution and authorization for re-conveyance of real property near Pierce, Idaho, back to the United States of America

APPLICABLE STATUTE, RULE, OR POLICY

Idaho State Board of Education Policies and Procedures, Section V.I.5.b.3.

BACKGROUND

In 1985 the United States of America granted 66.54 acres of unimproved forest land near Pierce, Idaho to the Board of Regents of the University of Idaho (UI). The property had been managed by the federal government as part of the Clearwater National Forest's Pierce Work Center, but was declared as surplus property and conveyed to the Regents at no cost. The conveyance restricted use of the property to continued use for educational purposes related to UI's anthropology programs. Per the deed, failure to comply with the restrictions could result in the University being obligated to pay market value for the subject property.

The property was of historic significance because of artifacts and land alterations related to the intensive methods of recovery of gold undertaken by Chinese miners during the local gold rush of the 1860's. The University has utilized the property for research and teaching opportunities in accordance with federal restrictions, and has made no improvements to the property.

DISCUSSION

UI programs related to the property have been of diminishing interest to students and faculty, and it has become more difficult for UI to demonstrate ongoing educational uses affiliated with the property. Consequently, the UI Department of Sociology, Anthropology, and Justice Studies has concluded that the property is no longer of benefit to its programs. Simultaneously, the United States Forest Service has contacted the University to inquire about the possibility of acquiring at least a portion of their formerly surplused property for purpose of constructing a storage building on this remote parcel for explosives related to wild land fire suppression.

The 1985 quitclaim deed and the federal program under which the property was granted to the Regents provides the option for UI to voluntarily return the surplus property if UI can no longer abide by the terms and conditions of the 1985 grant of real property. To comply with federal requirements for re-conveyance, the Board of Regents of the University of Idaho must adopt a resolution and send a letter to the Federal Real Property Assistance Program indicating a desire to voluntarily return the land back to the federal government. Upon receipt of the

INSTITUTION / AGENCY AGENDA UNIVERSITY OF IDAHO - continued

resolution and letter, the federal government will prepare a re-conveyance document in substantial consistency with the one attached to this agenda item for execution by the Board of Regents or its designee.

IMPACT

Adoption of the resolution and authorization of the re-conveyance of property will impose no costs on UI. The property has little remaining educational program value, and retention of the property will result in continued operation and maintenance costs, as well as costs for programs of diminishing academic interest. Failure to deliver programs consistent with the restrictions on the property could result in a determination of non-compliance with the federal grant and the payment of all or a portion of market value for the fee interest in this property that is no longer of programmatic value to the University.

STAFF COMMENTS AND RECOMMENDATIONS

Staff agrees with the statements and conclusions of the University regarding the university's need to re-convey this property to the Federal government due to a decline in educational use and the Federal government's need to reuse some of the property in question. The following documents are attached as follows:

٠	Resolution	Exhibit A
•	Deed of Voluntary Reconveyance	Exhibit B
•	US Department of Education Notice	Exhibit C
•	Letter Relinguishing Pierce Property	Exhibit D

Letter Reinquishing Pierce PropertyExhibit D
Location of Subject PropertyExhibit E

Staff recommends approval of the university request as noted in the following motion.

BOARD ACTION

A motion to adopt the attached resolution and authorize the Vice President for Finance and Administration at the University of Idaho to execute re-conveyance documents in substantial conformance to the form submitted to the Board.

Moved by _____ Seconded by _____ Carried Yes ____ No ____

REFERENCE: APPLICABLE STATUTE, RULE, OR POLICY

Idaho State Board of Education GOVERNING POLICIES AND PROCEDURES SECTION: V. FINANCIAL AFFAIRS I. Real and Personal Property Services

April 2002

5. Disposal of Real Property

a. Temporary Permits

Permits to make a temporary and limited use of real property under the control of an institution, school or agency may be issued by the institution, school or agency without prior Board approval.

- b. Board approval of other transfers
 - (1) Leases to use real property under the control of an institution, school or agency require prior Board approval - if the term of the lease exceeds five (5) years or if the lease revenue exceeds two hundred fifty thousand dollars (\$250,000).
 - (2) Easements to make a permanent use of real property under the control of an institution, school or agency require prior Board approval - unless easements are to public entities for utilities.
 - (3) The transfer by an institution, school or agency of any other interest in real property requires prior Board approval.

THIS PAGE INTENTIONALLY LEFT BLANK

RESOLUTION

The Board of Regents of the University of Idaho duly adopted the following resolution during a meeting held November 30, 2006.

WHEREAS, by Quitclaim Deed from the United States of America, acting by and through the Secretary of Education by the Acting Administrator for Management Services, to the Regents of the University of Idaho, dated September 11, 1985 and recorded as Instrument Number 142476 on October 28, 1985 in the records of the County of Clearwater (the "1985 Deed"), the United States conveyed unto the Regents of the University of Idaho a certain tract of land, containing 66.54 unimproved acres, designated as a portion of the former USFS Administrative Work Center in Pierce, Idaho (the "Property"); and

WHEREAS, the 1985 Deed specified that title to the Property is conveyed to the Regents of the University of Idaho subject to certain conditions subsequent (the "Conditions Subsequent"), grants the Regents of the University of Idaho the option to voluntarily return the Property if they can no longer abide by those terms and conditions; and if concurred to by the U.S. Department of Education and the General Services Administration; and

WHEREAS, the approval of the Board of Regents of the University of Idaho is required for the University of Idaho to exercise the voluntary return of the Property; and the Board of Regents of the University of Idaho hereby wish to exercise that option.

BE IT FURTHER RESOLVED that Lloyd E. Mues, Vice President of Finance and Administration for the University of Idaho, shall have the authority to execute those documents necessary to carry out the intent of this resolution.

BOARD OF REGENTS OF THE UNIVERSITY OF IDAHO

BY:__

Laird Stone President

ATTEST BY:

Milford Terrell Secretary

THIS PAGE INTENTIONALLY LEFT BLANK

DEED OF VOLUNTARY RECONVEYANCE

This AGREEMENT is made this ______ day of ______, 2006, between the BOARD OF REGENTS OF THE UNIVERSITY OF IDAHO, a body politic and corporate organized and existing under the constitution and laws of the State of Idaho("GRANTOR"), and the UNITED STATES OF AMERICA, pursuant to § 203(k) of the Federal Property and Administrative Services Act of 1949, as amended ("Act"), P. L. No. 81-152, 63 Stat. 377, 40 U.S.C. § 471 <u>et seq</u>., Reorganization Plan No. 1 of 1953, the Department of Education Organization Act of 1979, P. L. No. 96-88, 93 Stat. 668, 20 U.S.C. § 3401 <u>et seq</u>.

I. <u>RECITALS</u>

1. On September 11, 1985, the UNITED STATES OF AMERICA, acting by and through the Secretary of Education, acting by and through the Administrator for Management Services of the Department of Education, conveyed certain real property in Clearwater County, State of Idaho, formerly known as Pierce Work Center, near Pierce, Idaho, and consisting of approximately 66.54 acres of unimproved land ("Property") to GRANTOR, by Quitclaim Deed ("Deed"). The Deed was recorded on October 28, 1985 as Document 142476 in the Auditor and Ex-Officio Recorder's office in Clearwater County, Idaho. The property is more particularly described on Exhibit A of the Deed.

2. The conveyance was made expressly subject to certain covenants and conditions subsequent contained in the Deed, which the GRANTOR expressly assumed and agreed to observe and perform, including provisions that require that the Property be used for educational purposes in accordance with a program and plan ("Plan") set forth in the application of GRANTOR dated September 27, 1984, which had been approved by the Secretary of Education.

3. The Deed further provided that in the event of a breach of conditions set forth in the Deed, whether caused by the legal or other inability of the GRANTOR, its successors or assigns,

BAHR – SECTION II

TAB 9 Page 7

Exhibit B

to perform any of the obligations set forth, all right, title, and interest in and to the Property shall, at the option of the UNITED STATES OF AMERICA, revert to and become the property of the UNITED STATES OF AMERICA, which shall have an immediate right of entry thereon, and the GRANTOR, its successors or assigns, shall forfeit all right, title, and interest in and to the Property and in any and all of the tenements, hereditaments, and appurtenances thereunto belonging.

4. GRANTOR, determining that continued compliance with its Plan is no longer practical for achieving Grantor's educational mission, is desirous of reconveying the Property to the UNITED STATES OF AMERICA and was authorized to execute this Deed of Voluntary Reconveyance agreement by a resolution of the Board of Regents of the University of Idaho November 30, 2006.

II. <u>AGREEMENT</u>

5. In consideration of one dollar, the mutual agreements contained herein, other good and valuable consideration, the receipt of which is hereby acknowledged, and in performance of the agreements contained herein, GRANTOR grants and conveys the Property to the UNITED STATES OF AMERICA. GRANTOR warrants that it has not alienated, conveyed, or encumbered any interest in the Property, that on the date this instrument is executed and delivered, the above identified Property is and shall be free and clear from any claim, demand, lien or any other encumbrance whatsoever made, done or suffered by GRANTOR or any other person or entity claiming under it, and that it conveys all right, title, and interest to the Property to the UNITED STATES OF AMERICA.

6. GRANTOR warrants that the Property is in good order, condition, and repair, and is free from any waste.

7. GRANTOR warrants and agrees that it shall provide protection to and maintenance of the Property at all times until such time as the title is actually returned to and accepted by the UNITED STATES OF AMERICA. Such protection and maintenance shall conform to the standards prescribed by the General Services Administration in FMR 102-75.545 now in effect.

8. The UNITED STATES OF AMERICA, by its acceptance of this Deed of Voluntary Reconveyance agreement hereby releases GRANTOR from all of the obligations imposed by the Deed with respect to the Property.

IN WITNESS WHEREOF, the GRANTOR has caused this instrument to be executed in its behalf by its duly authorized officers and its seal affixed on this _____ day of _____, 2006.

IDAHO

BOARD OF REGENTS OF THE UNIVERSITY OF GRANTOR

By: _____ Lloyd E Mues Vice President for Finance and Administration University of Idaho

ACKNOWLEDGEMENT

LATAH COUNTY

STATE OF IDAHO

On this _____ day of _____, 2006, personally appeared before me, a Notary Public in and for the State of Idaho, Lloyd E Mues, who executed the foregoing instrument in his capacity as Vice President for Finance and Administration of the University of Idaho, who acknowledged the foregoing instrument to be his free act and deed in his said capacity and the free act and deed of the Board of Regents of the University of Idaho.

IN WITNESS WHEREOF, I have set my hand and seal on this _____ day of _____, 2006.

Notary Public

My Commission Expires: _____

K:\SHARED\FRPA\Barbara\FRPA\2006 Misc Ltrs\Jun\Centenary College Voluntary Reconveyance Deed.doc

December 5, 2006

Barbara L Shawyer Federal Real Property Assistance Program 400 Maryland Ave SW 2E117 Washington DC 20202-4553

Dear Ms Shawyer:

This letter is notification to the US Department of Education that the Board of Regents of the University of Idaho has decided to voluntarily return land conveyed to the University of Idaho on September 11, 1985 via Contract No. SA-X-ID-106. The US Forest Service has requested the property to be returned to achieve various management purposes.

Enclosed is a resolution of the University of Idaho's governing board which is the legal entity currently holding title to the subject property. It is my understanding that the Department of Education will prepare the necessary conveyance documents in accordance with the intent of the executed resolution. Please do not hesitate to contact Gerard Billington, University of Idaho Real Estate Officer, PO Box 443162, Moscow ID 83844-3162 if you have any questions or need additional information. Should you wish to utilize e-mail, you may contact Gerard at gerardb@uidaho.edu.

Sincerely,

Lloyd E Mues Vice President, Finance & Administration

Enclosure

C: Roberta Morin, Realty Specialist, Clearwater National Forest 12730 Hwy 12, Orofino ID 83544



College of Letters, Arts and Social Sciences P.O. Box 443154 Moscow, Idaho 83844-3154 208-885-6426 Fax 208-885-8964

October 11, 2006

MEMORANDUM

TO:	Lloyd Mues, Financial Vice President University of Idaho
FROM:	Katherine G. Aiken, Dean College of Letters, Arts & Social Sciences

RE: Relinquishing Pierce Property

In 1985, the University of Idaho accepted property near Pierce, Idaho exclusively for specific uses related to anthropology. The United States Department of Education's original grant of the property assigned programmatic responsibilities to the University of Idaho Department of Sociology, Anthropology, and Justice Studies. SAJS would like to be freed from these responsibilities and at the same time the Clearwater National Forest (the original owner) wants to use a portion of the property for construction of an explosives storage building.

Originally, it was intended that the property in question support several multiyear archaeological excavations. However, the site artifact density turned out to be quite low, making it impractical to implement these excavations. Sociology, Anthropology and Justice Studies has tried over the past few years to create a new program that would utilize the property without impacting adversely the archaeological materials that remain. This has not been a success.

As a result, the Pierce property is no longer of any significant programmatic interest to Sociology, Anthropology and Justice Studies. If faculty members and/or students desire to work on the site in the future, they would be able to gain access through the United States Forest Service Special Use Permit program. Consequently, returning this property will not harm the research interests of SAJS. Since the USFS is required to protect the archaeological site under federal legislation, the University of Idaho decision to give up the property will not result in an abrogation of responsibility to protect the site.

BAHR - SECTION II

TAB 9 Page 12



THIS PAGE INTENTIONALLY LEFT BLANK

INSTITUTION / AGENCY AGENDA LEWIS-CLARK STATE COLLEGE

SUBJECT

Lewis-Clark State College requests approval to enter into a management agreement with College Town Development Idaho, LLC, to manage a residence hall ("Clearwater Hall") financed by the developers.

APPLICABLE STATUTE, RULE, OR POLICY

Idaho State Board of Education Governing Policies & Procedures, Sections V.I.1. through V.I.3.

REFERENCE

October 2004	LCSC informed Board of shortage of residence hall space.
December 2004	Board asked for a needs presentation and competitive RFP.
January 2005	Board asked for review of private enterprise building any
	necessary residence halls and/or advantages of self-
	financing without a lease consideration.
March 2005	Board approved the sale of tax-exempt bonds to fund the
	construction of a residence hall. However, LCSC promised
	to provide adequate time for private industry to intervene
	prior to a financial commitment by the College.
October 2005	Board approved the sale of two lots to provide land for
	private development of a residence hall.
June 2006	Board approved the management agreement for the first of
	the two privately-developed residence halls ("College
	Place").

BACKGROUND

Lewis-Clark State College pursued private/public partnerships as a means of providing needed housing for LCSC students. During its January 2005 meeting, the Board had asked LCSC to pursue the possibility of allowing student housing needs to be met through the private sector renting directly to students without a financial obligation to the College for construction. In addition, the Board asked that LCSC pursue other funding alternatives rather than a lease arrangement guaranteeing a return to a private developer.

During the March 2005 meeting, the Board approved the issuance of a taxexempt bond to fund a new residence hall with the promise from LCSC it would pursue the possibility of encouraging private enterprise to construct a facility and rent directly to the students. The goal was to have new and affordable housing by Fall Term 2006.

In October 2005, the Board approved the sale of two lots owned by the College, adjacent to campus, that were needed by the developer to facilitate construction of the first of the two residence hall projects ("College Place"). A site for a second new residence hall ("Clearwater Hall") was obtained in downtown Lewiston by a second private developer (College Town Development Idaho, LLC).

Both College Place and Clearwater Hall were opened for occupancy by LCSC students at the beginning of the Fall 2006 semester and are now in operation. This agenda item seeks Board approval of the management agreement for the more recently-completed facility—Clearwater Hall.

DISCUSSION

Lewis-Clark State College has long needed additional student housing. New residence halls will help provide a safe and functional living environment for students and will help the College attain its goal of improved recruitment and retention of non-commuting students.

New student housing will provide the College with more flexibility for accommodating current students while older/decrepit residence halls are taken off-line to provide facility/system upgrades to address serious safety and efficiency needs. The new residence halls have enabled the College to provide housing to approximately 60 students who would otherwise have been housed in a local commercial hotel (an "emergency" arrangement used for the past three years) due to space limitations in existing housing. The new, privately-developed residence halls also will allow the College to plan for renovation of a dilapidated older facility ("Talkington Hall")—to address life safety and energy conservation needs

During the process of requesting zoning changes to allow a multi-family structure to be built at the location identified, the Lewiston City Council strongly encouraged the developers to: a) work with the College to secure a management agreement, and b) construct projects with the intent of allowing only LCSC students as tenants. Subsequently, the attached draft agreement was negotiated for the Clearwater Hall facility. The essence and main points of the agreement are as follows:

1. LCSC ("Manager") will manage the facility for the Owner and will rent the rooms to students during the school year and, during the summer, to both students ("Students") and to others ("Program Participants") attending courses and other events conducted on the LCSC campus. The initial lease rate paid to Owner will be \$390 per month per resident. This minimum "Monthly Rate" will be adjusted annually by mutual agreement of the parties.

- 2. As is done with current Residence Hall students, rental agreements will be issued by LCSC in its own name. LCSC will collect rents in advance at the beginning of the semester or other rental period.
- 3. The initial term of the Management Agreement will be 10 years.
- 4. As an inducement for the Owners to construct the projects, LCSC will agree to fill the College Place Residence (adjacent to campus on 4th Street) and the Clearwater Hall Residence (downtown at 5th Street and Main) on a pro-rata basis until both facilities have achieved 100% occupancy before allowing Students and Program Participants to reside in any other LCSC-owned or managed residence facilities; provided, however, that LCSC shall have the right to house up to 78 student athletes at Clark Hall, irrespective of the occupancy in the College Place Residence or Clearwater Hall Residence.
- 5. Manager shall bear certain maintenance and operating costs, including residence hall staffing, janitorial support, maintenance of the grounds, and maintenance of the interior of the facility. LCSC shall also be responsible for marketing and renting the rooms as part of LCSC's Residence Life Program.
- 6. Owner shall be responsible for taxes; insurance on the facility; and the maintenance and repair of roof, windows, doors, exterior of the building, building envelope, heating and cooling, and HVAC systems. Manager will be responsible for arranging for Owner's repairs, but will pay for those repairs from an Owner's Reserve Account which will be funded by Owner.
- 7. Manager shall be entitled to receive a management fee based on the gross revenues generated by the Clearwater Hall Residence. This management fee will be advanced to LCSC on a monthly basis after rent for the month is received from LCSC, except that if LCSC fails to obtain the equivalent of 85% of total occupancy, computed as hereinafter provided for a month, Owner will be under no obligation to advance the management fee to LCSC for that month. The parties shall determine the amount of the management fee to which LCSC is entitled in the following manner:
 - (a) 100% to 95% of total occupancy 4%
 - (b) Less than 95% to 90% of total occupancy 3%
 - (c) Less than 90% to 85% of total occupancy 2%

Presentation to the Board of the Clearwater Hall management agreement was somewhat delayed while each party (and legal counsel) ensured that all operational procedures were clearly stipulated in the agreement. The proposed management agreement for Clearwater Hall closely mirrors the management agreement for College Place, approved by the Board in June 2006. Daily operations of the new residence hall have conformed to the letter and spirit of the proposed management agreement since the facility was occupied at the end of August 2006. As stated in the proposed agreement, the Owner ("College Town Development Idaho, LLC") understands that the terms of the agreement are contingent upon State Board of Education approval.

IMPACT

The agreement would provide funding for the management of the residence hall project and provide a structure to incorporate the privately developed and constructed project into the residence hall program of Lewis-Clark State College.

ATTACHMENTS

Attachment 1: College Town Development Idaho LLC – Management Agreement for Clearwater Hall

STAFF COMMENTS AND RECOMMENDATIONS

Staff has reviewed the accompanying management agreement for Clearwater Hall and has compared that agreement with the one for College Place. The differences include the following:

- Term: Clearwater 10 years; College Place 5 years with multiple 5-year renewal terms
- Clearwater has commercial space on ground floor; College Place does not. LCSC has no responsibility for commercial space, although LCSC facilities and security personnel do have access to alarm and other utilities when required.
- Occupancy Fee: Clearwater \$390/month, College Place \$375/month Clearwater includes daily Summer Rent split 50/50 with owner
- Calculation of Management Fee:
 - College Park: annual calculation covering Fall, Spring, and Summer terms Not achieving occupancy rates in one term affects annual fee
 - Clearwater: separate calculation for Fall, Spring, and Summer terms. Not achieving occupancy rates in one term does not affect calculation in other terms
- Vending revenue: Clearwater LCSC, College Place owner

Staff recommends approval.

BOARD ACTION

A motion to approve the request by Lewis-Clark State College to enter into a management agreement with College Town Development Idaho, LLC, to allow LCSC to administer and rent a privately developed and funded residence hall, and to authorize the Vice President for Administrative Services to sign the enclosed management agreement and represent the Idaho State Board of Education on documents related to this transaction.

Moved by _____ Seconded by _____ Carried Yes ___ No ____

THIS PAGE INTENTIONALLY LEFT BLANK

COLLEGE TOWN STUDENT HOUSING

MANAGEMENT AGREEMENT FOR PROPERTY AT 402-418 MAIN STREET LEWISTON, IDAHO

	Page
SECTION 1. DEVELOPMENT	1
SECTION 2. TERM OF AGREEMENT	3
SECTION 3. MANAGEMENT OF RESIDENCES	4
SECTION 4. MANAGER'S COMPENSATION	7
SECTION 5. LEASING REQUIREMENTS	8
SECTION 6. USE OF RESIDENCES	13
SECTION 7. MAINTENANCE AND REPAIR OF RESIDENCES	14
SECTION 8. INSURANCE	17
SECTION 9. COMPLIANCE WITH LAWS	18
SECTION 10. GENERAL LIABILITIES	21
SECTION 11. DEFAULTS OF OWNER AND MANAGER AND RIGHT TO CURE	21
SECTION 12. INSOLVENCY – RESERVED	25
SECTION 13. DAMAGE AND DESTRUCTION	25
SECTION 14. CONDEMNATION	26
SECTION 15. ASSIGNMENT AND SUBLETTING	27
SECTION 16. SIGNS	28
SECTION 17. MISCELLANEOUS PROVISIONS	29
SECTION 18. BUILDING NAMES	34
SECTION 19. AUTHORITY OF MANAGER	35
SECTION 20. FUTURE MODIFICATIONS	35
SECTION 21. TERMINATION FOR FAILURE OF APPROPRIATIO	NS35
SECTION 22. STATE BOARD APPROVAL	35
BAHR – SECTION II	TAB 10 Page 8

TABLE OF CONTENTS

EXHIBITS:

- Exhibit A: Legal Description of Project
- Exhibit B: Project Plans and Specifications
- Exhibit C: Owner FF&E
- Exhibit D: Reserved.
- Exhibit E: Form of Project Management Financial Report
- Exhibit F: Form of Annual Budget
- Exhibit G: Form of Residence Program Terms, Conditions, Rules, and Regulations
- Exhibit H: Form of Rental Agreement
- Exhibit I: Signage Criteria
- Exhibit J: Student Tenant Permitted FF&E

MANAGEMENT AGREEMENT

THIS MANAGEMENT AGREEMENT ("**Agreement**") is made as of the 23rd day of August 2006 ("**Effective Date**"), by and between COLLEGE TOWN DEVELOPMENT IDAHO LLC, a Washington limited liability company ("**Owner**"), and STATE OF IDAHO, ACTING BY AND THROUGH THE STATE OF IDAHO BOARD OF EDUCATION AS BOARD OF TRUSTEES FOR LEWIS-CLARK STATE COLLEGE ("**Manager**").

RECITALS

A. Owner is the owner of record of that certain real property described in <u>Exhibit A</u> attached hereto and incorporated herein by this reference ("**Property**"), and the building and improvements located thereon as set forth on <u>Exhibit B</u> attached hereto and incorporated herein by this reference ("**Building**"). Except for the Retail Spaces (as defined below), the Property and the Building together with any other improvements now or hereafter located on the Property is herein collectively referred to as the "**Project**."

B. Owner desires to develop the Project for the purposes of retaining Manager to utilize and manage the Project as part of Manager's Residence Program for Student Tenants (as those terms are defined herein) attending Lewis Clark State College (the "**College**"), and Manager desires to manage the Project on behalf of Owner for the purpose of providing additional residential living options for students attending the College.

C. Owner is willing to develop the Project at its sole cost and expense in a form and manner acceptable to Manager based on Manager's willingness to enter into this Agreement, and Manager is willing to manage the Project in accordance with the terms and conditions set forth herein.

NOW, THEREFORE, in consideration of the above recitals and mutual covenants contained in this Agreement and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, Owner and Manager agree as follows:

AGREEMENT

1. <u>Development</u>.

1.1 <u>Scope of Development</u>. Owner agrees, at its sole cost and expense, to redevelop the Project substantially in conformance with <u>Exhibit B</u> attached hereto and incorporated herein by this reference. For purposes of this Agreement, the Project shall include all areas of the Building exclusive of the Retail Spaces, including without limitation, all Residences (as defined below), hallways, lobby areas, storage areas, and mailrooms of the Building, and each all as set forth in more detail on <u>Exhibit B</u>. As used herein, "**Residences**" shall comprise those portions of the Project intended as primary accommodation and housing for Student Tenants, including each single bedroom (each, a "**Residence Unit**"), a "**Common Room**", which adjoins a pod of from one (1) to five (5) Residence Units, a lounge area, and a

BAHR – SECTION II

study lounge, all as set forth in <u>Exhibit B</u>. Each of the Residences, and Residence Units, shall be equipped with the Owner FF&E (as defined below) which is set forth in more detail in <u>Exhibit C</u>.

1.2 **Development Period**. As of the Effective Date, Owner shall undertake development of the Project substantially in accordance with Exhibit B such that the same shall be Ready for Management (as those terms are defined below) on or before August 23, 2006 ("Target Delivery Date"). Notwithstanding the foregoing, the Manager shall not be required to accept delivery of the Project prior to the Target Delivery Date. For purposes of this Agreement, "Ready for Management" shall mean that certain date when (a) Owner obtains and provides to Manager temporary or permanent certificate of occupancy for the Project, (b) the Project is otherwise ready for possession and the Residences are ready and available for inhabitation by Student Tenants; provided that Owner shall be entitled to complete certain punch list items, landscaping, and other required items of completion to the Project that do not preclude Owner from obtaining the foregoing occupancy certificate and otherwise do not prohibit Manager from performing its obligations under the Rental Agreements, including without limitation delivering the Residences to Student Tenants for access and possession as contemplated in this Agreement and the Rental Agreements, and (c) Owner is otherwise ready, willing and able to deliver the same to Manager in accordance with this Agreement. The date of August 23, 2006, or the date when the Project is Ready for Management, whichever is later, is sometimes hereinafter referred to as the "Commencement Date" or the "Delivery Date".

1.3 **Delivery Date Contingency**. Notwithstanding anything to the contrary in Section 1.2 above, in the event Owner is unable to make the Residences Ready for Management on or before the Target Delivery Date, Owner shall not be liable to Manager for damages, claims or liability, including any indirect or consequential damages, arising from or related to any such delay or failure by Owner to make the Residences Ready for Management, and the delivery thereof to Manager, as provided for in Section 1.2 above. In the event Owner is unable to comply with the Target Delivery Date, and such failure to comply is not the direct or indirect result of any Manager Delay (as defined below), Manager's sole and exclusive remedies shall be to require Owner to (i) assist Manager in procuring and contracting temporary housing for all Student Tenants with valid Rental Agreements (as those terms are defined below) reasonably acceptable to Owner and Manager; (ii) to reimburse Manager not more often than monthly for any costs directly related to the procurement and use of such temporary housing, but only to the extent that such costs exceed the then-applicable aggregate rent received by Manager pursuant to any and all valid Rental Agreements in effect for which such temporary housing has been procured; and (iii) abate all Rent Installments ("Rent Installments" means the periodic payments of Minimum Monthly Rent or Minimum Daily Rent and any Owner's Additional Rent or Pro-Rata Rent payable to Owner under this Agreement) from the Target Delivery Date until such time as the Residences are Ready for Management.

For purposes of this <u>Section 1.3</u>, a "**Manager Delay**" shall mean any delay caused by Manager in the performance of any of its obligations hereunder, including without limitation the action or inaction of Manager, its employees and agents, or the gross negligence or willful misconduct by the same.

2. <u>Term of Agreement</u>.

2.1 <u>Initial Term</u>. The initial term of this Agreement ("Initial Term") shall commence on August 23, 2006 ("Commencement Date"), and shall continue thereafter until the latter of the last day of the one-hundred-twentieth (120th) month thereafter or expiration of the last Rental Agreement (as defined below) validly entered into by Manager ("Expiration Date"). Notwithstanding the foregoing, the Management Responsibility (as defined below) shall not arise, and Manager shall not be obligated regarding the same, until that certain date on which Owner makes the Residences Ready for Management and delivers the same to Manager as contemplated herein ("Management Commencement Date").

2.2 <u>Right of First Offer</u>.

2.2.1 Right of First Offer. Owner shall grant Manager a right of first offer (the "ROFO") to purchase the Project; provided that (a) this Agreement shall be in full force and effect, and (b) there shall not exist any Event of Default or any facts which with the giving of notice or passage of time, or both, would constitute an Event of Default. If at any time during the Initial Term or any Extended Terms Owner intends to offer the Project for sale, Owner shall provide written notice of such intention to Manager (the "ROFO Notice"). For sixty (60) days after receipt of the ROFO Notice (the "ROFO Negotiation Period"), Owner and Manager shall negotiate in good faith regarding terms for a sale of the Project to Manager. During the ROFO Negotiation Period, Owner will not actively market the Project and Owner will not negotiate with any potential purchasers of the Project other than Manager. If Owner and Manager agree on terms on which Owner will sell the Project to Manager during the ROFO Negotiation Period, such terms will be documented in a purchase and sale agreement ("PSA") to be executed between Owner and Manager. If Owner and Manager do not execute a PSA during the ROFO Negotiation Period, or if Manager is in monetary default under this Agreement at the time of the ROFO Notice, then the ROFO granted hereunder shall lapse and Owner will be free to market the Project for sale. Notwithstanding the foregoing, the ROFO shall remain in full force and effect in the event Owner fails to market the Project or removes the Project from sale for any reason.

2.2.2 <u>**ROFO Personal**</u>. The ROFO, and the respective provisions set forth in this <u>Section 2.2</u>, are personal to, and for the exclusive benefit of, the Manager, and no Transferee other than a Permitted Transferee (as those terms are defined below) shall have any right, title or interest whatsoever in the ROFO granted to Manager under this <u>Section 2.2</u>.

2.2.3 <u>Affiliate and Other Permitted Transfers</u>. Notwithstanding anything to the contrary herein, the ROFO shall not apply to any purchase, merger, consolidation, joint venture or reorganization of Owner (an "Ownership Change"), or any transfer of the Project to any affiliate of Owner. For purposes of this <u>Section 2.2.3</u>, "Affiliate" shall mean any corporation, partnership, or other entity: (a) which owns or controls the majority of ownership interests of Owner, either directly or indirectly through other entities; (b) the majority of ownership interests is owned or controlled by Owner; (c) the majority of whose ownership interests is owned or controlled by an entity described in (a); or (d) which owns or controls a majority of the ownership interests of Owner.

BAHR - SECTION II

2.3 <u>Voluntary Termination</u>. Either party may terminate this Agreement without cause by providing the other party with written notice of termination in accordance with <u>Section 17</u> herein on or before March 1st of any year during the Initial Term or any Renewal Term ("Termination Notice"). Following proper delivery of a Termination Notice, this Agreement shall terminate on August 14th of the then applicable calendar year, and thereafter, each of the parties shall fully and unconditionally released from the terms and conditions of this Agreement from and after the date of termination subject to any terms and conditions herein that expressly survive expiration or sooner termination of this Agreement.

3. <u>Management of Residences</u>.

3.1 <u>Delegation of Management Responsibility</u>. Commencing on the Commencement Date, Owner hereby engages and appoints Manager as the sole and exclusive manager and operator of the Residences, and Manager hereby accepts such appointment from Owner, which such appointment shall empower and obligate Manager to exclusively supervise, direct and control management and operation of the Residences in accordance with the terms and conditions set forth in this Agreement ("Management Responsibility"). Notwithstanding the foregoing, Manager's Management Responsibility under this Agreement or otherwise shall not include, nor shall Manager have any rights or responsibilities for, the management and/or operation of any of the retail spaces located on the first floor of the Building in which the Project is located, but specifically excluding that portion of the first floor of the Building identified as Manager's Management Responsibility, all as set forth in more detail on Exhibit B ("Retail Spaces"). As used herein, the term "Project" specifically excludes the Retail Spaces.

3.2 <u>Scope of Management Responsibility</u>. Pursuant to the Management Responsibility, and subject to the terms and conditions set forth in this Agreement, Manager shall have the exclusive responsibility, duty and obligation to service, promote, operate and manage the Residences on a day-to-day basis in an efficient and economical manner at Manager's sole cost and expense ("Manager Expenses"). Without limiting the foregoing, and in addition to other requirements and subject to any other limitations as may otherwise be set forth in this Agreement, Manager shall perform and otherwise provide the following services, and incur at its sole cost any Manager Expenses related thereto or arising thereunder, as follows:</u>

(a) hire, employ, manage and at all times maintain adequate staffing necessary to undertake and perform the Management Responsibility; as well as determine and implement personnel policies and practices relating to the management and operation of the Project generally, terms and conditions of employment, recruiting, screening, selection, hiring, training, compensation, employee benefits, supervision, discipline, dismissal and replacement;

(b) unless otherwise provided for and set forth herein, establish all relevant prices, price schedules, rates, rate schedules, rents, lease charges and concession charges for the Project and the Residences specifically;

(c) negotiate and administer Rental Agreements with Student Tenants pursuant to the Residence License (as those terms are defined below);

BAHR – SECTION II

(d) negotiate and administer in its own name, or in the name and on behalf of Owner as and where applicable, any leases and licenses for all appropriate areas of the Project;

(e) provide marketing and sales services for the Project in conformance with College requirements and standards;

(f) obtain and keep in full force and effect in accordance with applicable law, in its own name, or in Owner's name as applicable, any and all license and permits to the extent the same is within the control of Manager;

(g) negotiate, enter into, and administer in its own name, or in the name and on behalf of Owner as and where applicable, any contracts, licenses and purchase orders for services, inventories, provisions, and supplies that are necessary for maintenance and operation of the Project, and to use the same exclusively in the management and operation of the Project;

(h) institute in its own name, or in the name of Owner as and where applicable, any and all legal actions or proceedings necessary for, or incident to, operation and maintenance of the Project; provided, however, the Manager is not required to institute or to participate as a party in litigation with Owner's contractors, subcontractors, architects, engineers or agents in any dispute arising from the original development or construction of the Project by Owner;

(i) maintain the facilities associated with the Project in good repair and condition as set forth in more detail pursuant to <u>Section 7.1</u> herein, including without limitation the Owner Maintenance Obligations and Owner FF&E subject to the provisions of <u>Section 7.2</u> below;

(j) collect all rent, security deposits, charges, fees, sums and other amounts due from Student Tenants of the Residences in accordance with this Agreement and the Rental Agreement;

(k) maintain a comprehensive system of management records, books and accounts which shall belong to Owner. Owner and any party designated by Owner shall have at all times access to such records, accounts and books, including without limitation all files, rent rolls, invoices, receipts, and other materials pertaining to the Residences and/or this Agreement, all of which Manager covenants to keep safe, available and separate from any records not relating to the Residences, Manager's Management Responsibility and/or this Agreement;

(1) provide Owner on or before the fifteenth (15^{th}) day of each month during the Initial Term and any Renewal Terms a report substantially in the form attached hereto as <u>Exhibit E</u> and incorporated herein by this reference;

BAHR – SECTION II

(m) prepare not later than thirty (30) days following the Commencement Date and not later than March 1^{st} of each year of the Initial Term and any Renewal Terms thereafter, an annual budget substantially in the form attached hereto as <u>Exhibit</u> <u>F</u> and incorporated herein by this reference;

(n) Deliver to Owner not later than August 1st, January 1st, and May 15th respectively in each Lease Year of the Initial Term and any Extended Terms an "**Occupancy Schedule**" for the upcoming Rental Term (as hereinafter defined). The Occupancy Schedule shall set forth the anticipated occupancy of the Project for the upcoming Rental Term, the identifying number, but not the name, of each Student Tenants or other occupant, room number of the Residence Unit occupied by such person, and the Rental Rate to be paid for each such Residence Unit. The Occupancy Schedule shall be updated monthly by the Monthly Occupancy Reports to be submitted under Section 5.4.4, provided that, during the Summer Rental Term (as hereinafter defined), such update shall be retrospective;

(o) <u>Reserved;</u>

(p) use reasonable care to provide, at all times, for the safety and security of the Project, including without limitation the Project, Student Tenants and their personal property, guests and invitees;

(q) perform such other tasks as are customary and usual in the operation of a residential living facility of a class and standing equal to the Project and as otherwise required to operate and maintain the Project as contemplated under this Agreement or as may be reasonably specified by Owner from time to time; and

(r) negotiate and administer, in its own name and on its own behalf, concession agreements for all appropriate areas of the Project. Manager shall be entitled to receive and retain all commissions and other compensation payable as a result of those concession agreements.

3.3 <u>Reimbursement of Expenses and other Payment to Manager</u>. All Owner Expenses (as defined below) properly and validly incurred by Manager in accordance with the terms and conditions of this Agreement shall be for the account of, on behalf of, and at the expense of Owner. For administrative and economic efficiency, the parties agree that Owner Expenses will be accumulated by Manager until the total of those unpaid Owner Expenses equals at least One Hundred Dollars (\$100) before Manager submits an invoice for those accumulated Owner Expenses to Owner. For the same reasons, Manager agrees that it will not invoice Owner for minor repair costs which are an Owner's Expense where materials used have a value of less than Twenty Five Dollars (\$25) per job and are taken from Manager's existing inventory of lowcost materials and where any work is performed by Manager's salaried employees during their regular work day. Owner shall reimburse any Owner Expenses within thirty (30) days following submission of an invoice to Owner for such expenditure by Manager. Any invoice not paid within such thirty (30) day period shall bear interest from the date of the invoice at the Agreed Rate (as hereinafter defined) until paid. Other amounts payable by Owner to Manager under this Agreement (including Management Fees as hereinafter defined) shall bear interest from the due date of such payment at the Agreed Rate.

3.4 <u>Conditions to Management Responsibility</u>. Unless otherwise expressly provided in this Agreement, Manager shall be temporarily excused from its obligations under this Agreement:

(a) to the extent and whenever Manager shall be prevented from compliance with such obligation(s) by events of Force Majeure (as defined below); or

(b) to the extent that such obligations cannot reasonably be performed a result of any breach of any representation, warranty or covenant contained in this Agreement or default hereunder by Owner.

4. <u>Manager's Compensation</u>.

4.1 Management Fees. For purposes of this Section 4.1 specifically, and the Agreement generally, the following terms shall as follows: (a) the fall semester during which Student Tenants are in occupancy under fall semester Rental Agreements (as defined below) beginning on approximately August 20th and ending approximately December 31st, shall hereinafter be referred to as the "Fall Rental Term", (b) the spring semester during which Student Tenants are in occupancy under spring semester Rental Agreements beginning approximately January 1st, and ending approximately May 19th, shall hereinafter be referred to as the "Spring Rental Term," and (c) the period between the Spring Rental Term and the Fall Rental Term beginning May 20th and ending August 19th shall hereinafter be referred to as the "Summer Rental Term," each of which periods is hereafter sometimes referred to as a "Rental Term" or collectively as "Rental Terms", and in the aggregate which are collectively referred to as a "Lease Year"). During the Initial Term and any Renewal Terms, Owner shall pay Manager a management fee based upon the Rent Installments remitted to Owner under the terms of this Agreement. Within ten (10) days after the end of each Rental Term, Owner shall advance to Manager a certain percentage of the Rent Installments collected and paid to Owner during such Rental Term as a management fee ("Management Fee") based on the following methodology:

(a) Determine the gross Rent Installments paid to Owner during the preceding Rental Term.

(b) Determine the gross Rent Installments which would have been generated for the entire preceding Rental Term if Student Tenants had occupied one hundred percent (100%) of the Residence Units (excluding only one Residence Unit in the RD Apartment) for the entire preceding Rental Term and had paid Rent Installments for such entire Rental Term at the Minimum Monthly Rental Rate (as defined below) in effect for such Lease Year.

(c) Divided the amount computed under Item (a) by the amount computed under Item (b) to determine the actual percentage of total occupancy for the preceding

BAHR – SECTION II
Rental Term. If the resulting percentage is less than eighty-five percent (85%), Manager shall not be entitled to receive any Management Fee for the preceding Rental Term.

(d) If the percentage computed under Item (c) equals or exceeds eighty-five percent (85%), then the Management Fee shall be calculated as follows:

(i) Four percent (4%) of Rent Installments paid to Owner if the percentage is ninety-five percent (95%) or higher.

(ii) Three percent (3%) of the Rent Installments paid to Owner if the percentage is less than ninety-five percent (95%) but greater than or equal to ninety percent (90%).

(iii) Two percent (2%) of the Rent Installments paid to owner if the percentage is less than ninety percent (90%) but is greater than or equal to eighty-five percent (85%).

(e) For the purpose of computing the Management Fee for the Summer Rental Term, the entire Rent Installment to be remitted to the Owner on or before May 5 for the month of May will be considered part of the Rent Installments paid to Owner for the Spring Rental Term. Daily Rent collected and paid to Owner during the portion of the month of May which is included in the Summer Rental Term shall be considered rent paid to Owner for the Summer Rental Term. Daily Rent collected and paid to Owner for occupancy during the month of August will not be paid to Owner until September but will also be considered rent paid to Owner for the Summer Rental Term. The portion of the January Rent Installment to be paid in December under Section 5.5.1(g) will be considered part of the Rent Installments paid to Owner in the Spring Rental Term.

5. <u>Leasing Requirements</u>.

5.1 **<u>Rental Agreement</u>**. Subject to <u>Section 5.2</u> below, Manager shall lease available occupancy in the Residence Units of each of the Residences exclusively to registered students in good standing with the College and other persons participating in programs affiliated with the College and, during the Summer Rental Term, other persons to whom Residence Units are rented by Manager (all being hereinafter referred to as "Student Tenants"), all pursuant to the College's residence hall program and in accordance with the terms and conditions governing the same established and promulgated by the College from time to time ("Residence Program"), which such Residence Program's terms, conditions and guidelines are set forth in Exhibit G attached hereto and incorporated herein by this reference. In the event of any conflict between the Residence Program and this Agreement, this Agreement shall control to the extent of such conflict. For purposes of this Section 5 specifically and this Agreement generally, and subject to anything to the contrary herein, Owner hereby grants Manager an exclusive license during the Initial Term and any Extension Terms to enter into lease agreements with Student Tenants for occupancy of the Residences and related administration of the same as contemplated in this Agreement ("Residence License"). The residence License is also expressly subject to Owner's right to lease Residence Units reserved in Section 5.7 herein. For each and every leasing of

occupancy in the Residences to a Student Tenant, Manager shall, in its own name, enter into and execute a rental agreement in the form attached hereto as <u>Exhibit H</u> and incorporated herein by this reference ("**Rental Agreement**") at rental rates, not less than the applicable Minimum Rental Rates as set forth below. Manager shall provide Owner with a copy of each executed Rental Agreement not later than ten (10) days following execution of each such Rental Agreement. Subject to Owner's assumption rights/obligations pursuant to Section 5.8 herein, Manager may redact the names of the Student Tenants from the copies of the Rental Agreements provided to the Owner.

5.2 <u>Minimum Monthly and Daily Rental Rates</u>. The "Minimum Monthly Rent" for the first Lease Year shall be Three Hundred Ninety and no/100 Dollars (\$390.00) per Residence Unit, inclusive of the cost of local telephone and basic cable TV in the Common Room of each pod of Residence Units and broadband internet service in each Residence Unit. No later than February 15, 2007, and on or before February 15th in each subsequent Lease Year, Owner and Manager shall, by mutual agreement, determine the Minimum Monthly Rent for the upcoming Lease Year. Notwithstanding the foregoing, the Minimum Monthly Rent shall increase annually at the rate of not less than three percent (3%) per Lease Year. The "Minimum Daily Rent" shall be one thirtieth (1/30th) of the Minimum Monthly Rent.

5.3 <u>Additional Rent – Summer Term</u>. The parties expect that during the Summer Rental Term, Manager may be able to rent Residence Units at a daily rental rate which is in excess of the Minimum Daily Rent. The amount of such excess is hereinafter referred to as "Additional Rent," and such Additional Rent actually collected will be shared between Owner and Manager as follows:

(a) Owner shall be entitled to receive all Additional Rent from the Summer Rental Term actually collected by Manager until Owner has received an amount for the Lease Year equal to eighty-five percent (85%) of the product which results from multiplying the number of Residence Units in the Project (excluding only one Residence Unit in the RD Apartment) times the Minimum Monthly Rental Rate in effect for the Lease Year, times twelve (12) months.

(b) Thereafter, any Additional Rent for the Summer Rental Term actually received by Manager shall be paid one-half (1/2) to Owner and the remaining one-half (1/2) shall be retained by Manager.

5.4 <u>Additional Rent – Fall and Spring Terms</u>. The parties understand that any rent in excess of the Minimum Monthly Rent which is payable by Student Tenants during either the Fall or Spring Rental Terms, together with all forfeitures, unearned rent or other payments attributable to occupancy during either the Fall or Spring Rental Terms shall be retained by and belong to Manager.

5.5 <u>Remittance of Rent</u>.

5.5.1 Duty to Collect. Irrespective of Manager's ability or inability to actually collect any Rent Installments and any other sums due and payable from occupants under existing Rental Agreements, Manager hereby covenants that for the Fall Rental Term and Spring Rental Terms, it shall pay Owner the Minimum Monthly Rent for each Residence Unit occupied by a Student Tenant on the first day of each month during the Fall Rental Term and the Spring Rental Term (it being understood that September shall be considered the first month of the Fall Rental Term and that Minimum Monthly Rent payment for September will not be increased because that period includes a part of the month of August during which Student Tenants are in occupancy nor will the Minimum Monthly Rent payment for the month of May be reduced because that month includes a part of the month of May during which Student Tenants are not in occupancy), plus Pro-Rata Rent, as hereinafter defined. Monthly Rent Installments during the Fall and Spring Semester Rental Terms shall be computed by multiplying the number of Units occupied on the first day of each month, beginning with the month of September by the Minimum Monthly Rent as determined under Section 5.2. In addition to the foregoing, and after taking into account any adjustments reflected in the September 20th payment and the January 20th payment provided below, the Manager shall pay to Owner, as part of the applicable Rent Installment, the pro-rata portion of the Minimum Monthly Rent payable for the preceding month by a Student who occupies a Residence Unit after the first day of the preceding month ("Pro-Rata Rent"); Provided, further that Owner shall not be entitled to Pro-Rata Rent for a Residence Unit if Owner has received the Minimum Monthly Rent for that Residence Unit for the preceding month. Irrespective of when rent is collect by Manager, Monthly Rent Installments shall be remitted to Owner as follows:

September 5th – Seventy-five percent (75%) of Rent (a) Installment. for the month of September.

September 20th – The remainder of the September Rent (b) Installment, after adjustment for rent refunds and late registrations

> October 5th – Rent Installment for the month of October. (c)

November 5th – Rent Installment for the month of (d)

November.

December 5th – Rent Installment for the month of (e) December.

December 26^{th} – Seventy-five percent (75%) of the Rent (f)Installment for the month of January based upon occupancy for the month of December.

January 20th – The remainder of the January Rent (g) Installment after adjustment for non-returns, refunds and late registration

> February 5th – Rent Installment for the month of February. (h)

BAHR – SECTION II

TAB 10 Page 19

- (i) March 5^{th} Rent installment for the month of March.
- (j) April 5^{th} Rent Installment for the month of April
- (k) May 5^{th} Rent Installment for the month of May.

Manager shall collect and remit an entire Minimum Monthly Rent Payment for the month of December for each Student Tenant occupying a Residence Unit on December 1, even though school is not in session for the latter portion of December. Manager will also collect and remit an entire Minimum Monthly Rent Payment for the month of January for each Student Tenant occupying a Residence Unit on the first day of the spring semester even though that day will be after the 1st day of January.

5.5.2 <u>Summer Rental Term</u>. Irrespective of Manager's ability or inability to actually collect the Minimum Daily Rent, Manager hereby covenants that for the Summer Rental Term, it shall pay Owner the Minimum Daily Rent for each day a Residence Unit is occupied by a Student Tenant, plus an amount equal to any Owner's Additional Rent actually collected by Manager. Minimum Daily Rent and any Owner's Additional Rent owing for a month during the Summer Rental Term shall be remitted to Owner by the 10th day of the following month.

5.5.3 <u>Method of Payment</u>. The amounts to be remitted to Owner under this <u>Section 5</u> shall be placed is a separate Agency Account within the College's accounting system and shall be disbursed from said Agency Account as provided herein.

5.5.4 <u>Occupancy Reports</u>. At the time Rent Installments are remitted to Owner, Manager shall submit an "Occupancy Report" identifying each Residence Unit occupied during the period for which the Rent Installment is paid and, for the Summer Rental Term, the period the Residence Unit was occupied.

5.6 <u>Delinquent or Missed Rent Installments</u>. Any Rent Installment or other charges, sums or fees which are due and payable by Manager under the provisions of this Agreement and which are not paid when due shall bear interest from the date due until paid at an annual rate (hereinafter the "Agreed Rate") equal to five percent (5%) in excess of prime rate published in the Wall Street Journal as such rate changes from time to time.

5.7 <u>Lease-Up Schedule</u>.

(a) Manager hereby covenants and agrees that it shall use commercially reasonable efforts to market and fully lease-up the Residences each and every Lease Year.

(b) With regard to leasing-up of the Residences specifically, Manager shall lease occupancy in the Residences on a floor-by-floor basis beginning with the lowest floor in each Building and progressing to the next highest floor until all floors in the Building are fully occupied. Under no circumstances shall Manager be entitled to lease any Residence Unit on any

higher floor until occupancy on the immediately lower floor of the Building has been fully leased up pursuant to validly executed Rental Agreements for all of the available Residence occupancy on such immediately lower floors of each Building without first obtaining the Owner's written consent, which consent shall not be unreasonably withheld,, conditioned or delayed. By way of example, Manager would not be permitted to lease occupancy on the fourth floor of one Building until occupancy for the third floor of the Building had been fully leased in compliance with this Section 5.7(b). Additionally, in the event Manager is unable to fully lease ninety percent (90%) occupancy of the Residences for any given Rental Term, inclusive of the RD Apartment (as that term is defined below), then any and all remaining, unoccupied Residence Units not subject to valid Rental Agreements shall, after the forty-fifth (45th) day of that Rental Term, be unavailable to Manager for the then applicable Rental Term without Owner's prior written consent, which consent shall not be unreasonably withheld, conditioned, or delayed subject to Section 5.7(c) below. For purposes of this Section 5.7(b), Manager and Owner have designated Unit 212, A, B and C a pod composed of Units 212B and 212C and the adjoining Common Room (212A) as the residential director apartment ("RD Apartment"). One Residence Unit within the RD Apartment shall not be subject to, or included in the calculation of, Rent Installments, and shall not require execution of a valid Rental Agreement to be included in calculation of occupancy on any floor of the Building. The second Residence Unit in the RD Apartment shall not be rented by Manager to a Student Tenant and shall be used exclusively as an office and conference room for the Residence Director and the Residence Assistants. Manager shall have no obligation to pay Minimum Monthly or Daily Rent for this second Residence Unit while so used, but this second Residence Unit shall not be considered to be occupied for the purposes of calculating percentage of occupancy for any purpose under this Agreement.

(c) Owner's Right to Lease Residence Units. In the event Manager is unable in any given Rental Term to lease all of the Residence Units in the Project after the forty-fifth (45th) day of that Rental Term, Owner shall have the exclusive right, but not the obligation, to lease any such unoccupied Residence Units for the then current Rental Term or any remainder term thereof to tenants that Owner may solicit and accept, provided, that prior to placing any occupant in available Residence Units then managed by Manager, Owner must first provide Manager with a five-day (5) written notice of the occupant Owner intends to place in a Unit. In the event Manager refuses to accept the proposed occupant, Manager must respond in writing within such five-day (5) period as to why it rejects the proposed occupant and also within such five-day (5) period, Owner and Manager shall attempt to mutually agree as to whether the occupant shall be entitled to take procession of a Unit. Failure to agree shall result in a decision being made no later than five (5) days of the expiration of the initial five-day (5) notice period by LCSC's Vice-President for Administrative Services and a good faith decision by the Vice-President of Administrative Services shall be binding upon both parties. The exercise of any Owner Leasing Rights shall be documented by Owner executing a form of Lease Agreement prepared by Owner utilizing the same format that Manager is then using for all other Units. Manager shall have no responsibility for the collection of rent from Owner's tenants, nor shall Manager be responsible for the actions of said tenants or the repair or replacement of damage caused by said tenants.

5.8 <u>Owner's Right to Deal Directly with Student Tenants</u>. In the event of any Event of Default, early termination of this Agreement, or other event which results in the

removal or replacement of Manager, Owner shall have the right but not the obligation, in its sole discretion, to (a) assume any or all of the Rental Agreements with Student Tenants for the thenapplicable Lease Session and step into all of the rights, duties and obligations of Manager thereunder (and Manager shall draft any and all Rental Agreements with Student Tenants to authorize the same), or (b) enter into a new lease agreement with any or all of the Student Tenants for the then-remaining Lease Period. In the event Owner does not elect to assume the Rental Agreements or enter into new lease agreements with any or all of the Student Tenants for the then-applicable Lease Session, such Rental Agreements so affected shall be deemed terminated without any liability to Owner. Manager hereby covenants to include any and all relevant provisions in the Rental Agreements to effect the provisions of this Section 5.8. The rights set forth in this Section 5.8 shall explicitly be in addition to, and not subject to offset against, any default remedies provided to Owner pursuant to Section 11 of this Agreement. Notwithstanding the foregoing provisions of this Section 5.8, Owner agrees that if the termination occurs during the Fall or Spring Rental Terms, Owner will assume from Manager those Rental Agreements which conform to the provisions of Exhibit H, as that Rental Agreement may from time to time be amended with the approval of Owner, for the remainder of the then-current Rental Term, so long as the Rental Agreement provides for rent which is at least equal to the then applicable Minimum Monthly Rent. Manager shall promptly pay Owner all prepaid rent received by Manager upon the Rental Agreements assumed by Owner, which prepaid rent shall be determined by proration as of the termination date. Manager shall also remit to Owner the unexpended portions of security and other deposits made by Student Tenants whose Rental Agreements are assumed by Owner.

6. <u>Use of Residences</u>.

6.1 <u>Residential Area</u>. The Project, and the Residences and Residence Units, shall be used for the primary housing of Student Tenants and other incidental uses reasonably related thereto and in support thereof in accordance with the Residence Program or otherwise as Manager sees fit, provided that nothing herein shall authorize Manager to use or allow use of the Residences in any manner contrary to the terms and conditions of the Rental Agreement ("**Permitted Use**").

6.2 <u>Compliance with Applicable Laws and Prohibited Uses</u>. Manager shall not allow the Residences, or any portion thereof, to be occupied or used contrary to any applicable statute, rule, order, ordinance, requirement or regulation, or in any manner which would: (a) violate any certificate of occupancy affecting the same; (b) cause injury to the improvements or overload the floors; (c) constitute a public or private nuisance or waste; or (d) increase the cost of Owner's Insurance (as defined below) maintained by Owner relating to the Residences (collectively, the "**Prohibited Uses**"). Manager hereby covenants and agrees that it will promptly, upon discovery of any Prohibited Use, take all reasonable steps necessary to compel the discontinuance of such use. In the event that any Prohibited Use results in an increase in the cost or expense of Owner's Insurance, Manager shall pay any such increased costs or expenses in Owner's Insurance in connection with such Prohibited Use.

6.3 <u>**Prohibited Use of Retail Spaces**</u>. Owner hereby covenants not to lease or otherwise allow any of the Retail Spaces to be used as a pornographic bookstore or any other pornographic entertainment operation.

7. <u>Maintenance and Repair of the Residences</u>.

7.1. Manager Duties, Obligation and Liabilities.

7.1.1 <u>Manager's Maintenance Duties</u>. Pursuant to the Management Responsibility, Manager shall be responsible for, and the Management Expenses shall include, the obligation to maintain, repair and replace, as and when necessary, all aspects of the Project, including, without limitation, the Residences and all janitorial and trash removal services relating thereto ("Manager Maintenance Duties"). Additionally, Manager's Maintenance Duties shall additionally include any and all of Owner's Maintenance Duties (as defined below) in the event any of the same are caused by, result from, or are otherwise attributable, whether directly or indirectly, the negligence or willful misconduct of Manager, including, without limitation, any of the same arising under or as a result of the Management Responsibility.

7.1.2 <u>Manager's Changes and Alterations</u>. Manager, at its sole cost and expense as a Management Expense, may make any modifications, improvements, alterations or additions to the Residences ("Alterations"), provided that such Alterations (a) do not affect the exterior appearance of the Building; (b) do not or add or subtract from the square footage of the Residences, including without limitation any floor area of the Building and/or any Residences therein; (c) do not otherwise affect the structural, electrical or mechanical systems of the Building and/or Residences; and (d) do not in the aggregate cost in excess of Fifteen Thousand and no/100 Dollars (\$15,000.00) in any twelve (12) month period. Any other Alteration by Manager shall require Owner's prior written consent, which consent shall be in Owner's reasonable discretion to condition, delay or withhold. Notwithstanding the foregoing, prior to Manager undertaking any Alteration, Manager shall (w) notify Owner not less than ten (10) days prior to beginning such Alteration; (x) deliver to Owner a copy of the plans for such Alteration, if any; (y) properly obtain and deliver to Owner copies of any and all necessary permits required under Applicable Laws (as defined below) to undertake and complete such Alteration; and (z) perform and complete all Alterations in a good and workmanlike manner in compliance with Applicable Laws. Along with any request for Owner's consent, and prior to commencement of any Alteration or delivery of any materials to the Residences, Manager shall first furnish Owner with the names and addresses of all contractors performing work on the Alteration. Manager shall be liable for any damages and shall pay all costs and expenses, including Attorneys' Fees (as defined below) incurred by Owner in any way connected to or arising from claims and liabilities resulting from the negligence on the part of itself, its employees or its officers in the performance or completion of such Alteration. Unless otherwise notified by Owner in writing, Manager shall remove at its sole cost and expense any and all Alterations upon expiration of the Initial Term or any Renewal Terms, or the sooner termination of this Agreement, and shall repair any damage related thereto such that the Residences is restored to prior condition preceding each such Alteration.

7.1.3 Mechanic's Liens. Manager shall not suffer or permit any mechanic's lien or other lien to be filed against all or any portion of the Project because of any Authorized Repair, Alteration, or other work, labor, services, equipment or materials supplied or claimed to have been supplied to the Residences at the request of Manager, or anyone holding all or any portion of the Project through Manager, including any Student Tenants. If any such lien is filed against all or any portion of the Project, Owner shall give Manager immediate notice of the filing and Manager shall cause the lien to be discharged within thirty (30) days after Owner's demand. For the purposes hereof, the term "discharge" shall mean the payment of the lien or the posting of a bond with respect thereto allowed under Idaho law. If Manager fails to discharge such lien within such period, in addition to any other right or remedy Owner may have, after five (5) days prior written notice to Manager, Owner may, but shall not be obligated to, discharge the lien by paying to the claimant the amount claimed to be due or by procuring the discharge in any other manner that is now or may in the future be permitted by law. Any amount paid by Owner, together with all reasonable and actual costs, fees and expenses in connection therewith, including Owner's reasonable Attorneys' Fees, together with interest thereon at the Agreed Rate, shall be repaid by Manager to Owner on demand by Owner. Manager shall be liable for any damages and shall pay costs, damages, expenses, including Attorneys' Fees, and shall be solely liable for any liabilities, penalties, claims, demands and obligations, resulting from such lien.

7.2 <u>Owner's Duties, Obligations and Liabilities</u>.

(a) Unless otherwise expressly provided for herein, Owner shall be financially responsible for, but Manager shall undertake as part of the Management Responsibility, the maintenance, repair, and replacement, as and when necessary, of all aspects of the Project, expressly excluding any Management Responsibility and/or Manager's Maintenance Duties relating to the Project, but including without limitation (i) all structural and mechanical elements of the Project, including all foundations, exterior walls, roof structure, window, gutters, and exterior glazing, exterior doors, and Building envelope, (ii) all electrical, mechanical, HVAC, water, telecommunications, and other utility systems, lines and conduits located within the Building, (iii) the elevator, (iv) Owner's Additional Maintenance Obligation (as defined in Section 7.2(c)), and (v) all aspects of the Retail Spaces ("Owner Maintenance **Obligations**"). Manager shall be authorized to incur costs on account of the Owner without Owner's prior written consent for any Owner Maintenance Obligations which are less than five hundred dollars (\$500.00) per incident and less than five thousand dollars (\$5000.00) in the aggregate during any given Rental Term; provided, however, that these limitations shall not apply and Owner's prior written consent shall not be required if immediate repairs are reasonably necessary to prevent further damage to the Project. Owner also agrees that it will not unreasonably withhold, condition or delay any consent required under this Section 7.2(a). Costs incurred by Manager for Owner Maintenance Obligations are Owner Expenses and shall be reimbursed to Manager as provided in Section 3.3.

(b) Additionally, Owner shall, at its sole cost and expense, provide certain furnishings, fixtures, and equipment for the Residences and each Residence Unit as set forth on <u>Exhibit C</u>, ("**Owner FF&E**"). The parties recognize that the repair and replacement of the Owner FF&E, to the extent the costs of such repairs and replacement is not properly payable from tenant security deposits, is the joint obligation of the parties; provided that Manager shall

be responsible for undertaking all such maintenance, repair, and replacement obligations as part of the Management Responsibility. The Cost of such maintenance, repair and replacement of Owner's FF&E shall be paid one-half (1/2) by Owner and one-half (1/2) by Manager. Owner's share of such cost is an Owner Expense and shall be reimbursed to Manager as provided in Section 3.3.

(c) Owner shall also be responsible for the cost incurred by Manager to repair or replace any damage, deterioration or defects in the Project resulting from defects in the original design, materials or workmanship of the Project ("**Owner's Additional Maintenance Obligation**"); provided that Manager gives the Owner written notice of the damage, deterioration or defect within one (1) year of the date the Project is ready for Management. The cost of satisfying Owner's Additional Maintenance Obligation expense and shall be reimbursed to Manager as provided in Section 3.3.

7.3 <u>Maintenance of Landscaping</u>. Manager shall also be responsible for the maintenance of the landscaping and irrigation systems on the south side of the Building. Lawn mowing, trimming of shrubbery and other routine maintenance shall be done by Manager, at Manager's expense. All other costs, including, without limitation, servicing, repair or replacement of irrigation systems or their components, replacement of planting or poor drainage problems will be an Owner's Expense and will be reimbursed to Manager as provided in Section 3.3; provided that Owner shall not be responsible for any such costs resulting from Manager's negligence or willful misconduct in Manager's performance of Owner's landscaping obligations.</u>

7.4 <u>**Taxes.**</u> Owner shall pay, before delinquency all real estate and personal property taxes assessed against the land and improvements of which the Project is a part and against the Owner FF&E.

7.5 <u>Utilities.</u> Owner shall timely pay, when due, all charges and costs for utilities or services furnished to the Project, including water, sewer, gas and electricity, together with broadband internet, for each Residence Unit and basic cable and local telephone service to each Common Room.

7.6 <u>Common/Shared Costs</u>. In the event any of the Residences share any utilities, services, equipment, or the like ("Shared Resources") which fall under Owner's Maintenance Duties, and/or the Building, excluding the Residences but including without limitation the Retail Spaces, share any Shared Resources which fall under Manager's Maintenance Duties, then such Shared Costs shall be paid by Manager pursuant to Manager's Maintenance Duties on a pro-rata basis based on the proportion that the gross square footage of the Residences comprise of the Building, and by Owner pursuant to Owner's Maintenance Duties on a pro-rata basis based on the gross square footage of the Residences comprise of the Building, and by Owner pursuant to Owner's Maintenance Duties on a pro-rata basis based on the gross square footage of the Building excluding the Residences but expressly including the Retail Spaces.

7.7 Failure to Comply. Any failure by Owner or Manager to comply with this <u>Section 7</u> shall be deemed a default of this Agreement and shall be governed pursuant to and in accordance with <u>Section 11</u> below.

8. <u>Insurance</u>.

8.1 <u>**Owner's Insurance.**</u> Owner shall keep the Project insured in an amount equivalent to their full insurable replacement cost thereof against loss or damage by fire and such other risks as are customarily covered with respect to such property, including sprinkler leakage, windstorms, hail, explosions, vandalism, theft, malicious mischief, civil commotion and such other coverage capable of being covered by a standard coverage insurance policy. Owner shall additionally carry commercial general liability insurance covering the Project, which policy shall have a combined single limit of not less than One Million and no/100 Dollars (\$1,000,000.00), per occurrence, coverage on an occurrence basis, with a deductible of not more than Ten Thousand and no/100 Dollars (\$10,000.00) (collectively, "**Owner's Insurance**"). Such Owner's Insurance shall name Manager as an additional insured.

8.2 <u>Manager's Insurance</u>. Manager, at its sole cost and expense, shall keep any and all of Manager's inventory, equipment, furniture, fixtures and other personal property located in or used in connection with the Residences ("Manager's Personal Property") insured in an amount equivalent to the full insurable replacement cost against loss or damage by fire and such other risks as are customarily covered with respect to such property, including sprinkler leakage, windstorms, hail, explosions, vandalism, theft, malicious mischief, civil commotion and such other coverage capable of being covered by a standard coverage insurance policy. Manager may additionally, at Manager's option and, at its sole cost and expense, carry business interruption and extra expense insurance in a sufficient amount to protect Manager against any additional costs and lost income associated with interruption of its business from the Project for a period not less than one (1) year (collectively, "Manager's Insurance").

8.3 <u>General Provisions</u>. All policies required under <u>Sections 8.1</u> and <u>8.2</u> shall be written by companies licensed in the state in which the Project is located; provided, however, that Manager is authorized to insure its property through the State of Idaho Self-Insured Risk Program. Each such policy shall be evidenced by a certificate of insurance issued by the insurance company issuing the policy. The certificate or certificates of insurance for the policy or policies to be provided under Section 8.1, together with a copy of the endorsement naming Manager as an additional insured under Owner's commercial general liability insurance, shall be promptly provided to Manager by Owner. The certificate or certificates of insurance for the policy or policies to be provided under Section 8.2 shall be promptly provided to Owner by Manager. Each certificate shall provide the policy with respect to which the certificate is issued, shall not be cancelled unless at least twenty (20) days prior written notice of cancellation shall have been mailed to the party to whom the certificate is to be provided.

8.4 <u>Blanket Insurance Coverage</u>. Nothing in this <u>Section 8</u> shall prevent Owner from maintaining insurance of the kinds and in the amounts required by <u>Section 8.1</u> under a blanket insurance policy or policies covering other properties owned or operated by Owner in addition to the Project; provided, however, that (a) each policy contains the various provisions and coverage amounts required by this <u>Section 8</u>, and (b) the certificate for such policy includes a statement from the insurer setting forth the coverage maintained and the amounts exclusively allocated to the Project.

8.5 <u>Release and Waiver of Subrogation</u>. Notwithstanding any other provision of this Agreement, Owner and Manager each hereby waives, releases and discharges the other, its agents and employees from all claims whatsoever arising out of loss, claim, expense, damage or destruction covered or coverable by insurance required by <u>Sections 8.1</u> and <u>8.2</u> or covered by other casualty insurance it may carry (a "Loss"), notwithstanding that such Loss may have been caused by the other, its agents or employees. Owner and Manager each hereby agrees to look to its insurance coverage only upon such Loss. Owner's Insurance described in <u>Section 8.1</u> shall contain a waiver of subrogation clause as to Manager. Manager's Insurance required by <u>Section 8.2</u> shall contain a waiver of subrogation clause as to Owner.

9. <u>Compliance with Laws</u>.

9.1 <u>Generally</u>. As of the Delivery Date, Owner represents and warrants that the Project complies with all current laws, ordinances, orders, rules, regulations and requirements of all federal, state, municipal and other governmental bodies having jurisdiction over the Project as of the Commencement Date ("Applicable Laws"). Manager shall as of the date of the Delivery Date and throughout the Initial Term and any Renewal Term comply with any and all Applicable Laws, including without limitation any and all Applicable Laws with regard to the Residences and the Project generally, the Management Responsibility, Maintenance and Repair Obligations, Rental Agreements, and the Student Tenants and Residence Program.

9.2 <u>Compliance with Hazardous Materials Laws</u>. Manager shall comply with all federal, state and local laws, ordinances, codes, regulations, orders and decrees, as they now exist or are hereafter amended, including all policies, interpretations, guidelines, directions, or recommendations ("Hazardous Materials Laws") relating to industrial hygiene, environmental protection or the use, analysis, generation, manufacture, storage, presence, release, disposal or transportation of any petroleum products, flammable explosives, asbestos, urea formaldehyde, polychlorinated biphenyls, radioactive materials or waste, or other hazardous, toxic, contaminating or polluting materials, substances or wastes, including any materials defined as "hazardous substances," "hazardous wastes," "hazardous Materials" or "toxic substances" (collectively, "Hazardous Materials") under any such Hazardous Materials Laws. For purposes of this Agreement, any reference to Applicable Laws shall include any and all Hazardous Materials Laws as applicable.

Manager shall manage the Residences pursuant to its Management Responsibility in a manner designed to prevent the release of any Hazardous Materials. If any release of any quantity of Hazardous Materials occurs in, on, under or about the Residences and/or the Project generally of which Manager is or becomes aware, Manager shall promptly notify all appropriate governmental agencies and Owner. Manager shall promptly and fully investigate, remediate and remove all such Hazardous Materials released by Manager, its employees, agents or invitees, in accordance with all applicable governmental requirements and shall restore the affected portions of the Residences and/or Project. Manager shall promptly provide Owner with copies of all reports, analyses and correspondence in Manager's possession relating to such release and the remediation thereof. Upon expiration or earlier termination of this Agreement, Manager shall cause all Hazardous Materials located in, on, under or about the Residences and/or Project as a result of the acts or omissions of Manager to be removed from the Project and transported for

use, storage or disposal in compliance with all applicable Hazardous Materials Laws. Manager shall not take any remedial action in response to the presence of any Hazardous Materials in, on, about or under the Residences and/or the Project generally, nor enter into any settlement agreement, consent decree or other compromise with respect to any claims relating to the Residences and/or the Project generally without first notifying Owner of Manager's intention to do so and affording Owner ample opportunity to appear, intervene or otherwise protect Owner's interest with respect thereto. At the expiration or earlier termination of this Agreement, Manager shall remove all tanks or fixtures which were placed in the Residences (and the Project generally as and where applicable) by Manager during the term of this Agreement and which contain, have contained or are contaminated with, Hazardous Materials. Notwithstanding the foregoing, Manager shall not be financially responsible for any costs, including, but not limited to, remediation costs, resulting from the activities of Student Tenants, their guests or invitees, in the Project unless such costs also resulted from the negligence or willful misconduct of Manager, including, without limitation, Manager's duties, obligations and liabilities pursuant to the Management Responsibility and the Manager Maintenance Duties.

Manager shall notify Owner in writing immediately upon receiving notice of: (a) any enforcement, clean-up, removal or other governmental or regulatory action effecting the Residences instituted, completed or threatened pursuant to any Hazardous Materials Laws; (b) any claim made or threatened by any person against Manager, Owner or the Residences and the Project generally, relating to damage, contribution, cost recovery, compensation, loss or injury resulting from or claimed to result from any Hazardous Materials; and (c) any reports made to any environmental agency arising out of or in connection with any Hazardous Materials in, on or about the Residences and the Project generally or with respect to any Hazardous Materials removed from the Residences and the Project generally. Manager shall also provide to Owner, as promptly as possible, and in any event within five (5) business days after Manager first receives or sends the same, copies of all claims, reports, complaints, notices, warnings or asserted violations relating in any way to the Residences or Manager's use thereof. Upon written request of Owner, Manager shall promptly deliver to Owner copies of hazardous waste manifests, if Manager is required by applicable law to obtain such manifests, reflecting the legal and proper disposal of all such Hazardous Materials removed or to be removed from the Residences and the Project generally. All such manifests shall list Manager or its agent as a responsible party and in no way shall attribute responsibility for any such Hazardous Materials to Owner.

9.3 <u>Hazardous Materials Representation by Owner</u>. Owner warrants to its actual knowledge, that except as disclosed in the environmental assessment prepared by Jim Kolva and Associates, dated November 10, 2005, and the Asbestos Survey of the Adams Building, 405 Main Street, Lewiston, Idaho prepared by EnviroScience, dated December, 2006 (the "Environmental Assessments"), which Environmental Assessments Manager hereby acknowledges that it has received and reviewed, that no hazard substance, toxic waste, or other substance has been produced, disposed of, or is or has been kept on the Project. Owner further warrants that to Owner's actual knowledge, no items disclosed in the Environmental Assessments will subject the Manager to any damages, penalty, or other liability under any applicable local, state or federal law or regulation. Further, Owner represents to Manager that, to its actual knowledge, neither Owner nor any other person has caused the generation, storage or

release of Hazardous Materials in, on, about or upon the Project (including but not limited to the Residences), except as disclosed in the Environmental Assessments, the Good Faith Survey and/or in accordance with Hazardous Materials Laws and prudent industry practices regarding construction of the Property (including, but not limited to, the Residences).

9.4 Hazardous Substances Indemnification. Manager shall defend and hold harmless Owner and Owner's officers, directors, partners, managers, members, employees, agents, successors and assigns against any and all claims, and shall pay all costs and expenses arising from all liabilities, damages, penalties, forfeitures, losses, obligations, investigation costs, remediation and removal costs, natural resource damages and expenses (including Attorneys' Fees) (collectively "Damages") arising in whole or in part, directly or indirectly, from (a) the presence or release of Hazardous Materials, in, on, under, upon or from the Residences and the Project generally as a result of acts or omissions of Manager or its employees, agents or invitees; (b) the transportation or disposal of Hazardous Materials to or from the Residences and the Project generally by, at the request or with the permission of Manager, its employees, agents or invitees; (c) the violation of any Hazardous Materials Laws by Manager, its employees or agents; (d) the failure of Manager, its employees or agents to comply with the terms of this Section 9; or (e) the use, storage, generation or disposal of Hazardous Materials in, on or abut the Residences and the Project generally by Manager or its employees, agents contractors, assignees, sublessees or invitees during the Initial Term and any Renewal Terms. Owner shall defend (with legal counsel reasonably acceptable to Manager) and hold harmless Manager, its officers, directors, employees, agents, successors and assigns against any and all Damages arising out of the presence or release of Hazardous Materials in, on or about the Residences and the Project generally released by Owner, its employees, agents, contractors or invitees (other than Manager) or resulting from any breach of the representation and warranty contained in Section 9.3 above. The parties recognize that neither party is indemnifying the other for Damages arising out of acts of third parties not under either party's control. The respective rights and obligations of Owner and Manager under this Section 9 shall survive the expiration or earlier termination of this Agreement. Notwithstanding the foregoing, Manager shall have no indemnity obligations with respect to claims, costs or expenses resulting from the actions of Student Tenants, their guests or invitees in the Project, unless the matter giving rise to such claims, costs or expenses also results from the negligence or willful misconduct of Manager.

9.5 <u>Discrimination</u>. Owner assures and certifies that it will comply with all applicable provisions of the Americans with Disabilities Act of 1990 (42 U.S.C. §§ 12101-12213) and the Idaho State Law Against Discrimination, as well as the regulations adopted thereunder (collectively, the "ADA Rules & Regulations"). Owner, at Owner's expense, will provide that the Project complies with the minimum legal/regulatory requirements of the ADA Rules and Regulations at the time of acceptance by Manager and during the term of the Agreement. Manager shall have sole responsibility and liability for managing the placement of Student Tenants in rooms which are appropriate under ADA Rules and Regulations. Neither Owner nor Manager shall have any obligation to the other under this Agreement to make any part of the Project ADA-compliant outside the minimum legal/regulatory requirements, notwithstanding the fact that Manager may have more Student Tenants requiring such rooms than are available and/or that can be made available by relocating other Student Tenants.

10. <u>General Liabilities</u>.

10.1 <u>Indemnity by Manager</u>. Manager covenants and agrees to defend, indemnify and hold Owner, its agents, and employees harmless from and against all claims, losses, liabilities, damages, costs and expenses, including attorneys' fees and costs, arising out of or in connection with any breach of this Agreement, including but not limited to, (a) any breach of Manager's representation, warranties, covenants, obligations and duties under this Agreement, (b) any failure by Manager to comply with Applicable Laws, (c) any liabilities related to an Event of Default by Manager, and (d) any liabilities arising under Rental Agreements between Manager and Student tenants, including without limitation any liabilities arising as a result of Manager's removal and/or the termination of this Agreement by Owner; provided, however, that the indemnity obligations of Manager are subject to the limitations of liability contained in Idaho Code, Title 6, Chapter 9.

10.2 <u>Indemnity by Owner</u>. Owner covenants and agrees to defend, indemnify and hold Manager, its agents, and employees harmless from and against all claims, losses, liabilities, damages, costs and expenses, including attorneys' fees and costs, arising out of or in connection with any breach of this Agreement, including but not limited to, (a) any breach of Owner's representation, warranties, covenants, obligations and duties under this Agreement, (b) any failure by Owner to comply with Applicable Laws, and (c) any liabilities related to an Event of Default by Owner, provided, however, that the indemnity obligations of Owner are, by agreement of the parties, subject to the same monetary limitations of liability contained in Idaho Code, Title 6, Chapter 9 which would apply if Owner was a governmental entity under the laws of the State of Idaho.

11. Defaults of Owner and Manager and Right to Cure.

11.1 <u>Events of Default</u>. The occurrence of any of the following constitutes an "Event of Default" or "Manager Default" by Manager under this Agreement:

(a) Manager fails to take possession of the Project on August 20, 2006, or such later date when the Project is Ready for Management as provided for in <u>Section</u> <u>1.2</u> of this Agreement;

(b) Manager fails to collect and remit to Owner the amounts due to Owner under this Agreement when due and such failure continues for five (5) days after Owner notifies Manager of Manager's failure to pay such amounts when due;

(c) Manager fails to perform any of Manager's non-monetary obligations under this Agreement and the failure continues for a period of thirty (30) days after Owner notifies Manager in writing of such failure; provided that if Manager cannot reasonably cure its failure within such thirty (30) day period, Manager's failure shall not constitute an Event of Default if Manager commences to cure such failure within said thirty (30) day period and thereafter diligently pursues cure of the same;

(d) The existence at any time during the Initial Term and/or any Renewal Terms of any material misrepresentation, omission in any financial statements, correspondence or other information provided to Owner by or on behalf of Manager in regard to the following: (a) Manager's leasing of Residences to Student Tenants or others; (b) any proposed or attempted Transfer; or (c) any consent or approval Manager requests under this Agreement; and

(e) Manager makes a general assignment or general arrangement for the benefit of creditors; or otherwise files a Petition, voluntarily or involuntarily for adjudication of bankruptcy or for reorganization or loses its accreditation with the State Board of Education for the state of Idaho.

11.2 **<u>Remedies</u>**. Upon the occurrence of any Event of Default and failure to cure as provided for in <u>Section 12.1</u> above, at any time and from time to time, and without preventing Owner from exercising any other right or remedy, Owner may exercise any one or more of the following remedies:

Owner may terminate this Agreement effective on the date Owner (a) specifies in its written termination notice to Manager. In the event of such termination, unless otherwise assumed by Owner as provided below and subject to any required assumption under the provisions of Section 5.8, Manager shall, at its sole cost and expense, cause the removal of all Student Tenants from the Project and shall deliver the same to Owner in good condition and repair, less reasonable wear and tear, not later than thirty (30) days from the date of Owner's notice of termination. Additionally, Owner may recover from Manager and Manager will pay to Owner on demand all damages Owner incurs by reason of Manager's default, including, any amount necessary to compensate Owner for all reasonable out-of-pocket costs and expenses incurred by Owner proximately caused by Manager's failure to perform its obligations under this Agreement including costs related to relocation. Except as specifically set forth in this section, nothing shall limit or prejudice Owner's right to prove and obtain damages in an amount equal to the maximum amount allowed by Applicable Law, regardless whether such damages are greater than the amounts set forth in this section. In addition to, and not subject to offset against, any of Owner's other rights to collect damages as provided for herein, whether or not Owner terminates this Agreement as provided for under these remedies, Owner may recover from Manager, and Manager will pay to Owner, on demand, any amounts due to Owner which Manager has failed to collect and remit to Owner as required hereunder, including without limitation any Rent Installments which arise and/or relate to any period following termination during which Student Tenants reside in the Project prior to removal by Manager as provided for herein. In addition to, and not subject to offset against, any of Owner's other rights to collect damages as provided for herein, whether or not Owner terminates this Agreement as provided for under these remedies, Owner may recover from Manager, and Manager will pay to Owner, on demand, any amounts due to Owner which Manager has failed to collect and remit to Owner as required hereunder

(b) Subject to the provisions of Section 5.8 requiring assumption by Owner and in addition to Owner's termination rights set forth above, Owner may elect to terminate this Agreement and assume some or all of the then-outstanding Rental Agreements in accordance with Section 5.8 herein. In the event Owner elects to assume some or all of the then-

outstanding Rental Agreements or Owner is required to assume some or all of the outstanding Rental Agreements under the provisions of <u>Section 5.8</u>, Manager will immediately assign such Rental Agreements and surrender all management records to Owner (subject to Manager's right to retain copies of such records) and shall remit to Owner all tenant security deposits being held by Manager, and Manager shall have no further rights as Manager under this Agreement with respect to such Rental Agreements;

(c) Owner may perform any obligation on Manager's behalf which Manager has failed to perform, including Manager's obligations to remove Student Tenants following notice of termination, without waiving Owner's rights under this Agreement at law or in equity, and without releasing Manager from any obligation under this Agreement. The costs and expenses of Owner, including reasonable overhead and attorneys' fees, arising from or related to the performance of any such obligations of Manager shall be subject to reimbursement by Manager and otherwise expressly be included as part of any damages claims and/or right of recovery available to Owner under this Agreement or Applicable Law; or

(d) Exercise any other rights in law or equity under Applicable Laws, including remedies not explicitly provided for in this Agreement. Each remedy or right of Owner provided for in this Agreement will be cumulative and will be in addition to every other right or remedy provided for in this Agreement, or now or hereafter existing at law, in equity, by stature or otherwise. The exercise or the beginning of the exercise by Owner of any one or more of such rights or remedies will not preclude the simultaneous or later exercise by Owner of any or all other rights or remedies.

11.3 <u>Owner Default</u>. If the Owner defaults in the performance of any of its obligations under this Agreement, (a default includes failure to perform any of the required obligations of this contract and/or material misrepresentation or omission in any financial statement, correspondence or other information provided to Manager by or on behalf of Owner in connection with Owner's performance of this Agreement and any material misrepresentation made by Owner in this Agreement), which default continues for a period of more than five (5) days after receipt of written notice from Manager specifying such default if the default is the failure to make a payment to Manager when due or more than thirty (30) days after receipt of written notice from Manager specifying such default in the case of any other default, except that if the default within the thirty (30) days for remedy, then if Owner fails to commence curing such default within the thirty (30) day period or fails to thereafter diligently continue curing such default until completion, then Manager may elect one or more of the following remedies.

(a) Manager may terminate this Agreement effective on the date Manager specifies in its written termination notice to Owner. If Manager terminates this Agreement, Manager may recover from Owner and Owner will pay to Manager on demand all damages Manager incurs by reason of Owner's default, including, any amount necessary to compensate Manager for all reasonable out-of-pocket costs and expenses incurred by Manager proximately caused Manager by Owner's failure to perform its obligations under this Agreement including costs related to relocation of Student Tenants. Except as specifically set forth in this Section, nothing in this Section limits or prejudices Manager's right to prove and obtain damages

in an amount equal to the maximum amount allowed by the Laws, regardless whether such damages are greater than the amounts set forth in this Section. Manager may perform any obligation on Owner's behalf which Owner has failed to perform, without waiving Manager's rights under this Agreement, at law or in equity and without releasing Owner from any obligation under this Agreement.

(b) Except as specifically set forth in this Section, if the Owner breaches any of the provisions of this Agreement, Manager will be entitled to enjoin such breach and will have the right to invoke any right or remedy allowed at law, in equity, by statute or otherwise including summary proceedings and other remedies not provided for in this Agreement. Each remedy or right of Manager provided for in this Agreement will be cumulative and will be in addition to every other right or remedy provided for in this Agreement, or now or hereafter existing at law, in equity by statute or otherwise. The exercise or the beginning of the exercise by Manager of any one or more of such rights or remedies will not preclude the simultaneous or later exercise by Manager of any or all other rights or remedies, at its option and may incur any reasonable expense necessary to perform the obligation of the Owner and thereafter be reimbursed by Owner for all such costs and expenses incurred by Manager plus interest at the Agreed Rate, which shall be payable as a cash payment to Manager or in the alternative Manager may offset Owner's obligation by an appropriate reduction in the amounts next coming due from Manager to Owner under this Agreement.

Owner will reimburse and compensate Manager on demand for any costs, fees and expenses Manager incurs in connection with, resulting from or related to an Event for Default and Manager's exercise of any or all default remedies as provided for herein, and regardless whether suit is commenced or judgment is entered. Such loss includes all reasonable Attorneys' Fees (as defined below) which Manager incurs investigating, negotiating, settling or enforcing any of Manager's rights or remedies or otherwise protecting Manager's interests under this Agreement. In addition to the foregoing, Manager is entitled to reimbursement of all of Manager's fees, expenses and damages, including reasonable Attorneys' Fees Manager incurs in connection with protecting its interest in any bankruptcy or insolvency proceeding involving Owner, including any proceeding under any chapter of the Bankruptcy Code; by asserting or defending a claim; by defending a preference or fraudulent transfer action; by exercising and advocating rights under Section 365 of the Bankruptcy Code; by proposing a plan of reorganization and objecting to competing plans; and by filing motions for relief from stay. Such fees and expenses are payable on demand, or, in any event, upon assumption or rejection of this Agreement in bankruptcy.

11.4 <u>Mitigation of Damages</u>. Either party agrees to use its best reasonable efforts to mitigate its damages in connection with any damage claim it may have against the other arising out of or in connection with this Agreement.

12. <u>Insolvency</u>. [Intentionally Deleted]

13. <u>Damage and Destruction</u>.

Repair Obligations. Other than a Casualty (as defined below) caused by 13.1 the negligence or willful misconduct of Manager pursuant to its covenants, obligations and duties pursuant to this Agreement or otherwise, and subject to the terms of Section 13.2, in the event the Residences or any portion thereof is damaged by fire or other casualty (a "Casualty"), for which Owner is carrying or required by this Agreement to carry insurance, Owner shall repair such damage at its expense. In the event of a Casualty, Rent Installments shall be suspended or abated to the extent and for the period the Project is rendered uninhabitable for the Permitted Use and until such time that the Project has been restored to the condition immediately prior to such damage or destruction, which such condition shall render the Project Ready for Management as contemplated under this Agreement. If the Casualty is an insured loss or a loss required to be insured under the provisions of this Agreement, as long as Owner is diligently pursuing collection of the insurance proceeds, repair shall commence upon receipt of the proceeds, provided that, in any case, the repair shall commence as soon as reasonably possible after the event of damage, but in no event, subject to Force Majeure, later than one-hundred-twenty (120) days after the damage. Owner shall not be obligated to repair Manager's personal property, equipment or improvements installed by Manager or any personal property of any Student Tenants. Notwithstanding anything to the contrary contained herein, Owner's obligation to repair under this Section 13.1 shall be limited to the extent of the insurance proceeds available to Owner for such restoration. Additionally, in the event that any Casualty comprises more than fifty percent (50%) of the Project, then Owner, in its sole discretion, may terminate this Agreement forthwith by providing Manager with written notice of its intention to terminate within thirty (30) days after the date of the Casualty. In the event of a Casualty, Manager agrees that it will proceed with reasonable diligence at its sole cost to rebuild, repair and/or replace its signs, fixtures and equipment.

13.2 Termination Rights. If any damage or destruction to the Project cannot in Owner's reasonable judgment be repaired within one-hundred-eighty (180) days following the date of damage, Owner shall give Manager written notice thereof within thirty (30) days after the date of damage, and either Manager or Owner may terminate this Agreement by delivering written notice to the other within thirty (30) days after the date Manager receives Owner's notice. If (a) the cost to repair any damage to the Project exceeds fifty percent (50%) of the insurable replacement cost of the Project, or (b) a portion of the damage is uninsurable through the insurance policies required under this Agreement, or if the amount of insurance available to Owner is inadequate to restore the Project and Owner's FF&E to substantially their condition prior to the Casualty and Owner elects not to restore, then Owner shall give Manager written notice of the Owner's election within thirty (30) days after the date of such damage or destruction and thereafter either party shall have the right to terminate this Agreement by written notice to the other party given within sixty (60) days after the date Manager receives such notice of Owner's election not to repair such damage. If the cost to repair damage to the Project that occurs during the last twenty-four (24) months of the Initial Term or any Renewal Term exceeds thirty percent (30%) of the insurable replacement cost of the Project, either party shall have the option to terminate this Agreement by giving written notice to the other party within sixty (60)

days after the date of damage, unless Manager exercises an available Renewal Term option within ten (10) business days after receiving Owner's notice. If this Agreement terminates pursuant to this <u>Section 13</u>, Rent Installments shall be prorated as of the date of damage. Manager shall collect and remit to Owner the rent owed by Student Tenants to said date and Owner shall remit to Manager for payment to Student Tenants and Rent Installments which Owner has received for any period after said date.

14. <u>Condemnation</u>.

General Rights Upon Condemnation. If all or any portion of the Project 14.1 are taken under the power of eminent domain exercised by any governmental or quasigovernmental authority (a "Condemning Authority"), or are conveyed in lieu thereof (a "Condemnation"), this Agreement shall automatically terminate as to the portion condemned as of the date that possession and use are transferred to the Condemning Authority ("Condemnation Date"). The Owner shall have the exclusive right to grant possession and use to the Condemning Authority. Owner shall give Manager written notice (a "Condemnation Notice") specifying the extent of the taking and the anticipated Condemnation Date promptly after it receives such information. If more than thirty percent (30%) of the Project or more than thirty percent (30%) of the Residence Units are taken by Condemnation, or if as a result of Condemnation, Manager's access to the Project is terminated or materially and substantially interfered with, then in any of such cases Owner shall have the right to terminate this Agreement by giving written notice to Manager within sixty (60) days after the date of the Condemnation Notice. If more than thirty percent (30%) of the Building is taken by Condemnation and Owner elects not to repair, restore, alter or reconstruct the same, either party may elect to terminate this Agreement by giving written notice within sixty (60) days after the date of the Condemnation Notice. Any such termination shall be effective as of the Condemnation Date. If this Agreement is terminated pursuant to this Section 14.1, Owner and Manager shall be released from any liability arising after the termination date, but all Rent Installments and all other amounts payable under this Agreement shall be prorated and paid or remitted as of the termination date in the same manner as Rent Installments are to be prorated and paid or remitted under Section 13.2. If this Agreement is not terminated as provided for hereunder, Rent Installments, and all other amounts payable under this Agreement shall be reduced pro-rata based upon the percentage of the Project's net rentable floor area condemned, or if no floor area of the Project is taken, then the reduction shall be an equitable reduction reflecting the degree to which the Manager has been adversely affected by the taking. Owner shall make such repairs and alterations necessary to restore the portion of the Project not condemned to a condition reasonably satisfactory for Manager's use; provided that (i) at least three (3) years of the Initial Term or any Renewal Term remain, and (ii) that Owner's obligation shall be limited to the amount of the Condemnation Award (as defined below) available therefor after deducting all costs of obtaining the Condemnation Award. Within sixty (60) days after the amount of the Condemnation Award available for restoration is known, Owner shall give Manager reasonable notice of the repairs and alterations Owner anticipates making to restore the portion of the Project not condemned. If Manager reasonably determines that the anticipated repairs and alterations are insufficient to restore such portion of the Project to a condition reasonably satisfactory for Manager's use, Manager shall have the right to terminate this Agreement by giving notice to Owner within ten (10) days after receiving Owner's notice describing the repairs and alterations.

14.2 <u>Award</u>. Owner shall be entitled to receive the entire award in any Condemnation ("Condemnation Award"), including any award for the value of any unexpired portion of the Term of this Agreement. Manager shall have a right to claim and recover from the Condemning Authority, but not from Owner, such compensation as may be separately recoverable by Manager. Manager shall be entitled to claim an award in a condemnation proceeding for any loss resulting from loss of housing contracts with students, Manager's relocation expenses and/or the taking of Manager's personal property, to the extent such claims are recoverable.

15. Assignment and Subletting.

15.1 <u>Restriction on Transfer</u>. Manager shall not sublet all or any portion of the Project, and not assign, mortgage, pledge or otherwise encumber or transfer this Agreement, or any interest herein, or in any manner assign, mortgage, pledge, or otherwise encumber or transfer its interest or estate in all or any portion of the Project (each a "Transfer"), and each such party a ("Transferee") without obtaining Owner's prior written consent in each and every instance, which consent shall be in Owner's sole discretion to condition, grant or deny, specifically including Owner's right to condition its consent upon Manager agreeing to remain liable to Owner under the terms of this Agreement. This preclusion, however, shall not apply to Manager merging with or otherwise becoming a part of any other institution for higher education.

15.2 <u>**Transfer Requirements.**</u> In the event Owner consents to a Transfer by Manager as contemplated in <u>Section 15.1</u> above, such Transfer shall, at a minimum, comply with the following requirements:

(a) Any Transfer shall operate to transfer to the Transferee all of Manager's right, title and interest in this Agreement and all of Manager's estate or interest in the Project. Manager shall remain liable to Owner for all obligations under this Agreement, notwithstanding any Transfer of this Agreement.

(b) Any Transferee shall assume, by written, recordable instrument, all of Manager's obligations under this Agreement arising and related to a period subsequent to the date of Transfer. Such assumption agreement shall state that the same is made by the Transferee for the express benefit of Owner as a third party beneficiary. A copy of the agreement evidencing the Transfer, in a form and content satisfactory to Owner, fully executed and acknowledged by Transferee, together with a certified copy of a properly executed corporate resolution (if the Transferee is a corporation) authorizing the execution and delivery of such Transfer agreement, shall be sent to Owner at least ten (10) days prior the effective date of the Transfer.

(c) Any Transfer shall be subject to all the provisions, covenants and conditions of this Agreement. Manager and any Transferee shall remain liable for all obligations under this Agreement, as it may be amended from time to time, without notice to the Transferee.

(d) Manager shall reimburse Owner for any and all reasonable costs incurred by Owner, including reasonable Attorneys' Fees paid or payable to outside counsel, occasioned by such Transfer.

15.3 <u>Restriction From Further Assignment</u>. Notwithstanding any consent by Owner to any Transfer, no Transferee shall Transfer all or any portion of the Project, and no assignee shall further assign its interest in this Agreement or its interest or estate in all or any portion of the Project, or lease all or any portion of the Project, without Owner's prior written consent in each and every instance.

16. <u>Signs</u>.

16.1 **Proprietary Marks**.

16.1.1 <u>Owner Marks</u>. Subject to <u>Section 16.2</u> below, Manager will operate the Project under the Owner's proprietary marks ("Owner's Marks"), which such Owner Marks shall be used in any Advertising (as defined below) by Manager. For the purposes of this Agreement, Owner hereby grants Manager a temporary, non-exclusive, royalty-free, fully paid-up Renewal Extended Terms. Any violation of this license by Manager shall deemed to be an Event of Default.

16.1.2 <u>Manager Marks</u>. For purposes of Advertising the Project, Manager hereby grants Owner a temporary, non-exclusive, royalty-free, fully paid-up license to use the Manager's proprietary marks ("Manager's Marks," and together with the Owner Marks, the "Marks") in accordance with the terms and conditions of this Agreement during the Initial Term and any Renewal Terms.

16.1.3 <u>Protection of Marks</u>. The parties hereto acknowledge and agree that protection of the Owner Marks and Manager Marks is important; accordingly Owner and Manager shall immediately notify the other party of any infringement or dilution of or challenge to the use of the respective Marks granted hereunder, and will not, absent a court order or the other party's prior written consent, communicate with any person regarding any such infringement, dilution, challenge or claim. The parties covenant to cooperate to the fullest extent possible in any action taken by a party to protect and/or defend their respective Mark from any such infringement, dilution, challenge or claim.

16.2 <u>**Co-Branding**</u>. Owner and Manager shall each be permitted to advertise and promote the Residences strictly in accordance with the following:

16.2.1 <u>Owner Obligations</u>. Except in connection with the Retail Spaces, Owner shall, during the Initial Term and any Renewal Terms, prominently reference the Manager Marks in any advertisements, promotional materials, web sites, etc. ("Advertising") undertaken by Owner with regard to the Project and the Residences. **16.2.2** <u>Manager Obligations</u>. During the Initial Term and any Renewal Terms, Manager shall prominently reference the Owner Marks in any Advertising, including without limitation any materials associated with the Residence Program.

16.3 Signage.

Manager may, at its sole cost, erect signs on the interior or the exterior of the Project Building, provided that such sign or signs (a) do not cause any structural or other damage to the Project Building; (b) do not violate applicable governmental laws, ordinances, rules or regulations; (c) do not violate any existing restrictions affecting the Project; and (d) are approved in advance by Owner in accordance with Owner's signage criteria attached hereto as <u>Exhibit I</u> and incorporated herein by this reference ("**Signage Criteria**"). Manager, upon vacating the Residences and the Project generally or removing or altering its signs for any reason, shall repair, paint and/or replace the Building surfaces where its signs were attached. Manager shall not disturb signs placed by Owner on or in any part of the Building.

17. <u>Miscellaneous Provisions</u>.

17.1 Access to Residences.

17.1.1 Manager Access. For purposes of the Management Responsibility, Manager shall have free access to the Project twenty-four (24) hours per day, seven (7) days per week basis. Owner and its authorized representatives shall have the right at any and all times to enter any part of the Project for the purpose of inspecting the same, making any necessary changes or modifications to Building systems, making any repairs required or permitted by the terms of this Agreement or conducting any reasonable test or environmental audit of the Project or Manager's operation or use of the Project to determine Manager's compliance with this Agreement. The provisions of this Section 17.1.2 shall not be construed to impose upon Owner any obligations not otherwise explicitly provided for in this Agreement. Notwithstanding the foregoing, in the event Owner desires to enter any of the Residences pursuant to this Section 17.1.1, Owner shall provide Manager with notice of the same and Manager shall within three (3) business days provide notice to the Student Tenants residing in the applicable Residences of Owner's intended entry thereon. In the event any consent is required to be obtained from any Student Tenants pursuant to Owner's entry into any of the Residence, Manager shall obtain such necessary consents of the Student Tenants within not more than an additional three (3) business days. The consent requirement set out above shall not apply in the case of an emergency when Owner's immediate entry is reasonably necessary to prevent damage to deterioration of the Project. In the event of any Owner entry into the Residences, Owner shall at all times be accompanied by a representative of Manager. Owner may, during the progress of any such work in the Project and/or Property, take all material in, to and upon said Project and/or Property that may be required therefore, and store any necessary materials, tools and equipment at the Project and/or Property, without the same constituting an eviction of Manager or any Student Tenants in whole or in part (and any and all Rent Installments reserved shall not abate pursuant to exercise of any of Owner's rights hereunder, including but not limited to any loss or interruption of Manager's business or otherwise and Manager shall have no claim for damages); provided that such storage is done in a safe and secure manner and does not

otherwise unreasonably interfere in any material respect with Manager's and the Permitted Use by Student Tenant's of the Project. Owner shall not be liable for, and the obligations of Manager shall not be affected by, inconvenience, disturbance, loss of business or other damage caused by Owner in connection therewith, provided that Owner makes reasonable efforts to avoid interfering with the conduct of Manager's business, except to the extent caused by Owner's negligence or willful misconduct. The provisions of this <u>Section 17.1</u>, and any other provisions in this Agreement that require Owner notice to Student Tenant and/or mandate Student Tenant consent to Owner's entry shall be included as provisions of the Rental Agreement.

17.1.2 <u>Exhibition of Residences</u>. Owner shall have the right during normal business hours at any time during the Initial Term and any Renewal Term, following reasonable notice as defined in <u>Section 17.1</u>, to enter the Project for the purpose of exhibiting the same for any purpose in Owner's reasonable discretion.

17.3 Estoppel Certificates. Owner and Manager shall, each without charge at any time and from time to time, within ten (10) business days after written request by the other party, execute and deliver to the requesting party or any person whom the requesting party may designate, an estoppel certificate certifying: (a) that each of the terms of this Agreement is unmodified and in full force and effect (or, if there have been modifications, that the same is in full force and effect, as modified, and stating the modifications); (b) the dates to which the Rent Installments have been paid in advance; (c) whether or not there are then existing any breaches or defaults by such party or the other party known by such party and specifying such breach or default, if any; (d) whether or not there are then any setoffs or defenses against the enforcement of any term of this Agreement (and, if so, specifying the same and the steps being taken to remedy the same); and (e) such other statements or certificates as Owner, Manager or any Mortgagee may reasonably request. It is the intention of the parties that any statement delivered pursuant to this <u>Section 17.3</u> may be relied upon by any persons dealing with Owner, Manager or the Residences.

17.4 <u>Notices</u>. All notices, demands and requests which may be or are required to be given by either party to the other shall be in writing and shall be sent by United States certified mail, postage prepaid, or by a nationally recognized independent overnight courier service, addressed to Owner at 9 South Washington Street, Suite 600, Spokane, Washington 99201 with a copy to Shaun A. Gill, Cairncross & Hempelmann, P.S., 524 Second Avenue, Suite 500, Seattle, WA 98104, and addressed to Lewis-Clark State College, 500 8th Avenue, Lewiston, Idaho, 83501, Attention Vice-President for Administrative Services, with a copy to attorney, Robert Brown, 321 13th Street, Lewiston, Idaho, 83501, and any other address which is not a post office box as either party may from time to time designate by written notice to the other. Any notice, demand or request which is given as described shall be deemed to be given on the business day following the date it is delivered to a courier or three (3) days after it is deposited in the United States mail.

17.5 <u>Annual Statements</u>. Upon request by Owner, Manager shall promptly deliver to Owner a copy of Manager's rent roll (which shall not identify the residents by name consistent with the Family Educational Right and Privacy Act, 20 U.S.C. 1232g) and other information related to the operation of the Project and the Building. Such records may contain

confidential or proprietary information and shall only be used by the Owner to communicate the current financial status of the Project and/or Property and shall, except as follows, be kept confidential by the Owner. Owner may further disclose such financial statement(s) to its mortgage lender (or potential lender), accountants or other financial advisors or analysts, or any potential purchaser of the Project, provided that they have also agreed to the foregoing limit on use and confidentiality.

17.6 Owner's Continuing Obligations. The term "Owner," as used in this Agreement so far as obligations on the part of Owner are concerned, shall mean only the Owner or Owners of fee title to the Property at the time in question, and upon any transfer, the then grantor shall, subject to the conditions set forth below, be automatically freed and relieved after the date of such transfer of all liability for the performance of any obligation on the part of the Owner contained in this Agreement thereafter to be performed. As conditions precedent to the foregoing relief from liability: (i) any funds in the hands of such Owner or the then grantor at the time of such transfer, in which Manager has an interest, shall be turned over to the grantee; (ii) any amount then due and payable to Manager by Owner or the then grantor under any provision of this Agreement shall be paid to Manager; and (iii) the grantee shall agree, in writing, for the benefit of Manager that it will assume, pay and perform all payments, terms, covenants and conditions of Owner under this Agreement. The Owner's obligations contained in this Agreement shall, subject to the aforesaid, be binding on Owner's successors and assigns, during and in respect of their respective successive periods of ownership. Nothing herein contained shall be construed as relieving Owner of any of its obligations under this Agreement, or releasing Owner from any obligation to complete the cure of any breach by Owner during the period of its ownership of the Project.

17.7 <u>Intentionally Deleted</u>.

17.8 <u>Severability</u>. If any covenant, condition, provision, term or agreement of this Agreement shall, to any extent, be held invalid or unenforceable, the remaining covenants, conditions, provisions, terms and agreements of this Agreement shall not be affected thereby, but each covenant, condition, provision, term or agreement of this Agreement shall be valid and in force to the fullest extent permitted by law.

17.9 <u>Successors and Assigns</u>. Subject to provisions of this Agreement limiting assignment or sublease by Manager, the covenants contained in this Agreement shall bind and inure to the benefit of Owner, its successors and assigns, and Manager and its permitted successors and assigns.

17.10 <u>Relationship of Parties</u>. This Agreement creates the relationship of principal and agent and is not expressly or impliedly intended to create any partnership, joint venture, association or any other relationship between Owner and Manager.

17.11 <u>Prior Agreements</u>. This Agreement, together with the attached exhibits and the written agreements concurrently or hereafter executed and/or delivered pursuant to or in connection with this Agreement, embody the entire agreement between the parties relating to the subject matter hereof, and supersede all prior agreements and understandings between the Owner

and Manager, if any, relating to the subject matter hereof. Manager acknowledges that neither Owner nor any agent of Owner has made any representation or warranty not contained in this Agreement with respect to the Residences and the Project generally, the Building or the suitability or fitness of either for the conduct of Manager's business or operations or for any other purpose.

17.12 <u>No Waiver</u>. No failure by Owner or by Manager to insist upon the performance of any of the terms of this Agreement or to exercise any right or remedy available for a breach thereof, and no acceptance by Owner of full or partial rent from Manager or any third party during the continuance of any such breach, shall constitute a waiver of any such breach or of any of the terms of this Agreement. None of the terms of this Agreement to be kept, observed or performed by Owner or by Manager, and no breach thereof, shall be waived, altered or modified except by a written instrument executed by the waiving party. No express waiver shall waive any default other than the default specified in the express waiver. An express waiver waives the specified default only for the time and to the extent therein stated.

17.13 <u>Owner's Liability Limited</u>. Manager recognizes that Owner is a limited liability company. Manager expressly agrees, anything herein to the contrary notwithstanding, that each and all of the representations and agreements made by Owner are intended to bind only Owner's interest in the Project, and the rents and profits therefrom. No personal liability or personal responsibility is assumed by, nor shall at any time be asserted or enforced against, any of the members, managers, shareholders, partners, directors, officers, employees or agents of Owner on account of any agreements of Owner contained in this Agreement.

17.14 Intentionally Deleted.

17.15 <u>Force Majeure</u>. Whenever a period of time is herein prescribed for action to be taken by either party, said party shall not be liable or responsible for, and there shall be excluded from the computation of any such period of time, any delays due to ("Force Majeure"): strikes, riots, terrorist acts, acts of God, delay caused by the failure of a governmental agency to issue a building or occupancy permit despite diligent pursuit thereof, shortages of labor or materials because of priority or similar regulations or order of any governmental or regulatory body, war, or any other causes of any kind which are beyond the reasonable control of said party. Lack of funds or inability to obtain financing shall not be an event of Force Majeure.

17.16 <u>Survival</u>. All obligations (together with interest on monetary obligations at the Agreed Rate) accruing before expiration of the Initial Term or any Renewal Terms shall survive the expiration or other termination of this Agreement. Additionally, <u>Section 4.1</u> (Manager's Compensation), <u>Section 9</u> (Compliance with Laws), and <u>Section 10</u> (General Liabilities), will survive the expiration or other termination of this Agreement.

17.17 <u>**Quiet Enjoyment.**</u> Owner agrees that to its actual knowledge it is the fee simple owner and record title holder of all of the Project; that it has the full right, power and authority to execute this Agreement; that Manager, upon paying the Rent Installments and keeping all of the covenants of this Agreement, shall have the right (and all sublessees and all

other permitted transferees shall have the right) to lawfully and quietly hold, occupy, and enjoy the Project during the Initial Term or any Renewal Terms without any interference from Owner or anyone claiming by, from or under Owner.

17.18 <u>Brokers; Agency Disclosure</u>. Owner and Manager each represent to the other that it has not been represented by any brokers or finders in connection with this Agreement. Owner and Manager shall each defend and hold harmless the other against all claims of brokers, finders or any like third party claiming any right to a commission or compensation in connection with this Agreement by or through the acts of that party, to the extent permitted by law.

17.19 <u>Governing Law</u>. This Agreement shall be governed by the laws of the state of Idaho.

17.20 <u>Construction</u>. Words and phrases used in the singular shall be deemed to include the plural and vice versa. Nouns and pronouns used in any particular gender shall be deemed to include any other gender. When the word "including" is used in this Agreement, it shall mean "including, but not limited to." Whenever words such as "herein," "hereunder," etc., are used in this Agreement, they shall mean and refer to this Agreement in its entirety and not to any specific section, paragraph or other part of this Agreement. The word "person" includes any natural person, corporation, firm, partnership, limited partnership, limited liability company, trust, estate, unincorporated organization, or other legal or business entity, however designated or constituted. "Business day" means days when national banks are open in Lewiston, Idaho. The caption of each section of this Agreement is for convenience of reference only, and in no way defines, limits or describes the scope or intent of such section.

17.21 <u>**Time Is of the Essence.**</u> Time is of the essence with respect to the performance of every provision of this Agreement in which time of performance is a factor.

17.22 <u>No Oral Amendments</u>. This Agreement may be modified or amended only by an agreement in writing signed by the parties hereto. No receipt of money by Owner from Manager or any other person after termination of this Agreement, the service of any notice, the commencement of any suit or final judgment for possession of the Residences and the Project generally, shall reinstate, continue or extend the Term of this Agreement, affect any such notice, demand or suit, or imply consent for any action for which Owner's consent is required, unless specifically agreed to in writing by Owner. Any amounts received by Owner may be allocated to any specific amounts due from Manager to Owner as Owner determines.

17.23 <u>Exhibits</u>. All exhibits now or hereafter attached to this Agreement are incorporated into this Agreement in full by this reference. Each party agrees to perform any obligations to be performed by it pursuant to the provisions of all such exhibits.

17.24 <u>Recording</u>. Neither party shall record this Agreement. Promptly after request of either party, the parties hereto will execute and deliver to each other a memorandum of Agreement, in recordable form, setting forth (a) the date of this Agreement; (b) the parties to this Agreement; (c) the Commencement Date and Expiration Date of the Initial Term; and

(d) such other matters as are reasonably requested by either party to be stated therein. Either party may then record the memorandum of Agreement.

17.25 <u>Limitation on Right of Offset</u>. Except as herein provided, neither party shall have the right to offset and shall not reduce any amount owing to the other party under this Agreement, notwithstanding the existence of a bona-fide dispute between Owner and Manager, except that, if a party ("Delinquent Party") fails to pay any amount owing to the other party ("Non-Delinquent Party") on or before its due date on at least two occasions during any twelve (12) month period and the Non-Delinquent Party gives written notice after the second delinquency that it intends to exercise its right of offset, then, for a period of twelve (12) months following the second delinquency, any restrictions against offset contained in this Agreement shall not apply to Non-Delinquent Party and the Non-Delinquent Party may exercise against the Delinquent Party all rights of offset available at law or equity.

17.26 <u>Manager's Furniture and Equipment</u>. Only Owner and/or Manager shall have the right to transport into and maintain in the Residences furniture and equipment, whether owned or leased by Owner and/or Manager. Manager hereby covenants to include any and all relevant provisions in the Rental Agreement to effect the provisions of this <u>Section 17.26</u>. This section does not apply to the items of furniture and equipment described on Exhibit J.

17.27 <u>Public Records</u>. Owner and Manager acknowledge that all records, including but not limited to, design, construction, and Agreement documents will constitute public records. The Owner shall provide the Manager with a copy of all documents related to the design, construction and management of the Project.

17.28 <u>**Rules and Regulations.**</u> Manager shall perform, observe and comply with all reasonable nondiscriminatory rules and regulations established by Owner for the Residences and the Project generally from time to time.

17.29 <u>Reserved</u>.

17.30 <u>Attorneys' Fees.</u> If either party brings any lawsuit or arbitration proceeding to enforce or declare any rights under this agreement, including any appeal thereof, the party deemed as the prevailing party by the court or arbitrator hearing the matter shall be entitled to collect from the other party reasonable "Attorney's Fees". The Attorney's Fees award shall not be computed in accordance with any court schedule, but shall be such to fully reimburse all Attorney's Fees reasonably incurred by the prevailing party.

18. <u>Building Names.</u>

Owner shall have the exclusive right, in its sole discretion and from time to time, to name the Building and any part of the Residences and the Project generally, which such names shall be included in the Owner Marks.

19. <u>Authority of Manager</u>.

Manager warrants that execution of this Agreement by Manager has been duly authorized by Manager's board of trustees; and no further action, approval, legislative action or approval, or administrative action or approval, or any other action or approval of any kind, is necessary on the part of any branch or agency of the State of Idaho or on the part of Manager, to make this Agreement fully and completely binding upon Manager in accordance with its terms. If Manager does not have authority to enter into this Agreement in accordance with its terms, Manager shall be liable to Owner for all costs and damages arising from such breach of warranty and lack of authority described in this Section 19, and Owner shall have all remedies available to it as described for a Manager Event of Default as described in Section 11 of this Agreement.

20. <u>Future Modifications</u>.

This Agreement represents a new relationship between the parties and both parties recognize that mutually agreeable future modifications of some of its provisions because of unanticipated conditions, events or results, may be beneficial to both parties. Therefore, if a party believes that a provision of this Agreement should be modified, it will notify the other party of the proposed modification and the reason for that change. The parties will then discuss and consider the proposed modification in an effort to arrive at a mutually agreeable solution. However, unless and until the parties both execute a properly authorized and executed written amendment to this Agreement, nothing herein contained shall preclude a party from exercising all rights and remedies granted to that party under this Agreement, as validly amended, particularly including, but not limited to, the right to terminate this Agreement as provided herein.

21. <u>Termination for Failure of Appropriations</u>.

Owner understands that Manager and State of Idaho are governmental entities. Therefore, to the extent, if any, that Manager's obligations hereunder are payable from appropriated funds, this Agreement shall in no way or manner be construed so as to bind or obligate the Manager or the State of Idaho beyond the term of any particular appropriation of funds by the State's Legislature as may exist from time to time. The Manager and State each reserves the right to terminate this Agreement in whole or in part if, in its sole judgment, the Legislature of the State of Idaho fails, neglects, or refuses to appropriate sufficient funds as may be required for the Manager or State to continue such payments, or requires any return or "giveback" of funds required for the Manager to continue payments, or if the Executive Branch mandates any cuts or holdbacks in spending which results in insufficient funds. All affected future rights and liabilities of the parties hereto shall thereupon cease within ten (10) calendar days after notice to the Owner.

22. <u>State Board Approval</u>. This Agreement is subject to approval by the Idaho State Board of Education and shall not be effective or binding upon either of the parties until such approval is given. Once the necessary approval is given, this Agreement shall be effective from and after its Effective Date.

23. <u>Counterparts</u>. This Agreement may be executed in two or more counterparts, each of which shall be deemed an original, but all of which counterparts shall together constitute a single agreement.

[SIGNATURE PAGE FOLLOWS]

IN WITNESS WHEREOF, each of the parties hereto has caused this Agreement to be duly executed as of the day and year first above written.

OWNER:

COLLEGE TOWN DEVELOPMENT IDAHO LLC, a Washington limited liability company

- By: COLLEGE TOWN HOUSING MANAGEMENT LLC, a Washington limited liability company
- Its: Managing Member

By:

Robert C. Brewster, Manager/Member

MANAGER:

STATE OF IDAHO, ACTING BY AND THROUGH THE STATE OF IDAHO BOARD OF EDUCATION AS BOARD OF TRUSTEES FOR LEWIS-CLARK STATE COLLEGE

By: ___

Chet G. Herbst, Vice President for Administrative Services Lewis-Clark State College STATE OF WASHINGTON)) ss. COUNTY OF _____)

I certify that I know or have satisfactory evidence that **Robert C. Brewster, Jr.** is the person who appeared before me, and said person acknowledged that he signed this instrument, on oath stated that he was authorized to execute the instrument and acknowledged it as the manager/member of COLLEGE TOWN HOUSING MANAGEMENT LLC, a Washington limited liability company, managing member of COLLEGE TOWN DEVELOPMENT IDAHO LLC, a Washington limited liability company, to be the free and voluntary act of such party for the uses and purposes mentioned in the instrument.

Dated: _____, 2006.

(Signature of Notary Public)

(Printed Name of Notary Public) My Appointment expires

STATE OF IDAHO)) ss. COUNTY OF NEZ PERCE)

On this _____ day of _____ 2006, before me, the undersigned, a Notary Public in and for the State of Idaho, duly commissioned and sworn, personally appeared, CHET G. HERBST, known to me to be the Vice-President for Administrative Services of Lewis-Clark State College, whose name is subscribed to foregoing agreement and acknowledged to me that he executed the same for and on behalf of the Idaho State Board of Education.

IN WITNESS WHEREOF, I have hereunto set my hand and affixed my seal the day and year first above written.

(Signature of Notary Public)

(Printed Name of Notary Public) My Appointment expires _____

EXHIBIT A LEGAL DESCRIPTION

SITUATE IN NEZ PERCE COUNTY, STATE OF IDAHO TO WIT:

The East part of Lot 2, Block 29, of the City of Lewiston, according to the recorded official plat thereof now on file in the office of the County Recorder of Nez Perce County, Idaho, described as follows:

Commencing at a point on the South line of "E" Street, at the Northeast corner of said Lot 2, Block 29; thence West on the South line of "E" Street a distance of 28 feet; thence Southerly parallel with and 28 feet from the East line of Lot 2, Block 29, of said City of Lewiston, to the North line of Lot 4, of Acres known as Thatcher place; thence Easterly on the said North line of Lot 4 of Acres of said City of Lewiston to the Southeast corner of said Lot 2, Block 29 of said City of Lewiston; thence North on said East line of said Lot 2, Block 29, to the PLACE OF BEGINNING.

AND ALSO:

Part of Lot 2, Block 29, of the City of Lewiston, Nez Perce County, State of Idaho, according to the official plat of said City now on file in the office of the County recorder, of Nez Perce County, State of Idaho, commencing at a point on the South line of "E" Street at the Northeast corner of Lot owned by Eugene L. Alford in Lot 2, Block 29, of said City as disclosed by that certain Decree of Distribution in the matter of the Estate of A. H. Alford, deceased, recorded in Book 6 of Orders at page 173; thence East on said South line of "E" Street a distance of 32 feet to a point; thence Southerly parallel with and 28 feet from the East line of said Lot 2, Block 29 of said City of Lewiston to the North line of Lot 4 of Acres known as the Thatcher Place; thence Westerly on said North line of said Lot 4 of Acres to the Southeast corner of said Alford lot; thence Northerly on the East line of said Alford lot to the PLACE OF BEGINNING.

AND ALSO:

Lot 1, Block 29, ORIGINAL CITY OF LEWISTON, Idaho, according to the recorded plat thereof, ALSO; Commencing at the Southeast corner of Lot 1, Block 29 in the City of Lewiston, County of Nez Perce, State of Idaho; thence Easterly at right angles to the East line of said Lot 1 a distance of 16 feet; thence Northerly and parallel with the said East line of said Lot a distance of 30 feet; thence Northerly a distance of 94.5 feet to a point on said East line of said Lot 1, which point is also the present Southeast corner of the building known as the Weisgerber Building; thence Southerly along the East line of said Lot 1 in said Block 29 to the PLACE OF BEGINNING, said strip of land begin formerly a part of the 5th Street in said City of Lewiston.

EXHIBIT B PLANS AND SPECIFICATIONS





EXHIBIT C OWNER FF&E SCHEDULE

2

.
Clearwater Hall FFE Schedule

	T							_		-1									_					Т													
			-	-	4 	` ──	_	_						r—	-		ں _	, 	_						_		_			~	د 						Floor
Totals	Lounge	418	417	416	4 I J	41.5 C [+	412	4 1 1	411	410	Lounge	318	317	316	315	313	312	311	310	308	305	302	200	Lounge	817	217	216	512	213	212	211	210	207	206	205	204	Suite
117		4	4	. u) 4	4	2	- +	<u>,</u>	-		4	4	3	4	4	ω	4	-	4			+ ار		4		4	4	4		4	-	4				Beds
117		4	4	. 3	4	4	<u>،</u> ر	+ ,	× ,	-		4	4	ω	4	4	ω	4		4	. v		4		₽	. 4	4	4	4	2	4		4	4	S	s	Desks
117		4	4	3	4	. 4		4	<u> </u>	-		4	4	З	4	4	ω	4	.	4	. 0	ι u	4		4	4	4	4	4	2	4	1	4	4		5	Desk Chairs
117		4	4	ω	4	4	. ω	4		-		4	4	3	4	4	ω	4	-	4	S	S	4		4	4	4	4	4	2	4		4	4	5	5	s Wardrobes
27	1	1	1		1			-	•	-	<u> </u>					1		<u> </u>		1	-	-			-	,	1	1	-		1		1	1	1	1	Couch
8				1			1		_	-							1		<u> </u>													1					Loveseat
6	2									1	<u>،</u>													2											_		Chai
32		1	ľ	1	1	1	1	1		-	-			-	1	1	I	1	1	1	1	1			1	1	1	1	1	1	1		1		1	1	Desk Chairs Wardrobes Couch Loveseat Chair Dining Table
64		2	2	2	2	2	2	2	2)	~	J 1	- ب	2	2	2	2	2	2	2	2	2	2		2	2 1 1	2	2	2	2	2	2	2	2	2	2	e Dining Chairs
32		P-4		1-1	1	1	1	1	1			-						1-1		1		1	1		1	1	-				Ţ	,				0	Refrigerators
32	,		,				-1	Ι	1				-		_ ,	,	,	_			1	1	-			, 1			<u></u>			,	,		_ _		Microwawes

EXHIBIT D

"Intentionally Deleted"

BAHR - SECTION II

x

EXHIBIT E MONTHLY REMITTANCE

Lewis-Clark STATE

Connecting Learning to Life

College Town Monthly Remittance Form

To:	Name Mailing addr	ess
For:	The Month of	September
Date:	September 1	l, 200x
Check #	999999	

Headcount:	XX

Rate: \$xxx.xx

Total \$xx,xxx.xx

EXHIBIT F ANNUAL BUDGET

Lewis-Clark State College College Town Residence Hall Lease Year Budget

Revenues: Student Housing Charges Management Fee From Owner	\$XXXXXX XXXX
Total Revenues	\$XXXXXX
Expenditures:	
Rental Installments to Owners	\$XXXXXX
Payments to Owners for RA rooms	XXXXX
Custodial Supplies	XXXXX
Maintenance	XXXX
Trash Removal	XXXX
FF&E Repairs	XXXX
Miscellaneous	XXXX
Total Expenditures	XXXXXX
Net Operating Results	<u>\$ XXX</u>

EXHIBIT G RESIDENCE HALL TERMS AND CONDITIONS

×.

RESIDENCE HALL TERMS AND CONDITIONS

I. HOUSING DEPOSIT

A. A \$100 deposit must accompany the application for student housing. Deposit and contract must be received by before a room assignment will be made. Deposit will not be applied toward room or meal plan charges. When the student leaves the residence hall, the deposit will be refunded less any charges for loss and/or damage to college property or payments for room and board, student fees, other delinquent college obligations or premature termination of contract.

II. ACTIVITY FEE

A non-refundable activity fee of \$15 per semester is required of all residence hall students. This fee will be collected when the student pays for room and board.

III. ELIGIBILITY

An individual must be admitted by the College. A student must also be registered for and maintain a minimum of 12 credit hours at LCSC during the contracted period.

IV. PERIOD OF CONTRACT

A student may contract to live in a LCSC Residence Hall for the full academic year or one semester (See Section XVIII. Vacation Periods, for exception). Contractual period begins when LCSC receives a completed housing contract and ends as stated in the contract. All academic year contracts automatically expire on the final day of spring semester exams. Housing for the following year requires completion of a new contract

V. ROOM AND BOARD PAYMENT/COLLECTION

A. Room and board costs are due on the second official day of the Fall/Spring semester.

B. Late fees will be assessed beginning on the third day of the semester to accounts that are not paid in full or have a signed payment plan contract.

C. Failure to pay room and board costs may result in the withholding of meals AND the termination of this contract by the college

D. Delinquent accounts will be referred to a commercial collection agency. In addition to any balance owed to the college, you are responsible for the costs of collection, including interest, penalties, collection agency fees, court costs and attorney fees.

VI. CANCELLATION PRIOR to OCCUPANCY

All cancellations must be received in writing at the Office of Residence Life on the date designated below.

Fall Semester

Cancellation received by June 15	Full Refund
Cancellation received by June 16-July 15	\$50.00
Cancellation received after July 15	No Refund
Spring Semester	

Cancellation received by December 15 Full Refund Cancellation received by Dec. 16-Dec. 31 \$50.00 VII. OCCUPANCY

Occupancy occurs when student has been issued a key to his/ her room and/or completed and signed the room inventory form. Actual physical occupancy of the room by student and/ or placement of student's possessions within the room is not necessary to constitute occupancy.

VIII. TERMINATION OF CONTRACT BY COLLEGE The College may terminate the contract under the following conditions:

A. Student fails to comply with the rules and regulations set forth in this Contract, Student Code of Conduct, the Residence Hall Handbook, or other College policies, or any applicable, local, state or federal law

B. Student fails to pay room and board fees when due.

C. Student misrepresents information on this Contract or other Lewis-Clark State College documents.

D. Student breaches, violates, fails to perform or otherwise is in default of any of the terms and conditions of this Contract. The college specifically reserves the right to immediately remove any individual from the Residence Hall premises if the College, in its sole discretion, determines that the individual presents an immediate danger to himself/herself or others.

IX. TERMINATION OF CONTRACT (AFTER OCCU-PANCY

Students who wish to terminate this contract prior to the end of the contract period are subject to the following terms and conditions (in all cases, the deposit will be forfeited):

A. Under all conditions a Petition for Contract Release must be completed with the Office of Residence Life This includes students who are getting married, transferring, or withdrawing from the College. Approval or non-approval of the petition is contingent upon meeting the criteria set forth in the petition. Termination of the contract is not guaranteed.

B. A student whose petition is denied shall be obligated for the full amount of the room contract

C. Students who are required to leave the residence hall due to disciplinary action are still responsible for the entire semester or academic year room rent

D. Formal check-out with a staff member is required. X. ROOM REFUND SCHEDULE

This refund schedule is limited to Clark Hall and Parrish House for the 2006-2007 academic year. It does not include the suite-style residences. Refunds for approved housing terminations are based on the following schedule:

Refunds after date of official check-out for approved terminations of the contract, are subject to the following schedule;

FALL SEMESTER

August 25 - September 11	
September 12 - September 24	
September 25-October 23	40% refund
October 24 - November 23	
November 24 - end of semester	

SPRING SEMESTER

January 12 - February 12	60% refund
February 13 - March 12	
March 13 - April 16	
April 17 - end of semester	No refund

XI. GENERAL RESIDENCE HALL REGULATIONS

Occupancy in a hall is a privilege extended to the student by the College. The continuation of this privilege is dependent upon the student's reasonable and satisfactory personal conduct and observance of College regulations

No Alcohol: No alcohol is permitted in the residence halls. Students as individuals on or off campus are subject to the state law and city ordinances pertaining to the possession and consumption of alcoholic beverages and controlled substances. Students are also subject to the State Board of Education Alcohol policy, which is outlined in detail in the Residence Hall Handbook

Firearms/Weapons/Explosive Substances: Firearms, weapons, explosive substances & devices are prohibited in the residence halls and surrounding areas, including in private vehicles parked on College property. Weapons include, but are not limited to, knives, steel-tipped darts, pellet or BB guns, paintball guns, sling shots, arrows, axes, machetes, numchucks, swords, and throwing stars. Fireworks, gunpowder, explosives, gasoline and other hazardous chemicals or flammable liquids are also prohibited within the residence halls and surrounding areas.

Fire Safety Equipment: Students found to be tampering. damaging or in any way using fire equipment or safety apparatus for other than its intended use may be dismissed from the residence hall, fined and subject to further action by College authorities.

Group Billing: Where individual responsibility for damages to College property can be determined, the individual will be charged. Fines for common area damage will be assessed to the student(s) of the room/floor/hall involved. Willful and/or malicious damaging or stealing of College property is cause for contract cancellation under the terms listed in Section IX, Termination of Contract by College.

Occupants: The room is to be occupied only by the

Right of Entry: The College respects the student's right to privacy; however, the College reserves the right to enter a student's room at any time for health, safety, welfare, and maintenance purposes. Advance notice is given whenever possible. However, in cases where there is probable cause to believe that the student is or has been violating College and/ or residence hall regulations, the College can enter a student's room. This right is exercised with great discretion. XII. COLLEGE REGULATIONS

All students are held responsible for knowing the College regulations and information set forth in the College Catalog. Student Code of Conduct and Residence Hall Handbook. All College rules and regulations as well as residence hall regulations are specifically made a part of the contract by reference.

XIII. ROOM AND BOARD COSTS

All room and board costs, as approved by the State Board of Education, supersede all other information regarding costs. The College reserves the right to change rates. All parties to this agreement agree that room and board rates shall be those approved by the State Board of Education.

XIV. PERSONAL PROPERTY AND LIABILITY

The College shall not be liable for loss of or damage to student's personal property, wherever situated, due to fire, smoke, theft, water or any other casualty or cause. Students are encouraged to provide insurance against loss or damage to their personal property.

XV. FOOD SERVICE

A. Residents of Clark and Talkington Halls are required to maintain a meal plan during the rate period selected. B. Meal service is not provided during Thanksgiving, Winter, and Spring breaks.

C. Flex dollars Flex dollar plans are included in the suite-style residence halls.

XVI. ROOM ASSIGNMENT

A. The College reserves the right, at its sole discretion, to make assignments and re-assignments of space within the residence halls and to authorize or deny room and roommate changes and to relocate students to or from any room or residence hall

B. Students occupying a room as a single, but paying for a double will be required to move to a new room, acquire a roommate or pay for a double room as a single room, which will cost an additional \$325 per semester.

C. Any student who is assigned but fails to officially check into the hall by the first hour of classes each semester will forfeit the full amount of the deposit and the room will be reassigned.

XVII. VACATION PERIODS

The on-campus residence halls are officially closed between fall and spring semesters. If housing is offered during this time, students will be charged an additional room rent.

XVIII. ENFORCEMENT & RECOVERY OF ROOM

In the event the College should find it necessary to bring legal proceedings to recover possession of the student's room, or to enforce any of the terms of this contract, including the recovery of damages or charges, the student agrees to pay all reasonable attorney's fees, costs and expenses so incurred by the College.

XIX. CONTRACT TRANSFERS

Any student may, at any time, transfer their contract to another registered, continuing, off-campus student who has not otherwise contracted for housing with the College. Eligible parties must obtain approval for the transfer and complete the transfer process together at the Residence Life office. The physical transfer may take place only after proper approval has been granted and all applicable paperwork has been completed.

EXHIBIT H 2006-2007 HOUSING CONTRACT

.

BAHR - SECTION II

÷

TAB 10 Page 59

Residence Hall Contract

Instructions: Please read all sections of this contract before signing: **By signing below I acknowledge that I have read, understand, and agree to all terms and conditions of this contract.** Completion of this contract does not guarantee availability of a residence hall space. NO ALCOHOL is permitted in the residence halls.

Residence Hall Assignment:	Date Assigned:
To be assigned by Res	idence Hall Staff ONLY.
CONTACT INFORMATION Last/Family Name:	First/Given Name:
Permanent Mailing Address:	
City:State/Province:	Postal Code:
Country (if applicable): Hom	e Telephone (include area code):
Email Address:	Cell Phone No.:
Student ID#:	_
PERSONAL INFORMATION	
Date of Birth: Gend	er: 🗌 Male 📋 Female
Are you an LCSC athlete: 🗌 Yes 🗌 No Spor	t:
Please list any physical or medical conditions that shou or that might be necessary to know in case of an emerge CLASS: Freshman Sophomore Junior	gency:
PAYMENT OPTIONS LCSC Residence Halls are reserved by the <u>SEMESTER</u> or b need to pay the <u>full amount during the fall semester</u> . F SEMESTER YEAR Please indicate which semes	Please indicate your choice of payment below:
This section is for the Institute of Intensive English Lar need a residence hall room. IIE Student: □ Fall 1 □ Fall 2 □ Spring 1 □ Spring 2 □	
MEAL PLAN OPTIONS:	
Meal Plan A - Unlimited dinners and weekends through Meal Plan B - Unlimited dinners and weekends through Meal Plan C - Unlimited dinners and weekends through	out semester + \$350 Flex
Commuter Plan - Clearwater Hall, College Place, Parrisl	n House & CAMP House ONLY = \$500 Flex
Clark Hall residents must purchase Meal Plan A, B or C	
Rates are listed on the website at: <u>http://www.lcsc.ed</u>	u/reslife/Rates%202006-2007.htm
ROOMMATE PREFERENCE To accommodate your roommate preference, both you and yo same time period. I choose to live with:	ur roommate should send in your contracts within the

Would you be interested in living with someone of a different cultural or racial background?
YES No

Lewis-Clark

STATE

Please complete the Personal Preferences listed below. We attempt to match students with similar likes and dislikes to the best of our ability.

PERSONAL PREFERENCES

Words that Best D Outgoing/socia Value more qui	!		Intended Major
Sleeping Habits I am an early m I am a late nigh		rson.	Personal Interests
Preferred Room C Perfectly clean Reasonably stra Messy It doesn't matte	ight		Sports/Recreation Interests
Preferred Study Lo In my room In the library With music	ocation		Types of music I like
Smoking Do you smoke? Object to it?	Yes Yes	No No	
Drinking Do you drink alcoholi Object to it?	ic beverag Yes Yes	jes? No No	

In which Residence Hall would you prefer to live? Please rank 1- first choice; 2 - second choice, etc.

- College Place 4 person suites, 88 bed facility with shared kitchenette & living space (4 single bedrooms in each suite); located on 4th Street across from LCSC Bookstore.
- Clearwater Hall 4 person suites, 117 bed facility with shared kitchenette & living space (3-5 single bedrooms in each suite); Located in beautiful downtown and approximately 5 blocks from the campus, this hall has a private study lounge and furnished suites.

Parrish House (co-ed) Upperclassman; 3.0 GPA or higher – Houses 29 students

Clark Hall (co-ed) – Houses 78 students

Student Signature

Date

Parent/Guardian Signature (if under 18 years old) Date

Please mail this signed and completed contract with the \$100 room deposit (If paying by credit card, please remit \$103.00 to cover credit card charges. The refund amount will be \$100) to: CONTROLLER'S OFFICE--LEWIS-CLARK STATE COLLEGE--500 8TH AVENUE LEWISTON, ID 83501

For Cashier's Office Use Only:	For Residence Life Office Use Only:
Deposit \$ GR#	
Date	

EXHIBIT I SIGNAGE CRITERIA

Signage Criteria.

Exterior: Following Landlord's consent, Tenant shall have the right to place on the exterior of building and windows and at Tenant's sole expense, any signs which are permitted by applicable zoning ordinances and private restrictions. Landlord may refuse consent to any proposed signage that is in Landlord's opinion too large, deceptive, unattractive or otherwise inconsistent with or inappropriate to the Leased Premises or use of any other tenant. Landlord shall assist and cooperate with Tenant in obtaining any necessary permission from governmental authorities or adjoining owners and occupants for Tenant to place or construct the foregoing signs. Tenant shall repair all damage to the premises resulting from the installation or removal of signs installed by Tenant.

Interior: Tenant need not seek Landlord's consent for signage placed on the interior of the building (but excluding the windows), provided that said signage is installed only in that portion of the project administered by the Tenant. Tenant shall repair all damage to the premises resulting from the installation or removal of signs installed by Tenant.

EXHIBIT J

DESCRIPTIONS OF ITEMS OF PERMITTED FURNITURE AND EQUIPMENT

Student Tenants will be permitted to bring the following items and similar items of furniture and equipment into the Units and the adjoining Common Rooms occupied by them:

Electric Fan Stereo (but not an entertainment center cabinet) Small Television Computer, printer and similar computer peripherals Beanbag chair or small chair Desk lamp Clock Small chair, such as a folding chair Stand alone bookshelf – small (limited to room size) Card table & chairs

REFERENCE: APPLICABLE STATUTE, RULE, OR POLICY

Idaho State Board of Education GOVERNING POLICIES AND PROCEDURES SECTION: V. FINANCIAL AFFAIRS I. Real and Personal Property Services

April 2002

I. Real and Personal Property and Services

1. Authority

- a. The Board may acquire, hold, and dispose of real and personal property pursuant to Article IX, Section 2 and Article IX, Section 10, Idaho Constitution, pursuant to various sections of Idaho Code.
- b. Leases of office space or classroom space by any institution, school or agency except the University of Idaho are acquired by and through the Department of Administration pursuant to Section 67-5708, Idaho Code.
- c. All property that is not real property must be purchased consistent with Sections 67-5715 through 67-5737, Idaho Code, except that the University of Idaho may acquire such property directly and not through the Department of Administration. Each institution, school and agency must designate an officer with overall responsibility for all purchasing procedures.
- d. Sale, surplus disposal, trade-in, or exchange of property must be consistent with Section 67-5722, Idaho Code, except that the University of Idaho may dispose of such property directly and not through the Department of Administration.
- e. If the executive director finds or is informed that an emergency exists, he or she may consider and approve a purchase or disposal of equipment or services otherwise requiring prior Board approval. The institution, school or agency must report the transaction in the Business Affairs and Human Resources agenda at the next regular Board meeting together with a justification for the emergency action.
- 2. Acquisition of Real Property
 - a. Any interest in real property acquired for the University of Idaho must be taken in the name of the Board of Regents of the University of Idaho.

- b. Any interest in real property acquired for any other institution, school or agency under the governance of the Board must be taken in the name of the state of Idaho by and through the State Board of Education.
- c. This does not preclude a foundation or other legal entity separate and apart from an institution, school or agency under Board governance from taking title to real property in the name of the foundation or other organization for the present or future benefit of the institution, school or agency. (See Section V.E.)
- d. Acquisition of an option, lease, or any other present or future interest in real property by or on behalf of an institution, school or agency requires prior Board approval if the term of the lease exceeds five (5) years or if the cost exceeds two hundred fifty thousand dollars (\$250,000) annually.
- e. Appraisal.

An independent appraiser must be hired to give an opinion of fair market value before an institution, school or agency acquires fee simple title to real property.

f. Method of sale - exchange of property.

The Board will provide for the manner of selling real property under its control, giving due consideration to Section 33-601(4), applied to the Board through Section 33- 2211(5), and to Chapter 3, Title 58, Idaho Code. The Board may exchange real property under the terms, conditions, and procedures deemed appropriate by the Board.

g. Execution.

All easements, deeds, and leases excluding easements, deeds, and leases delegated authority granted to the institutions, school and agencies must be executed and acknowledged by the president of the Board or another officer designated by the Board and attested to and sealed by the secretary of the Board as being consistent with Board action.

- 3. Acquisition of Personal Property and Services
 - a. Purchases of equipment, data processing software and equipment, and all contracts for consulting or professional services either in total or through time purchase or other financing agreements, between two hundred fifty thousand dollars (\$250,000) and five hundred thousand dollars (\$500,000) require prior approval by the executive director. The executive director must be expressly advised when the recommended bid is other than the lowest qualified bid. Purchases exceeding five hundred thousand dollars (\$500,000) require prior Board approval.
 - b. Acquisition or development of new administrative software or systems that materially affect the administrative operations of the institution by adding new services must be reviewed with the executive director before beginning development. When feasible, such development will be undertaken as a joint endeavor by the four institutions and with overall coordination by the Office of the State Board of Education.

INSTITUTION / AGENCY AGENDA

SUBJECT

1st Reading of Proposed Amendments to Board Policy, Section III.T.4 – Intercollegiate Athletics.

APPLICABLE STATUTE, RULE, OR POLICY

Idaho State Board of Education Governing Policies & Procedures, Sections I.A.4. & 5.

Idaho State Board of Education Governing Policies & Procedures, Sections III.T.4.

BACKGROUND

The June Athletics Report shows each institution's original budget and estimate for the current fiscal year and the original budget for the subsequent year. The December Athletics Report shows each institution's revised estimates of revenues, expenditures and fund balance for the current fiscal year.

In most cases, the institutions revised estimates in the December reports are the same as the subsequent year budget in the June reports. Since the December reports are required to be finalized by the first week in November, the information for fall athletics programs is not readily available. Therefore, the institutions have not revised their estimates.

DISCUSSION

Board policy needs to be updated in order to reflect the current reporting schedule for the Athletic reports during the calendar year.

IMPACT

In order to provide a more meaningful estimate, the December Athletics Report should be moved to the February Board meeting. This will allow the institutions to include the results of their fall sports, including football, in the revised estimates.

STAFF COMMENTS AND RECOMMENDATIONS

Staff recommends updating Board policy to reflect the current schedule of Athletics reporting.

BOARD ACTION

A motion to approve for First Reading the changes to Idaho State Board of Education Governing Policies & Procedures, Section III.T.4. – Intercollegiate Athletics.

Moved by	Seconded by	Carried Yes	No	
----------	-------------	-------------	----	--

BAHR – SECTION II

TAB 11 Page 1

THIS PAGE INTENTIONALLY LEFT BLANK

CONSENT AGENDA - BUSINESS AFFAIRS AND HUMAN RESOURCES NOVEMBER 29 - DECEMBER 1, 2006

2nd Reading (as approved from 1st Reading)

Idaho State Board of Education GOVERNING POLICIES AND PROCEDURES SECTION: III. POSTSECONDARY AFFAIRS Subsection: T. Intercollegiate Athletics

June 2006

T. Intercollegiate Athletics

4. Financial Reporting.

The Board requires that the institutions adopt certain reporting requirements and common accounting practices in the area of intercollegiate athletic financing. The athletic reports shall contain revenues, and expenditures, in the detail prescribed by the Board office, including all revenue earned during a fiscal year. A secondary breakdown of expenditures by sport and the number of participants will also be required. The number and amounts of nonresident tuition waivers and the fund balances as of June 30 of the report year should be included in the report. The general format of the report will be consistent with the format used in recent years. The revenue and expenditures reported on these reports must reconcile to the NCAA Agreed Upon Procedures Reports that are prepared annually and reviewed by the external auditors. The institutions will submit the following reports to the Board:

- a. At the June Board meeting, the institutions shall submit an operating budget for the upcoming fiscal year beginning July 1 in a format prescribed by the Board office.
 - (1) Actual revenues and expenditures for the fiscal year most recently completed.
 - (2) Estimated revenues and expenditures for the current fiscal year.
 - (3) Proposed operating budget for the next budget year beginning July 1.
- b. At the <u>December February</u> Board meeting, the following fiscal year's financial information will be reported by each institution:
 - (1) Actual revenues and expenditures for the prior four (4) fiscal years
 - (2) Estimated revenues and expenditures for the current fiscal year.

THIS PAGE INTENTIONALLY LEFT BLANK

REFERENCE - APPLICABLE STATUTE, RULE, OR POLICY

Idaho State Board of Education **GOVERNING POLICIES AND PROCEDURES** SECTION: I. Governing Policies and Procedures Subsection: A. Policy-Making Authority

April 2004

4. Conformance with State and Federal Law

All Board Governing Policies and Procedures and the internal policies and procedures of its institutions, agencies and school will comply with and be in conformance to applicable laws.

5. Adoption, Amendment, or Repeal of Board Policies

a. Board policies may be adopted by majority vote at any regular or special meeting of the Board. The adoption, amendment, or repeal of a Board policy may be requested by any member of the Board, the executive director, or any chief executive officer. Persons who are Board employees, or students or student groups, must file a written request with the chief executive officer of an institution, agency or school, or his or her designee, to receive Board consideration. An Idaho resident, other than those described above, may file a written request with the executive director for Board consideration of a proposal. Regardless of the source, a statement of the proposed adoption, amendment, or repeal must be presented to the executive director for transmittal to the Board. If the subject matter of the presentation concerns an agency, institution, school, or department of the Board, the executive director will also notify the appropriate chief executive officer of the request.

b. Board action on any proposal will not be taken earlier than the next regular or special meeting following Board approval for first reading. During the interim between the first reading and Board action, the chief executive officers will seek to discuss and review the proposal with faculty, staff, or other Board employees and students or student groups, as appropriate. The chief executive officers will transmit summaries of oral statements and written comments on the proposal to the executive director. After thorough consideration, the proposal will be presented by the executive director to the Board for action.

c. The executive director is authorized to make nonsubstantive corrections and amendments to Board Governing Policies and Procedures as may be necessary in such areas as typographical errors, cross-references, and citations of state and federal statutes.

THIS PAGE INTENTIONALLY LEFT BLANK

INSTITUTION / AGENCY AGENDA

SUBJECT

Idaho Promise Scholarship – Category B Award

APPLICABLE STATUTE, RULE, OR POLICY

IDAPA 08.01.05.102.01 Sections 33-4305, 33-4307, and 33-4308, Idaho Code

BACKGROUND

The Idaho Promise Scholarship - Category B award is for all Idaho students attending college for the first time and who have a high school grade point average of at least 3.0 or an ACT score of 20 or above. This scholarship is limited to two years and to students younger than 22 years of age. Students must maintain at least a 2.5 GPA while taking an average of 12 credits to remain eligible for the scholarship. State law requires the State Board of Education to annually set the amount of the award based on the legislative appropriation and the number of eligible students.

DISCUSSION

Idaho's colleges and universities have identified eligible Promise Category B recipients for the fall 2006 semester at a rate of \$250/eligible student. Distribution of funds for fall 2006 students has occurred. Remaining funds available for distribution for the spring 2007 semester allow the Board to increase the spring 2007 award to \$275. This will allow for maximum use of the state appropriation for this program, and will assist eligible Idaho students in paying for postsecondary education.

IMPACT

The Idaho Promise Scholarship Category B provides a merit-based scholarship to Idaho high school graduates in an attempt to motivate students to excel in high school and attend a higher education institution in Idaho. The estimated number of students participating in the program for Fall 2006 is 7,732, and 8,765 for Spring 2007. By increasing the spring award, the estimated expenditure will be \$4,343,375. The State appropriation for this program is \$4,446,700, leaving a balance of approximately \$103,325.

STAFF COMMENTS AND RECOMMENDATIONS

Staff recommends approval of the Promise Category B scholarship in the amount of \$275 for the Spring semester, 2007. This would equal \$525 for the entire 2006-2007 academic year. A small amount would remain in the FY 2007 appropriation.

INSTITUTION / AGENCY AGENDA - continued

BOARD ACTION

A motion to approve the amount of the Idaho Promise Scholarship, Category B, to be \$275 for the Spring semester, 2007, for a total of \$525 for the 2006-2007 academic year, per eligible student for those current recipients who maintain eligibility, and for qualified first-year entering students under the age of 22 in the Spring 2007.

Moved by_____ Seconded by_____ Carried Yes_____ No_____

Promise B Scholarship Adjustment for Spring 2007 - December 2006 Board Meeting

Actual Fall Award Amount Proposed Spring Award Amount	\$250 \$315		
Institution	Fall <u>Students</u>	Fall Payments to Institutions	% of <u>Total \$\$</u>
Boise State University	1,823	455,625	22.83%
Idaho State University	1,200	312,500	15.66%
University of Idaho	1,500	395,000	19.79%
Lewis-Clark State College	215	53,000	2.66%
College of Southern Idaho	660	165,000	8.27%
North Idaho College	468	146,250	7.33%
Albertson College	221	52,500	2.63%
Eastern Id. Tech. College	49	15,000	0.75%
N. W. Nazarene University	141	37,500	1.88%
BYU-Idaho	<u>1,455</u>	<u>363,750</u>	<u>18.22%</u>
Total Fall 2006 Recipients:	7,732	\$1,996,125	100.00%
BYU-Idaho Summer session estimate	<u>672</u>	<u>168,000</u>	
Est. Spring Recipients, inc. BYU-I summer: (spring estimate Provided by Brenda Reeser@BYU-I)	8,404	\$2,164,125	
FY 2007 Appropriation, Promise B	\$ 4,446,700		
Actual Fall 2006 Students	7,732		
Total Fall 2006 Award @ \$250	1,933,000		
Available for Spring 2006	\$ 2,513,700	l	
Estimated Spring 2006 and BYUI summer Students		(see note)	
Proposed Spring 2006 Award	\$ 275		4,343,375
Proposed Spring 2006 Total	\$ 2,410,375		
Amount Remaining % of Original Appropriation Remaining	\$ 103,325 2.3%		

Note: Includes BYUI additional 361Spring 07 Promise Students

THIS PAGE INTENTIONALLY LEFT BLANK

BUSINESS AFFAIRS AND HUMAN RESOURCES NOVEMBER 29 - DECEMBER 1, 2006

REFERENCE - APPLICABLE STATUTE, RULE, OR POLICY

ADMINISTRATIVE RULES

IDAPA 08.01.05.102.01

102. MONETARY VALUE OF THE SCHOLARSHIP.

01. Monetary Value. The monetary value of each scholarship shall be set annually by the Board in accordance with Sections 33-4307(3) et seq., Idaho Code. (3-15-02)

02. Duration. The grant covers up to one (1) educational year or equivalent for attendance at an eligible postsecondary educational institution. (3-15-02)

REFERENCE - APPLICABLE STATUTE, RULE, OR POLICY - continued

IDAHO STATUTES

TITLE 33 EDUCATION CHAPTER 43 SCHOLARSHIPS

33-4305. PURPOSES. The purpose of this act is:

- (1) To establish a state scholarship program for the most talented Idaho secondary school graduates or the equivalent, consisting of category A students with outstanding academic qualifications and category B students with a cumulative grade point average for grades nine (9) through twelve (12) of 3.0 or better or achieving an ACT score of 20 or better or who become eligible after the student's first semester or who meet any other criteria as may be established by the state board of education and the board of regents of the university of Idaho, who will enroll in undergraduate nonreligious academic and professional-technical programs in eligible postsecondary institutions in the state; and
- (2) To designate the state board of education and the board of regents of the university of Idaho as the administrative agency for the state scholarship program.

33-4307. ELIGIBILITY -- MAXIMUM AMOUNTS -- CONDITIONS.

A grant may be awarded to an eligible student for matriculation at an eligible postsecondary educational institution in the state of Idaho if:

(1) The individual is accepted for enrollment as a full-time undergraduate or professional-technical student, as follows:

(a) In the case of an individual beginning his first year or freshman year of postsecondary education, he has satisfied the requirements for admission and has enrolled in an eligible postsecondary institution.
(b) In the case of an individual enrolled in an eligible postsecondary institution following the successful completion of the first term, he continues to meet the requirements of this act and has maintained such high standards of performance as may be required. Provided that high academic standards are maintained in accordance with requirements of this chapter, a student continues to be eligible when transferring from one (1) major program to another.

(c) In the case of an individual transferring from one (1) eligible postsecondary institution in Idaho to another eligible postsecondary institution in Idaho, he continues to meet the requirements of this act, is accepted and enrolled at the eligible postsecondary institution to which he is transferring, and has maintained such high standards of performance as may be required. (2) The grant for category A students is as follows:

(a) The grant payment to an individual per educational year for attendance on a full-time basis is not in excess of an amount determined annually by the state board of education or in excess of the total educational costs as certified by an official of the eligible postsecondary institution to be attended by the individual receiving the grant, whichever is less.

(b) The total grant payments over a period of six (6) years to an individual may not exceed four (4) annual grants or the total educational costs for four (4) educational years completed as certified by an official of the eligible postsecondary institution or institutions attended by the individual receiving the grant, whichever is less.

(c) The individual receiving such a grant signs an affidavit stating that the grant will be used for educational costs only.

(d) The grant is awarded on the basis of extraordinary performance in standardized, unweighted competitive examination and high school record.
(e) The individual receiving the grant is not precluded from receiving other financial aid, awards, or scholarships, provided the total of the grant and such other financial aids, awards or scholarships does not exceed the total educational costs for attendance at an eligible postsecondary institution as certified by an official of the eligible postsecondary institution to be attended by the individual receiving the grant.

(f) Grant payments shall correspond to academic terms, semesters, quarters or equivalent time periods at an eligible postsecondary institution; in no instance may the entire amount of a grant for an educational year, as defined in section 33-4306(10), Idaho Code, be paid to or on behalf of such student in advance.

(g) The individual has complied with such rules as may be necessary for the administration of this act.

(3) The grant for category B students is as follows:

(a) The grant payment to an individual per educational year for attendance on a full-time basis is not in excess of an amount determined annually by the state board of education and the board of regents of the university of Idaho and not to exceed one thousand two hundred dollars (\$1,200) per year including the required match.

(b) The total grant payments over a period of four (4) years to an individual may not exceed two (2) annual grants.

(c) The individual receiving such a grant signs an affidavit stating that the grant will be used for educational costs only.

(d) The grant is awarded on the basis of a high school record of a 3.0 grade point average or an ACT composite score of 20 or better and other criteria as may be established by the state board of education and the board of regents of the university of Idaho.

(e) The individual receiving the grant is not precluded from receiving

BUSINESS AFFAIRS AND HUMAN RESOURCES NOVEMBER 29 - DECEMBER 1, 2006

other financial aid, awards or scholarships except that category A student award recipients are not eligible for category B awards.

(f) Grant payments shall correspond to academic terms, semesters, quarters or equivalent time periods at an eligible postsecondary institution; in no instance may the entire amount of a grant for an educational year, as defined in section 33-4306(8), Idaho Code, be paid to or on behalf of such student in advance. The first grant payments pursuant to this section for category B students shall be made in the fall of 2001 or in the first fall academic term following an appropriation and when moneys are available to implement the category B scholarship program, whichever date is later.

(g) The individual has complied with such rules as may be necessary for the administration of this chapter.

(h) All eligible postsecondary institutions will report annually to the state board of education and the board of regents of the university of Idaho the number of students for each term receiving a grant award and the number of awards that were matched by the institution.

33-4308. MAXIMUM NUMBER OF GRANTS.

- The total number of grants to eligible category A students shall not exceed one hundred (100) per year, nor a cumulative total number of grants of four hundred (400) outstanding at any given time
- (2) The total number of grants to category B students will be determined annually by the state board of education and the board of regents of the university of Idaho based on the number of eligible students, the individual award amount and the availability of funds.

INSTITUTION / AGENCY AGENDA COLLEGE AND UNIVERSITIES OF THE STATE BOARD

REFERENCE

Annually

Regular December meetings of the State Board of Education

SUBJECT

Presentation of annual financial audit of the Colleges and Universities by the Board's external auditor.

APPLICABLE STATUTE, RULE OR POLICY

Idaho State Board of Education Governing Policies & Procedures, Section V.H.5.e.-f

BACKGROUND

The Board contracted with Moss Adams LLP, an independent certified public accounting firm, to conduct the annual financial audits of Boise State University, Idaho State University, University of Idaho, Lewis-Clark State College, and Eastern Idaho Technical College. FY 2006 is the second year that Moss Adams has conducted audits of the college and university financial statements.

The audits are conducted in accordance with Generally Accepted Government Auditing Standards (GAGAS), and include an auditor's opinion on the basic financial statements.

DISCUSSION

Along with this agenda, Board members will receive, for each institution audited, the following document from Moss Adams:

• "Financial Statements for the Year Ended June 30, 2006 and Independent Auditors' Reports" which contains the Independent Auditors' Report, Management's Discussion and Analysis (MDA), Financial Statements;

IMPACT

The State Board of Education will be informed, via published documents and the Moss Adams presentation, of the financial report regarding the five noted institutions for state Fiscal Year 2006. Institutions that have been audited will also be made aware of their particular financial condition, and recommended changes to procedures regarding financial matters.

INSTITUTION / AGENCY AGENDA COLLEGE AND UNIVERSITIES OF THE STATE BOARD - continued

STAFF COMMENTS AND RECOMMENDATIONS

The audited financial statements present the financial activity at each audited institution and include the following reports:

- Management's Discussion and Analysis (MD&A)
- Statement of Net Assets
- Statement of Revenues, Expenses and Changes in Net Assets
- Statement of Cash Flows
- Notes to the Financial Statements

While the MD&A and Notes help explain the financial activity and some trends, the audited financial statements, MD&A and notes do not attempt to measure the financial health of each institution. At the beginning of the third year of work, the Audit and Finance Committees will resume discussion regarding how best to identify and use various indicators (including financial ratios) for the purpose of evaluating the financial condition of the institutions based upon data collected for, and presented in, the financial statements. These discussions will also include institutional staff.

Moss Adams has performed in an exemplary manner with respect to keeping the Audit Committee and OSBE staff updated on the progress of the audit work throughout the year. This communication has included monthly written reports, and telephone conference calls or face-to-face meetings as needed. In mid-November Moss Adams conducted a preliminary review of the financial statements with members of the Audit Committee and Board staff.

There has been a much-improved level of communication and expectation between Moss Adams and the institutions compared to the first year (FY 2005) of the engagement. Last year the Audit Committee indicated to Moss Adams this was important to the overall contract, and Moss agreed.

BOARD ACTION

A motion to accept from the Audit Committee the Fiscal Year 2006 financial audit reports for Boise State University, Idaho State University, University of Idaho, Lewis-Clark State College, and Eastern Idaho Technical College, as presented by Moss Adams LLP.

Moved by_____ Seconded by_____ Carried Yes____ No____

REFERENCE - APPLICABLE STATUTE, RULE, OR POLICY

Idaho State Board of Education GOVERNING POLICIES AND PROCEDURES SECTION: V. FINANCIAL AFFAIRS H. Audits

August, 2005

5. Independent Auditors

e. Financial Statement Review

At the completion of the independent audit, the Committee shall review with institution management and the independent auditors each institution's financial statements, Management's Discussion and Analysis (MDA), related footnotes, and the independent auditor's report. The Committee shall also review any significant changes required in the independent auditor's audit plan and any serious difficulties or disputes with institution management encountered during the audit. The Committee shall document any discussions, resolution of disagreements, or action plans for any item requiring follow-up.

f. Single Audit Review

At the completion of the Single Audit Report (as required under the Single Audit Act of 1984, and the Single Audit Act Amendments of 1996), the Committee shall review with institution management and the independent auditors each institution's Single Audit Report. The Committee shall discuss whether the institution is in compliance with laws and regulations as outlined in the current Single Audit Act described in the U.S. Office of Management and Budget (OMB) Circular A-133 Compliance Supplement. The Committee shall report to the Board that the review has taken place and any matters that need to be brought to the Board's attention. The Committee shall document any discussions, resolution of disagreements, or action plans for any item requiring follow-up.

THIS PAGE INTENTIONALLY LEFT BLANK